



REFERENCE ONLY

UNIVERSITY OF LONDON THESIS

Degree

hD Year 2006 Name of Author DAPHNIS

COPYRIGHT

Danny Diamanh's This is a thesis accepted for a Higher Degree of the University of London. It is an unpublished typescript and the copyright is held by the author. All persons consulting the thesis must read and abide by the Copyright Declaration below.

COPYRIGHT DECLARATION

I recognise that the copyright of the above-described thesis rests with the author and that no quotation from it or information derived from it may be published without the prior written consent of the author.

LOANS

Theses may not be lent to individuals, but the Senate House Library may lend a copy to approved libraries within the United Kingdom, for consultation solely on the premises of those libraries. Application should be made to: Inter-Library Loans, Senate House Library, Senate House, Malet Street, London WC1E 7HU.

REPRODUCTION

University of London theses may not be reproduced without explicit written permission from the Senate House Library. Enquiries should be addressed to the Theses Section of the Library. Regulations concerning reproduction vary according to the date of acceptance of the thesis and are listed below as guidelines.

- A. Before 1962. Permission granted only upon the prior written consent of the author. (The Senate House Library will provide addresses where possible).
- In many cases the author has agreed to permit copying upon Β. 1962 - 1974. completion of a Copyright Declaration.
- C. 1975 - 1988. Most theses may be copied upon completion of a Copyright Declaration.
- D. 1989 onwards. Most theses may be copied.

This thesis comes within category D.



This copy has been deposited in the Library of .

This copy has been deposited in the Senate House Library, Senate House, Malet Street, London WC1E 7HU.

C:\Documents and Settings\lproctor\Local Settings\Temporary Internet Files\OLK8\Copyright - thesis (2).doc

Molecular and Cytogenetic Approaches to the Analysis of Chromosomes in Human Preimplantation Embryos.

by

Danny Diamantis Daphnis

A thesis submitted for the degree of Doctor of Philosophy at the University of London.

October 2006

The Human Genetics and Embryology Group Department of Obstetrics and Gynaecology University College London. UMI Number: U591674

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



UMI U591674 Published by ProQuest LLC 2014. Copyright in the Dissertation held by the Author. Microform Edition © ProQuest LLC. All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code.



ProQuest LLC 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106-1346

Dedication

To my mother, brother, mum and father for believing in me.

ACKNOWLEDGEMENTS

There are many people who deserve recognition for all their help and input in completing this thesis. Principally, I would like to thank my supervisor Dr. J.C. Harper for her invaluable help, concern and supervision throughout this study, and for her support during the time of the write up. I would also like to express my appreciation and gratitude to Prof J.D.A. Delhanty for her continuous advice, guidance, and encouragement during my time at UCL, and for inspiring me to embark on this wonderful journey of research.

I would like to specially acknowledge Dr. Mara Simopoulou for introducing me to the world of PGD and FISH, and for being a true friend to me offering her time and scientific expertise unconditionally. I am also grateful to Dr. Dagan Wells for introducing me to the joys of PCR and whose expertise and kindness have aided so many students over the years, not least myself.

My deep appreciation goes to all of the PGD team. Special thanks go to Elpida Fragouli ('lab director') for sharing the troubles of CGH with me (far too many tubes!!!) and offering her friendship unconditionally. Appreciation should be given to my 'student' Georgia Kakourou for working with me to develop the protocols for many PGD cases and even staying late for me. Gratitude to Anna Mantzouratou should be awarded for keeping the 'Chenies Mews team' sane. Credit should be given to 'Prof' Leigh Butt and 'Dr' Evelyn Dudman for their help with a 'misbehaving' prism.

Special gratitude should be awarded to the London Fertility Clinic team and especially Prof Ian Logan Craft for finding sponsorship for my PhD through Life-Force Ltd. Moreover, I would like to thank Jaco Geyer, Sonya Jerkovic and Kostas Economou for providing material whenever requested.

Also, a big thank you goes to all the staff of The Assisted Conception Unit, University College Hospital London. I would also like to extend my thanks to Alpesh Doshi the Lab manager and lead embryologist for his excellent embryo manipulation as well as the rest embryology team (Sarah²).

Finally, to all my family and friends a big thank you for your patience and for your constant support and encouragement during this time, I could not have done this without you.

Abstract

The original focus of the research for this thesis was concentrated on establishing strategies to detect chromosome imbalance as well as exploring the phenomenon of mosaicism and its underlying mechanisms in human preimplantation embryos. High levels of chromosomal mosaicism have been detected in human preimplantation embryos mostly by fluorescent *in situ* hybridisation (FISH) but also by comparative genomic hybridisation (CGH) and karyotyping. Mosaicism could arise through several mechanisms including abnormal cell divisions (mitotic non-disjunction or anaphase lag), failure of cytokinesis or endoreduplication. The FISH procedure has been criticised, as it is prone to failure. Two separate studies were developed and carried out in order to detect the level of mosaicism in embryos.

In the first study a FISH protocol for the use of two different probes per chromosome was developed. The aim was to gain information on mechanisms leading to aneuploidy mosaicism and its true incidence. Three colour FISH was performed in three sequential rounds. In the first and second round different probes were used for chromosomes 1, 11, 18. In the third round probes were used for chromosomes X, Y and 18. Each FISH procedure included a control slide to assess FISH efficiency in all rounds of FISH. Two groups of embryos were spread on day 5 of development; embryos grown in cleavage media throughout and embryos transferred to blastocyst media after day 3. A total of 21 embryos were analysed in each Group. The FISH results revealed one uniformly diploid and 20 mosaic embryos for Group I and 2 uniformly diploid and 19 mosaic embryos for Group II. Use of 2 different probes per chromosome was able to detect FISH artefacts and failure of hybridisation. Postzygotic chromosome loss was the predominant mechanism leading to aneuploidy mosaicism for both groups, followed by chromosome gain, with only a few examples of mitotic non-disjunction. The relatively high percentage of tetraploidy in the blastocyst medium group was considered to reflect normal embryonic development.

The use of CGH was investigated as an alternative strategy to detect the true level of mosaicism in the whole genome. The second part of the research for this thesis involved assessing the efficiency of CGH, improving the protocol for optimised use on single cells, and its application to human embryonic material. Results suggested

that CGH is a laborious and technically demanding technique however, can provide extra information when used as a research tool. CGH was combined with FISH in order to assess chromosomal abnormalities in day 3 and day 5 embryos respectively. CGH was employed in 1-2 biopsied cells from a day 3 embryo, which was grown up to day 5 and further analysed by multi-colour FISH. The aim of this study was to observe the full chromosomal status of 1-2 blastomeres biopsied at the cleavage stage (day 3) of development followed by FISH analysis of the rest of the embryo on day 5. This would allow the assessment of abnormalities in day 3 embryos by a full karyotype and then confirm whether the abnormality persists until day 5 using FISH for the chromosome(s) involved. In summary 30 embryos were fully analysed and only 3 (10%) were uniformly normal, while the rest were mosaic or chaotic. CGH was able to provide results in 83.3% of the embryos subjected to analysis. FISH and CGH showed either agreeing or complimentary results for all embryos analysed. The predominant mechanism of aneuploidy mosaicism was whole chromosome loss. Furthermore, partial aneuploidy was also detected, with partial chromosome loss being the principal mechanism.

In the final part of the thesis the development of PGD protocols for a single gene disorder, namely DM, were devised using polymerase chain reaction (PCR) techniques. Two PGD protocols were devised and employed clinically in two patients undergoing PGD for DM using fluorescent PCR. Due to the extensive workup needed to develop the specific PCR protocols for each patient, a universal-like protocol was researched. Such a protocol would involve production of a sufficient amount of DNA through whole genome amplification techniques i.e. DOP-PCR from a single cell to carry out subsequent analysis with F-PCR markers as well whole chromosome analysis using CGH. DOP-PCR amplified DNA was subjected to amplification of five markers that would have been used during a PGD workup for DM and also subjected to CGH analysis. Initially genomic DNA was tested which produced high fidelity of amplification. Single cell DNA was then utilised in order to assess the amplification rate, allele dropout (ADO) and contamination levels. It was shown that there was relatively low amplification and ADO rates of the five markers at the single cell DNA level compared to the results obtained when the markers were amplified directly from single cells during the development of the two F-PCR PGD protocols. However, CGH

analysis was successfully performed indicating that novel WGA methods might overcome the problem of low fidelity of the F-PCR markers.

In conclusion, several molecular and cytogenetic techniques were employed to analyse human embryos either to provide answers to phenomena such as an euploidy mosaicism and chaos or to select normal embryos in a clinical setting for PGD for patient carrying single gene defects.

Table of Contents

| Title. | 1 |
|--|----|
| Dedication. | 2 |
| Acknowledgements. | 3 |
| Abstract. | 4 |
| Table of Contents | 8 |
| List of Tables. | 15 |
| List of Figures. | 18 |
| List of Abbreviations. | 21 |
| Chapter 1: Introduction | 25 |
| 1.1 Human Development. | 26 |
| 1.1.1 Gametogenesis and Meiosis | 26 |
| 1.1.1.1 Spermatogenesis. | 26 |
| 1.1.1.2 Oogenesis. | 27 |
| 1.1.2. In Vivo Fertilisation. | 28 |
| 1.1.3. In Vivo Preimplantation Embryo Development. | 29 |
| 1.1.4. In Vitro Fertilisation | 32 |
| 1.1.4.1. Intracytoplasmic Sperm Injection (ICSI). | 33 |
| 1.2 Human Genetic Disease | 35 |
| 1.2.1 Chromosome Abnormalities | 35 |
| 1.2.1.1 Numerical Aberrations | 36 |
| 1.2.1.2 Structural Aberrations | 39 |
| 1.2.1.2.1 Robertsonian Translocations | 41 |
| 1.2.1.2.2 Reciprocal Translocations | 42 |
| 1.2.1.2.3 Inversions, Insertions and Ring Chromosomes | 42 |
| 1.2.1.3 Mosaicism | 44 |
| 1.2.1.4 Chromosome Abnormalities in Human Gamete Studies | 44 |
| 1.2.1.4.1 Karyotyping Studies | 44 |
| 1.2.1.4.2 FISH Studies | 45 |
| 1.2.1.5 Chromosomal Abnormalities in Embryo Studies | 46 |
| 1.2.1.5.1 Karyotyping Studies | 46 |
| 1.2.1.5.2 FISH Studies | 47 |
| 1.2.1.5.3 CGH Studies | 48 |
| 1.2.1.5.4 SKY-FISH and M-FISH | 49 |
| 1.2.2 Single Gene Defects | 51 |
| 1.2.2.1 Myotonic Dystrophies | 52 |
| 1.2.2.1.1 Clinical Pathology of DM | 52 |
| 1.2.2.1.2 Genetics of DM | 53 |
| 1.2.2.1.3 The Myotonin-Protein Kinase | 54 |
| 1.2.2.1.4 Mechanisms of DM Pathogenesis | 55 |
| 1.3 Molecular and Cytogenetic Techniques. | 56 |
| 1.3.1 Karyotyping. | 57 |
| 1.3.1.1 Karyotyping Principles | 57 |
| 1.3.1.2 Karyotyping Applications | 57 |
| 1.3.1.3 Technical Limitations of Karyotyping | 58 |
| 1.3.2 Fluorescence In Situ Hybridisation (FISH). | 58 |
| 1.3.2.1 FISH Principles | 58 |
| 1.3.2.2 FISH Applications. | 62 |
| 1.3.2.2.1 Interphase FISH Applications | 62 |

| 1.3.2.2.2 Metaphase FISH Applications | 63 |
|---|----------------------|
| 1.3.2.3 Technical Limitations of FISH | 64 |
| 1.3.3 Polymerase Chain Reaction (PCR) | 65 |
| 1.3.3.1 PCR Principles | 65 |
| 1.3.3.1.1 Multiplex PCR | 66 |
| 1.3.3.1.2 Whole Genome Amplification (WGA) | 66 |
| 1.3.3.2 Methods of Detection | 68 |
| 1.3.3.2.1 Fluorescent PCR | 68 |
| 1.3.3.2.2 Denaturing Gradient Gel Electrophhoresis | 69 |
| 1.3.3.2.3 Restriction Endonuclease Digestion | 69 |
| 1.3.3.2.4 Heteroduplex Analysis | 70 |
| 1.3.3.2.5 Single Stranded Conformational Polymorphism | 70 |
| 1.3.3.2.6 Amplification Refractory Mutation System | 70 |
| 1.3.3.3 PCR Applications | 71 |
| 1.3.3.4 Technical Limitations of PCR | 72 |
| 1.3.4 Comparative Genomic Hybridisation (CGH) | 72 |
| 1.3.4.1 CGH Principles | 73 |
| 1.3.4.2 Single-Cell CGH | 75 |
| 1.3.4.3CGH Applications | 76 |
| 1.3.4.3.1 Cancer Studies | 76 |
| 1.3.4.3.2 Prenatal Studies | 77 |
| 1.3.4.3.3 Embryo Studies | 77 |
| 1 3 4 3 Technical Limitations of CGH | 78 |
| 1 4 Prenatal Diagnosis | 78 |
| 1 4 1 Chorionic Villus Sampling (CVS) | 70 79 |
| 1 4 2 Amniocentesis | 80 |
| 1 4 3 Fetal Blood Sampling (FBS) | 80 |
| 1 4 4 Fetal Cells in Maternal Circulation | 80 |
| 1.4.5 Cytogenetics and Prenatal Diagnosis | 81 |
| 1 4 6 Prenatal Diagnosis of DM | 82 |
| 1 4 7 Prenatal Diagnosis and Mosaicism | 83 |
| 1 4 7 1 Confined Placental Mosaicism | 83 |
| 1 4 7 2 Uningerental Discomy | 85 |
| 1.5 Preimplantation Genetic Diagnosis (PGD) | 86 |
| 1.5.1 Sampling Strategies of PGD | 87 |
| 1.5.1 1 Polar Body Bionsy | 88 |
| 1.5.1.2 Cleavage-Stage Bionsy | 80 |
| 1.5.1.2 Cleavage-Stage Diopsy | 01 |
| 1.5.1.5 Diastocyst Diopsy | 02 |
| 1.5.2 Chromosonial Adnormannes | 92 |
| 1.5.2.2 Structural Abarrations | 92 |
| 1.5.2.2 Structural Abertations | 93 |
| 1.5.2.2.2 Robertsonian Translocations | 9 4 05 |
| 1.5.2.2.2 Recipiocal Halistocations 1.5.2.2.3 Inversions Insertions and Ping Chromosomes | 95 07 |
| 1.5.2.2.5 Inversions, insertions and King Chromosomes | 97 |
| 1.5.2.4 Single Coll FISH Limitations | 100 |
| 1.5.2.4 Single Cent FISH Limitations | 100 |
| 1.5.5 Single Gene Disolders | 101 |
| 1.5.3.1 MICrosalenne DNA (STK S) 1.5.2.2 Single Coll F DCD | 103 |
| 1.5.3.2 Single Cell F-FCK | 104 |
| 1.5.3.3 Single Cell Multiplex PCR | 104 |

| 1.5.3.4 Single Cell WGA | 105 |
|--|-----|
| 1.5.3.5 WGA and STR's | 106 |
| 1.5.3.6 Single Cell Mini-Sequencing | 107 |
| 1.5.3.7 Single Cell PCR Limitations | 108 |
| 1.5.3.7.1 Contamination | 108 |
| 1.5.3.7.2 Amplification Failure | 109 |
| 1.5.3.7.3. Allele Dropout | 109 |
| 1.5.4 PGD for DM | 110 |
| 1.6 Mosaicism in Human Preimplantation Embryos | 111 |
| 1.6.1 Mosaicism and Karyotyping | 113 |
| 1.6.2 Mosaicism and FISH | 116 |
| 1.6.3 Mosaicism and CGH | 118 |
| 1.6.4 Types and Mechanisms of Mosaicism | 121 |
| 1.6.4.1 Aneuploidy Mosaicism | 122 |
| 1.6.4.2 Ploidy Mosaicism | 123 |
| 1.6.4.3 Chaotic Mosaicism | 125 |
| 1.6.4.4 Multinucleation | 125 |
| 1.6.4.5 Mosaicism and Translocations | 126 |
| 1.7 Aims of Study | 127 |
| | |
| Chapter 2: Materials and Methods | 129 |
| 2.1 Materials | 130 |
| 2.1.1 FISH Materials | 130 |
| 2.1.1.1 Materials for Lymphocyte culture | 130 |
| 2.1.1.2 Materials for Embryo Spreading | 130 |
| 2.1.1.3 Materials for Nick Translation (commercial kit) | 130 |
| 2.1.1.4 Materials for the FISH procedure | 131 |
| 2.1.2 PCR Materials | 131 |
| 2.1.2.1 Materials for Single-cell isolation | 131 |
| 2.1.2.2 Materials for DNA extraction | 131 |
| 2.1.2.3 Materials for PCR procedure | 132 |
| 2.1.2.4 Materials for gel electrophoresis | 132 |
| 2.1.2.5 Materials for ABI Prism [™] 310 and 3100 | 132 |
| 2.1.3 CGH Materials | 133 |
| 2.1.3.1 Materials for DNA Extraction | 133 |
| 2.1.3.2 DOP-PCR Materials | 133 |
| 2.1.3.3 Nick-Translation Materials | 133 |
| 2.1.3.4 CGH Procedure | 133 |
| 2.1.4 Embryo Materials | 134 |
| 2.1.4.1 Ethical Approval | 134 |
| 2.1.4.2 Grading Criteria for the Embryos | 134 |
| 2.1.4.3 Categorisation of Chromosomal Abnormalities in Embryos | 134 |
| 2.1.4.4 Embryo Culture | 135 |
| 2.1.4.5 Embryo Freezing | 135 |
| 2.1.4.6 Embryo Thawing | 136 |
| 2.1.4. / Embryo Spreading | 136 |
| 2.1.4.8 Embryo Classification and Scoring of Embryos | 136 |
| 2.1.4.9 Mosaicism and Events | 136 |
| 2.1.5 Probes and Primers | 157 |
| 2.1.3.1 DNA Probes | 131 |

.

| 2.1.5.2 PCR Primers | 138 |
|--|-----|
| 2.1.6 Equipment | 140 |
| 2.1.6.1 FISH and CGH Equipment | 140 |
| 2.1.6.2 PCR and CGH Equipment | 140 |
| 2.2 Methods | 140 |
| 2.2.1 FISH Methods | 140 |
| 2.2.1.1 Control Lymphocyte Preparation | 141 |
| 2.2.1.1.1 Harvesting | 141 |
| 2.2.1.1.2 Preparation of control lymphocytes | 142 |
| 2.2.1.2 Day-5 Embryo Spreading | 143 |
| 2.2.1.3 Plasmid Probe Preparation | 143 |
| 2.2.1.4 Probe Labelling | 144 |
| 2.2.1.4.1 Nick Translation | 145 |
| 2.2.1.5 FISH Procedure | 145 |
| 2.2.1.5.1 Slide Pre-treatment | 146 |
| 2.2.5.1.2 Probe Preparation | 146 |
| 2.2.1.5.3 Separate Denaturation | 146 |
| 2.2.1.5.4 Post-Hybridisation Washes | 146 |
| 2.2.1.5.5 Detection, Capturing and Scoring of the Probes | 147 |
| 2.2.1.5.6 Sequential FISH Procedure | 147 |
| 2.2.1.6 FISH Protocol for Chapter 4 | 148 |
| 2.2.1.6.1 FISH Protocol Selection and Optimisation | 148 |
| 2.2.1.6.2 Final FISH Protocol | 149 |
| 2.2.1.6.3 Stringency Conditions | 149 |
| 2.2.2 PCR Methods | 150 |
| 2.2.2.1 DNA Extraction from Blood | 151 |
| 2.2.2.2 Single Cell Isolation | 151 |
| 2.2.2.2.1 Preparation for Single-cell Isolation | 151 |
| 2.2.2.2.2 Single Buccal Cells | 152 |
| C C | |
| 2.2.2.2.3 Isolation and Tubing of Single Human | |
| Blastomeres | 152 |
| 2.2.2.3 PCR Procedure | 153 |
| 2.2.2.3.1 DOP-PCR | 153 |
| 2.2.2.3.2 Fluorescent PCR | 153 |
| 2.2.2.4 PCR Analysis | 155 |
| 2.2.2.4.1 Gel Electrophoresis | 155 |
| 2.2.2.4.2 Automated laser DNA analyser | |
| (ABI Prism TM 310 and 3100) | 155 |
| 2.2.4.3 Optimisation of PCR protocol | 155 |
| 2.2.2.5 PGD | 156 |
| 2.2.2.5.1 PGD Consultation | 156 |
| 2.2.2.5.2 IVF Treatment and Manipulation of Embryos | 156 |
| 2.2.2.5.3 Embryo Biopsy (UCL) | 156 |
| 2.2.2.5.4 Embryo Biopsy (LFC) | 157 |
| 2.2.2.6 PGD Strategy for DM Using F-PCR | 157 |
| 2.2.2.6.1 Genomic DNA Workup | 157 |
| 2.2.2.6.2 Single Cell Workup | 158 |
| 2.2.2.6.3 Clinical DM PGD Cases | 158 |
| 2.2.2.7 PGD Strategy for DM Using WGA | 159 |
| | |

| 2.2.2.7.1 Procedure on Genomic DNA | 159 |
|---|-----|
| 2.2.2.7.2 Procedure on Single Cells | 160 |
| 2.2.2.7.3 CGH on DOP-PCR Amplified Single Cells | 160 |
| 2.2.3 CGH Methods | 160 |
| 2.2.3.1 DNA Extraction from Frozen Fibroblasts | 160 |
| 2.2.3.2 Single Cell isolation | 161 |
| 2.2.3.3 DOP-PCR | 161 |
| 2.2.3.4 CGH Procedure | 161 |
| 2.2.3.4.1 Labelling of Probe DNA | 161 |
| 2.2.3.4.2 Preparation of Labelled DNA | 161 |
| 2.2.3.4.3 Denaturation of Metaphase Chromosome | |
| Spreads and Probe Hybridisation | 162 |
| 2.2.3.4.4 Post-Hybridisation Washing | 162 |
| 2.2.3.4.5 Microscopy and Image Analysis | 162 |
| 2.2.3.5 CGH Procedure for Chpater 5 | 163 |
| 2.2.3.5.1 WGA Protocol | 163 |
| 2.2.3.5.2 Optimisation of WGA Protocols | 163 |
| 2.2.3.5.3 CGH Protocol | 164 |
| 2.2.3.5.4 Optimisation of CGH Protocol | 164 |
| 2.2.3.5.5 FISH Protocol | 164 |
| 2.2.3.5.6 Optimisation of FISH Protocol | 165 |
| 2.2.3.6 Patient Details for Chapter 3 | 166 |
| 2.2.3.7 Patient Details for Chapter 4 | 166 |
| 2.2.3.8 Patient Details for Chapter 5 | 168 |
| | |
| Chapter 3: Novel approaches for the preimplantation genetic | 1(0 |
| alagnosis of Myotonic Dystropny (DM1) | 109 |
| 3.1 Alms 2.2 Deculta | 170 |
| 3.2 Results 2.2.1 Droliminary Washing of Familias | 1/1 |
| 3.2.1 Preliminary workup of Families | 1/1 |
| 3.2.2 POD workup | 170 |
| 3.2.2.1 Analysis of Control Single Cells | 179 |
| 3.2.2.2 Analysis of Patient Single Cens | 1/9 |
| 3.2.3 Childar FOD Results | 101 |
| 3.2.3.1 Falling O 3.2.3.2 Family U | 102 |
| 3.2.3.2 raining ri 3.2.4 DOP PCP on Conomia DNA | 104 |
| 3.2.4 DOP-FCK on Genomic DIVA | 100 |
| 3.2.5 DOI -1 CK on Single Cens 3.2.6 CGH Results from DOP-PCR Products | 195 |
| 3.3 Discussion | 100 |
| 3.3.1 Strategy for PGD | 199 |
| 3.3.2 Clinical DM PGD Cases | 202 |
| 3 3 3 Universal PGD Protocol for DM | 202 |
| 3 3 3 1 DOP-PCR on Genomic DNA | 200 |
| 3.3.3.2 DOP-PCR on Single Cells | 207 |
| 3.3.3.3 CGH and DM | 210 |
| 3.3.4 Conclusions and Future Work | 211 |

| <u>Chapter 4:</u> Detailed FISH analysis of day 5 human embryos reveals the | |
|---|-----|
| mechanisms leading to mosaic aneuploidy | 213 |
| 4.1 Aims | 214 |
| 4.2 Results | 214 |
| 4.2.1 Preliminary Work | 214 |
| 4.2.2 FISH Study | 216 |
| 4.2.2.1 FISH Analysis of Controls | 216 |
| 4.2.2.2 FISH Analysis of Embryos | 217 |
| 4.3 Discussion | 226 |
| 4.3.1 Preliminary Work Assessment | 226 |
| 4.3.2 Distinction between Artefacts and True Results | 228 |
| 4.3.3 Group I Embryos | 229 |
| 4.3.4 Group II Embryos | 231 |
| 4.3.5 Types of Mosaicism | 232 |
| 4.3.6 Mechanisms of Aneuploidy Mosaicism | 235 |
| 4.3.7 Conclusions and Future Work | 237 |
| Chapter 5: Analysis of the development of chromosome abnormalities | |
| in human embryos from day 3 to day 5 using CGH and FISH | 241 |
| 5.1 Aims | 242 |
| 5.2 Results | 242 |
| 5.2.1 WGA Optimisation | 242 |
| 5.2.2 CGH Optimisation | 244 |
| 5.2.3 FISH Optimisation | 250 |
| 5.2.4 CGH Analysis of Embryos | 251 |
| 5.2.5 FISH Analysis of Embryos | 253 |
| 5.2.6 Interpretation of CGH and FISH Results | 254 |
| 5.2.6.1 Cytogenetic Events Leading to Mosaicism | 267 |
| 5.3 Discussion | 269 |
| 5.3.1 WGA Optimisation | 270 |
| 5.3.2 CGH Optimisation | 271 |
| 5.3.3 CGH and FISH on Human Embryos | 272 |
| 5.3.3.1 Group 1 Embryos | 275 |
| 5.3.3.2 Group 2 Embryos | 277 |
| 5.3.4 Mosaicism and Chaos | 279 |
| 5.3.5 Mechanisms of Aneuploidy Mosaicism | 280 |
| 5.3.6 Conclusions | 282 |
| 5.3.7 Future Work | 283 |

.

| Chapter 6: Conclusion 6.1 Final Thoughts 6.2 Ethical Considerations Towards PGD | 285 287 293 |
|--|-------------------|
| Chapter 7: References | 295 |
| Chapter 8: Appendix | 353 |

List of Tables

| 1.1 | Classification of human embryonic mosaicism adapted from Delhanty <i>et al</i> (1997) | 112 |
|------|--|-----|
| 1.2 | Chromosomal patterns in cleavage stage embryos as described by Munne and Weier (1996) | 112 |
| 1.3 | Frequency of mosaicism in human preimplantation embryos after karyotypic analysis | 115 |
| 1.4 | Summary of studies carried out on normally developing human cleavage-stage embryos by applying FISH to investigate the extent of mosaicism | 117 |
| 1.5 | Summary of CGH data carried out on human preimplantation embryos revealing the level of mosaicism. | 119 |
| 2.1 | List of FISH DNA probes used in the FISH and CGH/FISH study | 137 |
| 2.2 | List of PCR primers used in the PCR study | 139 |
| 2.3 | Multi-colour FISH-probe combinations | 148 |
| 2.4 | Probe combinations and conditions for all three rounds of sequential FISH | 150 |
| 2.5 | Conditions for the thermal cycler using different polymerase enzymes | 154 |
| 2.6 | Annealing temperatures and primer concentrations for the singleplex reactions for each polymorphic marker | 159 |
| 2.7 | FISH probe combinations and stringency conditions | 165 |
| 2.8 | Patient details for couples undergoing PGD for DM | 166 |
| 2.9 | Patient details for Chapter 4 | 167 |
| 2.10 | Patient details regarding their IVF/ICSI cycle and their embryo details | 168 |
| 3.1 | List of polymorphic markers tested on genomic DNA for family A and their allele sizes | 171 |
| 3.2 | List of polymorphic markers tested on genomic DNA for family B and their allele sizes | 172 |
| 3.3 | List of polymorphic markers tested on genomic DNA for family C and their allele sizes | 172 |

| 3.4 | List of polymorphic markers tested on genomic DNA for family D and their allele sizes | 173 |
|------|---|-----|
| 3.5 | List of polymorphic markers tested on genomic DNA for family E and their allele sizes | 173 |
| 3.6 | List of polymorphic markers tested on genomic DNA for family F and their allele sizes | 174 |
| 3.7 | List of polymorphic markers tested on genomic DNA for family G and their allele sizes | 174 |
| 3.8 | List of polymorphic markers tested on genomic DNA for family H and their allele sizes | 175 |
| 3.9 | List of polymorphic markers tested on genomic DNA for family I and their allele sizes | 175 |
| 3.10 | List of polymorphic markers tested on genomic DNA for family J and their allele sizes | 176 |
| 3.11 | Results of amplification efficiency and ADO in control clumps and single cells | 179 |
| 3.12 | DM1 and D21S1414 amplification results on single buccal cells from the parents of family G | 180 |
| 3.13 | DM1 and D19S112 amplification results on single buccal cells from the parents of family H | 181 |
| 3.14 | PGD results of family 'G' | 182 |
| 3.15 | Confirmatory results from spare embryos after PGD for family 'G' | 184 |
| 3.16 | PGD results of family 'H' | 185 |
| 3.17 | Confirmatory results from spare embryos after PGD for family 'H' | 187 |
| 3.18 | Amplification results for two different individuals on single cell DNA level | 194 |
| 4.1 | Various probe combinations that were attempted in order to find the most efficient 3-colour FISH in both rounds | 215 |
| 4.2 | Probe efficiencies scored in 200 interphase nuclei of each control slide whilst carrying sequential 3-round FISH | 217 |
| 4.3 | Results of FISH analysis of Group I embryos after three sequential rounds of hybridisation with probes for chromosomes 1, 11, 18, X and Y | 218 |

| 4.4 | Aneuploidy mosaicism mechanisms for Group I embryos | 220 |
|-----|--|-----|
| 4.5 | Mechanisms of mosaicism by chromosome for Group I embryos | 221 |
| 4.6 | Results of FISH analysis of Group II embryos after three sequential rounds of hybridisation with probes for chromosomes 1, 11, 18, X and Y | 222 |
| 4.7 | Aneuploidy mosaicism mechanisms for Group II embryos | 224 |
| 4.8 | Mechanisms of mosaicism by chromosome for Group II embryos | 225 |
| 4.9 | Failure rates for each probe in the embryonic nuclei for both Groups | 225 |
| 5.1 | Probe efficiencies scored in 200 interphase nuclei of each control slide whilst carrying 2-round FISH | 249 |
| 5.2 | CGH results from six embryos that did not have any blastomeres available for FISH analysis. The CGH results are presented as excess of GR (green) or RD (red) fluorescence in part of the chromosome or in the whole chromosome | 251 |
| 5.3 | Embryo results showing normal CGH findings (Group 1) followed by sequential FISH analysis | 253 |
| 5.4 | Embryo results showing variable CGH findings (Group 2) followed by sequential FISH analysis | 258 |
| 5.5 | Mechanisms leading to aneuploid mosaicism detected by FISH analysis for the embryos which revealed cells with normal karyotypes after CGH (Group 1). | 265 |
| 5.6 | Mechanisms leading to aneuploid mosaicism for the embryos, which revealed a least one cell with abnormal karyotype after CGH (Group 2). | 266 |

LIST OF FIGURES.

| 1.1 | Demonstration of blastocyst implantation (picture taken from Larsen, 1997) | 30 |
|-----|---|-----|
| 1.2 | Illustration of an ICSI procedure | 34 |
| 1.3 | Segregation at meiosis of a single pair of chromosomes | 38 |
| 1.4 | Eight types of balanced rearrangements are illustrated | 40 |
| 1.5 | Illustration of a FISH probe | 59 |
| 1.6 | Different types of FISH probes | 61 |
| 1.7 | Comparative genomic hybridisation on tumours | 74 |
| 1.8 | Illustrating cleavage stage embryo biopsy. (a): Embryo attached to holding pipette and positioned. (b): Acid Tyrodes is used to create a hole in the zona pellucida. (c): A single blastomere is carefully removed. (d): The embryo following the biopsy (Harper and Doshi, 2000) | 89 |
| 3.1 | Illustration of preferential amplification, stutter peaks and the +A artefact on a dinucleotide repeat region | 178 |
| 3.2 | Analysis results for embryos G1a, G5a and the parents Gm and Gf | 183 |
| 3.3 | Analysis results for embryo H1 (blastomeres H1a and H1b) and the parents Hm and Hf | 186 |
| 3.4 | Analysis results for the DM primer for family E using genomic DNA | 189 |
| 3.5 | Analysis results for the D21S1414 primer for family E using genomic DNA | 190 |
| 3.6 | Analysis results for the D18S535 primer for family E using genomic DNA | 191 |
| 3.7 | Analysis results for the D13S305 primer for family E using genomic DNA | 192 |
| 3.8 | DOP-PCR amplification results run on 1% agarose gel of individual X | 193 |
| 3.9 | DOP-PCR amplification results run on 1% agarose gel of individual Z | 194 |

| 3.10 | Illustration of allele dropout of the D18S535 locus | 195 |
|------|---|-----|
| 3.11 | Illustration of allele dropout of the D19S112 locus | 196 |
| 3.12 | Increased occurrence of 'stutter effect' for the D21S1414 locus | 196 |
| 3.13 | Illustration of the +A artefact affecting the DM locus | 197 |
| 3.14 | CGH results from DOP-PCR amplified single buccal cell of individual C against control male DNA | 198 |
| 4.1 | Results of three sequential rounds of FISH analysis on a single female embryonic nucleus from embryo 9.1 | 219 |
| 4.2 | Results of three sequential rounds of FISH analysis on a single male embryonic nucleus from embryo 13.2 | 223 |
| 4.3 | Results of three sequential rounds of FISH analysis on a single female embryonic nucleus from embryo 15.4 | 223 |
| 5.1 | DOP-PCR amplification results from single cells, clumps of 3-4 buccal cells and genomic DNA | 241 |
| 5.2 | Agarose gel results from (A) Eppendorf and (B) Omnigene thermal cyclers | 241 |
| 5.3 | Illustration of a control CGH experiment between male genomic DNA (red) acting as control against female genomic DNA (green) acting as test | 243 |
| 5.4 | Illustration of a positive control CGH experiment between male single cell (red) acting as control DNA trisomic for 13 (green) acting as test | 244 |
| 5.5 | Illustration of a positive control CGH experiment between male single cell (red) acting as control against DNA trisomic for 18 (green) acting as test | 245 |
| 5.6 | Illustration of a positive control CGH experiment between female single cell (red) acting as control against a single cell trisomic for 21 (green) acting as test | 246 |
| 5.7 | Illustration of a positive control CGH experiment between male single cell (red) acting as control against DNA trisomic for 22 (green) acting as test | 247 |
| 5.8 | Illustration of a positive control CGH experiment between male single cell DNA (red) acting as control against triploid DNA (green) acting as test | 248 |

| 5.9 | Illustration of a control CGH experiment between a male single buccal cell (red) acting as control against blastomere 7.1a (in green) | 255 |
|------|---|-----|
| 5.10 | CGH and FISH analysis results from embryo 7.2 | 261 |
| 5.11 | CGH and FISH analysis results from embryo 8.1 | 262 |
| 5.12 | CGH and FISH analysis results from embryo 9.6 (blastomere a) | 263 |
| 5.13 | CGH and FISH analysis results from embryo 9.6 (blastomere b) | 264 |

LIST OF GRAPHS

| 5.1 | Cumulative results from the 37 embryos that were donated for research | 250 |
|-----|---|-----|
| 5.2 | Cumulative analysis results from 30 embryos (41 blastomeres) | 252 |

Abbreviations

List of Abbreviations

| μg | microgram |
|-----------------|--|
| μl | microlitre |
| μΜ | micromolar |
| 6-FAM | 6-carboxyfluorescein |
| AMA | advanced maternal age |
| ARMS | Amplification Refractory Mutation System |
| ART | assisted reproduction techniques |
| bp | base pair |
| CCD | coupled charged device |
| CGH | comparative genomic hybridisation |
| cm ³ | cubic centimetre |
| CPM | confined placental mosaicism |
| CVS | chorionic villus sampling |
| DAPI | 4', 6-diamidino-2-phenylindole |
| DGGE | Denaturing Gradient Gel Electrophoresis |
| DM | Myotonic dystrophy |
| DMD | Duchenne muscular dystrophy |
| DNA | deoxyribonucleic acid |
| dNTP | deoxyribonucleoside triphosphate |
| EDTA | ethylene-di-amine-tetra-acetic acid |
| FBS | fetal blood sampling |
| FISH | Fluorescent in situ hybridisation |
| FSH | Follicle-stimulating hormone |
| g | gram |
| h | hour |
| HA | heteroduplex analysis |
| HEX | 4,7,2',4',5',7'- hexachloro-6-carboxyfluorescein |
| HFEA | the Human Fertilisation and Embryology Authority |
| GnRH | Gonadotrophin-releasing hormone |
| ICM | inner cell mass |
| ICSI | intracytoplasmic sperm injection |
| IU | international unit |
| IVF | in vitro fertilisation |

| Kb | kilobase |
|--------|--|
| LH | Luteinizing hormone |
| MCC | maternal cell contamination |
| mg | milligram |
| min | minute |
| ml | millilitre |
| mm | millimetre |
| mM | millimolar |
| MNB | multinucleated blastomeres |
| Ν | normal |
| NP-40 | nonidet-40 |
| NT | nuchal translucency |
| PB | polar body |
| PBS | phosphate-buffered saline |
| PCR | polymerase chain reaction |
| PGD | preimplantation genetic diagnosis |
| pmol | picomole |
| pМ | picomolar |
| PZD | partial zona dissection |
| QF-PCR | quantitative fluorescent polymerase chain reaction |
| RFLP | restriction fragment length polymorphism |
| RIF | recurrent implantation failure |
| RM | recurrent miscarriage |
| RNA | ribonucleic acid |
| rpm | rounds per minute |
| SD | standard deviation |
| SDS | sodium dodecyl sulphate |
| SE | standard error |
| sec | second |
| SKY | spectral karyotyping |
| SNP | single nucleotide polymorphism |
| SSCP | Single Strand Conformational Polymorphism |
| SSC | standard saline citrate |
| STR | short tandem repeat |

| TAMRA | N, N, N', N' - tetramehtyl-6-carboxyrhodamine |
|-------|--|
| Taq | Thermus Aquaticus |
| TE | trophectoderm |
| TET | 4, 7, 2', 7' – tetrachloro-6- carboxyfluorescein |
| UPD | uniparental disomy |
| WGA | Whole genome amplification |
| YAC | yeast artificial chromosome |
| ZP | zona pellucida |

Chapter 1 -Introduction

CHAPTER 1

Introduction

1.1 Human Development

Humans, compared with other species, display low fecundity since the chance of conception in any ovulatory cycle is low, being 25% for women under 35 who are trying to conceive and are of proven fertility (Wilcox *et al*, 1988). Despite the advent of *in vitro* fertilisation (IVF) in 1978 (Steptoe and Edwards, 1978) and the major advances in reproductive medicine, there still remains a relatively high failure rate after embryo transfer (Wells and Delhanty, 2000).

1.1.1 Gametogenesis and Meiosis

Gametogenesis is the process of meiosis and cytodifferentiation that converts germ cells into mature male and female gametes (Larsen, 1997). During the fourth week of embryonic development primordial germ cells differentiate within the yolk sac and actively migrate to the posterior body of the embryo. These cells populate the developing gonads and differentiate into the gamete precursor cells which are known as spermatogonia in the male and oogonia in the female. When these cells produce gametes, the process is known as gametogenesis (spermatogenesis in the male and oogenesis in the female).

Meiosis involves a single round of DNA replication followed by two successive chromosome segregations; meiosis I and meiosis II. Meiosis I involves chromosome pairing and recombination between non-sister chromatids, and yields two haploid daughter cells. In meiosis II the duplicated chromosomes divide, yielding four haploid daughter cells. The chromosome number is restored to its diploid state with the fusion of the male and female gametes at fertilisation. The recombination of homologous chromosomes during meiosis leads to exchange of material between maternally and paternally derived chromosomes. This process called crossing over is the source of new combinations of genes in the next generation.

1.1.1.1 Spermatogenesis

Spermatogenesis is a unique process of continuing differentiation since the DNA content of the product is half that of the progenitor cells (Cooke *et al*, 1998). Furthermore, spermatogenesis is a non-stop process involving many mitotic divisions,

possibly around 20-25 per annum, and is initiated at puberty. Spermatogenesis is a relatively rapid process with an average duration of 60-65 days and most steps are closely controlled by a hormone called testosterone. In the initial stages, spermatogonia undergo mitotic divisions, giving rise to primary and secondary spermatocytes, the cell type in which the first and second meiotic divisions occur. The haploid products of meiosis are round spermatids, which elongate during spermiogenesis and with the aid of the Sertoli cells they compact their chromatin into the sperm head and produce further sperm components. Spermiogenesis, a process of sperm cell differentiation, requires about 24 days (Metz and Monroy, 1985; Eddy and O'Brien, 1993). A spermatozoon consists of a head, a midpiece and a very long tail. The head contains the condensed nucleus and is capped by an apical vesicle filled with hydrolytic enzymes. This vesicle, the acrosome, plays a significant role during fertilisation. The midpiece contains mitochondria and is responsible for generating power to the spermatozoon to swim. The tail contains microtubules, which form part of the propulsion system of the spermatozoon. The final step of sperm maturation is known as capacitation and it involves changes mainly in the acrosome that prepare it to release the enzymes required to penetrate the zona pellucida (ZP). Capacitation is thought to take place within the female genital tract.

The process of spermatogenesis is defective for around 2% of the population resulting in abnormally low sperm counts ($<20 \times 10^6$ /ml, oligozoospermia) or total absence of sperm (azoospermia). The discovery that some of these men had a deletion for a region of Yq termed AZF (azoospermia factor) (Tiepolo and Zuffardi, 1976) lead to characterisation of gene families involved in spermatogenesis, with those mapped including Deleted AZoospermia (DAZ) (Reijo *et al*, 1995) and RNA-binding motif (RBM) (Ma *et al*, 1993). Moreover, a study by Mahadevaiah *et al* (1998) suggested that although in mice RBM deficiencies cause sperm abnormalities, in men deletion of the functional copies of RBM is associated with meiotic arrest rather than sperm anomalies.

1.1.1.2 Oogenesis

Oogenesis is a discontinuous process and begins during fetal life. In the female, all primary oocytes that the individual will ever possess are produced during fetal life.

An estimate of 200,000 germ cells are available for the reproductive life span at puberty when recruitment of some of these primordial follicles begins. However, over 99% of follicles undergo atresia rather than ovulation, a degenerative process leading to cell death. Mature ova develop from oogonia by a complex series of intermediate steps. Oogonia originate from primordial germ cells by a process involving 20-30 mitotic divisions early in embryonic life (Siracusa et al, 1985). Between the third and fifth months of intra-uterine life the oogonia mature into primary oocytes which start to undergo meiosis. Shortly after beginning meiosis, however, these cells enter a state of dormancy and meiotic arrest that will persist until after puberty (Baker, 1963). The primary oocytes remain suspended in the meiotic stage at dictyotene until puberty. After puberty, a few oocytes and their enclosing follicles resume development each month in response to the monthly production of pituitary gonadotrophic hormones. These hormones include gonadotrophin-releasing hormone (GnRH), folliclestimulating hormone (FSH) and luteinizing hormone (LH). The hormones above play an indispensable part in controlling folliculogenesis, ovulation and the condition of the uterus. Only one follicle matures fully (others undergo atresia) and undergoes the process of ovulation, which is the expulsion of the secondary oocyte from the follicle. The final maturation stage of the oocyte, where the oocyte will complete meiosis, can be concluded only if it is fertilised by a spermatozoon.

1.1.2 In Vivo Fertilisation

Fertilisation, the process by which the male and female gametes fuse, occurs in the fallopian tube. This multi-step process begins with the specific recognition of complementary receptors on the surfaces of the two gametes. When a spermatozoon encounters an ovulated oocyte it forces its way through the cumulus mass. The zona matrix is made up of at least three major glycoproteins, ZP1, ZP2 and ZP3 where the sperm are able to bind (Longo, 1997; Acevedo and Smith, 2005). If the spermatozoon reaches the ZP, the sperm binds in a human-specific interaction with the glycoprotein sperm receptor molecule (ZP3) and then the sperm acrosome is induced to release degenerative enzymes that allow the sperm to penetrate the ZP. As soon as the spermatozoon penetrates the ZP and reaches the oocyte, the cell membranes of the two cells fuse. This causes a cascade of events including release of specific substances that interact with the ZP in such a way as to alter the sperm receptor molecules,

rendering the ZP impenetrable to other spermatozoa. Moreover, the fusion of the spermatozoon cell membrane with the oocyte membrane causes the oocyte to resume meiosis where the oocyte will consequently release the second polar body.

The chromosomes of the oocyte and sperm are enclosed within female and male pronuclei respectively, which merge to produce the single, diploid, 2N nucleus of the fertilised zygote. The moment of zygote formation is taken as the beginning of embryonic development. The position of the chromosomes in the sperm nucleus has been found to be significant since location could influence paternal gene expression (Foster *et al*, 2005).

1.1.3 In Vivo Preimplantation Embryo Development

Twenty-four hours after merging of the male and female pronuclei, the zygote starts dividing through mitotic cell division known as cleavage. The zygote is subdivided into smaller daughter cells called blastomeres. The first cleavage division divides the zygote in line with the polar bodies, whereas subsequent divisions are asynchronous. The embryo should have divided into 4 cells by 48 hours and 6-8 cells by 72 hours (day 3) post ovulation. By day 4 embryos should reach the morula stage and by day 5-6 embryos form a hollow ball of about 100 cells. At this point the preimplantation embryo, now called a blastocyst, enters the uterine cavity and begins to implant into the endometrial lining of the uterine wall (Figure 1.1).

During these initial stages of embryo development (0-6 days), the zygote travels down the oviduct and undergoes cleavage that subdivides the zygote without increasing its size. In fact, with each division the resulting blastomeres are half the size of the parent cells, and they become increasingly tightly connected as compaction occurs to form the 16-32 cell embryo with an appearance of a small mulberry therefore called a morula (90-120h post-fertilisation) which then becomes a blastocyst (Trounson *et al*, 1982; Gardner and Lane, 2005). The cells of the blastocyst are differentiated into two types: the inner cell mass (embryoblast), which will give rise to embryo proper and its attached membranes and the outer cell mass (trophoblast), which will be the main source of the placenta and related structures. This is a time characterised by intense DNA synthesis and replication.



Figure 1.1. Demonstration of blastocyst implantation (picture taken from Larsen, 1997)

During oogenesis, a supply of maternal proteins and mRNA is produced which, following fertilisation, supports the initial development of the mammalian zygote (Daniels and Monk, 1997; Houghton and Leese, 2004). As development proceeds, these maternally derived proteins and mRNAs are degraded and replaced by those produced by the zygote itself. Several studies have tried to determine the timing at which the 'switch' from maternal to embryonic control of development occurs (Tesarik *et al*, 1986; Braude *et al*, 1988; Leese, 1998). Studies involving the incubation of human preimplantation embryos from the 1-cell stage in a medium containing α -amanatin, which inhibits transcription, have shown that development of human embryos may proceed to the 4- to 8-cell stage, presumably supported only by maternal transcripts (Tesarik *et al*, 1986). However, Braude and colleagues (1988) postulated that beyond this stage embryonic transcripts are required for further development (Gardner and Lane, 2005). This is in contrast to the earlier 2-cell stage embryonic genome initiation in the mouse (Flach *et al*, 1982) and later 16-cell stage in bovine embryos (Telford *et al*, 1990).

Preimplantation embryos in vivo develop in an undefined, complex milieu containing nutrients, including pyruvate, glucose, lactate and amino acids, ions and macromolecules (Houghton and Leese, 2004). The metabolic requirements of the embryo during the early preimplantation period are linked with the activation of the embryonic genome. Prior to embryonic genome activation lactate and pyruvate are the primary energy sources. However, following the activation of the embryonic genome, metabolism shifts to glucose (Gardner and Sutherland, 1996). The precise amino acid requirements of the preimplantation embryo are unknown but they certainly vary before and after genome activation (Gardner, 1998). Several studies have investigated the effect of single or small numbers of amino acids on preimplantation embryo development such as the positive effect of glutamine (Devreker *et al*, 1998) as well as the osmolyte effect of taurine (Dumoulin *et al*, 1997). Houghton and Leese (2004) developed an HPLC-based technique to measure non-invasively the turnover of a physiological mixture of 18 amino acids by single human embryos and found that leucine is depleted completely from the medium.

Recently, the contribution of mitochondrial activity to embryo competence has been investigated in the human (Jansen, 2000; Cummins, 2002; Van Blerkom, 2004). The mitochondrial complement size, mitochondrial DNA copy numbers and defects and stage-specific spatial distribution, have been found to influence the developmental normality and viability of preimplantation embryos (Van Blerkom, 2004).

During embryo growth and more significantly embryo differentiation two types of intracellular junctions have been described as communicative devices between blastomeres, appearing from the 8-cell stage onwards (Fleming *et al*, 2000). Firstly there are the structural tight junctions and desmosomes forming in the outer most cells, anchoring the cells together and forming a permeability seal isolating the interior of the embryo from the external environment. Tight junction construction initiates at compaction and is dependent upon uvomorulin adhesion (Fleming *et al*,

1993). Certain integral and peripheral proteins such as occluding and cingulin (ZO-1) compose the tight junctions (Fleming *et al*, 1993). Desmosomal proteins are points of intercellular contact that 'bolt' the cells together. Secondly, there are the low resistance junctions known as gap junctions allowing intracellular connection between the cytoplasm of two cells so that small proteins and ions can be exchanged (Levin and Mercola, 1998; Cronier *et al*, 2001). Direct transfer through the gap junctions includes metabolites and second messangers (cAMP) (Bennet *et al*, 1997). In human embryos gap junctions are not apparently well developed until the early blastocyst stage when intracellular communication is clearly seen between inner cell mass (ICM) cells (Dale *et al*, 1991). Gap junctions have also been linked with the early generation of left–right assymetry in Xenopus embryos (Levin and Mercola, 1998) as well as differentiation of the human placenta (Cronier *et al*, 2001)

1.1.4 In Vitro Fertilisation

The first successful birth following in vitro fertilisation (IVF) treatment was reported in 1978 by Steptoe and Edwards (Steptoe and Edwards, 1978). This medical breakthrough heralded the beginning of real hope for thousands of infertile couples. For almost three decades all aspects of IVF treatment have constantly been fine-tuned in order to improve pregnancy rates and help couples conceive their dream: a baby. Superovulation regimens have been changed to yield a high number of viable and normal oocytes; sperm preparation methods have been improved; culture techniques and media are continuously evolving in association with our understanding of the needs of the early embryo; moreover, laboratory settings have been re-designed to accommodate the gametes and embryos 'normal' growth. The pregnancy rate varies between centres according to the patient group treated but overall the livebirth per cycle or 'take home baby' rate is about 25%, which declines rapidly as maternal age increases over 36 years (HFEA Annual Report, 2004). Of all pregnancies following IVF, 29% are multiple with a significant proportion of triplets (5%). Obstetric and perinatal complications associated with IVF are largely attributable to the high incidence of multiple pregnancies at risk of pre-term delivery and low birthweight. There is no evidence that fetal abnormalities are more common in children conceived after assisted conception (about 2-3%), compared to their peers conceived normally. In longitudinal studies, child development appears normal (SART, 1992; FIVNAT,

1995). Human births from cryopreserved/thawed embryos do not differ significantly from those from fresh IVF embryos or natural conceptions (Wennerholm *et al*, 1997).

1.1.4.1 Intracytoplasmic Sperm Injection

Micro-assisted fertilisation techniques (MAF) based on IVF technology have been developed since the late 1980s specifically to address the problem of male factor infertility. These include partial zona dissection (PZD) where a breach is made in the zona by mechanical dissection to facilitate sperm entry (Malter and Cohen, 1989), and sub-zonal insemination (SUZI) involving placement of two to five single sperm into the perivitelline space (Laws-King et al, 1987). Both techniques mentioned above have been replaced with intracytoplasmic sperm injection (ICSI). ICSI involves direct microinjection of a single sperm into the cytoplasm of a mature metaphase II oocyte that has been denuded of all surrounding cumulus and corona cells (Palermo et al, 1992; Figure 1.2). This procedure enables men who were previously thought to be irreversibly infertile, the chance to have their own child. However, the clinical application of ICSI has evoked an intense debate on possible associated risks (Cummins and Jequier, 1994; Griffin et al, 2003), because bypassing natural mechanisms of sperm selection has been suspected to support the propagation of gene mutations and chromosomal abnormalities. These concerns are based on the assumption that the ZP stops the penetration of genetically abnormal sperm. Furthermore, ICSI has been widely used as the only method of fertilisation in PGD cases for chromosomal abnormalities and single gene defects (Harper et al, 2000). The underlying reason is to reduce the chance of maternal or paternal contamination during the biopsy procedure since only one sperm is injected into an oocyte where the surrounding cells i.e. cumulus and corona cells have been removed using hyaluronidase

| Figure 1.2. Illustration of an IC | SI procedure. |
|-----------------------------------|---------------|
|-----------------------------------|---------------|

A: 1) Suction applied by a glass holding pipette keeps the oocyte from moving during the injection. A single sperm is picked up in a tiny micro-needle. 2) The needle has a sharp tip and can be gently pushed through the shell of the oocyte and into the cytoplasm and 3) and (B) The sperm is deposited deep inside the oocyte and the empty needle is withdrawn (www.google.com/images)

These concerns appeared to be somewhat confirmed with reports of an increase in sex chromosome aberrations in ICSI babies (Bonduelle *et al.*, 1995; 1998; In't Veld *et al*, 1995; Tournaye *et al*, 1995; Scholtes *et al*, 1998) and the inheritance of paternal chromosome rearrangements (Testart *et al*, 1996; In't Veld *et al*, 1997; Meschede *et al*, 1997) and Y deletions (Kent-First *et al*, 1996). However, other studies have concluded that the chromosomal abnormality rate is not higher in ICSI compared to IVF children (Palermo *et al*, 1996; Engel *et al*, 1996). These discrepancies between studies may be due to differences among laboratories in scoring criteria as well as intrinsic (e.g. age) and extrinsic (e.g. environment) factors may play a role (Griffin *et al*, 2003). Recently, Griffin *et al* (2003) prepared a questionnaire for IVF staff assessing their views whether or not to screen patients for sperm aneuploidy before undergoing ICSI treatment. It was found that staff was not against screening for sperm chromosomal abnormalities and that there would benefits in doing so, however, others argued that most would undergo the ICSI procedure regardless of the screening results (Griffin *et al*, 2003).
1.2 Human Genetic Disease

1.2.1 Chromosome Abnormalities

In 1956 the human chromosome number was identified by Tijo and Levan being 46, and three years later Lejeune *et al*, (1959) connected genetic disease with deviations from the right number of chromosomes. Chromosomal abnormalities comprise cytogenetically detectable alterations in the normal karyotype, and may be either structural involving physical rearrangements of chromosomes, or numerical involving the loss or gain of individual chromosomes (aneuploidy) or whole chromosome sets (polyploidy). Furthermore, in cases where more than one genetically distinct cell line is present, if they arise from one zygote it is called mosaicism whereas if it results form more than one zygote it is referred as chimerism.

Chromosome abnormalities are a major category of genetic disease. They account for a large proportion of reproductive wastage, congenital malformations and mental retardation. The incidence of abnormalities of the autosomes and sex chromosomes are approximately the same (Jackson, 2002). Approximately 60% of spontaneous abortions in the first trimester and 5% of stillbirths are caused by chromosomal abnormalities (Hassold et al, 1986; Eiben et al, 1990). There has been a variety of factors linked to the high early pregnancy failures, with chromosome abnormalities, present or induced, being the most important (Munne et al, 1999). Embryonic aneuploidy, numerical chromosomal abnormality involving the gain or loss of one or more chromosomes, is one of the major causes of reproductive failure in women above 35, at least following IVF (Warburton et al, 1986; Munne et al, 1995a; Munne, 2002). The chance of a fetus being affected with a chromosomal disorder increases with the age of mother from approximately 1:250 at age of 35 to 1:65 at 40 and 1:20 at 46 years of age (Jackson, 2002). It has been postulated that the high level of early embryonic death must contribute significantly to the observed low fecundity (Wells and Delhanty, 2000). The importance of chromosome abnormalities was demonstrated by karyotyping aborted foetuses, which showed that >60% of spontaneous abortions at <12 weeks carried a chromosomal abnormality (Boue et al, 1985).

The major type of chromosomal abnormality found at all gestational ages is aneuploidy, either trisomy or monosomy, for the X chromosome. Trisomies for all the chromosomes have been found in abortuses, including chromosome 1 (Dunn *et al*, 2001) but monosomies with the exception of monosomy X are extremely rare (Boue *et al*, 1985). The incidence of the most common live-born aneuploidies has been shown to increase with maternal age (Nicolaidis and Petersen, 1998). Trisomy 21 is the most common aneuploidy found at birth and the majority of cases are the result of non-disjunction occurring at maternal meiosis (Nicolaidis and Petersen, 1998), whereas only 5% are attributable to mitotic error (Yoon *et al*, 1996). Furthermore, it has been suggested that aneuploidy of gonosomes and disomy 18 by non-disjunction can be due to increasing paternal age (Griffin *et al*, 1995)

1.2.1.1 Numerical Aberrations

Aneuploidy involves the loss or gain of an extra chromosome causing monosomies, trisomies and tetrasomies (Griffin, 1996). The incidence of aneuploidy in humans is an order of magnitude greater than for other animals and most commonly leads to a high rate of mental retardation and pregnancy wastage between 6 and 20 weeks of gestation (Mahmood et al, 2000). Trisomy is the most common class of chromosome abnormality occurring in at least 0.3% of all newborns (Bond et al, 1983) and 25% of spontaneous abortions (Hassold et al, 1985). Approximately 30% of pregnancies result in fetal wastage due to aneuploidy (Wilcox et al, 1988). The most frequently surviving trisomies include trisomy 21 (Down syndrome), 13 (Patau syndrome), 18 (Edwards' syndrome) and aneuploidies involving the sex chromosomes such as monosomy X (Turner syndrome). Trisomy 21 is the most common aneuploidy among liveborns with an incidence of 1:800 births and increases markedly with maternal age, whereas trisomy 16 is the most frequent cause of pregnancy loss (Hassold et al, 1986). The incidence of trisomy 21 at conception is far higher but only 24% survive to term (Hassold et al, 1996). Furthermore, translocations with an emphasis on Robertsonian translocations, accounting for 5% of trisomy 21, are another cause of aneuploidy. Edwards' syndrome has an incidence of 1:10,000 and Patau's syndrome has an incident of 1: 20,000 livebirths (Hassold et al, 1996). Sex trisomies, such as 47XXY or 47XYY, have an incidence of 1:1100 and approximately 1 in 500 liveborn males have either a 47XXY or a 47XYY chromosome constitution (Hassold et al,

1998). The only monosomy reported to show compatibility with life is monosomy XO, termed Turner's syndrome, with an incidence of 1:5,000-10,000 female births. This represents only a fraction of all conceptions with this condition, as 98-99% are spontaneously aborted and only 0.3% are born alive (Hassold *et al*, 1996).

There are essentially three developmental stages when chromosomal defects, and especially aneuploidy, can arise: gametogenesis, fertilisation and embryogenesis (Delhanty et al, 1995). Errors in gametogenesis are usually considered to be meiotic in origin (Delhanty et al, 1995), whereas fertilisation errors are caused due to dispermy which occurs in 1% of spontaneous miscarriages in vivo (Jacobs et al, 1978). Mitotic errors in embryogenesis are implicated in 5-20% of cases, but this may be underestimated as errors in pre-mitotic divisions during gametogenesis may also appear as being of meiotic origin (Antonarakis et al, 1993). Aneuploidy largely arises as a result of an error of chromosome segregation at cell division called nondisjunction (Griffin, 1996) (Figure 1.3). The classical model for the mechanism of nondisjunction in maternal meiosis I is the failure of homologous chromosomes to segregate properly to opposite poles during meiosis resulting in the production of gametes that have an incorrect chromosome complement (Day et al, 1998). Nondisjunction of bivalents leading to loss or gain of dyads during meiosis I results in a disomic and a nullisomic gamete (Nakaoka et al, 1998). According to Angell and colleagues, premature division of the chromosome centromere during anaphase I leads to loss or gain of monads (Angell et al, 1994). Studies in oocytes have revealed that during anaphase I univalents can migrate either as a whole chromosome towards the meiotic spindle, or divide prematurely and move as separate chromatids, with subsequent random segregation (Angell et al, 1994). Both types of nondisjunction have been confirmed in subsequent studies (Cozzi et al, 1999).



A: Non-disjunction occurring during meiosis I at anaphase I will lead to four unbalanced gametes. B: Non-disjunction occurring during meiosis II at anaphase II will lead to two disomic gametes and two unbalanced gametes. (www.google.com/images)

Lamb *et al*, (1996) proposed a two hit system to explain the predominance of maternal MI errors in human trisomy. The first event establishes a susceptible pairing configuration in fetal meiosis whilst the second event is an age-related impairment of the meiotic process, such as defective spindle apparatus (Battaglia *et al*, 1996), which gives an increased risk of nondisjunction. This general interpretation is broad enough to encompass other factors that may contribute to spindle disturbances linked to aneuploidy such as hormonal imbalance and reduced intrafollicular vascularity (Gaulden, 1992; Van Blerkom, 1998).

There is a long list of factors known to affect the process of non-disjunction such as: parental age, recombination of chromosomes, chromosome mover components, differential chromosome susceptibility and certain chemicals. The most significant factor affecting nondisjunction is advanced maternal age and subsequent aberrant recombination (Griffin, 1996). Errors in maternal meiosis I account for 75% of trisomy 21 cases (Antonarakis *et al*, 1998) and all (100%) of trisomy 16 (Hassold *et al*, 1996). In trisomy 18 cases most arise due to errors occurring during maternal meiosis II (Fisher *et al*, 1995).

Polyploidy (multiple copies of all chromosomes) is the second most common group of chromosomal abnormalities resulting in a spontaneous abortion in the first or second trimester as progression to term is rare (Book and Santesson, 1960; Cassidy *et al*, 1977). Errors in cell division can result in polyploidy. However, it is mainly due to polyspermic fertilisation (Hassold *et al*, 1980; Angell *et al*, 1986, Zaragoza *et al* 2000), with dispermy being the most common cause of triploidy. Complete non-disjunction at MI or MII can also lead to triploidy. In digynic cases this meiotic failure can be caused by non-extrusion of a polar body which then becomes incorporated into the embryo (Penrose and Delhanty, 1961; Zaragoza *et al*, 2000). Following DNA replication any meiotic or post-zygotic mitotic failure in cytokinesis can cause tetraploidy or higher orders of ploidy

1.2.1.2 Structural Aberrations

Structural chromosomal abnormalities arise from chromosome breakage with subsequent reunion in a different configuration. They can be balanced, where the chromosome complement is complete with no loss or gain of genetic material or unbalanced, where the chromosome complement contains an incorrect amount of genetic material. Structural abnormalities include translocations (reciprocal or Robertsonian), deletions, inversions, insertions, ring chromosomes and isochromosomes (Figure 1.4). Structural chromosomal aberrations may be familial or de novo in nature with an estimated mutation rate of $1/1 \times 10^{-3}$ (Jacobs, 1981) and are seen in 5% of spontaneous abortions (Hassold et al, 1986; Eiben et al, 1990) and between 0.2-0.6% of livebirths, the majority balanced (Hook and Hamerton, 1977; Nielsen and Wohlert, 1991).

Chapter 1 -Introduction



Figure 1.4: Eight types of balanced rearrangements are illustrated, Chromosomal translocations a) Reciprocal, b) Robertsonian, c) Insertions (i) interchromosomal, (ii) intrachromosomal between-arm and (iii) intrachromosomal within-arm. d) Chromosomal inversions (i) pericentric and (ii) paracentric. e) Ring chromosome. A, B – Normal homologue. der, ins, inv, r – Rearranged chromosome. (Therman *et al.*, 1989)

Translocations are the most common structural abnormality in man (approximately 1.6 in 1000 newborns) and carriers of balanced translocations show an increased risk of having chromosomally abnormal offspring in addition to infertility and spontaneous abortions due to the formation of unbalanced gametes during meiosis (Iwarsson *et al*, 2000). Impaired spermatogenesis has also been frequently seen in male balanced translocation carriers. Most of the male translocation carriers that have reduced fertility are considered to be phenotypically normal (Sutton, 1980).

Most structural anomalies are only identified if a couple presents with a history of chromosomally abnormal conceptions. These reproductive problems arise due to abnormal meiotic segregation and/or production of recombinant chromosomes at gametogenesis, the degree of which varies according to the chromosomes involved, the size and location of the rearranged segments, whether complete synapsis is achieved and the position of crossover events and the individual (Gardner and Sutherland, 1996). Furthermore, the fact that certain couples carrying a chromosomal rearrangement suffer only a slight reduction in fertility and others a detrimental one is a consequence of the multifactorial nature governing the production of viable or nonviable gametes as the production of unbalanced gametes of a particular type is influenced by the size of the imbalance (Cohen *et al*, 2000). In addition, it is generally known that infertile men have a higher incidence of de novo chromosomal abnormalities in sperm than fertile men.

1.2.1.2.1 Robertsonian Translocations

Robertsonian translocations occur with a prevalence of ~ 1 in 1000 in the general population (Gardner and Sutherland, 1996). Robertsonian translocations result from the breakage of two acrocentric chromosomes (13, 14, 15, 21 and 22) at or close to their centromeres, with subsequent fusion of their long arms (Figure 1.4b). The short arms of each chromosome are lost, which is of no clinical importance and the total chromosome number is reduced to 45. Since there is no loss or gain of essential genetic material this is a functionally balanced rearrangement (Conn et al, 1999). The most common forms are non-homologous, involving two different acrocentric chromosomes such as two chromosomes from the D or G group (13, 14 and 15 or 21 and 22 respectively) or a D group and G group chromosome. There are three ways for the formation of a Robertsonian translocation: fusion at the centromere (centric fusion), giving a monocentric chromosome, a whole arm reciprocal translocation, with breakage in one short arm and one long arm giving a monocentric chromosome, and finally, union following breakages in both short arms giving a dicentric chromosome (or after the suppression of one centromere, a monocentric). The most common Robertsonian translocation is between chromosomes 13 and 14 and makes up 75% of all Robertsonian translocations (Gardner and Sutherland, 1996).

1.2.1.2.2 Reciprocal Translocations

Reciprocal translocations are characterised by the exchange of chromosomal genetic material between two non-homologous chromosomes (Figure 1.4a). They are very common in the general population, occurring in 1 in 500 livebirths (Hook *et al*, 1977). Carriers of these translocations are nearly always phenotypically normal, as no loss or gain of genetic material is involved (abnormalities can arise if breakpoints disrupt important genes) (Scriven *et al*, 1998). Translocations are detected when the patient presents with recurrent pregnancy loss or phenotypically abnormal offspring due to the production of unbalanced gametes (Scriven *et al*, 1998; Iwarsson *et al*, 2000). The risk of miscarriages, stillbirths or chromosomally unbalanced live births with multiple congenital abnormalities depends on the probability of different types of unbalanced gametes being produced and, after fertilisation of such gametes, on the probability of different types of unbalanced embryos being able to survive (Van Assche *et al*, 1999). Impaired spermatogenesis is also frequently seen in male balanced translocation carriers.

1.2.1.2.3 Deletions, Inversions, Insertions, Ring Chromosomes and Isochromosomes

Deletions involve loss of a chromosomal segment, resulting in chromosome imbalance. A carrier of a chromosomal deletion (with one normal homologue and one deleted homologue) is monosomic for the genetic information on the corresponding segment of the normal homologue. The clinical consequences generally reflect haploinsufficiency (Jackson, 2002) and appear to depend on the size of the deleted segment and the number and function of the genes that it contains (Gardner and Sutherland, 1996). Deletions have an incidence of approximately 1 in 7000 births and can be terminal or interstitial. Deletions may originate by chromosome breakage and loss of the centric fragment or due to unequal crossing over between misaligned homologous chromosomes or sister chromatids. Furthermore, it has been suggested that deletions are also derived due to abnormal segregation from a balanced translocation or inversion (Iwarsson *et al*, 2000).

Insertions or insertional translocations involve three breaks; the first two breaks release an interstitial segment, which is then inserted into the gap created by the third break (Figure 1.4c). If a single chromosome is involved, this can be described as a

shift. In the interchromosomal insertions a segment of one chromosome is inserted interstitially into another chromosome. The insertion could be direct when the inserted segment is inserted with the same orientation with respect to the centromere, or it could be inverted when the orientation towards the centromere is reversed (Harnden et al., 1985). Interchromosomal insertions are rare chromosomal rearrangements with an incidence of 1:80000, of which nearly 80% are referred because of congenital abnormalities and mental retardation (Van Hemel and Eussen, 2000). In the case of intrachromosomal insertion either within or between an arm crossing over can occur in one or the other insertion loop following complete synapsis. The risk of having a child with an unbalanced karyotype when carrying an intrachromosomal insertion has been estimated to be 15% by Madan and Menko (1992). Insertions are amongst rearrangements implying the highest reproductive risk. This risk is greater for the small segment insertion than the large segment insertion however in the case of the meiotic recombination product carrying a duplication there appears to be no clear relationship between the phenotype and the length of the duplicated segment (Wilson et al, 1985).

Inversions involve a two-break rearrangement in a chromosome and the segment formed by the breaks is reversed. If the inverted segment includes the centromere then the inversion is pericentric [Figure 1.4d(i)], if however, the centromere is not involved the rearrangement is known as paracentric [Figure 1.4d(ii)]. Excluding the common inversions of heterochromatin 1gh, 9gh, and 16gh generally considered normal variants, inversions occur in about 1 in 2-5000 births (Gardner and Sutherland, 1996). In pericentric inversions the smaller the inverted fragment the greater the chromosomal imbalance and the likelihood of miscarriage of the conception. Conversely, with a large inverted segment, the unbalanced regions will be small and offspring carrying the recombinant chromosomes may be viable (Daniel, 1981). In the case of paracentric inversions the crossing-over during synapsis at meiosis will lead to the formation of recombinant chromosomes that are either acentric, and incompatible with survival or dicentric and unstable during cell division and therefore are not compatible with life either (Worsham, 1989). Therefore, the risk of the birth of an abnormal child resulting from a carrier parent of a balanced paracentric inversion is almost non-existent.

Ring chromosomes are formed by breakage in both arms of a chromosome, with fusion of the points of the fracture and loss of the distal fragments (Figure 1.4e). Thus, this would result in a partial monosomy for the distal short arm and the distal long arm. Ring chromosomes are often unable to complete mitotic division so that it is not unusual to find the ring chromosome in only a proportion of cells. In carriers of this abnormality where reproduction is an option, meiotic and mitotic disturbances presumably lead to high levels of gametic chromosomal imbalance and impaired fertility, although reports of stable familial rings show that this is not inevitable (McGinniss *et al*, 1992).

Isochromosome is a chromosome where one arm is missing and the other is duplicated in a mirror-image fashion, thus a person carrying an isochromosome has partial monosomy of one arm and partial trisomy of the other. The basis for isochromosome formation is not precisely known. However, the most probable explanation is that the centromere has divided transversely rather than longitudinally. The isochromosome for the long arm of the X chromosome, i(Xq), is the most common isochromosome and is found in some individuals with Turner's syndrome.

1.2.1.3 Mosaicism

Mosaicism is defined as the presence in an individual or in a tissue of two or more cell lines, which differ in their genetic constitution but are derived from the same zygote. The phenomenon of mosaicism will be analysed in section 1.6

1.2.1.4 Chromosomal Abnormalities in Human Gamete Studies

1.2.1.4.1 Karyotyping Studies

Karyotypic analysis of 710 oocytes (with or without polar bodies) revealed that cytogenetic abnormalities are closely associated with abnormal gametes, parthenogenesis, cytoplasm immaturity and division asynchrony after fertilisation (Benkhalifa *et al*, 1996). Angel (1997) after analysis of 200 oocytes by karyotyping postulated that there are two major factors regarding oocyte aneuploidy and maternal age: a) Vulnerable recombinants arise by chance distribution of recombination events at the pachytene stage of meiosis in fetal life and b) the physical structure of all chromosomes in the oocyte gradually deteriorates during the extended dictyate stage.

1.2.1.4.2 FISH Studies

Studies on sperm have tried to approach aneuploidy and especially trisomy by employing FISH (Hixon *et al*, 1998). Sperm studies are advantageous in studying nondisjunction over female studies since they have: a) accessibility to the origin of aneuploidy, the gametes, b) limitless supply of the products of meiosis, making possible to detect subtle differences in nondisjunction among different individuals and c) examination of nondisjunction in a setting that is unaffected by the maternal age effect. Recently, Tempest *et al* (2004), was able to find an association between infertile male individuals (of variant infertility) and sperm disomy for the sex chromosomes and chromosome 21. Overall, it was concluded that oligozoospermic male have significantly elevated levels of sex chromosome disomy only, whereas asthenozoospermic males have elevated levels of disomy 21 and not the sex chromosomes.

Studies on metaphase II oocytes have revealed an unexpected class of chromosome abnormalities, which manifest as single chromatids, rather than whole chromosomes comprising a dyad of chromatids. Mahmood et al (2000), determined that extra material in 127 oocytes and 57 polar bodies (PB) was seen in chromosomes 13, 16, 18, X and notably 21, but none were observed in chromosomes 1 and 9. Furthermore, the previous group found clues suggesting that an additional mechanism of maternal aneuploidy operating at anaphase II, might be the underlying reason of prematurely separated chromatids in the metaphase II oocyte Dailey et al (1996), used FISH on oocytes to investigate the involvement of specific chromosomes in aneuploidy, especially chromosomes 13, 15, 16, 18, 21, 22, X and Y. Sequential multi-probe FISH was employed for chromosomes 1, 9, 12, 13, 16, 18, 21 and X by Cupisti and colleagues (2003) to investigate the chromosomal status of unfertilised polar bodies and associated polar bodies. The same group found that mechanisms leading to aneuploidy include nondisjunction of whole univalents. In a recent study by Martini and co-workers (2000) on oocytes analysed by FISH, it was revealed that almost half (44% overall) of the material examined was aneuploid with chromosome specific patterns. Furthermore, Durban et al (1998) described an oocyte spreading procedure which was able of allowing detection of aneuploidy as well as structural chromosomal abnormalities during PGD using FISH.

1.2.1.5 Chromosomal Abnormalities in Embryo Studies

1.2.1.5.1 Karyotyping Studies

Ideally, cytogenetic investigation would involve analysis of metaphase chromosomes from the blastomeres or polar bodies (Ruanvutilert *et al*, 2000b). Karyotyping embryos has been applied using conventional techniques including culture synchronisation, disruption of the mitotic spindle and G-banding (Papadopoulos *et al*, 1989; Clouston *et al*, 1997; Clouston *et al*, 2002), however it is considered to be technically challenging.

Cytogenetic analyses of human preimplantation embryos have revealed extremely high levels of chromosome imbalance at this early stage of development (Papadopoulos *et al*, 1989; Clouston *et al*, 1997). Many embryos are karyotypically normal, however >50% of embryos are reported to carry a variety of chromosomal abnormalities including aneuploidy and chromosome breakage, polyploidy (addition of one or more complete haploid complements) and haploidy (one set of chromosomes instead of two) (Voullaire *et al*, 2000). These chromosome abnormalities arise during gametogenesis and/or fertilisation. Alternatively, they can be present in a proportion of nuclei due to post-zygotic errors during mitosis. This percentage indicates that pregnancy loss is frequently a result of natural selection against chromosomal aberrations

Limited, useful information can be extracted from processing intact day 2 and day 3 embryos (Angell *et al*, 1986; Papadopoulos *et al*, 1989; Jamieson *et al*, 1994) after analysis using banding techniques. Jamieson *et al* (1994) published a study of 816 embryos, however was able to karyotype only 195 (23.9%), finding 19.1% being aneuploid, 3.5% being tetraploid, 2.3% being triploid and only 0.6% being haploid. This was due to reduced quality of metaphases and G-banding and long colcemid exposure times resulting in highly contracted chromosomes. Clouston and colleagues (1997) devised a novel and inexpensive method for obtaining better quality G-banded metaphases from human blastocysts. They were able to provide full ploidy analysis on 64% (55/86) of the embryos and reported high levels of diploid embryos (67%); however, this percentage included uniformly abnormal karyotypes. Several studies reported chaotic findings while trying to detect mosaic and non-mosaic chromosome

abnormalities (Jamieson *et al*, 1994; Clouston *et al*, 2002) and were able to show that mitotic non-disjunction is significant for the production of aberrations in human preembryos (Papadopoulos *et al*, 1989). Furthermore, karyotypic studies on preimplantation embryos were among the first to reveal significant levels of tetraploidy, usually mosaic of about 19-23% (Papadopoulos *et al*, 1989; Clouston *et al*, 1997). Papadopoulos *et al* (1989) observed variable levels of structural chromosome damage, which was later confirmed by Clouston and co-workers (1997), including chromosome branching, chromosome breakage, anomalous chromatid pairing and apparent interchanges. However, it has been suggested that damage in early cell divisions may be a consequence of the ovarian stimulation stage of the IVF process rather than a characteristic of early preimplantation embryos per se (Eibling and Colot, 1985).

1.2.1.5.2 FISH Studies

studies of chromosomal abnormalities Although karyotyping on human preimplantation embryos had already provided important results, it was the advent of FISH that enabled the examination of every cell within the embryo that revealed the true extent of chromosomal abnormality in human development. The widespread use of IVF for treatment of infertility has stimulated research into chromosomal abnormalities in human preimplantation embryos. Most of the studies on embryos have been carried out using FISH. Nowadays, several FISH protocols have been devised for complex translocations (Simopoulou et al, 2003) as well as aneuploidy detection of up to 13 chromosomes (Abdelhadi et al, 2003). FISH studies have been able to demonstrate that the incidence of aneuploidy, haploidy, polyploidy and mosaicism in embryos is much higher than the incidence observed in clinically established pregnancies (Bielanska et al, 2002a). FISH has been carried out in arrested (Munne et al, 1995a; Bahce et al, 1999), frozen-thawed (Munne et al, 1997; Iwarsson et al, 1999) and fragmented (Munne et al, 1994a,b; 1995b) embryos, as well as embryos of good quality (Harper et al, 1995; Delhanty et al, 1997; Bielanska et al, 2000).

An essential part of FISH is its ability to be carried out on the same sample sequentially or as it is know in the circles of FISH laboratories, re-FISHing. The need

for sequential rounds of FISH on the same tissue/cell came about due to the lack of fluorescent colours of DNA probes thus the number of chromosomes that can be examined simultaneously e.g. in a blastomere is limited to a maximum of five (Gianaroli *et al*, 1997b) or six (Munne *et al*, 1998c). Liu and colleagues (1998a) were able to devise an accurate FISH protocol analysing more than six chromosome in 3 consecutive FISH rounds in a clinical setting. It has been shown that by using centromeric or telomeric probes in all three rounds of FISH, an efficiency of 88-94% and 87-96% respectively can be achieved (Liu *et al*, 1998b).

Delhanty and co-workers (1993) showed that surprisingly even apparently normally developing IVF embryos were often chromosomally abnormal, in agreement with the original karyotyping data (see section 3.1.1). Several factors have been proposed from studies on embryos for this level of abnormality. These factors including maternal age, embryo culture conditions (Munne *et al*, 1995), multinucleation and freezing of human embryos (Laverge *et al*, 1998), ovarian stimulation regimes and patient specificity (Ruangvutilert *et al*, 2000a) have been shown to affect chromosomal constitution of embryos. Almost all studies carried out on preimplantation embryos regarding chromosome abnormalities have tried, through their FISH results, to explain the mechanisms behind these aberrations. Staessen *et al* (1998) whilst analysing the genetic constitution of multinuclear blastomeres using M-FISH, revealed that more than half of bi-or multinuclear blastomeres were abnormal, however, 45% of the those blastomeres had a diploid status.

1.2.1.5.3 CGH Studies

More recently high rates of aneuploidy have been reported whilst applying metaphase CGH in human preimplantation embryos (Wells and Delhanty, 2000; Voullaire *et al*, 2000; Wilton *et al*, 2001; Malmgrem *et al*, 2002; Voullaire *et al*, 2002; Wilton *et al*, 2003; Trussler *et al*, 2004) as well as polar bodies and MII oocytes (Wells *et al*, 2002; Gutierez-Mateo *et al*, 2004). CGH on single cells from human preimplantation embryos has been performed and has provided the opportunity to assess the copy number of all chromosomes and thus the genuine abnormality and mosaicism level at this stage (Wells and Delhanty, 2000).

The application of single cell CGH, although being laborious, provides the opportunity to assess all the chromosomes and thus identify the true level of

mosaicism (Wells and Delhanty, 2000, Voullaire et al, 2000, Wilton et al, 2001; Trussler et al, 2004). CGH has allowed the investigation of chromosomal aberrations in human preimplantation embryos generated by IVF. Wells and colleagues (1999) revealed a WGA protocol that was powerful enough to generate sufficient quantities of DNA from a single cell to perform CGH as well as >90 independent amplification reactions. Instantly, two published studies showed the extent of chromosomal abnormalities in preimplantation human embryos. Voullaire et al (1999) obtained results for 63 blastomeres from 12 cleavage stage embryos and found that the proportion of chromosomally normal embryos was 25%, lower than that found from FISH studies (Delhanty et al, 1997; Munne et al, 1998d). Wells and Delhanty (2000), in a similar study found high levels of chromosomal mosaicism, non-mosaic aneuploidy, and chromosome brakeage. Furthermore, it was postulated that there was a low number of uniformly normal embryos (normal chromosome numbers in every cell), which may have superior developmental potential, but their low frequency might explain the relatively low success rates in assisted conception in humans (Wells and Delhanty, 2000). In a recent study which combined CGH and FISH, the high numbers of abnormal embryos were thought to have arisen as a result of culture artefact or inadequate cell cycle surveillance, rather than meiotic error (Trussler et al, 2004). CGH has also been performed in the context of PGD for an uploidy screening (see section 1.5.2.3).

1.2.1.5.4 SKY-FISH and M-FISH

To overcome the limitations of FISH, it has been suggested that metaphase transformation by cell fusion would allow the differentiation between normal, balanced and unbalanced karyotypes (Munne and Cohen, 1998). Methods to obtain metaphase stage chromosomes by interphase conversion have been published by two teams. They are based on the fusion of blastomeres to bovine eggs or mice zygotes (Verlinsky and Evsikov, 1999; Willadsen *et al*, 1999). The Willadsen approach has been used for two PGD clinical cases for translocations resulting in chromosomally normal offspring (Willadsen *et al*, 1999) whereas the Verlinsky and Evsikov approach has been performed in the context of clinical PGD in 19 patients (Verlinsky and Evsikov, 1999). The methodology of Verlinsky and Evsikov (1999) was performed by fusing individual human blastomeres with enucleated or intact mouse zygotes. After blastomere-cytoplasm fusion, heterokaryons were fixed at metaphase of the first

cleavage division or treated with okadaic acid to induce premature chromosome condensation. The effectiveness of the proposed technique with blastomeres from day 3 diploid embryos was reported to be 91%, since 63 metaphases were obtained from 69 blastomeres (Verlinsky and Evsikov, 1999). However, interphase conversion is extremely labour intensive and has to provide an interpretable metaphase spread from a single cell, which in turn will render a number of metaphases unsuitable due to poor chromosome morphology, overlapping chromosomes and even loss of chromosomes (Harper and Wells, 1999). Furthermore, this technique might cause ethical tension in some countries where it would be impossible to fuse human blastomeres or polar bodies with enucleate oocytes from other species (Harper and Wells, 1999).

Multiplex-FISH (M-FISH) (see section 1.3.2.2.2) and spectral karyotyping are whole genome screening techniques that have been successfully applied for cytogenetic diagnostics of constitutional chromosomal abnormalities (Speicher et al, 1996; Schröck et al, 1996; 1997). Both techniques utilise 24 chromosome-specific paint probes labelled with different combinations of fluorochromes, thus all chromosomes can be analysed at one time. Schröck et al (1996) employed a dedicated customdesigned imaging spectrometer system, in a method known as spectral karyotyping or SKY-FISH. This finely samples each pixel for all fluorescence across the spectrum simultaneously producing a set of interferograms, before data processing to form a spectral image. Analysis is based on comparison of the interference pattern for each pixel with stored data on chromosome interference spectra; the pixel is then allocated to a matching chromosome (Schröck et al, 1996). SKY-FISH has been employed to successfully identify particular de novo supernumerary marker chromosomes as well as de novo unbalanced structural rearrangements, proving to be of beneficial role for diagnostic and counselling purposes, due to its reliability and speed (Haddad et al, 1998). SKY-FISH has already been used to examne chromosomes from oocytes and polar bodies (Marquez et al., 1998). Sandalinas et al (2002), whilst carrying out SKY-FISH on fresh non-inseminated oocytes found increased non-disjunction, increased balanced and unbalanced predivision in the group of patientsars of over 35 years of age, Furthermore, all types of metaphase spreads, including metaphases obtained after nuclear conversion, can be analysed with M-FISH or SKY-FISH. Hence, SKY-FISH was the preferred method to analyse blastomeres and polar bodies where the nucleus

had been converted to a metaphase (Evsikov & Verlinsky, 1999; Willadsen et al., 1999).

However, both SKY-FISH and M-FISH require expertise and are unable to detect small deletions, duplications and translocations (Kirchhoff *et al*, 2000). Also, they are relatively time-consuming and fail to yield metaphase chromosomes of suitable quality for routine diagnostic purposes. Ultimately, a method capable of a full chromosome analysis in single cells without the need for metaphases has been devised, based on the technique known as comparative genomic hybridisation (CGH).

1.2.2 Single-Gene Defects

Analysis of syndromes with Mendelian inheritance patterns has identified over 7000 single-gene defects. They are categorised as autosomal dominant (>3000), autosomal recessive (>3000) and X-linked disorders (>1000). A series of these mutations would be expected to affect the function of essential housekeeping genes and may therefore be lethal either in preimplantation development following the initiation of embryonic gene transcription and loss of maternally inherited products or in post-implantation development. Most *de novo* mutations causing inherited disease are paternal in origin (McKusick, 1992). This is thought to be related to differences between spermatogenesis and oogenesis. Since, in the females, the mitotic expansion of primordial germ cells continues until the late gestation when oogonia enters meiosis and arrest at the dictyate stage of meiosis I until menstrual cycles are initiated at puberty (see section 1.1.1.2). However, in males mitotic division of spermatogonia is continuous throughout life (see section 1.1.1.1). Therefore, the number of mitotic divisions preceding gametogenesis is much greater in the male which consequently may increase the risk of replication errors (Delhanty and Handyside, 1995).

An autosomal dominant trait is one which manifests in the heterozygous state, i.e. in a person possessing both the abnormal or mutant allele and the normal allele. Dominantly inherited diseases are generally caused by mutations resulting in gain of function protein alterations (Ranum and Day, 2002). Autosomal dominant traits can involve only one organ or part of the body and can have a multisystemic effect (pleiotropy) which has been seen in myotonic dystrophy (Ranum and Day, 2004) as

well as tuberous sclerosis (Ess *et al*, 2005). During this study a triplet repeat dominant disorder will be discussed, namely myotonic dystrophy 1.

1.2.2.1 Myotonic Dystrophies

Steinert as well as Batten and Gibb in 1909 identified myotonic dystrophy (DM [MIM 160900 and MIM 602668]) as a multisystemic disorder. Myotonic dystrophy (DM1) also know as dystrophia myotonica is the most common form of inherited muscular dystrophy seen in adults, with an overall incidence of approximately 1 in 8000 (Shaw and Harper, 1989). DM1 shows autosomal dominant inheritance with anticipation and an early onset form with rather different clinical features. The clinical picture of DM1 is well established but exceptionally variable (Harper, 1989). One of the striking features of this disorder is the variability of phenotype both within and between families (Harper, 1989).

Shortly after genetic testing became available, a second multisystemic disorder was identified referred to as either proximal myotonic myopathy (PROMM) or myotonic dystrophy type 2 (DM2) (Ricker *et al*, 1994; Thornton *et al*, 1994). The DM2 mutation was linked to a 3cM region of the 3q21 and it is caused by a transcribed but untranslated CCTG repeat expansion located in intron 1 of the zinc finger protein 9 (*ZINF9*) gene (Ranum *et al*, 1998; Liquori *et al*, 2001).

During this study only DM1 was investigated therefore for simplicity DM1 will be referred to as DM.

1.2.2.1.1 Clinical Pathology of DM

In contrast to most forms of muscular dystrophy, in individuals suffering from DM the clinical features are not limited exclusively to the neuromuscular system. Patients with DM have been found to suffer in other organs in addition to myopathy i.e. bone, skin, eyes, gastrointestinal organs and the endocrine system. Most commonly persons with DM present in their adult life with slowly progressive weakness and myotonia. The latter term refers to tonic muscle spasm with prolonged relaxation, which can manifest as a delay in releasing the grip of shaking hands. Other clinical abnormalities can include cataract, cardiac conduction defects, disturbed gastrointestinal peristalsis,

frontal balding and testicular atrophy. The age of onset can be very variable and the disorder usually runs a benign course. In the congenital form affected babies present at birth with hypotonia, talipes and respiratory distress, which can prove life threatening (Harper, 1975). Children who survive tend to show a lack of facial expression with delayed motor development and mild retardation. The congenital form of DM is mostly maternally transmitted (Brook *et al*, 1992; Zeesman *et al*, 2002). The phenomenon of anticipation in which the disease symptoms become more severe and age at onset earlier in successive generations, is often manifested in a family producing a congenitally affected child (Brook *et al*, 1992).

Early diagnosis is based on clinical symptoms including myotonia, muscular weakness and atrophy, cataract and hypogonadism. In contrast to other dystrophies, DM attacks the distal and then proximal muscles of extremities. Originally the diagnosis of DM patients was based on electromyography, which has now been superseded by mutation analysis (Shaw and Harper, 1989).

1.2.2.1.2 Genetics of DM

Initial biochemical studies failed to identify the defective protein in DM, although some have highlighted a link with abnormalities in calcium transport (Seiler and Kuhn, 1970), membrane fluidity (Butterfield *et al*, 1974) as well as apamin receptor expression (Renaud *et al*, 1986). It was Brook and colleagues (1992) that employed positional cloning strategies in order to identify the CTG triplet repeat that undergoes expansion in DM patients.

The molecular genetic defect of DM involves an unstable expansion of a CTG repeat at the 3' untranslated region (exon 15) of the DM gene on the long arm of chromosome 19 (19q13.3) (Brook *et al*, 1992; Fu *et al*, 1992). The number of repeats relates to the age of onset and the severity of the disease (Harley *et al*, 1993). Normal individuals possesses 5-35 repeat copies, patients with 36-49 copies are said to have the premutation, patients with 50-150 copies are mildly affected, patients with 100-1,000 copies are severely affected and the most severe or neonatal form have 2,000 copies or more [The International Myotonic Dystrophy Consortium (IDMC), 2000]. The mutant gene is almost always transmitted from the mother; however, paternal transmission cases have been reported (Nakagawa *et al*, 1994; de Die-Smulders *et al*, 1997). Variation in the DM triplet repeat has been observed in sperm and somatic cells of the same individual (Jansen *et al*, 1994), between muscle and lymphocyte cells (Anvret *et al*, 1993) as well as in affected identical twins (Dubel *et al*, 1992). The length of the expanded CTG repeat remains fairly homogenous in different tissues during embryonic development (Mankodi and Thornton, 2002). However, during postnatal life a marked variability of repeat length emerges in different cells and tissues of an individual (Thornton *et al*, 1994). This somatic heterogeneity has been found to increase with age (Thornton *et al* 1994). During PGD cases for DM significant increases have been found in the number of repeats in embryos from affected female patients and in their immature and mature oocytes, whereas, in spermatozoa and embryos from affected male patients, smaller increases have been detected (De Temmerman *et al*, 2004).

The estimated risk of any heterozygous woman with DM to have a congenitally affected child is 4-9%, though the risk is significantly increased during conception of a second child to 20-37% (Koch *et al*, 1991). Another feature of DM is the reported tendency for healthy individuals who are heterozygous for DM alleles in the normal size range to preferentially transmit alleles which are greater than 19 CTG repeats in size (Magge and Hughes, 1998). Such phenomenon might help maintain the mutant DM alleles in the population. However, recently Zunz *et al* (2004) found no evidence of statistically significant deviation of the frequency of transmission of the mutated alleles from the 50% expected in autosomal dominant disorders and suggested that previous studies showed ascertainment bias.

1.2.2.1.3 The Myotonin-Protein Kinase

The myotonin-protein kinase (DMPK) gene, which consists of 15 exons occupying over 13,000 bases of genomic DNA, encodes a polypeptide of 624 amino acids that functions as a member of a protein kinase family (Shaw *et al*, 1993). At present it is not known how or indeed if the DMPK gene causes muscular weakness or other clinical problems. This is due to the fact that the expanded repeat lies in the 3' untranslated region of the DMPK gene, a region that is transcribed into RNA, but not translated into protein. Rather surprisingly it has been demonstrated in mice that the over-expression and under-expression of the DMPK gene does not display the typical features of DM (Harris *et al*, 1996). However, it has been suggested that the RNA

produced by the mutant *DMPK* gene may influence the cellular processing of RNA produced by other genes i.e. the gene encoding DM locus-associated homeodomain protein (DMAHP) which is located immediately downstream to the CTG repeat (Harris *et al*, 1996). It has been recently proposed that the RNA produced from the mutant DMPK gene has a toxic effect on muscle cells (Mankodi *et al*, 2000). Cell culture models have been used to demonstrate that mutant DM mRNA takes on a gain-of-function and inhibits myoblast differentiation (Amack *et al*, 1999). Although the molecular mechanism(s) by which this mutant mRNA disrupts myogenesis is not fully understood, recent findings suggest that anomalous RNA-protein interactions have downstream consequences that compromise key myogenic factors (Amack and Mahadevan, 2004). Several studies have attempted to determine the effects of the expansion in nearby genes. Fillipova *et al* (2001) suggested that changes in chromatin structure, and the resulting misregulation of genes in the vicinity could be relevant to the defects of muscle and brain development that occur in congenital DM patients.

1.2.2.1.4 Mechanisms of DM Pathogenesis

Most dominant disorders are caused by the expression of an abnormal protein with an altered function, it has not been clear how the multisystemic clinical features of dominantly inherited DM could be caused by a trinucleotide repeat that did not affect the protein coding portion of a gene (Tapscott, 2000). Four different mechanisms have been proposed for the explanation of the DM pathogenesis (Ranum and Day, 2004):

1. Haploinsuffiency of *DMPK*; Earlier studies suggested that alteration of the *DMPK* expression might cause the multisystemic features of the disease (Fu *et al*, 1992), however, studies in *DMPK* knockout mice showed only a very mild, late-onset myopathy (Jansen *et al*, 1996).

2. Haploinsuffiency of *SIX5* and Neighboring genes: A second mechanism is that the expanded repeat affects the expression of multiple genes in the region. It has been proposed that the mutation interferes with expression of multiple genes in the DM1 region, possibly through regional effects produced by repeat-induced alterations in chromatic structure (Otten and Tapscott, 1995). Studies using *Six5* knockout mice provided some support to this mechanism (Klesert *et al*, 2000) however was not conclusive.

3. RNA pathogenesis: Another proposed mechanism has been that the enlarged CUGcontaining transcripts accumulate as intranuclear foci and disrupt cellular function which came from a transgenic mouse model (Mancodi *et al*, 2000).

4. Additive model: This model of DM1 has been suggested where all the above mechanisms contribute to disease pathogenesis (Larkin and Fardaei, 2001).

1.3 Molecular Cytogenetic Techniques

Extensive research in the field of chromosome analysis has lead to constant development of laboratory techniques available to detect abnormalities. Nowadays, scientists demand techniques to be powerful, accurate, reproducible and able to provide results in the shortest amount of time possible. Clinical cytogenetics is the study of chromosome structure and behaviour in relationship to clinical syndromes. Karyotyping was the first technique allowing analysis and identification of all 23 pairs of chromosomes with respect to number and morphological structure. Karyotyping is still considered the most widely applicable technique in clinical and research cytogenetics. Fluorescent in situ hybridisation (FISH) is a relatively novel technique, which combines cytogenetics and molecular technology. FISH combines DNA hybridisation techniques with fluorescent microscopy, allowing direct visualisation of a specific DNA sequence onto metaphase chromosomes, interphase nuclei or DNA strands. The combination of FISH and karyotyping has lead to the development of innovative techniques such as spectral karyotyping (SKY-FISH) (Schrock et al, 1996) and multi-fluorochrome karyotyping (M-FISH) (Speicher et al, 1996) (see section 4.1). The Polymerase Chain Reaction (PCR) using oligonucleotide primers and a thermostable DNA polymerase (Saiki et al, 1985), is another key technique designed to enrich a DNA sample for a specific fragment, amplifying it to a level at which it can be visualised and subjected to further genetic analysis. However, the most novel technique which has united traditional cytogenetics and modern molecular techniques is comparative genomic hybridisation (CGH). CGH allows in a single hybridisation the copy number of all 23 pairs of chromosomes to be assessed in situations that do not allow standard methods of chromosomal analysis to be used (Kallionemi et al, 1992).

1.3.1 Karyotyping

1.3.1.1 Karyotyping Principles

Tijo and Levan in 1956 discovered that the normal human somatic cell contains 46 chromosomes rather than 48. The methods they used, with certain modifications, are now employed in all cytogenetic laboratories to analyse the chromosome constitution of an individual, known as a karyotype. Any tissue with living nucleated cells which undergoes division can be used to study human chromosomes. These cells are cultured with nutrients for 3 days to stimulate the T lymphocytes (for whole blood), chorionic villi (for CVS) and amniotic fluid cells to divide e.g. phytohaemogglutinin. While in culture, colchicine is added to prevent the formation of the spindle, thus arresting cell division during metaphase. During metaphase the chromosomes are maximally condensed and therefore more easily visible.

There are several different staining methods that can be utilised to identify individual chromosomes: 1) G or Giemsa banding is the most common method used. The chromosomes are treated with trypsin, denaturing their protein content and giving each chromosome a characteristic pattern of light and dark bands. 2) Q or Quinacrine banding gives a similar banding pattern to G-banding and requires examination of the chromosomes under ultraviolet fluorescence. 3) R or Reverse banding shows reverse light and dark banding patterns to G-banding since the chromosomes are heated before staining. 4) C or Centromeric Heterochromatin banding preferentially stains highly repetitive DNA sequences such as centromeres and heterochromatic regions, by treatment of chromosomes with acid followed by alkali prior to G-banding.

1.3.1.2 Karyotyping Applications

Karyotyping has been widely used in prenatal diagnosis for genetic analysis of amniocentesis and chorionic villus sampling (CVS) in high risk women. Karyotyping is widely in the analysis of different types of cancers e.g. sarcoma (Ravi and Wong, 2006), leukaemia (Scandura, 2005) and myelomas (Schilling *et al*, 2005) Furthermore, it has been used for research purposes on oocytes, polar-bodies, spermatozoa and embryos (Martin *et al*, 1986; Plachot *et al*, 1988; Zenzes *et al*, 1991; Jamieson *et al*, 1994; Clouston *et al*, 1997; 2002). Karyotyping was amongst the first

technique to be used for embryo analysis. Karyotyping has been used extensively for investigating the mechanisms of nondisjunction in human oocytes and polar bodies.

1.3.1.3 Technical Limitations of Karyotyping

Karyotyping provides valuable information at the research and diagnostic level. Karyotyping has been considered the 'mother of all techniques'. However, technical difficulties have created the need to develop newer and superior techniques. Problematic fixation methods resulting in loss of chromosomes and over dispersed or poorly spread cells which are not in a single focal lane restricting the potential of analysis are the main disadvantages (Martin et al, 1996). Moreover, long colcemid exposure times are able to increase the mitotic index but this produces highly contracted chromosomes that exhibit chromatid separation and G-band poorly compromising the information obtained (Jackson et al, 2002). Furthermore, karyotyping for prenatal diagnosis requires the isolation of metaphase chromosomes and takes 7-10 days for a final result (Reid et al, 1992), which can increase the emotional burden on the patient (Goel et al, 1998). The most important limitation during G-banding is that only a few metaphases can be obtained hence only a small proportion of the cells can be analysed. In the study of preimplantation embryos basic technical difficulties of handling individual embryos compromises the potential quality of preparations and hence limits the amount of information that can be gained (Harper et al, 1995).

1.3.2 FISH

1.3.2.1. FISH Principles

The combination of chromosome banding techniques with the revolutionary technique of *in situ* hybridisation has aided in the research and diagnosis of structural abnormalities not previously possible. *In situ* hybridisation was first introduced in 1969 using radiolabelling detection (Pardue and Gall, 1969; John *et al*, 1969). In 1986, Pinkel *et al* and Cremer *et al* reported FISH using non-radioactively labelled probes. Fluorescent *in situ* hybridisation (FISH) is a molecular cytogenetic technique for enumerating chromosomes combining DNA hybridisation techniques with fluorescent microscopy. FISH utilises fluorescently labelled DNA probes, which are essentially single stranded DNA fragments capable of binding to their complementary sequences in an interphase nucleus or a metaphase spread that are fixed and located on a microscope slide (Pinkel *et al*, 1986; Tonnies, 2002) (Figure 1.5).

Figure 1.5. Illustration of a FISH probe

A single-stranded DNA fragment, tagged with fluorescent molecules, which is able to bind to its complementary sequence on a chromosome

(www.google.com/images)

idded to the probe DNA, anding and blocking the repeat sequences and

FISH probes are quite sensitive DNA molecules being able to detect regions as small as 0.5 kb on metaphase chromosomes (McNeil and Ried, 2000). Cloned DNA fragments are usually the constitution of FISH chromosomal probes, which are characterised by the ability to anneal only to their complementary DNA sequences (Pinkel et al, 1986). Probes employed for FISH can be directly or indirectly labelled. The indirectly-labelled probes are labelled with a hapten such as biotin and digoxygenin, which are detected using affinity reagents labelled with fluorophorelinked immunoglobin reagents. This immunocytochemical detection of 'indirectlylabelled' probes allows the flexibility to amplify a weak FISH signal, necessary by sequential applications and is especially useful for visualising smaller probes. However, indirect labelling is a time-consuming process with non-specific background fluorescence potentially making interpretation difficult. The development of directly labelled probes, where the fluorochrome is attached to the probe itself, reduced the time-frame of the FISH experiments and further improved the specificity of the fluorescent detection (Wiegant et al, 1993). The most common fluorophores include those based on fluorescein (green) or rhodamine (red), although this selection has widened considerably in the last few years with the introduction of a new generation of commercial fluorophores such as the Cyanine, Blue, and Gold dyes (Yurov et al, 1996).

There are three types of probes that are used in FISH with respect to their size, location and how they bind onto the chromosome.

- a. Locus specific probes (Figure 1.6a), are probes able to bind to a unique sequence of a particular chromosome. Often called band-specific, they require more than six hours to hybridise and can be used in interphase and metaphase nuclei. Depending on the type of clone, these probes may range from small cDNA fragments in plasmids, to much larger blocks of genomic DNA in YACs. When using probes containing unique DNA sequences (as well as repetitive non-specific sequences) suppression of the hybridisation of the non-chromosome specific repetitive DNA is accomplished by chromosome in situ suppression (CISS) using co-hybridisation with labelled probe and unlabelled repetitive sequences (Lichter *et al*, 1988). Currently, CoT-1 [coefficient of temperature (Ekong and Wolfe, 1998)] DNA is added to the probe DNA, binding and blocking the repeat sequences and consequently allowing the probe to bind to the unique sequence that it targets.
- b. Repetitive probes (Figure 1.6b&c), bind to repetitive sequences located in the centromeric, heterochromatic and telomeric regions of a chromosome. These include the widely used α -satellite/centromeric probes (b), which hybridise specifically to the centromeres of individual chromosomes (Willard, 1985). Heterochromatic probes include those binding to the heterochromatin usually situated below the centromere, seen in chromosomes 1, 9 and Y. Telomeric probes specifically recognize the repetitive sequence TTAGGG, and can be used for the simultaneous identification of all telomeres. They can be used in interphase and metaphase chromosomes and require only one hour to hybridise. The signals obtained are very bright and the whole FISH procedure can be performed within two hours (Harper *et al*, 1994c)
- c. Chromosome paints (Figure 1.6d), paint the whole chromosome and can only be used in metaphase preparations. They contain a cocktail of DNA sequences hybridising to an entire chromosome or chromosome arm and are derived either from a pool of clones picked from a chromosome specific library (Fuscoe *et al*, 1989) or from flow-sorted chromosomes amplified by PCR (Vooijs *et al*, 1993).
- d. Whole genome probes. Label the whole chromosome whilst the cells are in metaphase in one fluorescent colour either red or green. These types of probes are widely used whilst carrying out CGH on metaphase chromosomes.





(a) Illustration of a locus-specific probe labelled in Spectrum Green; (b) Illustration of a centromeric probe labelled in Spectrum Orange. Illustration of two sub-telomeric probes; one for the short arm and one for the long arm, labelled in Spectrum Green and Spectrum Orange respectively and (c) Illustration of chromosome paint. (Harper, *et al*, 2000)

FISH protocols vary between laboratories; however the same basic principles apply. The tissue/cells are obtained and placed on a microscope slide by some sort of fixation, either by methanol/acetic acid fixation or by applying adhesive agents on the actual slide prior to fixation. The slide with the tissue/cells is pre-treated with proteolytic enzymes and /or RNase digestion which render the nucleus of the sample accessible to the DNA probe. Denaturation of the slide and probe follows, causing the double DNA strands to separate. The sample is then left to re-anneal under optimal conditions, the length of time depending on probe type. The stringency with which the probe binds is controlled during post-hybridisation treatment by varying the temperature, ionic strength and concentration of formamide washes to destabilise and remove imperfectly base-paired probe/target duplexes. In this way only probe bound stably to its complementary target remains. After hybridization the slide is observed under ultraviolet light of the proper wavelength, and any region where the labeled DNA fragment has bound fluoresces. The scoring criteria that are most frequently applied are adopted from Hopman et al (1991) which state that signals must be a minimum of a signal's width apart in order to be scored as 2 individual signals.

TEN is an alterapt to attain cannot of the infertile states of these patients withermore, the effects of physical inspirately such as charactheology or reducibeitagy

1.3.2.2. FISH Applications

1.3.2.2.1 Interphase FISH applications

FISH has allowed chromosome enumeration to be performed on interphase nuclei without the need for culturing cells or preparing metaphase spreads. Therefore, the FISH technique has overcome the problems that have arisen during karyotypic analysis and it has been possible to investigate the presence of specific numerical and structural abnormalities in sperm, oocytes, and embryos as well as prenatal samples (chorionic villi and amniotic fluid). FISH has allowed the study of the level of chromosomal abnormalities and understanding of the mechanisms of chromosomal mosaicism in human preimplantation embryos (see section 1.6), a phenomenon whereby two or more cell lines are present in a tissue, which differ in their genetic constitution but are derived from the same zygote.

The growing availability of commercially produced probes and probe-cocktails has enabled FISH to become an outstanding routine diagnostic tool in the cytogenetic laboratory. The wide selection of existing centromeric probes allows determination of chromosome-specific ploidy on solid tumours, fresh or archived biopsy specimens, and on cultured and uncultured preimplantation, prenatal, postnatal and hematologic samples. In malignant solid tumors, genomic imbalances resulting from either gains or losses of whole chromosomes or segments of chromosomes are frequently observed using FISH. FISH has enabled the observation and further understanding of how the presence of the large genomic abnormalities are believed to contribute to the disease phenotype by alteration of normal patterns of gene expression. Examples of these genomic aberrations include the aneuploidy seen using FISH in breast, bladder and prostate tumors (Hopman *et al*, 1991; Cornelisse *et al*, 1992) as well as chromosomal rearrangements seen in hematopoietic cancers (Hilgenfeld *et al*, 1999).

In gynaecology and fetal medicine, multicolour FISH has assisted in investigating fertility for example in women with endometriosis (Shin *et al*, 1997) and men suffering from Klinefelter's syndrome (Guttenbach *et al*, 1997) where both conditions cause infertility. The chromosomal constitutions of these patients were analysed by FISH in an attempt to attain causes of the infertile status of these patients. Furthermore, the effects of clinical treatment such as chemotherapy or radiotherapy

on chromosomes has been examined by FISH, whereby a sperm sample was taken before and after treatment and its effects of studied (Monteil *et al*, 1997).

The FISH technique has been found to be highly effective for rapidly determining the number of specified chromosomes in interphase cells. FISH thus seemed to be especially appealing for the prenatal detection of chromosomal aberrations. The first developed probes were derived from DNA of flow-sorted whole chromosomes and used for prenatal diagnosis of trisomy 13, 18 or 21 by Kuo *et al* (1991) on uncultured amniotic fluid cell nuclei. Interphase FISH with a specific probe set for familial rearrangements also allows rapid exclusion of an unbalanced translocation in the fetus of a balanced translocation carrier (Kilby *et al*, 2001).

Aneuploidy and chromosomal rearrangements diagnosed by FISH can be performed on preparations made from around 1.0-1.5ml of amniotic fluid or chorionic villus samples (sections 1.4.1 and 1.4.2). Many prospective studies have been carried out examining the efficacy and accuracy of the FISH probes and subsequently the FISH technique and have revealed more than 90% accuracy (Hulten *et al*, 2003). The most significant point of doing this is a rapid diagnosis without having to wait for culture and karyotype. It has also been shown the risk of over- or under-diagnosis of aneuploidy for the target chromosomes is limited by interphase FISH compared to the 'gold standard' of karyotyping. FISH on uncultured interphase amniocytes and chorionic villus cells, might aid diagnosis of constitutional mosaicism (Feldman *et al*, 2000). Quilter *et al* (2001) stipulated that interphase FISH may be of special advantage with respect to the problem of confined placental mosaicism in chorionic villus sample.

1.3.2.2.2 Metaphase FISH applications

Speicher *et al*, (1996) attempted combinatorial labelling of twenty-four chromosome paints using five fluorophores, reporting the first study of Multiplex-FISH (M-FISH). Following hybridisation, a monochrome CCD camera with multiple optical filters was used to capture each channel sequentially before merger to form a final image. The basis for M-FISH analysis depends on the presence or absence of probe signal for each fluorophore at each pixel that is then compared to the labelling strategy to identify the chromosome (Speicher *et al*, 1996). However, many groups have tried to

maximise the number of probes that can be used employing combinatorial or Boolean labelling i.e. combining labelled probes in different proportions much earlier that 1996. Nederlof and colleagues (1990) attempted FISH utilising three fluorophores to detect four chromosomes and Wiegant *et al* (1993) developed combinatorial labelling, to the detection of six loci on a single chromosome. Combinatorial labelling remains a strategy that although in theory many combinations are possible, in practice the number of combinations is dependent on the sensitivity of the imaging system.

Chromosomal microdeletions are associated with a number of syndromes including Di-George and Williams syndromes. Microdeletion probes consist of a probe specific for the locus or region of deletion associated with the microdeletion syndrome, as well as a control probe for accurate identification of the chromosome of interest. Their use has proved very useful for the detection of a range of microdeletion syndromes, providing rapid same-day results (Ligon *et al*, 1997). Idiopathic mental retardation accounts for approximately 36% of the moderately to severe handicapped population and are associated with subtle subtelomeric chromosomal rearrangements (Raynham *et al*, 1996). Although these abnormalities remain undetected by traditional cytogenetics, they can be readily detected using a panel of FISH probes specific for the subtelomeric regions of all non-acrocentric chromosomal arms (Knight *et al*, 1999)

1.3.2.3. Technical Limitations of FISH

FISH has been hailed as one of the most robust and efficient techniques however, it has certain limitations. There are conflicting reports as to the true sensitivity and specificity of FISH diagnoses with ranges of diagnostic accuracy reported between 70-98% (Christensen *et al*, 1993; Ward *et al*, 1993; Pergament *et al*, 2000). The availability of FISH probes has increased in recent years but it is still not possible to obtain probes that cover all different parts of chromosomes rendering some specific chromosomal rearrangements impossible to diagnose. FISH is unable to distinguish between samples of balanced and normal chromosomal complements in structural aberrations in interphase nuclei (Warburton, 1991; Conn *et al*, 1998). The lack of fluorescent dyes also limits the scope of diagnosis since not many different chromosomes can be analysed simultaneously.

1.3.3 Polymerase Chain Reaction

1.3.3.1 PCR Principles

In 1983, Dr Kary Mulis conceived a novel concept of performing a test-tube process of repetitive DNA synthesis, termed polymerase chain reaction (Mulis *et al*, 1986; Mulis, 1990). The polymerase chain reaction (PCR) is used to amplify a sequence of DNA using a pair of oligonucleotide primers each complementary to one end of the DNA target sequence. These are primers extended towards each other by a thermostable DNA polymerase in a reaction cycle of three steps: denaturation, primer annealing and polymerisation or primer extension. PCR is a powerful molecular technique for quickly amplifying a particular DNA fragment to a stage that can be further analysed by other methods (Saiki *et al*, 1985). The success of PCR in achieving this objective has enabled it to become one of the most important methods in genetic testing having numerous applications in basic research and medicine.

The first step, denaturation, is able to separate the two strands. The temperature is then reduced depending on the primer length and sequence to allow the specific primers to anneal. After annealing, the temperature is increased for optimal polymerisation which uses a mix of deoxynucleoside 5'-triphospates (dNTP's) and a polymerase, which are substrates for DNA synthesis and Mg²⁺. In the first polymerisation step, the target is copied from the primer site for various distances on each target molecule until the beginning of cycle 2, when the reaction is heated up again in order to denature the newly synthesized molecules. In the second annealing step, the other primer can bind to the newly synthesized strand and during polymerisation can only copy till it reaches the end of the first primer. Therefore, at the end of cycle 2, some newly synthesized molecules of the correct length exist. The successive cycles of DNA synthesis result in an exponential amplification of the target DNA sequence leading to a 10^5 - 10^6 -fold increase in the amount of target DNA. If the PCR technique is 100% efficient, one target molecule would become 2^n after n cycles. In practice, 20-30 cycles are carried out for a PCR experiment, however, during single-cell PCR up to 50 cycles have been reported (Piyamongkol et al. 2001a).

PCR primers usually need to be about 18-30 base pair long (bp) and to have similar G+C contents in order to anneal to their complementary sequences at similar temperatures. The primers are designed to anneal on opposite strands of the target sequence so that they will extend towards each other by addition of nucleotides to their 3'-ends. Short target sequences amplify more easily, so often this distance is less than 500bp, however, with optimisation, PCR can amplify fragments over 10kb in length (Reiss and Cooper, 1990).

Thermostable DNA polymerases have been mostly isolated and cloned from a number of thermophilic bacteria. The most common is *Taq* polymerase, which is isolated from *Thermus Aquaticus*. It survives the high temperature denaturation step, usually 95° C, having a half-life of more than two hours at this temperature. However, Taq polymerase is known to introduce errors when it copies DNA since it does not have associated 3' to 5' proofreading exonuclease activity. Therefore, other thermostable DNA polymerases have been introduced with greater accuracy including AmpliTaqTM.

1.3.3.1.1 Multiplex PCR

The technique of amplifying multiple loci concurrently is termed multiplex PCR (Findlay *et al*, 1995a, b; Pertl *et al*, 1996; Sherlock *et al*, 1998). This can facilitate the diagnosis of a specific genetic disease or of multiple diseases since it provides information for multiple loci at the same time. One example of multiplex PCR in a non-single-cell PCR is the amplification of nine regions of the DMD gene in a single reaction (Chamberlain *et al*, 1988). Each multiplex PCR needs to be optimised for the combination of primers involved with regard to primer concentrations, annealing temperature, the reaction buffer utilised and number of cycles carried out. Multiplex PCR is usually achieved using F-PCR, as several primers can be multiplexed together employing different fluorescent dyes even if the product ranges overlap each other (Kimpton *et al*, 1993). Strategies for multiplex PCR reactions have been widely reported in prenatal (see section 1.4.5) and PGD (see section 1.5.3) setups.

1.3.3.1.2 Whole Genome Amplification

An alternative method to multiplex PCR that can be used for amplification of low copy numbers of the entire genome is whole genome amplification (WGA) (Zhang *et*

al, 1992; Snabes *et al*, 1994). With WGA a single genome can be amplified numerous times, thus providing sufficient DNA templates for many independent PCR amplifications (Wells and Sherlock, 1998). There are three types of WGA described:

Primer Extension Preamplification (PEP) is a WGA technique whereby at least 70% and 90% of the genome is amplified more than 30 times according to Zhang *et al*, (1992), and Wells *et al* (1998) respectively. PEP can be viewed as essentially a prediagnostic PCR treatment, since the PEP product can be subsequently used in a further PCR to diagnose a specific disorder. It has been suggested that PEP is unsuitable for clinical PGD (Findlay, 2000) (see section 1.6.3.3).

Degenerate Oligonucleotide Primed PCR (DOP-PCR) has also been considered as a valuable WGA technique (Telenius *et al*, 1992). DOP-PCR amplifies a similar proportion of the genome to PEP, but to a much more significant level. DOP-PCR followed by comparative genomic hybridization (CGH) analysis of tumour DNA has been described by several investigators (Speicher *et al*, 1993; James and Varley, 1996). A significant drawback of DOP-PCR and of other WGA techniques is that amplification of repetitive DNA sequences, such as short tandem repeats is error-prone when performed on WGA products (Wells *et al*, 1998). In some studies over 50% of fragments amplified are found to differ from their expected size, presumably due to slippage of the DNA chain during product generation (Wells and Sherlock, 1998). The low annealing temperatures that characterize all WGA protocols may underlie this problem (Wells *et al*, 1998).

Multiple Displacement Amplification (MDA) is a rolling-circle amplification (Lizardi *et al*, 1998) method that was developed for amplifying large circular DNA templates such as plasmid and bacteriophage DNA (Dean *et al*, 2002). Using $\varphi 29$ DNA polymerase and random exonluclease-resistant primers, DNA is amplified in a 30°C reaction not requiring thermal cycling. This is made possible in part by the great processivity of $\varphi 29$ DNA polymerase, which synthesises DNA strands 70kb in length (Blanco *et al*, 1989). Dean and colleagues (2002) showed that MDA-generated DNA product is >10kb, and its performance can be demonstrated for a variety of applications including single nucleotide polymorphism (SNP) analysis, restriction fragment length polymorphism (RFLP), Southern blotting, DNA sequencing as well

as comparative genomic hybridisation (CGH). However, in the same study the capability of MDA was not assessed on single cells. More recently Spits *et al* (2006), was able to carry out 22 locus-specific PCR's from minute quantities of DNA.

1.3.3.2 Methods of Detection

PCR is designed to enrich a DNA sample for one specific fragment, amplifying it to a level where it can be visualised and subjected to further genetic analysis (Wells and Sherlock, 1998). The methods of mutation analyses most commonly used are referred to as the 'scanning' methods, which can be used for diagnostic purposes and applied for searching of uncharacterised mutations. Scanning methodologies are optimal for diagnosis of inherited disorders caused by a heterogeneous spectrum of mutations.

1.3.3.2.1 Fluorescent PCR

The traditional methods for visualizing PCR products following electrophoresis include ethidium bromide or silver staining. Both techniques benefit from nested PCR to ensure sufficient amplified fragments for visualization. The utilisation of radioactively labelled primers or nucleotides can be employed for visualization for greater sensitivity however it has been characterised as a time consuming technique as well as harmful. Furthermore, all of the above techniques are limited by their qualitative nature, with quantities of DNA as well as products of a similar size being indistinguishable (Wells and Sherlock, 1998). The advent of fluorescent PCR technology (Hattori et al, 1992) has enabled more far reaching diagnostic applications to be considered especially at the single cell level. Fluorescent-PCR (F-PCR) is a modification of PCR technology using fluorescent primers and an automated DNA sequencer and has improved both PCR accuracy and sensitivity (Ziegle et al, 1992). The application of oligonucleotide primers attached to fluorescent molecules gives rise to amplified products labelled with a fluorescent dye. When these F-PCR products migrate under electrophoresis to the position where the laser bisects the products, the fluorescent molecules are activated by the laser and give a signal of a specific wavelength, subsequently detected by a CCD detector and analyzed by computer software. The fluorescence dye is detected at a much lower threshold level than conventional agarose or acrylamide gel analysis. The size analysis is as precise as a single base pair difference. It is of major importance that fluorescent PCR is

compatible with heteroduplex analysis (Rommens *et al*, 1990; section 1.3.3.2.3), single strand conformational polymorphism (SSCP) (Ellison *et al*, 1993; section 1.3.3.2.4) as well as amplification refractory mutation system (ARMS) (Sherlock *et al*, 1998; section 1.3.3.2.5).

The main advantage of fluorescent PCR is its sensitivity compared to conventional PCR techniques. F-PCR is able to detect minute amounts of the fluorescent product such that for a single cell only 35-40 cycles of PCR are required. This in turn eliminates the need for nested PCR, so with one round of PCR amplification products even from a single cell can be analysed. F-PCR has been used extensively for prenatal (Hulten *et al*, 2003) and preimplantation genetic diagnosis worldwide (Sermon *et al*, 1998a; Piyamongkol *et al*, 2001a and b; Dean *et al*, 2001). It has been postulated that F-PCR assays are powerful adjuncts to conventional cytogenetic techniques and can be applied for the rapid and accurate prenatal diagnosis of the most frequent aneuploidies (Pertl *et al*, 1999). Fluorescent PCR has been used for analysis of amniotic fluid and chorionic villus sampling (see section 1.4.5) as well as multiple single gene defects in the context of PGD (see section 1.5.3)

1.3.3.2.2 Denaturing Gradient Gel Electrophoresis

Denaturing gradient gel electrophoresis (DGGE), involves mixing a radiolabelled single stranded DNA probe with double stranded DNA being screened, which has been previously heated to make it single stranded. The mixture is electrophoresed on a denaturing gradient gel. The differences in DNA are detected by virtue of altered melting characteristics, which affect the DNA strand as it passes through the polyacrylamide gel with an increasing concentration of a denaturant. Recently, DGGE has been used for the detection of the major histocompatibility complex (Knapp, 2005).

1.3.3.2.3 Restriction Endonuclease Digestion

Differences in DNA sequence e.g. caused by mutation can be shown using restriction endonucleases to digest the DNA. The restriction endonucleases are bacterial enzymes which recognise specific DNA sequences and cleave the DNA strand near the recognition site. Therefore, if the mutation and DNA sequence are known, a restriction endonuclease will be chosen and will cleave the normal strand, whereas the mutant will remain undigested. The results are visible after electrophoresis.

1.3.3.2.4 Heteroduplex Analysis

Heteroduplex analysis (HA) is based on the ability of heterozygous DNA to anneal, after denaturation, to its complementary strands which will associate recreating the original homoduplexes. Hybrid molecules will form generated by the association of partially complimentary strands from mutant and normal alleles. The latter heteroduplexes have an area of mismatch, since at the site of mutation the two alleles differ in DNA sequence. The area of mismatch does not anneal and retards heteroduplex migration during electrophoresis. Hence, heterozygocity of a sample can be established by the existence of homoduplexes and heteroduplexes (White *et al*, 1992). Recently, sensitivity has been improved by 97% by combining a high-resolution sieving matrix and nucleosides as additives (Weber *et al*, 2006)

1.3.3.2.5 Single Strand Conformational Polymorphism

Single strand conformational polymorphism (SSCP) is able to detect point mutations as well as DNA polymorphisms in DNA fragments ranging in size from 100-500bp (Orita *et al*, 1989). This is accomplished by denaturing double stranded PCR products into single strands of DNA. These single strands will form stable, sequence-specific conformations and alterations in base sequences (e.g. different alleles) will give distinct conformations, which will subsequently migrate in different rates to nondenaturing electrophoresis gel. Hence, different alleles can be distinguished (see section 1.3.3.1.1). However, SSCP is inefficient at detecting mutations with increasing size of the PCR product tested which limits its ability to analyses PCR products >300bp (Frayling, 2002)

1.3.3.2.6 Amplification Refractory Mutation System

Amplification refractory mutation system (ARMS) is based on the annealing of three different allele-specific oligonucleotides. With ARMS these oligonucleotides serve as primers for PCR and are not directly detectable. For example in a nested PCR reaction one of the primers can be designed to contain the mutation site during the outer reaction. The other two primers can be part of two different inner amplifications, one containing the normal and the other the mutant allele. By amplifying both normal and
mutant alleles a 'safety net' is created in case one of the inner reaction does not amplify (see section 1.3.3.1.1).

1.3.3.3 PCR Applications

PCR has become an essential tool in molecular biology as an aid to cloning and gene analysis. A variety of PCR-modified techniques have been developed for different purposes, including forensic analysis, PND and PGD of single gene disorders. More sophisticated and modern techniques have been devised to better suit the needs of its applications.

The advent of PCR has helped in the genetic linkage of genes and diseases. For example, a tetranucleotide repeat region around the gene for myelin basic protein, after amplification, demonstrated the presence of eight different alleles among 14 multiple sclerosis families and was used to exclude a linkage between this candidate gene and the disease (Rose et al, 1993). The usefulness of PCR in the elucidation and understanding of genetic diseases can be measured by the number of diseases that have been identified and diagnosed by utilising the PCR technology including myotonic dystrophy (Caskey et al, 1992), cystic fibrosis (Collins, 1992) as well as inherited colorectal cancer (Lunch et al, 1994) and breast cancer (Mikki et al, 1994). Furthermore, PCR can facilitate the diagnosis of infectious diseases caused by viral, bacterial, fungal, protozoal and other infectious agents. Using PCR on peripheral blood mononuclear cells from HIV affected individuals, HIV-1 sequences were detected in 100% of specimens that contained discernible virus by culture (Ou et al. 1988). However, PCR has provided extreme aid in the identification and diagnosis of genetic diseases including β-thalassaemia, Tay-Sach's disease, Duchenne muscular dystrophy in addition to deducing and cloning of cancer-causing genes such as the *Rb* gene associated with retinoblastoma, APC gene associated with adenomatous polyposis coli and mismatch repair genes for nonpolyposis colorectial cancer.

PCR has an advantage over FISH in prenatal diagnosis of aneuploidies, since it can accommodate smaller sized samples (Hulten *et al*, 2003). Quantitative Fluorescent PCR (QF-PCR) has been described as more amenable to automation, and a large number of samples can be handled simultaneously, allowing substantial economy of

scale (Grimshaw *et al*, 2003). It has been postulated that it is less time consuming and labour intensive compared to FISH in a prenatal or preimplantation diagnostic setup (Findlay, 2000).

1.3.3.4 Technical Limitations of PCR

Because of the extremely sensitive nature of the PCR process, contamination from carrying over of previously amplified PCR products in the same laboratory can lead to significant problems. The problem of contamination affects both the diagnostic and quantitative PCR and may be the most formidable obstacle in PCR application. Contamination can occur either by intersample contamination during sample processing and reagent contamination from carryover (section 1.5.3.5.1). Precautionary measures are required such as a separate work bench, pipettes, sterile tips and diligent practice of aliquoting all PCR primers and reagents beforehand to reduce carryover contamination. Furthermore, it has been suggested that use of ultraviolet light (254+300nm) to inactivate as much as 30ng of contaminating doublestranded DNA can alleviate the problem (Sarkar and Sommer, 1993). Nested-PCR has also been found to significantly reduce contamination (Wells and Sherlock, 1998) whereas the utilisation of polymorphic markers can detect contamination (Ma, 1995). The human genome contains many dispersed tandem-repetitive DNA sequences that are polymorphic due to variation in the copy number of tandem repeats. Those with longer motifs are called minisatellites (Jeffreys and Thein, 1985), while those with shorter motifs are called microsatellites (Weber et al, 1989). These polymorphic markers can act as contamination markers whilst examining a genetic disease in prenatal and preimplantation diagnosis. An important limitation of PCR and specifically single cell PCR is termed allele dropout (ADO) and poses a serious threat of false positive or false negative results (section 1.5.3.5.3)

1.3.4. Comparative Genomic Hybridisation

CGH has been described as an alternative method to interphase FISH or G-banding (Kallioniemi *et al*, 1992). It was originally applied for the detection of chromosome copy number changes in tumours and has further identified a variety of unbalanced chromosome complements. As a result quantitative fluorescent analysis using a digital imaging system is able to detect duplications or deletions in the sample DNA mapped

to a target metaphase; ploidy abnormalities and balanced rearrangements however remain undetected.

1.3.4.1 CGH Principles

CGH is a powerful technique that utilises the advantages of the FISH and karyotyping techniques. Even when DNA quantities are minute CGH is able to merge karyotyping, FISH and PCR to view its chromosome complement under a fluorescent microscope. The tissue is tested against a control DNA sample, either a single cell or genomic DNA from a normal individual. The "control" DNA is labelled in spectrum Red and the "test" DNA (tissue under investigation) is labelled in spectrum Green. The test and control samples are mixed and co-hybridised on a target metaphase slide of a normal male individual. The slide is viewed under a fluorescent microscope. Several metaphases are captured, which are analysed using specific software able to give a ratio of the green and red spectra (test and control samples respectively) of the chromosomes therefore, allowing detection and examination for all the chromosome complement. Equal hybridisation of red and green DNA produces a yellow colour. However, if the test sample contains an excess of chromosomal material, trisomy 21 for example, then green DNA fragments for this chromosome will be more abundant than their red equivalents and will out compete the red DNA fragments for hybridisation sites. This effect is only seen on the over-represented chromosome and results in a greenish coloration. Conversely if the test sample has a deficiency of chromosomal material, such as a monosomy, then a predominantly red coloration is seen on the chromosome in question (Figure 1.7).

Figure 1.7: Comparative genomic hybridisation on tumours.



Sample DNA (labelled green) and 46,XY reference DNA (labelled red) are simultaneously hybridised to normal chromosome spreads. The ratio of red:green fluorescence along the axis of each chromosome is calculated by computer. An excess of red fluorescence indicates a deficiency in the sample for the chromosome under analysis, while an excess of green reveals regions of the genome that are over-represented. (www.bu.us; University of Boston)

In Figure 1.7, it can be observed that the ratio of red:green fluorescence along the axis of each chromosome is calculated by a computer. Equal quantities of any loci (disomy) are assigned a profile ratio of 1.0. Changes in the copy number will deviate from this by a factor of 0.5 X n (where n is the number of copies of any locus). Therefore, chromosome gain (trisomy) will produce a ratio of 1.5 and a loss (monosomy), a ratio of 0.5 (Chang and Mark, 1997). The standard resolution of this technique is in the range of 10-40Mb but varies according to the regions analysed and degree of amplification (Kallioniemi et al, 1992; 1994; Voullaire et al, 1999). However, recent studies have reported sensitivity of their CGH protocols detecting deletions of <10Mb (Kirchhoff et al, 1999; Tonnies et al, 2001) by averaging CGH profiles. Since the smallest autosome is in excess of 50Mb, CGH analysis provides a very powerful and sensitive method for detecting duplications and deletions of significant size (Wells and Levy, 2003). However, certain regions of the karyotype including centromeres and heterochromatic segments show variation in the profile and these regions are normally excluded from CGH (Kallionemi et al, 1994). Certain chromosomes (1p, 17, 19 and 22) are prone to show frequent enhancement of the test signal and are also excluded from the analysis (Moore et al, 1997).

1.3.4.2 Single Cell CGH

Most genomic DNA CGH protocols utilise between 100 ng and 1 µg of test DNA, equivalent to more than 10,000 cells. CGH can be applicable when only a single cell, or a few cells are available for analysis such as in the case of PGD when a single blastomere is analysed (Griffin *et al*, 1993) as well as for non-invasive prenatal diagnosis using fetal cells obtained from the maternal circulation (Simpson and Elias, 1995). In these cases universal amplification, using specific PCR, of the minute amount of starting template DNA is required to produce a sufficiently large DNA sample for CGH. Amplification of the whole genome from single cells or a few cells has been reported by performing PEP, DOP, Alu PCR and MDA (Snabes *et al*, 1994; Wells *et al*, 1999; Dean *et al*, 2002; Lasken and Egholm, 2003; Handyside *et al*, 2004; see section 1.6.3.3).

It has been shown that between PEP, DOP, and Alu PCR the most reliable WGA technique is DOP since it provides the most complete coverage of the genome (Wells et al, 1999). Wells et al (1999) reported successful application of DOP-CGH for diagnosing chromosome imbalance in single cells, which was the first reliable application of CGH in a research context on single cells from fibroblasts, buccal cells, amniocytes, and blastomeres from human preimplantation embryos. Similar results from single fibroblasts and amniocytes were described by Voullaire et al (1999) where it was demonstrated that diagnosis of chromosomal aneuploidy in single cells is feasible using CGH with DOP-PCR amplified DNA (see section 1.6.2.3). Single cell CGH has also been successfully employed to assess clonal evolution of genetic variants in complex populations, by subjecting single micrometastatic cells isolated from bone marrow of cancer patients to CGH (Klein et al, 1999). Since the first report of successful application of single cell CGH, continuous investigations aim to improve the technique's fidelity. Huang et al (2000) conducted a comparative study aiming to define the optimal protocol for single cell CGH. The study employed differently labelled probes and hybridisation combinations and concluded that DOP-PCR-CGH homo-hybridisation (amplified test DNA vs. amplified reference DNA), especially when combined with labelling by nick translation is reliable and reproducible. However, the most important product of the first successful single cell CGH application was the potential use of CGH as a diagnostic tool for detecting chromosomal abnormalities in human preimplantation embryos (see section 1.6.4). DOP-PCR has been successfully employed as a means to amplify single human blastomere DNA allowing comparative genomic hybridisation to be undertaken by groups worldwide (Wells and Delhanty, 2000; Voullaire *et al*, 2000; Malmgren *et al*, 2002).

1.3.4.3 CGH Applications

CGH was originally developed as a cancer research tool (Figure 1.5), the initial application of CGH involved direct analysis of genomic DNA from tumour specimens. Rising above the usual problems encountered with the use of conventional cytogenetics, CGH did not require time-consuming and laborious techniques of cell culture, or obtaining poor quality metaphase spreads consisting of short chromosomes of low mitotic index unsuitable for G- banding analysis. Therefore, the major advantage of CGH to highlight chromosomal regions of amplification or deletion is that it can be applied to any sample from which DNA can be extracted in a rapid single step, without reliance on the cytogenetic preparations needed for standard FISH or karyotyping..

1.3.4.3.1 Cancer Studies

The use of CGH for the analysis of tumours revealed a number of new recurring chromosomal gains, amplifications, losses and deletion sites. There have been over 1400 articles published that have employed CGH to delineate cytogenetic changes in cancer specimens (Wells and Levy, 2003). A number of groups have employed CGH to show previously unrecorded areas of presumed tumour suppresser gene deletion and oncogene amplification in cell lines and solid tumours (Kallioniemi *et al.*, 1994; Kokkola *et al.*, 1997; Van Roy *et al.*, 1997). Some of the tumours studied included; Uveal melanomas (Becher *et al.*, 1997), small-cell lung carcinomas (Ried *et al.*, 1994; Levin *et al.*, 1995) and breast cancer (Kallionemi *et al.*, 1994). CGH has also shown its prognostic value by detecting chromosomal alterations in neoplasms including node-negative breast cancer (Isola *et al.*, 1995), renal cell carcinomas (Moch *et al.*, 1996), Uveal and cutaneous melanomas (Prescher *et al.*, 1996; Wiltshire *et al.*, 1995) and bladder cancer (Kallionemi, *et al.*, 1995). CGH has also been applied to neoplastic samples relevant to reproductive medicine, such as: prostate cancer (Joos *et al.*, 1995;

Visakorpi et al, 1995), testicular germ cell tumours (Korn et al, 1996) and ovarian tumours (Kiechle et al, 2001).

1.3.4.3.2 Prenatal Studies

Similarly in clinical cytogenetics CGH has been applied to prenatal and paediatric samples as well as mitotically inactive cells derived from products of conception (Bryndorf *et al*, 1995; Levy *et al*, 1997; Wells and Levy, 2003) to characterise marker chromosomes, cryptic deletions and complex rearrangements, and the origin of intrachromosomal duplications (Benzaken *et al*, 1998; Daniely *et al.*, 1999; Kirchhoff *et al*, 2001). Lapierre and co-workers (2000) carried out the first prospective study of CGH on uncultured amniocytes and concluded that CGH is a valuable alternative to interphase FISH for the rapid detection of unbalanced chromosomal aberrations. CGH has also been applied to material from spontaneous abortions (Daniely *et al.*, 1998), where it was able to detect 48% of the abnormalities usually detected by conventional karyotyping including trisomies, monosomies and partial gains and losses. Tabet *et al* (2001) used CGH to analyse trophoblast cells from spontaneous abortions, intrauterine fetal death, and malformed fetuses, circumventing the need for culture. The investigations suggested that the contribution of chromosome aberrations to first trimester pregnancy loss is nearly 70%.

1.3.4.3.3 Embryo Studies

Recently PGS has been carried out on using CGH (Wilton *et al*, 2001; Wells *et al*, 2002). The application of CGH for PGS is not straightforward as after embryo biopsy there is only a narrow window of time for the diagnosis to be made. As most protocols for CGH require 72 hours for hybridisation alone, two strategies have been proposed. Embryos can be frozen following biopsy and thawed after the CGH analysis has been completed (Wilton *et al*, 2001) or alternatively Wells *et al* (2002) tried an accelerated protocol following polar body biopsy on the day of fertilisation. Both studies suggested that the major limitation of CGH for PGS is the investigation takes time due to the laborious analysis of template chromosomes. These difficulties will be overcome when the template chromosomes are replaced by microarrays where hundreds or thousands of chromosome specific probes are spotted onto a glass slide (Wilton, 2002). In the study carried out by Wilton *et al* (2001), the need for cryopreservation reduces embryo viability and implantation potential by 30% (Edgar

et al, 2000), whereas in the study by Wells *et al* (2002), post-zygotic abnormalities, which account for more than half of abnormalities, as well as paternally derived aneuploidies are not detected

1.3.4.3 Technical Limitations of CGH

One of the few limitations of CGH is that it only detects relative alterations of chromosome copy number and cannot detect changes that involve the entire set of chromosomes, therefore it is unable to detect abnormalities of ploidy (Wells and Delhanty, 2000). Although providing information on imbalances of all chromosomes, unlike conventional karyotyping, CGH fails to provide information on chromosomal architecture. Furthermore, CGH analysis is suboptimal for both telomeric and pericentromeric regions, because of low fluorescence intensities and highly polymorphic regions that are blocked by COT-1 DNA (Daniely *et al*, 1998). The most significant limitation in utilizing CGH in a clinical setting and especially in IVF, relates to the technical complexities of the technique, as the method has been described as time consuming, labour intensive, and requiring expertise with several cytogenetic and molecular genetic techniques (Wells and Levy, 2003).

1.4 Prenatal Diagnosis

Prenatal diagnosis with a view to identifying fetal genetic disorders started in the early 1970's. Prenatal tests can be divided into two groups: a) screening techniques such as those involving measurements of chemicals in maternal blood and imaging the fetus and b) diagnostic techniques including invasive tests to remove tissues of fetal origin. Women who are at known risk e.g. carrying a translocation or a single gene disorder and women who are found to be at risk due to abnormal screening results undergo invasive tests, which will allow karyotyping or identification of the single gene defect. The appropriate technique is selected based on the evaluation of gestational age, the urgency to obtain a result, the risk of the procedures and the *a priori* risk for a disorder in the fetus. Fetal cells for chromosome diagnosis are obtained invasively either by chorionic villus sampling (CVS) at about 9-11 weeks gestation or amniocentesis at about 15-20 weeks of gestation. Both procedures are invasive and carry an associated risk for induced abortion of 0.5-2% of women tested (Hulten *et al*, 2003).

1.4.1 Chorionic Villus Sampling

CVS can be carried out from 9-11 weeks of gestation. The timing of diagnosis can be crucial in minimising the trauma of an elective abortion caused to the patient when unbalanced chromosomal abnormalities as well as single gene defects are diagnosed. The procedure is done either transcervically or transabdominally. For transcervical sampling different catheters are used including curved biopsy forceps as well as the catheter originally designed by Ward and colleagues (1983). For the transcervical approach, a bendable polyethylene catheter with a metal obturator is introduced through the cervix and guided to the chorionic frondosum under ultrasound surveillance. With an attached syringe (partly filled with medium) a vacuum is applied and approximately 10-50mg are aspirated and rinsed into a Petri dish (Rodeck and Whittle, 1999). The transabdominal technique was initially developed in Denmark by Smid-Jensen and Hahnemann (1984). It can be carried out either as a free-hand ultrasound-guided fine-needle aspiration or with a needle guide ultrasound transducer. In CVS, some fetal cells are spontaneously dividing and cells at metaphase can be obtained. However, the resolution, which is of great importance for the detection of structural abnormalities, may be low due to the condensed nature of the dividing cells, hence cells are cultured.

Two major problems are encountered in fetal karyotyping using cultured cells from chorionic villi: the relatively slow growth of these cells in culture, which subsequently delays the diagnosis and the occurrence of maternal cell contamination (MCC) (Goumy *et al*, 2004). However, chorionic villi are considered an excellent source of DNA supplying sufficient amounts for most molecular genetic techniques without prior culture. Thus, CVS is the method of first choice in pregnancies at risk for monogenic diseases (Stranc *et al*, 1997). Biochemical testing after CVS is possible for most metabolic disorders and is advantageous over amniotic fluid cells due to the high recurrence risks and the usually feasible use of fresh uncultured villi (Holzgreve *et al*, 1999). Confined placental mosaicism (Section 1.4.7.1) has been detected in 2% of viable pregnancies (Kalousek and Vekemans, 1996; Grati *et al*, 2006) and can give false positive results which need to be confirmed by amniocentesis (Lacroute *et al*, 2004)

1.4.2 Amniocentesis

Amniocentesis can be carried out at gestational ages between about 14 weeks and term. Traditionally this test is offered between 15-17 weeks of gestational age when the total amount of amniotic fluid is approximately 200ml and the uterus can be reached transabdominally without major risks of transversing the bladder or bowel. Nowadays, the correct line of insertion of the needle is chosen by ultrasound and subsequently the puncture and aspiration are performed under continuous ultrasound surveillance (Holzgreve *et al*, 1999). In most cases, culture of amniotic fluid cells is required for prenatal diagnosis and this takes a further 1-4 weeks, depending on the number of cells required and individual variations in the speed of cell growth. This is due to the fact that amniotic fluid samples do not contain any fetal cells in division and have to be grown *in vitro* to obtain cells at the metaphase stage

Three problems can occur during analysis of amniocentesis samples including complete failure of culture, maternal cell contamination leading to false-negative results and the incidence of mosaicism. Several studies have been carried out trying to determine the risk of spontaneous abortion after amniocentesis and it is found to be 0.5-1.5% (Tabor *et al*, 1986; Kapel *et al*, 1987).

1.4.3 Fetal Blood Sampling

Fetal blood sampling is usually obtained antenatally by fetoscopy, however, this procedure has been almost abandoned in favour of cordocentesis. Cordocentesis is a sampling technique whereby fetal blood is obtained by direct puncture of the umbilical vein.

1.4.4 Maternal Cells and DNA in the Fetal Circulation

During pregnancy, the fetal and maternal circulations are separated by the placental membranes. However, a variety of evidence has pointed towards the incompleteness of this barrier of cellular trafficking. Fetal nucleated cells were first demonstrated in the maternal circulation by Walknowska *et al* (1969) and have now been widely pursued as potential substrates for non-invasive prenatal diagnosis (Bianchi, 1999). The isolation and analysis of fetal cells in maternal blood for genetic diagnosis has been a matter of great interest in the world of prenatal diagnosis. Attention is being

directed on choice of the best fetal cell type, selection of optimal cell enrichment, consistency and reproducibility of cell recovery and analysis.

1.4.5 Cytogenetics and Prenatal Diagnosis

Prenatal diagnosis of chromosomal abnormalities is routinely accomplished by standard cytogenetic techniques. The major disadvantage of these procedures is the fetal cells must be cultured for up to two weeks (for amniocentesis) before analysis and that cultivation must be prolonged in advanced stages of pregnancy (Pertl *et al*, 1999). This time interval between sampling and diagnosis places a considerable emotional burden on the prospective parents (Tercyak *et al*, 2001). A rapid diagnosis is essential when an ultrasound examination suggests an abnormal fetus. The two most common types of rapid molecular method for prenatal diagnosis of chromosome disorders are fluorescent *in situ* hybridisation (FISH) and quantitative fluorescent polymerase chain reaction (QF-PCR). Both methods are used, but, QF-PCR is applied routinely for rapid and simple diagnosis of aneuploidy (numerical chromosome abnormalities) including trisomies 21, 13 and 18 which give rise to Down, Patau and Edward syndromes respectively and sex chromosome abnormalities.

In studies where either FISH or QF-PCR have been compared to karyotyping it has been postulated that both rapid techniques are more economical, rapid and easier to perform, however, they cannot be carried out independently (Eiben *et al.* 1998; Billi *et al.* 2002). Eiben *et al.* (1998) concluded that all FISH analysis should be followed by karyotyping in order to cover the 35% of aberrations which cannot be detected by FISH. In a recent study on uncultured chorionic villus samples using FISH for chromosomes 13, 18, 21, X and Y showed that FISH is able to minimize maternal cell contamination and provide rapid diagnosis which reduces parent anxiety (Goumy *et al.* 2004). Feldman *et al.* (2000) found 100% sensitivity, specificity and predictive values for their routine FISH analysis as a method to detect aneuploidies of chromosomes 13, 18, 21, X and Y in amniocytes. The same group, however, quoted that even with 100% accuracy of their test, routine FISH analysis will miss about 25-30% of the abnormalities detected by standard cytogenetic techniques, therefore, the latter must always act as a backup to FISH. Several studies have shown a variety of detection percentages whilst using FISH in normal uncultured amniotic fluid and

uncultured CVS, such as 70-100% (van Opstal *et al*, 1995; Bryndorf *et al*, 1997) and 36-100% (Bryndorf *et al*, 1996) respectively. This substantial range of FISH efficiency is probably due to: maternal cell contamination; true low level mosaicism; the types of probes used; error in scoring' and the efficiency of hybridisation, especially in nuclei interphase (Ruangvutilert *et al*, 2000b).

Pertl and co-workers (1999) after carrying out QF-PCR analyses on 247 chorionic villus samples concluded that the QF-PCR assay provides a fast and economical method for the prenatal diagnosis of major chromosome defects, but it is not a substitute for conventional cytogenetic analysis. Billi *et al* (2002), found no discordance between the results of QF-PCR and karyotyping after CVS where only 4/1100 cases were missed by QF-PCR, indicating the reliability of QF-PCR as a supporting prenatal diagnostic method. Furthermore, it has been shown that QF-PCR is a valuable tool for prenatal diagnosis of multiple pregnancies since it allows the detection of all aneuploid fetuses in just a few hours after sampling, as well as the determination of fetal zygocity in all cases (independently of chorionicity and fetal sex) (Cirigliano *et al*, 2003).

1.4.6 Prenatal Diagnosis of DM

The development of a reliable molecular diagnostic test for DM has meant that presymptomatic testing, prenatal testing as well as PGD for DM can be offered to those families for whom it is appropriate and acceptable. The severe congenital form of DM has a prevalence of 2.5-5.5 per 100,000 live births (Geifman-Holtzman and Fay, 1998).

In congenital DM affected pregnancies polyhydramnios, decreased fetal movement and prematurirty often complicate the pregnancies. Since the triplet repeat is expanded to hundreds or even thousands of copies, standard PCR protocols are unable to amplify such long products (Monckton *et al*, 1995). Therefore, the CTG expansion is detected by Southern blot performed from extracted DNA of CVS and amniotic fluid samples digested with the appropriate enzyme using a radioactive probe (Geifman-Holtzman and Fay, 1998). More recently several groups have reported the development of non-radioactive methods to determine the CTG repeat expansion in DM patients involving the use of fluorescein-labelled probes (Brugnoni *et al*, 1998; Zuhlke *et al*, 2000). The combination of PCR and hybridisation reduces the time required for prenatal diagnosis from 10 days to 2-3 days (Zuhlke *et al*, 2000). Furthermore, smaller amounts of DNA are required compared to conventional Southern blot analysis (Brugnoni *et al*, 1998). Moreover, Amincucci and co-workers (2000) attempted to isolate fetal cells from maternal plasma for PND of DM of an unaffected woman whose husband was affected by DM and concluded that this noninvasive method allowing first-trimester PND, can become an alternative procedure in selected cases.

1.4.7 Prenatal Diagnosis and Mosaicism

Mosaicism, defined as the presence of two or more cell lines with different chromosome constitution in an individual, presents a prognostic dilemma since it influences the postnatal phenotype. Mosaicism has been found present in amniotic cells as well as chorionic villus samples (Eisenberg and Wapner, 2002). In order to identify samples with mosaicism the minimal demand is:

a) Two or more identical aberrant cells in villi analysed by the direct (short term culture) method (CVS)

b) At least one identical aberrant cell in each of the two or more independent culture dishes (CVS/amniotic cells)

c) At least one identical cell found both by the direct and the culture methods (CVS+amniotice cells) (Phillip and Bryndoff, 1998).

The confirmation or exclusion of mosaicism detected prenatally by CVS requires repeat sampling of fetal cells (amniotic fluid or fetal blood) (Pergament, 2000). The phenomenon of mosaicism in prenatal diagnosis has also been linked to intrauterine growth retardation (IUGR), uniparental disomy as well as confined placental mosaicism.

1.4.7.1 Confined Placental Mosaicism

In most pregnancies the chromosomal complement detected in the fetus is also present in the placenta. The detection of an identical chromosomal complement in both the fetus and its placenta has always been expected as both develop from the same zygote. However, in approximately 2% of viable pregnancies studied by chorionic villus sampling (CVS) at 9 to 11 weeks of gestation, the cytogenetic abnormality, most often trisomy, is confined to the placenta (Kalousek and Vekemans, 1996). This phenomenon is known as confined placental mosaicism (CPM). It was first described by Kalousek *et al* (1991) in term placentas of infants born with unexplained intrauterine growth restriction (IUGR). Contrary to generalised mosaicism, which is characterised by the presence of two or more karyotypically different cell lines within both the fetus and its placenta, CPM may be due to a postzygotic nondisjunction event generating a trisomic cell line in an initially normal conceptus (mitotic origin) or the postzygotic loss of one chromosome in an initially trisomic conceptus (meiotic origin and trisomy rescue) (Grati *et al*, 2006).

Cytogenetically, CPM can assume three different forms. In type I the trophoblast is aneuploid, in type II aneuploidy is present in chorionic stroma and in type III both trophoblast and chorion are aneuploid. In trisomic zygote rescue, either reverse type I CPM, when trophoblast is diploid and the rest of the placenta and the fetus are nonmosaic trisomic, or type III are seen (Kalousek, 2000). It has been shown that the effects of CPM on development depend on the origin of the extra chromosome in the placenta as well as the specific chromosome involved. Meiotic origin is highly correlated with type III CPM and increased risk of pregnancy complications, whereas mitotic origin, more frequently found in types I and II, shows a lower risk of pregnancy complications (Robinson *et al*, 1997).

Specific chromosomal trisomies have been observed in CPM more frequently than others, with trisomy of chromosomes 7, 16, and 18 being the most prevalent (Wolstenholm, 1996). Previous studies have shown that the majority of the CPM for autosomes 9, 16, and 22 are meiotic in origin, whereas CPM for autosomes 2, 7, 8, 10, and 12 are predominantly somatic in origin (Robinson *et al*, 1997)

Uniparental disomy (UPD), the inheritance of the two copies of a chromosome from the same parent, may sometimes be associated with CPM. UPD can result from gamete complementation, chromosome loss in trisomy, or duplication in monosomy (with or without residual mosaicism) and somatic recombination (Engel, 1993). In isodisomy, the uniparental pair is a duplicate of a same chromosome DNA template and causes an increased risk of recessive disorder by reduction to homozygosity. In heterodisomy, the pair remains heterozygous, made up of 2 non-recombinant homologous segments. But both iso- and heterodisomy may also cause disruption of the genomic imprints needed for differential expression of some maternal and paternal genes crucial to growth and development (Engel, 1997).

Conventional cytogenetic methods for detection of CPM depend on the availability of dividing cells and analysis of a large number of metaphase cells (Lestou *et al*, 1999). FISH has been established as a molecular cytogenetic technique for the detection of chromosomal aneuploidy and mosaicism in placental tissues (Lomax *et al*, 1994). More recently, it has been shown that CGH can be reliably performed on DNA obtained form placental (either trophoblast or stroma cells) and fetal tissues in order to detect aneuploidy (Lestou *et al*, 1999; Lestou *et al*, 2000; Barrett *et al*, 2001).

1.4.7.2 Uniparental Disomy

Uniparental disomy (UPD) is the occurrence of both homologous chromosomes from one parent. If an individual inherits two copies from one parent, through an error in meiosis II this is named uniparental isodisomy. If, however, the individual inherits two different homologues from one parent through an error in meiosis I, this is termed uniparental heterodisomy. Recent insights have revealed that the molecular basis for the clinical features of UPD are specific human genes that are only monoallelically active, depending on whether they are located on the paternal or maternal chromosome. UPD will lead to an imbalanced expression of these imprinted genes and cause abnormal development (Eggermann *et al*, 2002). Specific syndromes have been found to be associated with UPD, these include Prader-Willi syndrome (maternal UPD15/mUPD15), Angelman syndrome (paternal UPD15/pUPD15), (transient) neonatal diabetes mellitus (pUPD6), Silver-Russell syndrome (mUPD7), Beckwith-Wiedemann syndrome (pUPD11) and the mUPD14 syndrome.

The Prader-Willi (PWS) and Angelman (AS) syndromes are two clinically distinct syndromes which result from lack of expression of imprinted genes within chromosome 15q11-q13. These two syndromes result from 15q11-q13 deletions, chromosome 15 uniparental disomy (UPD), imprinting centre mutations and, for AS, probable mutations in a single gene. The differential phenotype results from a paternal

genetic deficiency in PWS patients and a maternal genetic deficiency in AS patients. Within 15q11-q13, four genes (SNRPN, IPW, ZNF127, FNZ127) and two expressed sequence tags (PAR1 and PAR5) have been found to be expressed only from the paternally inherited chromosome, and therefore all must be considered candidate genes involved in the pathogenesis of PWS (Glenn et al, 1997; Liehr *et al*, 2005)The mechanisms of imprinted gene expression are not yet understood, but it is clear that DNA methylation is involved in both somatic cell expression and inheritance of the imprint

Maternal UPD(16) is the most often reported UPD other than UPD(15); almost all cases are associated with confined placental mosaicism (CPM). Most of maternal UPD(16) cases are characterised by intrauterine growth retardation (IUGR) and different congenital malformations. Maternal UPD(16) has therefore been suspected to have clinical effects: however, the lack of uniqueness and specificity of the birth defects observed suggests that the phenotype may be related in parts to placental insufficiency (Eggermann *et al*, 2004)

1.5 Preimplantation Genetic Diagnosis

It has been 15 years since the first PGD baby was born (Handyside *et al*, 1989). PGD has been offered for a variety of single gene defects (Verlinksy *et al*, 1994; Sermon *et al*, 1997; Abou-Sleiman *et al*, 1999; Sermon *et al*, 2004), chromosomal abnormalities (Munne *et al*, 1998e; Conn *et al*, 1999; Scriven *et al*, 2001; Braude *et al*, 2002) and sexing (Griffin *et al*, 1991; 1992; 1994; Staessen *et al*, 1994; Handyside and Delhanty, 1997) and offers an alternative to traditional methods of prenatal genetic testing (CVS and amniocentesis), and allows genetic analysis to be performed on early embryos prior to implantation and pregnancy. This provides couples at a risk for certain genetic diseases the opportunity to know that any pregnancy they achieve should be unaffected (Handyside, 1998). Patients requesting PGD undergo in vitro fertilisation (IVF) treatment because in this way multiple embryos can be generated *in vitro* giving an increased probability that a disease free embryo will be identified (Wells and Delhanty, 2001).

FISH and PCR, following extensive groundwork for their suitability in clinical application, are the methods of choice and now form the basis of most PGD strategies used today for a spectrum of genetic defects ranging from gross chromosomal aberrations to single base-pair mutations. Although classical cytogenetic techniques can also be successfully applied to single blastomeres with a view to PGD, the efficiency with which analysable metaphase preparations can be produced per biopsied cell is notoriously low (Kola and Wilton, 1991; Wells and Levy, 2003).

The scope of PGD has been widened and apart from the diagnosis of genetic diseases (Handyside, 1998) and the detection of chromosomal abnormalities, it has been used to try and improve pregnancy rates for certain groups of IVF patients (Munne *et al*, 2000; 2005). Preimplantation genetic screening (PGS) is offered to couples with recurrent IVF failure, recurrent abortions and screening for aneuploidy in older aged women (Egozcue *et al*, 2000; Abelhadi *et al*, 2003).

PGD is a clinical diagnostic procedure that has evolved from the substantial advances both in assisted reproduction technology and molecular genetic analysis. Patients undergo routine IVF procedures which will produce multiple embryos. Polar bodies (first and second) and day 3 or day 5 embryos are biopsied (section 1.5.1) and are analysed either by FISH (section 1.5.2), or PCR (section 1.5.3) or CGH depending on the nature of the aberration. Embryos free of the specific inherited disorder are replaced in the uterus, which will give the parents the chance of starting a 'normal' pregnancy. As the embryos subjected to PGD must be IVF generated, certain difficulties are encountered. Significant limitations are presented involving the number and the subsequent quality of the embryos provided, that have a direct effect on the success rate of PGD. Thus, the success rate of PGD is relative to the highest possible success rate of IVF (Egozcue *et al*, 2000).

1.5.1 Sampling Strategies for PGD

Genetic analysis of preimplantation developmental embryos prior to replacement into the uterus inevitably involves removal of some embryonic cellular material from one of these stages.

1.5.1.1 Polar Body Biopsy

Biopsy of the first polar body has been carried out when the carrier of the mutation or the chromosomal aberration is the female partner (Verlinsky *et al*, 1990; 1997; Munne *et al*, 2000; Strom *et al*, 1997). The first polar body (PB) is biopsied within six hours of oocyte retrieval to preserve optimum chromosome morphology (Verlinsky *et al*, 1990). PB biopsy involves a breech in the zona which can be performed by mechanical (Verlinsky *et al*, 1997) or laser (Montag *et al*, 1998) drilling followed by polar body aspiration. Direct penetration through the ZP is mainly performed using a bevelled pipette (Roudebush *et al*, 1990; Verlinksy and Cieslak, 1993). The bevelled pipette allows mechanical perforation of the ZP, and once inside the perivitilline space, the first and second PB can be aspirated (Gianarolli, 2000).

First PB removal does not seem to interfere with normal fertilisation and the percentage of the embryos entering cleavage (Verlinsky *et al*, 1992). No effect on the viability of the resulting embryos was observed in a mouse model when both polar bodies were removed (Kaplan *et al*, 1995). Moreover, Strom *et al* (2000a and b) did not observe any deleterious effect in a follow-up study of 109 children born after first and second PB biopsy. The previous studies were reassuring for the clinical application and genetic evaluation of human oocytes, which has resulted in clinical pregnancies and livebirths (Verlinsky and Kuliev, 1996b). Munne *et al* used PB biopsy for translocations of female origin, significantly reducing spontaneous abortion rate (Munne *et al.*, 1998a; 1998b; 2000).

However, the main disadvantage of PB sampling is that the paternal contribution cannot be evaluated. Furthermore, the presence of post-zygotic mitotic errors cannot not be detected. Furthermore, this technique is extremely laborious (Wells and Delhanty, 2001), there is a risk of misdiagnosis in case of crossing over during meiosis I when only the first polar body is analysed and there is no information about the paternally derived genetic make-up of the embryo (Ruangvutilert *et al*, 2000b). First and second maternal meiotic errors can be excluded only if information on both polar bodies is obtained (Angell, 1994b). In cases where the paternal contribution needs to be evaluated, biopsy at post fertilisation stages will be required.

1.5.1.2 Cleavage-Stage Biopsy

At present, most of PGD analyses are performed by employing cleavage-stage biopsy, where 1-2 cells are extracted from each embryo (ESHRE PGD Consortium, 1999; 2000; 2001; 2002). The biopsy procedure involves making an opening in the ZP through which a cell can be removed. This can be accomplished either mechanically involving partial zona dissection (PZD) by direct piercing or cutting with a micropipette (Grifo et al, 1990; Cieslak et al, 1999), chemically by the localised application of acid Tyrode's solution (pH 2.3) (Hardy et al, 1990; Inzunza et al, 1998) or by employing an extremely precise laser system to puncture a hole into the zona (Palanker et al, 1991; Veiga et al, 1997; Montag et al, 1998; Joris et al, 2003). The latter two techniques, using Acid Tyrode's solution and laser, are the most commonly used methods for cleavage or blastocyst stage biopsy (Figure 1.8). Chemical zona drilling followed by blastomere aspiration is normally performed with separate pipettes (one drilling pipette with inner diameter $\pm 5-7$ µm, and one aspiration or biopsy pipette with inner diameter $\pm 40 \ \mu m$) using a double-holder setup (de Vos and Steirteghem, 2001). The resultant local acidification can cause subtle damage to the embryos and may interfere with further embryo development. Hence, Cohen et al (1992) suggested immediate washing of micromanipulated embryos. The use of a laser for zona drilling in cases of PGD is an easier procedure and results in more intact blastomeres (Veiga et al, 1997). Zona opening by laser drilling is performed by exposing the ZP to laser light. The hole size can be chosen precisely by varying the irradiation time, however extra care should be taken not to harm the embryonic cells with the laser shot. Since similar pregnancy rates are obtained from studies comparing laser and the acid Tyrodes technique (De Vos and Van Steirteghem, 2001). Recently, it was shown that there was no difference in blastomere viability, level of chromosome abnormalities and cytoskeleton damage when comparing embryos biopsied either by acid tyrodes or laser (Chatzimeletiou et al, 2005).



Figure 1.8. Illustrating cleavage stage embryo biopsy. (a): Embryo attached to holding pipette and positioned. (b): Acid Tyrodes is used to create a hole in the zona

pellucida. (c): A single blastomere is carefully removed. (d): The embryo following the biopsy. (Harper and Doshi, 2000)

Prior to cleavage-stage biopsy, embryos are placed into Ca^{2+}/Mg^{2+} -free medium to reduce the tight junctions that occur between human blastomeres at compaction (Dale *et al*, 1991). Blastomeres at the 6-8 cell stage show a strong tendency to adhere to each other, which in turn might cause a high rate of cell lysis whilst performing the biopsy. Hence, Santalo and colleagues (1996) used Ca^{2+}/Mg^{2+} -free medium to loosen the tight junctions of preimplantation mouse embryos, which allowed easier removal of the blastomeres and reduced the biopsy time. The use of Ca^{2+}/Mg^{2+} -free medium thus allows for an easier biopsy procedure during pre-implantation genetic diagnosis, while it does not result in a loss of developmental potential of the embryo to the blastocyst stage (Dumoulin *et al*, 1998).

Early studies focused on measuring the effect of biopsy on human embryonic development revealed that two cells could be removed from 8-cell embryos on day three post-insemination without reducing the number blastulating or disturbing cleavage rates (Hardy *et al*, 1990). De Vos and Van Steirteghem (2001) suggested that at day 3 of embryo development the blastomeres are totipotent and embryo biopsy could be carried out even if the embryo was compacting. Unfortunately, not all embryos reach the 7- or 8-cell stage by the morning of day 3. Six cell embryos might also be included for one or two cell removal. However, biopsies performed at the 4-cell stage may alter the ratio of inner cell mass to trophectoderm cells, if more than one cell is removed, which may be detrimental to embryo development (Tarin *et al*, 1992). Consequently two-cell biopsy procedures should only be carried out on day three post-insemination at the 6-8 cell stage, when up to a quarter of the blastomeres can be removed without disturbing subsequent development (Handyside, 1991; Hardy *et al*, 1990), as the biopsied cells are still undifferentiated (Harper *et al*, 1996).

The decision as to whether one or two cells should be removed from a cleavage stage embryo is controversial. It has been suggested that the removal of two cells reduces the cellular mass and could potentially reduce its developmental capacity (Braude *et al*, 2002). However, diagnosis of inherited diseases can suffer from technical limitations. Single cell FISH and PCR are not 100% accurate and can give false

positives and negatives. Furthermore, the phenomenon of mosaicism, poses a serious threat of misdiagnosis for PGD of chromosome abnormalities (see section 3.1.1). Hence it has been postulated that the accuracy of the diagnosis can be enhanced if embryos are replaced when results from two cells are concordant (Van de Velde *et al*, 2000). For this reason many groups prefer to base their diagnosis on the result of two biopsied cells, particularly for chromosomal analysis and dominant disorders (Delhanty *et al*, 1994; Delhanty and Handyside, 1995; Kuo *et al*, 1998; Van de Velde *et al*, 2000; Simopoulou *et al*, 2003).

1.5.1.3 Blastocyst Biopsy

The limited amount of tissue available in PB and cleavage-stage sampling could be overcome with the use of blastocyst biopsy. Trophectoderm biopsy from blastocyst stage embryos has been attempted in animal models (mouse, Gardner, 1971; cattle, Betteridge et al, 1981; monkey, Summers et al, 1988), as well as humans (Dokras et al, 1990) showing no adverse effect on further development. The technique of zona slitting (similar to PZD) is used to open the ZP with a microneedle (Dokras et al, 1990), however, laser zona opening has also been performed (Veiga et al, 1997). This will allow a generous amount of embryonic material in order to provide a reliable genetic analysis, overcoming the lack of material made available when carrying out cleavage stage biopsy. As these cells are from the trophectoderm there is no decrease in the ICM and therefore in the embryo proper. De Boer et al (2004) showed that blastocyst biopsy permits up to five or six cells to be genetically tested, leaving the inner cell mass intact and enabling embryos to be electively transferred one at a time without diminishing the chance of pregnancy compared with cleavage-stage biopsy and testing. The main drawback of this approach is the limited number of embryos available for biopsy and diagnosis, since only half or fewer of the IVF generated embryos are able to reach the blastocyst stage (Jones et al, 1998). Moreover, the time for diagnosis will be severely limited if the biopsy is postponed to this later preimplantation stage, posing serious restrictions in the time available for genetic analysis (De Vos and Van Steirteghem, 2001).

1.5.2 Chromosomal Abnormalities

Chromosomal analysis of embryos is one of the most important research fields in the area of preimplantation development. It has been suggested that fewer than 50% of naturally conceived embryos reach full term with most lost before or shortly after implantation (Edwards an Gardner, 1967). With the advent of IVF and more recently PGD studies to confirm chromosomal abnormalities by analysing embryos at early stages have revealed similar rates of embryonic mortality, where a significant portion of embryos arrest in development during the first days after fertilisation. Chromosomal abnormalities especially in the form of aneuploidy, usually have an adverse effect on the developing embryo by altering the dosage of hundreds of expresses genes (Wells and Levy, 2003). Chromosome analysis of human embryos has shown higher rates of aneuploidy than those reported for prenatal testing (Jamieson *et al*, 1994), suggesting that considerable numbers of chromosomally abnormal embryos are eliminated early in development (Wells and Delhanty, 2000; Sandalinas *et al*, 2001).

1.5.2.1 X-Linked Disorders

Sexing the embryo to avoid X-linked disease was the first application of FISH in this context (Griffin et al, 1991; 1992; 1994) and is one of the major indications for PGD (ESHRE PGD Consortium, 1999, 2000 and 2002, 2004). X-linked recessive diseases account for 6-7% of single gene defects and include conditions such as Duchenne muscular dystrophy (DMD), haemophilia, and various metal retardation syndromes. The first application of PGD was to avoid X-linked disease carried by the mother by the selection of female embryos for transfer following diagnosis of embryo sex by employing PCR (Handyside et al, 1990). Primers specific for a sequence derived from the long arm of the Y chromosome were designed and diagnosis was performed by negative selection i.e. the embryos for which no amplification was present were diagnosed as female and recommended for transfer. In the second series, of seven fetuses tested following sexing by PCR amplification, one singleton was male and the pregnancy was terminated (Handyside and Delhanty, 1993). This error presumably arose from amplification failure of a XY blastomere, subsequently shown to occur in 15% of cells tested, although biopsy of an anucleate or haploid blastomere would give the same result (Kontogianni et al, 1996). Following this, protocols were developed

for the simultaneous detection of both X and Y chromosomes, using either combinations of specific primers (Kontogianni *et al*, 1991; Grifo *et al*, 1992) or more reliably common primers for homologous sex chromosome sequence such as amelogenin (Nakahori *et al*, 1991) steroid sulphatase (Liu *et al*, 1994) or ZFX/ZFY (Chong *et al*, 1993).

FISH with biotinylated probes was introduced Griffin *et al* (1991). Moreover, the same group simultaneously hybridised a biotinylated X probe and two digoxigenin labelled Y probes using FISH to establish dual-colour FISH as the preferred method of for embryo sexing (Griffin *et al*, 1992; 1994). Indirectly labelled probes originally employed were soon replaced by directly labelled probes reducing the time of the FISH procedure from 7 to 2 hours (Harper *et al*, 1994a; Harper and Delhanty, 1996). Nowadays, to misdiagnose a normal male embryo as a normal female embryo two errors must occur; the signal for chromosome Y must be lost and an extra signal for chromosome X must be generated. A further possibility for misdiagnosis can arise in the case when the embryo analysed is chaotic or grossly mosaic and the cells biopsied are not representative of the whole embryo (Kuo *et al*, 1998). However, only one FISH misdiagnosis occurred among 78 cycles of social sexing as reported to the ESHRE PGD consortium (2002).

1.5.2.2 Structural Aberrations

Structurally abnormal chromosomes are formed from the rearrangement, deletion or duplication of chromosomal segments leaving the karyotype either genetically balanced or unbalanced. Structural abnormalities include translocations (reciprocal or Robertsonian), deletions. inversions, insertions, ring chromosomes and isochromosomes (see Figure 1.6; section 1.2.1.2). Balanced translocations occur in 0.2% of the neonatal population, however are at a higher rate among infertile couples and patients with recurrent abortions (Munne, 2002). In a study, it was reported that balanced translocations were found in 0.6% of infertile couples, 3.2% of couples that had failed over 10 IVF cycles and 9.2% among fertile couples experiencing three or more consecutive first trimester abortions (Stern et al, 1999).

With the advent of PGD, carriers of balanced translocations can circumvent repeated miscarriages or prenatal diagnosis and termination of pregnancy (TOP) of unbalanced fetuses and opt for PGD. The aim of PGD for translocations is to reduce the rate of spontaneous abortions and to minimise the risk of conceiving an unbalanced offspring. PGD for structural aberrations has been attempted using a variety of approaches with the aid of FISH. A number of reports concerning PGD have been published involving Robertsonian and reciprocal translocations, inversions, insertions, microdeletion syndromes and gonadal mosaicism (Conn *et al*, 1998, 1999; Iwarsson *et al*, 1998; Reubinoff *et al*, 1998; Scriven *et al*, 1998, 2001; Van Assche *et al*, 1999; Simopoulou *et al*, 2003).

Different approaches have been tried to identify structural chromosomal abnormalities including: a) probes spanning the breakpoints of each translocation or inversion (Cassel et al, 1997; Munne et al, 1998a and b) use of probes distal to the breakpoints or telomeric probes in combination with proximal or centromeric probes (Conn et al, 1998; 1999; Munne et al, 1998g). Munne et al (1998a) employed FISH in polar bodies in order to identify the translocations using chromosome painting probes for the chromosomes involved in the translocation. This technique was later modified by the same group using telomeric probes to further enhance the regions not covered by the chromosome-paint probes (Munne et al, 1998b). Furthermore, spectral imaging has been applied in polar bodies to identify all 23 chromosomes, though it was found to be laborious and needed well-spread chromosomes in order to distinguish each one (Marquez et al, 1998). The drawback of polar body analysis for translocations is the occurrence of crossing over and predivision of chromatids (Munne et al, 1998a, b and f; Marquez et al, 1998). The result of the second meiotic division in both cases is unclear and either the second polar body or the blastomeres should be analysed. Moreover, Munne et al (1998f) detected interstitial crossover with subsequent segregation of balanced and unbalanced sets of chromosomes during the second meiotic division, which might cause problems during diagnosis.

1.5.2.2.1 Robertsonian Translocations

PGD for Robertsonian translocations has been carried out on biopsied polar bodies as well as blastomeres (Conn *et al*, 1998; Munne *et al*, 1998g; Scriven *et al*, 2001; Ogilvie and Scriven, 2004; Sermon *et al*, 2004). To detect Robertsonian translocations

chromosome enumerator probes are used to count the chromosomes in the interphase nuclei in the cases of cleavage blastomere biopsy (Conn et al, 1998; Munne et al, 1998g). Probes can be chosen which bind to any point on the long arm of each chromosome involved in the translocation. Studying infertile couples carrying Robertsonian translocations using FISH revealed two factors leading to infertility in some of these cases. Firstly the aneuploid segregation of Robertsonian translocations carried by the parent and secondly a factor acting at the post-zygotic level provoking an uncontrolled chromosome distribution in early cleavage stages giving rise to chaotic embryos (Conn et al, 1998). However, this latter factor was not supported by a recent study by Scriven et al (2001), which concluded that Robertsonian translocations do not predispose to embryos with abnormal cleavage division. Nevertheless all studies on Robertsonian translocation carriers have reported a high incidence of mosaicism on their resulting blastomeres of >60%. It has been shown that the most common mode of segregation is the alternate (85%) (Iwarsson et al, 2000). This finding is in line with studies of meiotic segregation in sperm of male carriers of a Robertsonian translocation which displayed an incidence of 91% (Pellestor *et al*, 1987). However, recent data has shown that the pregnancy rates for maternal or paternal reciprocal or Robertsonian translocations are similar (Sermon et al, 2004)

1.5.2.2.2 Reciprocal Translocations

PGD is appropriate for those at high reproductive risk. Due to the complex nature of reciprocal translocations, each case is usually unique, hence difficult to treat by PGD (Harper and Bui, 2002). For reciprocal translocations the prevalence of unbalanced gametes is estimated to be between 50-70% (Gardner & Sutherland, 1996).

As mentioned above (see section 1.5.2.2) two types of FISH protocols have been devised for the investigation of reciprocal translocations; the use of spanning (Munne *et al*, 1998g) or flanking probes (Conn *et al*, 1998; Munne *et al*, 1998e). Breakpoint spanning probes used in interphase nuclei can detect normal, balanced or unbalanced karyotypes resulting from a reciprocal translocation (can also be used for inversions, deletions and duplications). The disadvantage with spanning probes and their seldom use in PGD is due to laborious, time-consuming and expensive probe development required for each breakpoint for each translocation. Methods used to produce these

probes have been described by Fung *et al* (1998). For the flanking-probe approach, two probes would be used flanking the breakpoint of one chromosome and the third probe would be specific for the other chromosome. Conn and co-workers (1998) employed this strategy for studying a reciprocal translocation involving chromosomes 6 and 21, being able to detect all segregation patterns apart from free trisomy and monosomy of chromosome 6, which is not viable. The disadvantage of using flanking probes is that balanced and normal embryos cannot be distinguished. This approach however, is the simplest owing to the commercialisation of sub-telomeric probes for most p and q arms. Scriven *et al* (1998), suggested a generalized strategy involving chromosome specific sub-telomeric probes specific for the subtelomeric regions of the translocated segments, combined with proximal probes in order to provide a fast and reliable approach to PGD for cases of reciprocal, Robertsonian translocations, inversions and other complex chromosomal rearrangements.

Malmgrem *et al* (2002) performed CGH on embryos diagnosed as abnormal after PGD for two Robertsonian and four reciprocal translocations. The group was able to confirm the results of CGH supported by the PGD results in 11 out of 15 embryos. It was revealed that all the embryos (100%) were mosaic and it was further observed that some couples were more prone to generate chaotic embryos than others as shown previously by Delhanty *et al* (1997). However, imbalances involving a translocation with a very distal breakpoint could be missed as CGH resolution was limited and unreliable ratio changes appeared at the telomeric regions.

Several groups have revealed high rate of mosaicism and chaos during PGD for reciprocal translocations. Van Assche *et al* (1999) after analyzing 35 embryos from carriers of the most common reciprocal translocation, involving chromosomes 11 and 22 (see section 1.3.1.2.2), revealed that use of four-colour FISH analysis allowed the identification of all the possible segregation modes using commercially available probes. The most common segregation pattern in reciprocal translocations is alternate, although considerable variations in the proportions of segregants in reports of male carriers exist (Iwarsson *et al*, 2000). Estop *et al* (1995) reported an average of 47% of alternate segregation mode and a very low rate of adjacent-2 segregation.

It has been postulated that two mechanisms exist giving rise to the production of unbalanced gametes in patients with reciprocal translocations: a) one from meiotic crossing over involving the critical region between the centromere and the breakpoint and b) the other from abnormal meiotic segregation (Brandriff *et al*, 1986; Scriven *et al*, 1998). By contrast, Robertsonian translocations result in unbalanced gametes only as a consequence of abnormal meiotic segregation because there is no critical region (Munne *et al*, 2000).

1.5.2.2.3 Inversions and Insertions.

The majority of patients requesting PGD for inversions carry a pericentric inversion with a very large inverted segment. The FISH protocol devised for inversions utilises breakpoint spanning probes and were originally presented by Cassel and co-workers (1997) whilst performing PGD for inversions. The probe strategy devised for such cases involves a probe for one of the distal segments of the chromosome involved, which can then detect chromosome imbalance from either of the two possible recombinant chromosomes. Additional probes, with different fluorophores, on the same or different chromosomes can be added as a control as reported by Iwarsson *et al* (1998b).

PGD for insertions can be rather problematic since depending on whether synapsis takes place or not, several types of meiotic behaviour are possible that have to be taken into consideration while devising a strategy. The complexity of such cases is increased as the orientation of the inverted segment is often impossible to detect. Therefore, the use of a subtelomeric probe along with a locus specific for the inserted segment is essential.

1.5.2.3 Numerical Aberrations

The major causes of the decline in implantation observed with increased maternal age and embryo incompetence is an euploidy. Navot *et al* (1994), transferred embryos from younger women to women >40 years of age and observed a high implantation rate suggesting that the latter group's ability to become pregnant is largely unaffected, whereas their oocyte quality is compromised. Altered oocyte metabolism such as ATP production (Van Blerkom *et al*, 1995) and excessive deposition of ZP glycoproteins

(Garside et al, 1997) have also been linked with advanced maternal age. The increase in aneuploidy and maternal age seen in spontaneous abortions and live offspring after genetic analysis (Antonarakis et al, 1991; Fisher et al, 1991) has also been observed in unfertilised oocytes (Dailey et al, 1996) as well as human cleavage-stage embryos (Munne et al, 1995a and b). However, in the case of preimplantation embryos and oocytes, the rate of chromosomal abnormalities was significantly higher, suggesting that a proportion of chromosomally abnormal embryos are eliminated before any prenatal diagnosis (Munne et al, 1999). Close correlation between aneuploidy and declining implantation rates, prompted the hypothesis by Munne and co-workers (1993b) that selection of chromosomally normal embryos could reverse this trend. However, while some research groups have shown a positive attitude towards this kind of selection (Verlinsky and Kuliev, 1996b; Gianaroli et al, 1997a; 1999) other have doubted its value and feasibility (Egozcue, 1996; Reubinoff and Sushan, 1996). As well as being labour intensive PGD reduces the number of embryos available for transfer and the error rate is estimated to be 9-15% (Munne et al, 1998c). Clear benefits of this technique in terms of live birth rate per initiated cycle have yet to be shown in any large-scale prospective controlled study that would properly evaluate this technology and its effects (Braude et al, 2002). An international clinical trial is needed, with a suitably matched control group to determine if this procedure benefits older-aged IVF patients. Such trial would involve clinical settings with similar (identical if possible) stimulation protocols, IVF media, diagnostic procedures in different laboratories. The trial would consist of three groups of patients: women <35 years of afe, women >35 years of age and the last "blind" group would have a mixture of ages. All patients should be assessed for the type of infertility and should be "healthy" individuals.

Couples opting for PGD of aneuploidy are mainly infertile and undergoing IVF/ICSI to overcome their infertility. There are three main indications for aneuploidy screening: 1) advanced maternal age (AMA), 2) recurrent miscarriage (RM) and 3) recurrent implantation failure (RIF). These patients are thought to be predisposed to producing aneuploid embryos which would die at or before the time of implantation (Wilton, 2002).

1) Patients >39 years of age show the highest increase of an uploidy (Marquez *et al*, 2000), however more recent data has suggested that PGD for an uploidy can display an increase in implantation even for women between 35-39 years of age (though, not shown to be statistically significant) (Munne *et al*, 2003). Staessen *et al* (2004) in clinical randomised study showed that PGD-AS does not improve clinical outcome per initiated cycle in patients with AMA when there are no restrictions in the number of embryos to be transferred.

2) RM in patients with a normal karyotype is defined as three or more consecutive spontaneous abortions of less than 20-28 weeks' gestation (Stephenson, 1996). PGD–AS has been performed in this category of patients (Pellicer *et al*, 1999; Rubio *et al*, 2003), however, no differences in pregnancy or implantation have been observed.

3) RIF is defined as three or more failed IVF attempts or implantation failure after the replacement of more than 10 embryos (Munne *et al*, 2002). A study by the ESHRE PGD Consortium Steering Committee (2004) which covers cases from 25 centres, reported a pregnancy rate for RIF of only 7% per retrieval, compared to 28% for PGD of aneuploidy cases with the indication of AMA or RM. Gianaroli *et al* (2001b) reported results on 66 PGD cycles of RIF patients, which did not find any statistical significance between the implantation rate of the PGD group (17%) and the control group (10%). From those studies it can be postulated that there is no clear indication that RIF patients have benefited from PGD for aneuploidy.

FISH allows enumeration of chromosomes on preimplantation embryos of common aneuploidies (13, 14, 15, 16, 18, 21, 22, X and Y) testing either blastomeres from cleavage-stage embryos or oocyte polar bodies (Munne *et al*, 1993b; 1995a,b; 1998c,e; Verlinsky *et al*, 1996b; Gianaroli *et al*, 1997a). Currently, probes for chromosomes 13, 14, 15, 16, 18, 21, 22, X and Y are being used simultaneously (with re-probing)(Bahçe *et al*, 2000), with the potential of detecting 70% of the aneuploidies detected in spontaneous abortions. Inclusion of probes for chromosomes 16 and 22 is of particular importance as trisomies of these two chromosomes are the two most common autosomal aneuploidies observed in spontaneous abortions (Wilton, 2002). Thus far, more than 2000 cases of PGS have been performed, resulting in >400 chromosomally normal babies (Sermon *et al*, 2004). Misdiagnoses have been reported after PGD for an euploidy (Munne *et al*, 1998c; Gianaroli *et al*, 2001; PGD Consortium 2005). In all of those cases, reanalysis of the misdiagnosed cells with probes binding to a different locus confirmed prior results, indicating that the errors were probably caused by mosaicism (Munne, 2003; see section 3.1.1). Other sources where misdiagnosis can occur include false monosomies produced by signal overlaps due to the failure of the FISH technique to display the remaining signals (Ruangvutilert *et al*, 2000a and b). Causes of reduced FISH hybridisation efficiency can be attributed to loss of DNA during denaturation or fixation, poor probe penetration, insufficient binding of detection reagents or overlap of chromosome-specific signals when multiple probes are used (Ruangvutilert *et al*, 2000b; Munne *et al*, 2002)

Munne and colleagues (1999) revealed that couples undergoing PGD for common aneuploidies showed a significant decrease in cases of spontaneous abortions (from 23% to 9%), whereas the ongoing pregnancies and deliveries in the PGD group of patients increased from 10.5% to 16.1%. Furthermore, several studies have displayed that after PGS, the incidence of multiple embryos being transferred has been significantly reduced (Munne *et al*, 2003; Werlin *et al*, 2003). However, this might be due to the fact that there are fewer chromosomally normal embryos to select after PGD. After 10 years of PGS, only recently has it been shown that PGS can increase implantation rates, when nine chromosomes were screened using FISH (Munne *et al*, 2003a). In this study the PGD group showed a 20% implantation rate compared to a 10% rate in the control group in women with an average age of 40 years (Munne *et al*, 2003).

Recently PGS has been carried out on using CGH (Wilton *et al*, 2001; Wells *et al*, 2002; section 1.4.1.2).

1.5.2.4 Single Cell FISH Limitations

FISH has been hailed as one of the most robust and efficient techniques however, it has certain limitations. When using FISH to analyse chromosome constitutions, several obstacles emerge, including failure of hybridisation, probe inefficiency, signal overlapping yielding false negative results and split/diffused/patchy signals (Munne *et*

al, 1998a; Ruangvutilert *et al*, 2000a). Most importantly FISH is limited by the number of probes that can be simultaneously applied due to an increasing chance of FISH artefacts and FISH failure and lack of colours, since it is not possible to look at all chromosomes in one interphase (Ruangvutilert *et al*, 2000b). However, as many as six different chromosomes have been simultaneously analysed by Munne *et al* (1998c) and Gianaroli and colleagues (1999), achieving a total of nine chromosomes studied per cell with a second sequential hybridisation.

1.5.3 Single Gene Disorders

Almost any source that contains one or more intact target DNA molecules can, in theory, be amplified by PCR, providing appropriate primers can be designed. This has been of great interest to researchers as well as clinical geneticists, forensic scientists and even scientists involved in archaeological biological findings. Single cell PCR has enabled geneticists to carry out PGD. By employing PCR, the 5-10pg of DNA in a single blastomere is amplified in order to make the DNA amenable to conventional forms of mutation analysis (Wells and Delhanty, 2001). Over 8,000 disorders caused by single gene defects have been described. The first autosomal single gene disorder to be analysed by PGD was cystic fibrosis more than ten years ago (Handyside et al, 1992). Since then, methods and procedures have improved, allowing more and more PGD protocols to be devised for monogenic diseases. To date many single gene defect diseases can be treated with PGD. Fluorescent dyes, multiplex reactions as well whole genome amplification techniques have been employed during PGD with PCR (see sections 1.5.3.1-3). Several inherent difficulties that are associated with single cell DNA amplification have become evident. These include potential sample contamination, total PCR failure, allelic dropout and preferential amplification, all of which should be minimised for any PGD PCR protocol before clinical application (see section 1.5.3.4). In addition, the chosen method must reliably and accurately characterise the genotype of the embryo relative to the disorder under investigation.

Initial work to develop PGD for single gene diseases began with the common disorders for which mutation profiles and prenatal diagnostic tests were already well established such as CF and the haemoglobinopathies (Monk *et al*, 1988; Monk and Holding, 1990; Liu *et al*, 1992). The first single gene defect to be diagnosed was

cystic fibrosis which was accomplished using simple heteroduplex analysis to identify unaffected homozygous normal and heterozygous embryos for transfer (Handyside *et al*, 1992). Since then increasingly sophisticated PCR-based protocols have been developed and applied for many single gene defects on a list which is growing steadily along with patient demand and technological advances (Wells and Delhanty, 2001). These include Tay-Sachs disease (Gibbons *et al*, 1995), Duchenne muscular dystrophy (Lui *et al*, 1995), Marfan syndrome, (Harton *et al.*, 1996), spinal muscular atrophy (Dreesen *et al.*, 1998), Lesch-Nyhan syndrome (Ray *et al.*, 1999), sickle cell anaemia (Xu *et al.*, 1999) as well as the triplet repeat expansion disorders Huntingtons disease (Sermon *et al.*, 1999), myotonic dystrophy (Sermon *et al.*, 1998a,b; Piyamonkol *et al.*, 2001), Fragile X (Sermon *et al.*, 1999; Apessos *et al.*, 2001) and inherited cancer syndromes, familial adenomatous polyposis coli (FAPC) (Ao *et al.*, 1998), Li-Fraumeni syndrome (Verlinsky *et al.*, 2001) and neurofibromatosis type-1 and type-2 (NF-1 and NF-2) (Abou Sleiman *et al.*, 2002).

PGD on the 1st polar body (PB) was first introduced by Verlinsky et al (1992). However, this analysis was not sufficient since the genotype of embryos resulting from diagnosed heterozygous oocytes was not predictable and testing of the 2nd PB was required. Therefore, Verlinsky and co-workers (1997) carried out 1st and 2nd PB analysis for PGD of sickle cell disease, haemophilia B and cystic fibrosis. Furthermore, the same technique was later performed on patients at risk of producing offspring affected with Neurofibromatosis type-I (NF-1) (Verlinsky et al, 2002). The same group have continued using the 'two-step PB analysis of oocytes' as their preferred method for PGD of single gene disorders, because they have reported that ADO is at least half as frequent in PB's than in blastomeres (Rechitsky et al. 1998), which has not been by corroborated by any other study. Dean and co-workers (2003) introduced the prospect of PGD for heritable mitochondrial diseases on polar bodies using PCR. The group studied the levels of heteroplasmy on polar bodies, oocytes and embryos from a heteroplasmic mouse model to assess the relative proportions of each of the two mitochondrial DNA genotypes in gametes and early cleavage stage embryos. They concluded that PGD for mitochondrial DNA diseases is feasible, although it should be approached with caution (Dean et al, 2003). Currently, most PGD cases involving single gene defects are performed on blastomeres of cleavagestage human embryos, thus making PB analysis at the single gene level scarce.

1.5.3.1 Microsatellite DNA (STR's)

Microsatellite DNA is a subclass of tandemly repeat DNA found throughout the human genome mostly in non-encoding loci and is also referred to as short tandem repeats (STR's) (Ellegren, 2004). The most common type of microsatellite DNA are dinucleotide repeats, comprising 0.5% of the genome, whilst the total microsatellite DNA comprises approximately 2-3% of the genome (IHGSC, 2001). Other types of microsatellite DNA include mono-, tri- and tetra-nucleotide repeats. A significant feature of microsatellite DNA is that its polymorphic nature i.e. the variability in repetitive DNA is in the number of repeat units, which confers variability in length rather than sequence. STR's have been used extensively for PGD of single gene disorders to overcome the problems of ADO and contamination (Sermon *et al*, 2001; Piyamongkol *et al*, 2001a and b). This has been achieved by carrying out multiplex reactions incorporating one to two STR's in addition to the mutation marker (Piyamongkol *et al*, 2001a)

During STR amplification, the incidence of so-called 'shadow bands' have been reported by many groups especially during single cell PCR. It has been suggested that such bands, termed stutter bands, represent mutations in the repeat sequence and are thought to have arisen due to replication slippage (Hauge and Litt, 1993). Replication slippage occurs due to mispairing between two complementary DNA strands during DNA replication. One strand dissociates from its complementary strand during replication, only to re-associate to its complementary strand incorrectly, causing an increase or a decrease in one repeat unit length. It has been noticed that stutter bands are more prominent in amplification of mono- and dinucleotide repeats and are usually one repeat unit length shorter than the main allele (Ellegren, 2004). Stutter bands usually cause problems when allele scoring of heterozygote individuals whose alleles are close in size.

Multiplex PCR reactions during single cell work are designed to alleviate the problem of ADO and contamination. The need to design protocols which include amplification

of more than one (singleplex) primer has mostly been the case during PGD cases. Incorporation of STR's linked to the disease gene or unlinked i.e. located in another chromosome require alterations in the PCR protocol. Laborious and time-consuming testing is required especially whilst performing on single cell DNA in order to acquire analysable results from more than 2 primers when multiplexing. Primers must be designed, ideally, to have similar annealing temperatures and must not be competing for reagents such as the dNTP's or the Mg⁺². Furthermore, when designing PCR protocols for single gene disorders, a variety of STR's may be informative for different families. Thus, the need to develop several multiplex protocols according to each family and in single cell DNA can be time consuming and not cost-effective.

1.5.3.2 Single Cell F-PCR

The introduction of F-PCR (see section 1.3.3.2.1) has boosted single cell analysis driving scientists to achieve analysis as well as diagnosis of single gene mutations accurately. It has been concluded that F-PCR was an efficient tool for PGD of Steinert's disease (myotonic dystrophy) (Sermon *et al*, 1997; Piyamongkol *et al*, 2001a). Currently, most diagnoses before implantation for inherited monogenic diseases are carried out using multiplex F-PCR (see section 1.5.3.2). In this type of diagnosis it amplifies, short tandem repeat (STR) markers which are highly polymorphic loci of differing numbers of a 2-5 base pair repeated unit. The polymorphic nature of STR markers allows the determination of the origin and purity of DNA amplified e.g. from single cells. Hence, if the STR allele sizes are known, any impurities or unexplained sizes can be distinguished. Furthermore, in extreme cases of preferential amplification, with conventional PCR product detection techniques it may seem as ADO, whereas F-PCR may still demonstrate the presence of both alleles with one greatly amplified relative to the other (Sherlock *et al*, 1998).

1.5.3.3 Single Cell Multiplex PCR

Single-cell PCR has the disadvantage that just one amplification reaction can be undertaken. When several different primers are combined within a single PCR assay, each can multiply independently and sufficiently, allowing simultaneous diagnoses to be performed. Furthermore, multiplex PCR can alleviate problems caused by ADO (Kuliev *et al*, 1998). When a disease-causing mutation marker and an informative polymorphism for the respective mutation are multiplexed, there are two chances of for the detection of a chromosome carrying a mutant gene. Thus, if one primer does not amplify due to ADO, there is another primer for backup. It has been postulated that ADO is independent for each fragment amplified in a multiplex PCR assay and the probability of ADO affecting both the mutation site and the linked polymorphism are very low (Ao *et al*, 1998). Lewis et al (2001), developed a a mathematical model to explore accuracy of PGD using PCR by including both extrinsic technical errors and intrinsic errors related to nuclear and chromosomal abnormalities. It was suggested in that study that a linked marker or a second biopsied cell reduces the probability of replacing an affected embryo (Lewis *et al*, 2001)

1.5.3.4 Single Cell WGA

WGA can be used for amplification of low copy numbers of the entire genome (Zhang *et al*, 1992) and provide sufficient DNA templates for many independent PCR amplifications (Wells and Sherlock, 1998). Hence several studies have been performed assessing the use of WGA in the context of PGD.

Ao *et al* (1998) was able to perform PGD for familial adenomatous polyposis coli (FAPC), after amplifying two biopsied blastomeres from each embryo with PEP and subsequently carrying out nested PCR to amplify two APC fragments. Following PEP the APC mutation and the linked polymorphism achieved 87.5% and 75% amplification respectively.

Applying DOP-PCR to single fibroblasts, buccal cells, amniocytes and human blastomeres, Wells *et al* (1999) showed that it could provide DNA sufficient for performing 100 subsequent PCR amplifications as well as CGH analysis. Voullaire and colleagues (1999) showed that DOP-PCR was able to successfully amplify single cells in such quantities able to provide diagnosis of aneuploidy using CGH. Despite hopes that WGA strategies might reduce the incidence of allele drop out, Wells and Sherlock (1998) found ADO rates after PEP and DOP-PCR to be comparable to those obtained by direct amplification of single cell loci.

MDA is a method of whole genome amplification that utilises the bacteriophage $\varphi 29$ DNA polymerase for isothermal displacement amplification (Dean et al, 2001; see section 1.3.3.1.3). Recently, Handyside *et al* (2004) reported that isothermal WGA from single and small numbers of lymphocytes and blastomeres isolated from cleavage stage embryos yielded microgram quantities of amplified DNA. 20 different loci were successfully analysed, but a relatively high ADO of 31% at heterozygous loci was found (Handyside *et al*, 2004).

1.5.3.5 WGA and STR's

Whole genome amplification (WGA) has been reported by Wells and colleagues (1999), as an efficient means to generate sufficient quantities of DNA for as many as 90 independent amplification reactions from a single cell. Thus, numerous specific loci and subsequently copy number of every chromosome (using comparative genomic hybridisation, CGH) can be assessed in a single cell.

WGA may be defined as the non-specific amplification of all sequences in the genome (Wells and Sherlock, 1998). There are different types of WGA, with three techniques being the most notable in context of embryo research and PGD, namely PEP, DOP-PCR and MDA (sections 1.3.3.1.3 and 1.6.3.3). PEP has been used clinically in the context of PGD of familial adenomatous polyposis coli (Ao et al, 1998). DOP-PCR has been found to be the choice of method, especially for CGH since it results in larger amounts of DNA being produced (Wells et al, 1999; Chapter 4). It has been proposed that WGA would be beneficial for patients undergoing PGD for single gene disorders who are also of advanced maternal age since detection of the single gene disorder as well as chromosome complement of the embryos can be performed (Wells, 2004). WGA could reduce the significance of the risks of ADO and achieve enhanced detection of contamination without the need to perform multiplex reactions. However, this is based on the assumption that a single cell can be amplified in full, without bias, such that one could rely on subsequent tests performed on the DOP-PCR product. By performing microsatellite marker genotyping of the DOP-PCR amplified product, this would allow testing for coverage of the genome.
Cheung and Nelson (1996) first attempted to amplify different types of STR's after WGA with DOP-PCR. The study was carried out on genomic DNA and it was concluded that DOP-PCR provides relatively good coverage of the genome, however, is dependent on the amount of starting DNA. Wells and co-workers (1999) studied DOP-PCR on single cells, in addition to other WGA techniques, in order to investigate the most suitable WGA protocol for CGH analysis. The idea of producing enough DNA from a single cell and then perform separate PCR reactions would be ideal for PGD of single gene disorders.

1.5.3.6 Single Cell Mini-Sequencing

The amount of time and resources required for the design and validation of a novel PGD strategy limits the number of protocols that a laboratory can develop. Furthermore, developing a PGD strategy is not cost-effective especially for rare mutations which can only be used for a handful of families; hence, a universal methodology applicable to several mutations would accelerate the rate of which new methods can be developed and reduce the cost (Bermudez et al, 2003). Minisequencing has recently been proposed as an alternative technique able to tackle disease involving a heterogeneous spectrum of mutations, such as cystic fibrosis, β thalassaemia or haemophilia A in PGD (Fiorentino et al, 2003). The basis of minisequencing is similar to that of sequencing apart from the fact that the nucleotides added to the reaction are of the labelled dideoxy type (ddNTP) i.e. only a single nucleotide can be added to the primer thus preventing further extension. The primers designed for mini-sequencing anneal to the template at the 3' end, one nucleotide upstream of a known mutation/polymorphism. The ddNTP added to the primer is able to reveal the identity of the base of this site. Bermudez et al (2003) assessed minisequencing as a method for single cell analysis and concluded that although expensive compared to established methods, it is rapid and accurate in diagnosing diseasecausing mutations in single cells and the near-universal applicability of this method could shorten the time required for devising PGD protocols. Fiorentino and coworkers (2003) were able to interpret results from 96.5% of the 887 blastomeres tested, including 55 PGD cases. This technique may be particularly useful in cases where the mutation involved is difficult to assess by restriction analysis or other commonly used methods.

1.5.3.7 Single Cell PCR Limitations

1.5.3.7.1 Contamination

Single cell PCR assays usually employ a large number of cycles to be carried, especially for sufficient amplification of a single cell. Contamination can be caused by cumulus cells of maternal origin, sperm, culture media and the PCR products present in the laboratory environment (Delhanty, 1998). To improve the PCR technique the protocol has changed over the years to reduce contamination. The introduction of ICSI, where the oocyte is fertilised by the injection of a single sperm into the cytoplasm, thus avoiding excess sperm often left embedded in the zona pellucida following IVF, reduces paternal contamination (Wells and Delhanty, 2001). In addition some PGD protocols attempt to detect contaminants by amplifying a highly polymorphic locus for DNA fingerprinting (Findlay, 2000; Piyamongkol *et al*, 2001a). Single-cell PCR should be set up in a DNA-free environment away from the analysis area, which can reduce the chance of 'carry over' contamination.

Nested PCR was developed to increase sensitivity and specificity (Monk and Holding, 1990) addressing the problem of "carry over" contamination. Nested PCR is widely used in PGD and is based on the use of two sequential amplification reactions in order to enhance the specificity of PCR and reduce the risk of contamination caused by the accidental amplification of DNA fragments (Wells and Sherlock, 1998). The use of nested PCR by aliquoting the first amplified products for use as the target template in a second reaction using another set of primers (markers) situated internally to those used in the first reaction was suggested to prevent carry over contamination (Wells and Sherlock, 1998). Furthermore, Multiplex-PCR may substantially decrease the possibility of misdiagnosis (Lewis et al, 2001), by providing the added assurance of a partial "fingerprint" of the embryo, and confirming that the amplified fragment is of embryonic origin (Findlay et al, 1995). Piyamongkol et al (2000a) showed that the use of DNA fingerprinting can increase the detection of the contaminants. Ideally linked (Rechitsky et al, 1998; Dressen et al, 2000; Piyamongkol et al, 2001a and b) or unlinked polymorphic markers, which are informative for the family undergoing PGD should be employed (Kuliev et al, 1998; Ioulianos et al, 2000; Harper et al, 2002). For these markers the parents should have four different alleles e.g. A and B (mother), C and D (father); hence, during analysis of the embryo it should have one allele from

each parent e.g. AC, AD, BC or BD. Any other pattern e.g. ABD or ACD would indicate presence of contaminants. Moreover, multiplex F-PCR is considered approximately 1000-fold more sensitive than conventional agarose or acrylamide gel analysis (Findlay, 2002). Despite all efforts, paternal and maternal contamination has caused misdiagnosis (Sermon *et al*, 1998; Harper and Delhanty, 2000).

1.5.3.7.2 Amplification Failure

The problem of AF of single cell PCR emerged when a misdiagnosis from the first series of PGD for X-linked disorders was reported (Hardy and Handyside, 1992). Whilst amplifying single cells, even an experienced PCR practitioner might face the obstacle of total failure of amplification. AF can between 10-20% (Kontogianni *et al*, 1996; Wells and Sherlock, 1998; Piyamongkol *et al*, 2003) and the underlying reasons of this relatively high incidence are difficult to determine. The cause of AF cannot be distinguished empirically. However the loss of the isolated single cell during transfer into the PCR tube or the chance that the cell is anucleate or in the process of degeneration have been considered as likely candidates. Furthermore, DNA in the cell nucleus might not have been made accessible to the PCR reagents due to failure of cell lysis (Piyamongkol *et al*, 2003).

1.5.3.7.3 Allele Dropout

ADO is the amplification of only one of the two parental alleles present in the single cell, which is usually caused due to sub-optimal PCR conditions and rapid degradation of the target DNA during thermocycling (Handyside *et al*, 1997). Several factors have been linked with the struggle against ADO including increased denaturation temperature (Ray *et al*, 1996; Lissens and Sermon, 1997), different lysis methods (Sermon *et al*, 1995) and F-PCR (Findlay *et al*, 1995a). When ADO is encountered it gives the perception that the cell is homozygous which can lead to misdiagnosis when applying PCR for PGD analyses in dominant disorders. If the affected allele does not amplify, the embryo will be diagnosed as normal, when actually it is affected. For autosomal recessive diseases in couples carrying the same mutation, the consequences of ADO are minimal; the loss of the normal allele in carrier embryos will result in an apparently affected embryo that will not be replaced. In the case where the affected allele is lost, this will lead to diagnosis of a homozygous normal embryo instead of a carrier (Fasouliotis and Schenker, 1998). In

a recent large study of single blastomeres and single buccal cells all possible aspects of PCR were investigated regarding their effect on allele drop-out. It was revealed that ADO is affected by amplicon size, amount of DNA degradation, freezing and thawing, the PCR programme and the number of cells simultaneously amplified (Piyamongkol *et al*, 2003). In the same study, factors which had little or no affect on ADO were local DNA sequence, denaturing temperature (94°C or 96°C) and type of cell (buccal or blastomere). Fluorescent PCR technology (See section 1.2.3.1.2) is having a wide impact on the PGD of single gene disorders proving to be more sensitive, reliable, accurate, and fewer cycles are required, thereby reducing the time taken to reach diagnosis (Findlay *et al*, 1996; Sermon *et al*, 1998). Furthermore, F-PCR technology has been found to be amenable to automation.

A different approach, which involves the combination of FISH and PCR techniques, has been reported by Thornhill and colleagues (1994) termed cell recycling. Cell recycling is performed by fixing a cell onto a glass slide and initially carrying out PCR and then FISH. These combined analyses can serve independently for sexing for X-linked diseases or for PGD of single gene defects. The combined efficiency of both techniques ranges from 65-85% (Monk and Thornhill, 1996; Rechitsky *et al*, 1996). However, the ADO rates were reported to be considerably higher than in conventional single cell PCR, in both studies above.

1.5.4 PGD for DM

Sermon and colleagues (1997) were the first group to develop a single cell PCR assay able to detect DM affected embryos during clinical application of PGD. It was reported, in this first attempt of PGD for DM, that ADO affected 24% of the biopsied cells in the clinical setting and 21% of the cells in the research setting (Sermon *et al*, 1997). Although, these figures are considered high, there was no use of multiplex or fluorescence PCR techniques. The same group was able to increase their amplification efficiency from 87% to 95% during another study where F-PCR and automated fragment analysis was performed and significantly decrease their ADO rate (4.5%) in a clinical setting (Sermon *et al*, 1998a). However, the same group reported a misdiagnosis case during the latter study (Sermon *et al*, 1999). Piyamongkol *et al* (2001a) was among the first to carry out PGD on DM patients using the combination

of multiplex and fluorescent PCR. By incorporating informative polymorphic markers (single tandem repeats), which are linked to the disease gene and a marker for the mutation, affected embryos can be distinguished thus reducing ADO and AF rates. Furthermore, polymorphic markers, which not linked to the disease gene, can act as contamination markers able to detect contamination of any sort. In an ideal PCR protocol a mutation marker, a linked marker and a contamination marker should be multiplexed (as long as they are informative for the respective patients) to minimise the risks associated with single cell PCR. Dean *et al* (2001) were able to reduce the ADO rates (during clinical PGD) using a hemi-nested multiplex F-PCR protocol to 0% for the DM locus and 18% for the linked polymorphic marker.

1.6 Mosaicism in Human Preimplantation Embryos

The advent of IVF as a treatment for infertility has created the opportunity to study the chromosomal constitution of surplus human preimplantation embryos. An increasing body of evidence suggests that the incidence of chromosomal abnormalities in embryos is extremely high (Wilton, 2002) and even good embryo morphology does not necessarily exclude an abnormal chromosome constitution (Magli *et al*, 2000).

Mosaicism is defined as the presence in an individual or in a tissue of two or more cell lines, which differ in their genetic constitution but are derived from the same zygote. Little is known about the mechanisms of mosaicism. Sometimes mosaicism is described for chromosomal trisomy and diploidy however, the abnormal cell line may be represented by other chromosomal abnormalities such as sex chromosome monosomy, triploidy or structural rearrangement (Kalousek, 2000).

Delhanty *et al* (1993) first noted the presence of mosaicism as a common feature in preimplantation embryos whilst analysing the X and Y chromosomes using interphase FISH. A high degree of mosaicism has been detected in several FISH studies (Delhanty *et al*, 1993, 1997 (Table 1.1); Harper *et al*, 1994, 1995; Munne and Weier, 1996 (Table 1.2); Munne *et al*, 1998c, d, e; Iwarsson *et al*, 1999; Ruangvutilert *et al*, 2000a), karyotyping studies (Jamieson *et al*, 1994; Clouston *et al*, 1997, 2002) and CGH studies (Wells and Delhanty, 2000; Voullaire *et al*, 2000; Malmgrem *et al*,

2002). The frequency of chromosomal abnormality in early cleavage embryos has also made a major impact in IVF and human reproductive biology. It is well known that in routine IVF, at the cleavage stage, each transferred embryo has only about a 1 in 5 chance of implantation (Delhanty and Harper, 2000).

| Table 1.1 | Classification | of human | embryonic | mosaicism | adapted | from | Delhanty of | et al |
|-----------|----------------|----------|-----------|-----------|---------|------|-------------|-------|
| (1997) | | | | | | | | |

| Explanation | | | | |
|---|--|--|--|--|
| Embryo uniformly normal for the | | | | |
| chromosomes tested | | | | |
| Embryo uniformly abnormal for the | | | | |
| chromosomes tested | | | | |
| Majority of embryo euploid but one or a | | | | |
| few cells differ (i.e. aneuploid, polyploid | | | | |
| or haploid). | | | | |
| Majority of embryo uniformly aneuploid | | | | |
| or polyploid but one or a few cells differ | | | | |
| Chromosome constitution varies | | | | |
| randomly from cell to cell and status of | | | | |
| original zygote cannot usually be | | | | |
| determined | | | | |
| | | | | |

Table 1.2. Chromosomal patterns in cleavage stage embryos as described by Munne and Weier (1996)

- (i) **Normal** Embryos whose chromosome constitution is >90% either normal or aneuploidy or haploid or polyploidy.
 - Mosaic embryos containing a majority of diploid cells*
 - a. Diploid mosaics
 - i. Majority of diploid embryos with few being aneuploid, haploid or teraploid
 - ii. Embryos which contain a diploid and a polyploid cell line (2n/4n)
 - b. Polyploid mosaics embryos with a polyploid cell line
 - c. Haploid mosaics embryos with a haploid cell line
- (iii) Mosaic

(ii)

- a. Extensive embryos with >38% of abnormal cells
- b. Limited embryos with <38% of abnormal cells
- (iv) Chaotic nuclei showing randomly different chromosome complements

*This category includes mosaic embryos that contain a majority of abnormal cells (e.g. due to presumptive meiotic error and also embryos in which the majority of cells are normal).

It has been suggested that all types of mosaicism have the same impact on embryo development (Sandalinas *et al*, 2000). Delhanty *et al* (1997) classified the chromosome patterns in cleavage stage embryos into four groups: uniformly normal, uniformly abnormal, mosaic and chaotic (Table 1.1). These observations were confirmed by Clouston *et al*, (1997), who karyotyped nuclei from 6- to 8-day human blastocysts and proposed that all four groups were also observed at that stage. Furthermore, the same patterns have been observed in human blastocysts (Ruangvutilert *et al* 2000a).

The delineation of the extent and nature of mosaic and non-mosaic chromosome abnormalities in early human preimplantation embryos is important for understanding the origins and selective processes leading to the anomalies seen later in gestation. Therefore it is important to try to understand the underlying mechanisms responsible for mosaic cell lines. The different types of mosaicism and their underlying mechanisms are better understood by reviewing the different types of analysis that have been used to investigate embryos

1.6.1 Mosaicism and Karyotyping

Karyotyping was amongst the first technique to be used for embryo analysis. It is a robust staining technique using a variety of dyes including DAPI (4, 6-diamino-2-phenylindole) and Giemsa to produce characteristic bands along each chromosome in order to carry out chromosome enumeration and structural analysis (Ronne *et al*, 1990; Glassman, 1997; section 1.3.1).

Angell *et al* (1983) whilst examining 8-cell stage embryos revealed a high incidence of chromosomal abnormalities which she suggested was contributing to early embryonic loss and to the high failure rate after embryo transfer. This was the first report on haploid human embryos with an incidence of 20% suggesting parthenogenic activation of the oocyte, which was later confirmed by Plachot (1985). Angell *et al* (1986) carried out another karyotyping study trying to assess the contribution of lethal chromosome abnormalities to implantation failure and found non-disjunction giving rise to trisomy, monosomy, nullisomy, as well as structural abnormalities, haploidy and triploidy. In the same study it was revealed that chromosomally abnormal embryos could not be distinguished on morphological criteria from embryos of normal chromosomal constitution based on similar cleavage rates, which was also supported by Jamieson et al (1994). However, it was later concluded that only the embryos which were very fragmented and degenerated were shown to display a higher rate of chromosomal abnormalities (78%) when compared with morphologically healthy embryos (Plachot et al, 1989). Similar findings were observed by Pellestor et al (1995) which showed that the rate of abnormalities were significantly higher in dysmorphic embryos (86.6%) than in good quality embryos (36.6%). Hence, these findings confirm the prognostic value of the grading system as a means of eliminating a large proportion of chromosomally abnormal embryos (Pellestor et al, 1995). The most significant findings whilst carrying out cytogenetic analysis on human preimplantation embryos was the correlation of chromosomal aberrations and maternal age. Maternal age was found to be directly proportional to aneuploidy frequency (Angell et al, 1986; Plachot et al, 1987; 1989), particularly affecting aneuploidies of the small satellited chromosomes of Group G (Zenzes and Casper, 1991).

Chromosomal abnormalities on human preimplantation embryos have also been linked to certain parameters of IVF such as oocyte retrieval and constitution of culture media (Angell *et al*, 1983; Pellestor *et al*, 1995). Ovarian hyperstimulation might be involved in the immaturity or overmaturity of the oocytes retrieved having a serious effect on the fertilization process (Testart *et al*, 1989) as well as delayed IVF might also lead to triploidy, abnormal cleavage, and fragmentation (Plachot *et al*, 1988). Moreover, delay in gamete fusion could also lead to asynchronisation of both formation and migration of the male and female pronuclei, resulting in cleavage disturbance and chromosome set fragmentation (Ron-El *et al*, 1991).

Several karyotyping studies reported chaotic findings while trying to detect mosaic and non-mosaic chromosome abnormalities (Angell *et al*, 1986; Plachot *et al*, 1989; Papadopoulos *et al*, 1989; Jamieson *et al*, 1994; Clouston *et al*, 1997; Clouston *et al*, 2002) (Table 1.3). It has been proposed that hypodiploidy, hyperdiploidy and structural chromosome damage reflect the same basic phenomenon, termed uncontrolled or chaotic division (Clouston *et al*, 1997), predicted by FISH studies (Harper *et al*, 1995). Studies have shown significant levels of tetraploidy, usually mosaic, whilst karyotyping of about 19-23% (Angell *et al*, 1987; Jamieson *et al*, 1994; Clouston *et al*, 1997). From karyotypic analysis of human blastocysts it was suggested that mosaic tetraploidy might result due to failure of cytokinesis (Hardy *et al*, 1993). The production of mosaic tetraploidy as a common event in early embryogenesis was also confirmed by a recent study on blastocysts (Clouston *et al*, 2002). Mosaic monosomy was not detected in a study by Clouston *et al* (2002), due to technical difficulties and haploid cells were completely lacking in the same study

| Study | Papadopoulos | Jamieson et | Clouston et al, | Clouston et al, |
|----------------------|---------------------|-------------|------------------|------------------|
| Observation | <i>et al</i> , 1989 | al, 1994 | 1997 | 2002 |
| No. of embryos | 35 | 178 | 73 (blastocysts) | 182(blastocysts) |
| observed | | | | |
| Diploid | 40% | 22.5% | 67% | 68% |
| (including abnormal | | | | |
| cells) | | | | |
| Mosaic polyploid | 3% | 2.2% | 18% | 28% |
| (mainly tetraploidy) | | | | |
| Mosaic aneuploid | 9% | 19% | 7% | 5% |
| (mainly trisomies) | | | | |
| Chaotic cells | 5/35 | 6/178 | 1/73 | - |
| Structural | 26% | 1.1% | 26% | % |
| chromosome damage | | | | |

Table 1.3. Frequency of mosaicism in human preimplantation embryos after karyotypic analysis

* Some of the studies do not add up to 100%. This is due to the presence of abnormalities that cannot be classified

Although results observed from karyotypic analysis have shown that a certain level of mosaicism exists, several technical difficulties have limited the conclusion that can be drawn from these studies. These technical difficulties include problematic fixation

methods resulting in loss of chromosomes and over dispersed or poorly spread cells which are not in a single focal lane restricting the potential of analysis (Harper *et al*, 1995). Also, long colcemid exposure times are able to increase the mitotic index but this produces highly contracted chromosomes that exhibit chromatid separation and G-band poorly compromising the information obtained (Jamieson *et al*, 1995). The most important limitation during G-banding is that only a few metaphases can be obtained hence only a small proportion of the cells can be analysed.

Spectral karyotyping (SKY) is a technique which employs 24 chromosome specific probes and has been considered an alternative to conventional cytogenetic analysis (Schrock *et al*, 1996). Each probe is labelled with different proportions of five separate fluorochromes and observed by spectral imaging, providing a different colour for each human chromosome. It has been applied to human oocytes and polar bodies, being able to simultaneously detect specific aneuploidies as well as de novo structural abnormalities, such as acentric fragments, translocations and marker chromosomes (Marquez *et al*, 1998). The analysis on first polar bodies provided useful data for polar body genetic diagnosis (Marquez *et al*, 1998). However, currently the technique is known to be fairly unreliable especially as it requires good quality chromosome spreads (Wells and Levy, 2003). Therefore, it would be more suitably employed in a research rather than in a clinical setting

1.6.2 Mosaicism and FISH

FISH has been widely applied in studies of oocytes, polar bodies, spermatozoa, blastomeres and blastocysts (sections 1.3.2.2). High levels of mosaicism and chaotic embryos have been reported in all embryonic stages (Delhanty *et al*, 1997; Clouston *et al*, 1997). Chaotics or those with a majority of abnormal cells are unlikely to survive beyond implantation (Harper and Delhanty, 2000; Table 1.1). Chromosomal abnormalities and mosaicism have been described in arrested and morphologically abnormal embryos (Munne and Cohen, 1998) as well as in normally developing embryos (Harper *et al*, 1995; Delhanty *et al*, 1997). Different types of mosaicism have been reported through FISH analysis including aneuploid mosaics, olyploidy and haploid mosaics, chaotic mosaics and multinucleation. Thus each type should be explored individually to understand their underlying mechanisms. Each type of

mosaicism observed in human preimplantation embryos using FISH is explored in section 3.1.4, in order to fully understand the different forms of mosaic embryos and explain the mechanisms that various FISH studies have revealed. In Table 1.4, the data obtained from several studies carrying out FISH in human cleavage stage embryos and blastocysts is summarised. The table reveals high levels of mosaicism found in several studies both on cleavage stage embryos and human blastocysts.

| stage entoryes | by apprying I | | sugare me | | | | |
|---|------------------------------|---------|-----------|----------------------|------------------|-----------------------|---------|
| Study/Chromos | omes | NO. Of | | Abnormal | Mosaic | Mosaic | |
| observed | | embryos | Normal | | Diploid | Abnormal ³ | Chaotic |
| Delhanty et al (1 | 993) / X, Y | 4 | 50% | 25% | 25% | - | 0 |
| Munne et al (199 | 94) / | 67 | 0 | 2% | 52%# | 40% | 6% |
| 18, X & Y | | | | | | | |
| Coonen <i>et al</i> (19 | 94) / | 37 | 38% | 8% | 55% | - | - |
| Harper $et al$ (199 | (95)/1.7 | 35 | 54% | 9% | 23% | 0 | 14% |
| Kligman <i>et al</i> (1) | 996) / | 47 | 23% | 21% | 51% | 4% | 11/0 |
| 13. 18. 21. X and | 1 Y | | 2370 | 2170 | 5170 | | |
| Delhanty <i>et al</i> (1 1, 7, X and Y | 997) / | 93 | 48% | 2% | 19% | 4% | 26% |
| Munne <i>et al</i> (199 13, 18, 21, X and | 97) / 1 Y | 138 | 66% | 9% | 12%# | - | 14% |
| Laverge et al (19 | 997) / | 97 | 40% | 25% | 23% | -l | 12% |
| I, X and Y | | | 0.60.6 | | 100 /# | 1 1 2 2 4 | |
| Munne <i>et al</i> (199 13, 16, 18, 21, X | 98d) / , Y | 381 | 36% | 12% | 40%" | 12% | |
| Laverge <i>et al</i> (1998) / 1 X and Y [¥] | | 60 | 20% | 45% | 15% | - | 20% |
| Evsikov & Verli | nsky (1998) / | 86 | ** | 4% | 86% | 7% | 3% |
| Iwarsson <i>et al</i> (1 15, 16, 17, 18, X | 999) / and Y | 40 | 28% | 2% | 45% | 10% | 15% |
| Staessen <i>et al</i> (19 18, X and Y | 999) / | 94 | 35% | 11% | 40% | 3% | 11% |
| Veiga <i>et al</i> (1999)/ | Blastocysts | 8 | 12.5% | 0 | 75% | 12.5% | 0 |
| 13, 18, 21, X, and Y * | Arrested Embryos | 8 | 37.5% | 0 | 25% | 0% | 37.5% |
| Ruangvutilert et al (2000a) | Blastocysts | 19 | 10.5% | - | 68.5% | 10.5% | 10.5% |
| /13, 18, 21, X and Y * | Arrested Embryos | 20 | 20% | - | 70% | - | 10% |
| Harrison <i>et al,</i> (7, 8, 9, 13, 16, 18 | 2000) / 1, 4, 8, 21, X, Y | 6 | 0% | 17% | 50% | 17% | 17% |
| Sandalinas <i>et al,</i> 1, 13, 15, 16, 18, | (2001) / 21, 22, X, Y | 216 | 15% | 47% ^{&} | 15% [#] | | 23% |

Table 1.4. Summary of studies carried out on normally developing human cleavagestage embryos by applying FISH to investigate the extent of mosaicism

| Gonzalez-Merino <i>et al,</i> (2003) / 13, 18, 21, X and Y | 50 | 10% | | 68% | 20% | 2% |
|---|-----|-----|----|-----|-----|------------------|
| Baart <i>et al</i> , (2004) / 1, 7, 13, 15, 16, 18, 21, 22, X, Y | 22 | 45% | 5% | 45% | - | 5% |
| Coonen et al, (2004) / 18, X and Y* | 295 | 25% | - | 26% | 31% | 17% [@] |

* These studies were carried out on human blastocysts

[¥] The embryos were analysed after the freezing and thawing process

* No diploid/euploid nuclei present with no evidence of chaotic division

[§] Major cell line has an abnormal chromosome complement but diploid cells present

** The study did not distinguish between the normal and diploid mosaic

[#] Diploid mosaics are embryos with >38% of abnormal cells (Table 3.2)

[&] Including the 2n/aneuploid cell

[@] Including embryos classified as unexplained

When using FISH to analyse chromosome constitutions, several obstacles emerge, including failure of hybridisation, probe inefficiency, signal overlapping yielding false negative results and split/diffused/patchy signals (Munne et al, 1998; Ruangvutilert et al, 2000a, b). Most importantly FISH is limited by the number of probes that can be simultaneously applied due to an increasing chance of FISH artefacts and FISH failure and lack of colours, since it is not possible to look at all chromosomes in one interphase (Ruangvutilert et al, 2000b). As many as six different chromosomes have been simultaneously analysed by Munne et al (1998c) achieving a total of nine chromosome studied per cell with a second sequential hybridisation. However, by analysing many chromosomes simultaneously the efficiency of the FISH technique drops (Conn et al, 1998; Ruangvutilert et al, 2000b) thus producing false negative and positive results. Conn et al (1999) proposed that double locus analysis might decrease the number of chromosomes analysed per FISH analysis will however, provide true information about the extent of mosaicism present in human preimplantation embryos. Magli and co-workers (2001) carried out double locus analysis of chromosome 21 for PGD to reduce any false positives and/or negatives and thus the risk of misdiagnosis

1.6.3 Mosaicism and CGH

To date only a small number of embryos have been studied using CGH to analyse every cell (Table 1.5). Two studies were carried out on a series of good quality cleavage stage embryos. Both studies aimed to reveal the true extent of chromosomal abnormalities. Combining the results of the two similar studies conducted by Wells and Delhanty (2000) and Voullaire *et al* (2000), the most striking finding was that mosaicism was found to be extremely common affecting 67% and 64.6% respectively. However, Trussler *et al* (2004) observed a lower incidence of mosaicism of 50% whilst analysing 40 embryos. It was concluded in the latter study that the difference in normality in the three studies might be due to variations in maternal age, embryo quality, stimulation protocols and culture (Trussler *et al*, 2004).

Table 1.5. Summary of CGH data carried out on human preimplantation embryos revealing the level of mosaicism.

| Study | No. of embryos | Normal | Abnormal ^a | Mosaic Diploid | Partial Aneuploidy ^b | Chaotic |
|------------------------------|-------------------|--------|-----------------------|-------------------|------------------------------------|---------|
| Wells & Delhanty (2000) | 12 | 25% | 8% | 50% | - | 17% |
| Voullaire et al, (2000) | 12 | 25% | 8.3% | 58.3% | - | 8.3% |
| Wilton <i>et al</i> , (2001) | 5 | 20% | 40% | - | - | 40% |
| Malmgrem et al, (2002) | 28 | 0% | - | 14% ^c | 54% ^d | 32% |
| Voullaire et al, (2002) | 126 | 40% | 25% ^a | - | 6% | 29% |
| Trussler et al, (2004) | 40 | 42.5% | 7.5% | 37.5% | 5% | 7.5% |

^a Including mosaic aneuploid complements

^b Partial aneuploidies or structural damage induced by chromosome breakage

^c The embryos were mosaic diploid or mosaic aneuploid but balanced regarding the chromosomal rearrangement

^d The embryos were mosaic diploid or mosaic aneuploid but unbalanced regarding the chromosomal rearrangement

However, all three CGH studies have confirmed the FISH and karyotyping studies performed on cleavage embryos. In the study by Wells and Delhanty (2000) only one embryo (8.3%) was completely aneuploid where all six cells were trisomic for chromosome 21 and monosomic for the X chromosome and was thought to have arisen due to meiotic errors. The same results were obtained from Voullaire *et al* (2000), though Trussler and co-workers found slightly lower (7.5%) consistent abnormality resulting from a meiotic error. In both early studies 3/24 (12.5%) embryos revealed evidence of chromosome breakage resulting in imbalance of specific regions, rather than whole chromosomes, were detected, with one embryo showing reciprocal gains and losses of regions of chromosomes 2 and 7 in sibling blastomeres (Wells and Delhanty, 2000). In the recent study of Trussler *et al* (2004) chromosome breakage was again reported (7.5%). Furthermore, such chromosome

breakage has also been recorded in karyotyping studies (Papadopoulos et al., 1989; Zenzes and Casper, 1992; Clouston et al., 1997).

Voullaire et al (2002) used CGH for an euploidy screening and detected chromosome abnormality in 60% of single blastomeres biopsied prior to implantation from 20 women with repeated implantation failure. The abnormalities included aneuploidy for one or two chromosomes (25%) and complex chromosomal abnormality (29%). Mosaicism involving a complex abnormality (i.e. chaotic) is a more frequent occurrence in these patients than in the previously studied cohort of surplus embryos (Voullaire et al, 2000), and is therefore likely to be related to the history of recurrent implantation failure. This study supports the observation that some individuals are more prone to chaotic embryos than others as suggested by previous FISH studies (Delhanty et al, 1997; Harrison et al, 2000). In addition, the complex abnormality seen in morphologically normal and actively dividing embryos supports the idea that mitotic checkpoints may not function in the cleavage embryo (Delhanty and Handyside, 1995; Wells and Delhanty, 2000; Harrison et al, 2000), and it suggests that disturbance of the normal early embryonic cell cycle might be a pathology associated with infertility and implantation failure (Wilton et al, 2003). Recently, Wells et al (2005), after carrying out gene expression studies on human oocytes and embryos found that BUB1, MAD2 and APC genes were expressed in low quantities in 2-4cell stage embryos. These findings prompted the author to suggest that this might be significant for the level of mosaicism since these genes are involved in producing proteins that interact in the spindle assembly checkpoint which ensures accurate chromosome segregation.

A recent study by Malmgren *et al* (2002) analysed 94 blastomeres from 28 embryos generated from 13 couples carrying a balanced chromosomal rearrangement undergoing PGD. The single cell CGH confirmed most of the unbalanced translocations detected by PGD. As the embryos made available for this study were previously diagnosed as unbalanced regarding the chromosomes involved in the translocation or were considered unsuitable for transfer for other reasons a higher degree of mosaicism was expected in comparison to the other CGH studies (Wells and Delhanty, 2000; Voullaire *et al*, 2000). Indeed all of the embryos (100%) were classified as mosaic (containing more than one chromosomally uniform cell line) or

chaotic. In this study, a tendency for some couples to be more prone to generate chaotic embryos than others was also seen, as previously described by Delhanty *et al* (1997). In the study by Malmgrem *et al* (2002) there was a significant reduction in the efficiency of the CGH technique, compared to the previous studies (Wells and Delhanty, 2000; Voullaire *et al*, 2000) from 98% and 97% respectively to 70%. However, the dissimilarity was attributed to the fact that normal IVF generated embryos were used in the earlier studies in contrast to the unbalanced or not suitable for transfer embryos analysed in the Malmgrem study (Malmgrem *et al*, 2002). However, it is more likely the underlying reason was the use of different CGH techniques between the Malmgrem study (Malmgrem *et al*, 2002) and the other two early CGH studies (Wells and Delhanty, 2000; Voullaire *et al*, 2000; Voullaire *et al*, 2000).

Following the application of CGH on single blastomeres, concerns about the reliability of PGD using FISH to identify chromosomally normal from abnormal embryos were raised. In the case of FISH, probes target a defined region on a chromosome so the status of the rest of the chromosomes is simply assumed to be normal, but lacking conclusive proof. This fact strengthens the argument for the adaptation of CGH for clinical screening of embryos (Wells and Levy, 2003). However, hypothetically more than half of the abnormalities found in the CGH embryo studies could have been excluded using a limited FISH probe set (13, 18, 21, X and Y). Recently, in a study carried out by Trussler *et al* (2004) the combination of CGH and FISH was assessed in cleavage stage embryos. A total of 1-4 cells were biopsied from 40 embryos and analysed with CGH and their sibling blastomeres were examined by FISH. From the forty embryos investigated FISH results were in agreement with the CGH results in all 22 embryos where both tests were informative (Trussler *et al*, 2004).

1.6.4 Types and Mechanisms of Mosaicism

Different types of mosaicism have been reported in preimplantation embryos including aneuploid mosaics, polyploid and haploid mosaics, chaotic mosaics and multinucleation. Thus each type should be explored individually to understand their underlying mechanisms.

1.6.4.1 Aneuploidy Mosaicism

Aneuploid mosaicism is considered to be the most frequent form of mosaicism observed in human embryos (Munne *et al*, 1994b). Aneuploid mosaicism arises as a somatic postzygotic event. However molecular studies of the origin of the extra chromosome in the trisomic cell line indicate it is of two types: meiotic and somatic (Kalousek, 2000). Hence aneuploid mosaicism is of two types. Meiotic mosaicism is where a loss of the trisomic chromosome occurs in a trisomic fetus producing a diploid cell line as well as the trisomic cell line, whereas somatic mosaicism is where a trisomic cell line arises in a normal diploid embryo, giving two or three cell lines.

Munne et al (1997) analysed 138 normally fertilised human cleavage-stage embryos and found that 14% were chaotic suggesting that there maybe a relationship between embryo abnormalities and different drug regimes and embryo culture conditions. Furthermore, the same group reported that mosaicism due to aneuploid cells interfered with embryo development for some unknown reason (Sandalinas et al, 2000), since aneuploidy combined with extensive mosaicism had a stronger effect in that study resulting in none of the human embryos developing to blastocysts. In Gonzalez-Merino et al (2003) it was found that all types of blastocysts (eiher pre-expanded or expanded) were all diploid/mosaic with >70% of the cells being diploid and the author suggested that this finding might be a normal finding in *in-vitro* embryos. Delhanty et al (1997) observed 19% of human cleavage-stage embryos were diploid mosaic, however relatively high amounts of chaotic embryos (26% out of 93 embryos) were found (Table 3.3). It was proposed that the mechanism of mosaic aneuploidy were probably mitotic non-disjunction, which causes a reciprocal loss or gain in addition to anaphase lag (Delhanty et al, 1997). Moreover, Delhanty and Handyside (1995) suggested that due to the absence of cell-cycle checkpoints mosaic aneuploid embryos could form, which may be specific to the cleavage stage of development. The latter was also postulated by Harrison et al (2000), which observed a mirror-image distribution about the plane of attachment of the signals in each sequential hybridisation, indicating premature decondensation during anaphase, which is consistent with lack of checkpoint control. The possibility that cell-cycle checkpoints do not fully operate during cleavage of the human embryo may also

explain the relatively high incidence of various nuclear abnormalities which have been observed (Winston *et al*, 1991; Hardy *et al*, 1993) For example binucleate blastomeres are seen in 15% of human embryos, and are frequently associated with chromosomal abnormalities and appear to result from failure of cytokinesis (Winston *et al*, 1991; Hardy *et al*, 1993; Kligman *et al*, 1996; Staessen and Van Steirteghem, 1998). Most recently, work on inbred mouse strains with elevated levels of Y chromosome nondisjunction has shown that malsegregation in this system is largely restricted to the earliest mitotic divisions (Bean *et al*, 2001). This suggests that mammalian embryos are indeed susceptible to mitotic nondisjunction in early cleavage stages (Bean *et al*, 2001; 2002) and fits the lack of checkpoint control model.

Another study which analysed 161 embryos for sex determination using FISH, observed examples of three different mechanisms in aneuploid mosaic embryos: (i) involvement of an aneuploid gamete which finally lead to an aneuploid mosaic embryo; (ii) reciprocal mitotic non-disjunction leading to mosaic embryos with monosomic, disomic and trisomic blastomeres for the chromosomes involved; and (iii) chromosomal loss leading to a combination of monosomic and disomic blastomeres (Staessen *et al*, 1999) confirming the previous reports by Delhanty *et al* (1997)

Veiga *et al* (1999) revealed a high prevalence of mosaicism in both blastocyst and arrested embryos showing 87.5% and 62.5% respectively. Moreover, Ruangvutilert *et al* (2000a) showed that 30% of the day 5 arrested embryos and 21% of the blastocysts were aneuploid mosaics, which supports the hypothesis of the selection against chromosomal abnormalities through culture to the blastocyst stage proposed by Sandalinas and co-workers (2001). Coonen *et al* (2004) analysed 295 blastocysts and after finding high levels of mosaicism (57%) and chaos (17%) concluded that anaphase lagging appeared to be the major mechanism through which human embryos acquire a mosaic aneuploid pattern.

1.6.4.2 Ploidy Mosaicsim

Ploidy mosaics have also been frequently reported in cleavage stage embryos, blastocysts (Harper et al, 1995; Delhanty et al, 1997; Munne et al, 1997; Staessen et

al, 1999; Ruangvutilert et al, 2000a; Sandalinas et al, 2001) and those of other animal species (Long and Williams, 1982; Murray et al, 1986). Ploidy mosaics may well play a role in normal early development particularly when associated with TE lineages (Angell et al, 1987) with tetraploid or haploid predominating. It has been proposed that probably tetraploid cells may be a normal feature in the development of the trophectoderm (Angell et al, 1987). Tetraploid trophectoderm cells may arise as a result of endoreduplication or endomitosis and possibly play a role in embryo implantation (Drury et al, 1998). Although other studies have linked mosaic tetraploidy to poor quality arrested embryos (Wells and Delhanty, 2000). Laverge et al (1997) while studying 97 human cleavage-stage embryos reported that possible mechanisms for polyploid mosaics could involve endoreduplication of a binucleate cell which would subsequently form two mononucleate daughter cells with polyploid nuclei. The fusion of nuclei in binucleate blastomeres or less frequently blastomere fusion may also lead to polyploidy (Balakier et al, 2000).

Less common mosaicism findings includes haploid and triploid nuclei. The presence of a haploid cell in a mosaic embryo is difficult to explain, however the underlying mechanism maybe associated with binucleate cell production with a meiotic type of segregation (Delhanty *et al*, 1997) or maybe an incorporation of a polar body into the embryo (Staessen *et al*, 1999). Haploid/diploid mosaics can be found from pronuclear zygotes and are believed to arise due to the activation of the oocyte (Staessen and Van Steirteghem, 1998). Haploid and triploid cells have been characterised as less viable and less actively dividing than tetraploid cells (Ruangvutilert *et al*, 2000a); with the exception of some triploid cells persisting until later in development as in cases of mosaic diploid/triploid (Edwards *et al*, 1994).

The origin of diploid/triploid mosaicism is not clear. The underlying mechanism that leads to diploid/triploid mosaics may be due to an incorporation of another gamete or its genome into one of the daughter cells derived after the first mitotic division or later. The extra gamete might be a polar body (Mueller *et al*, 1993). Kuo *et al* (1998) also suggested that diploid/triploid mosaicism could result from fusion of a diploid zygotic nucleus with an extra sperm nucleus or the extrusion and degeneration of a haploid nucleus to produce a diploid cell line in a triploid embryo. In a recent prenatal

study of four cases of triploid/diploid mosaics, the three different mechanisms of origin for these apparent mosaics were detected: i) chimaerism with karyotypes from two separate zygotes developing into a single individual, ii) delayed digyny, by incorporation of a pronucleus from a 2^{nd} PB into the embryonic blastomere and iii) delayed dispermy, similarly by incorporation of a 2^{nd} sperm pronucleus into one embryonic blastomere (Daniel *et al*, 2003).

1.6.4.3 Chaotic Mosaicism

Chaotic embryos have been reported in many studies and involve embryos where all nuclei show a different chromosome complement. Chaotic embryos have been found in cleavage-stage embryos (Harper *et al*, 1995) as well as at the blastocyst stage (Evsikov and Verlinsky, 1998). It has been hypothesised that they originate due to uncontrolled "chaotic" division, which is possibly related to centriole or spindle deficiencies and disturbance of pronuclear syngamy (Klingman *et al*, 1996). During FISH analysis of human cleavage-stage embryos it was suggested by Delhanty and Handyside (1995) that chaotic embryos may result form the absence of cell cycle checkpoints leading to chaotic segregation of chromosomes. Furthermore, the frequency of chaotic embryos appears to be a patient related phenomenon (Delhanty *et al*, 1997).

1.6.4.4 Multinucleation

A normal human embryo should have a single nucleus in each blastomere (only visible during interphase when the nuclear membrane is present). The presence of multinuclear blastomeres has been reported for both *in vivo* (Hertig *et al*, 1954) and *in vitro* (Winston *et al*, 1991) developing embryos. Its frequency ranges from 17-69% in human embryos (Plachot *et al*, 1987; Hardy *et al*, 1993; Munne *et al*, 1994a) however, it might be significantly higher when associated with other morphological abnormalities (Munne *et al*, 1995a). Recently, Meriano et al (2004) found 24% multinucleation in 770 embryos derived from ICSI. The author suggested – after time-lapse photography – that multinucleated blastomeres were subject to dissolution of their nuclear membrane suggesting an asynchrony the nuclei and a possible interruption in proper nuclear and cell division (Meriano *et al*, 2004). Multinucleation first occurs at the 2-cell stage, but it has been shown to occur most frequently at the 8-

cell stage (Hardy et al, 1993). Kligman et al (1996) suggested that if multinucleation occurs at the first embryonic division all the embryonic cells might be affected. Laverge et al (1997) observed multinucleation in 15 out of 39 normal diploid embryos. This is in agreement with Munne and Cohen (1993) who concluded that multinucleation occurs frequently in arrested as well as normally developing embryos. In a study of 1885 embryos by Balakier and Cadesky (1997) it was observed that 44% of patients possessed embryos with at least one multinucleated blastomere (MNB) and 15% of embryos contained MNB, however, there was no correlation with maternal age. It has been proposed that asynchrony between karyokinesis and cytokinesis leads to fragmentation and production of multinucleate blastomeres (Lopata et al, 1983; Evsikov and Verlinsky, 1998). Furthermore, partial fragmentation of nuclei or defective migration of chromosomes during mitotic anaphase have also been linked as mechanisms leading to MNB (Tessarik et al. 1987; Wiston et al. 1991; Pickering et al, 1995). However, Staessen and co-workers (1998) after analysing 101 3- to 8-cell embryos developing from 2-cell embryos (where both blastomeres were bi- or multinucleate), reported that the genetic constitution of binucleate and multinucleate blastomeres and the daughter cells developing from them are not always abnormal.

1.6.4.5 Mosaicism and Translocations

A study carried out by Iwarsson *et al* (2000) in preimplantation embryos from translocation carriers during PGD cycles revealed a high degree of mosaicism for the chromosomes involved in the translocations (65%), compared to control chromosomes (35%). However, the degree of mosaicism within each embryo differed between the chromosomes involved in the translocation and the control probes. In order to explain these relatively high frequency of mosaicism Iwarsson *et al* (2000) proposed three hypothetical explanations: (i) Acrocentric chromosomes (13, 14, 15, 21 and 22) have a higher tendency to malsegregate during meiosis as well as mitosis; (ii) The translocation itself may predispose to malsegregation and; (iii) The difference between the chromosomes involved in the translocations with the control chromosomes could be methodological. Although it was concluded that this last hypothesis cannot be regarded as a major cause.

1.7 Aims of the Study

This study is comprised of three parts employing molecular and cytogenetic approaches to the analysis of human chromosomes in human preimplantation embryos.

The aim of the first part was to devise novel fluorescent PCR protocols for PGD acting as well as attempting to develop a universal-like protocol i.e. methodology applicable to different DM patients requiring little optimisation, for PGD for DM. Initially, several F-PCR polymorphic markers, linked or unlinked, were investigated to assess their efficiency at the single cell level and test whether they are informative for two couples. PGD was carried out for both couples using two different single step multiplex F-PCR protocols. A third protocol was devised and tested on single cells using whole genome amplification in addition to F-PCR. The purpose of employing whole genome amplification in this study was primarily to examine whether carrying out DOP-PCR on a single cell was able to amplify regions within the genome that would match the regions where specific F-PCR polymorphic primers hybridise. As a consequence this would eliminate the need for multiplexing and one would be able to carry out several single F-PCR procedures with different F-PCR markers using the DOP-PCR product as a template. Hence, during a PGD case of a single gene disorder such as DM, there would be no need to optimise for a multiplex PCR protocol. If a DOP-PCR of the single blastomere was carried out and then separate but simultaneous singleplex PCR reactions with different F-PCR markers could be performed informative for the parents involved. Also, enough DNA would be available to perform CGH analysis which would allow chromosomal examination of the blastomere. Therefore, from one blastomere the single gene defect as well as its chromosomal status could be identified.

The second part of this study was to develop a reliable FISH-based protocol for the analysis of chromosome abnormalities in day 5 human embryos in order to reveal the level of chromosomal mosaicism for the five chromosomes studied (1, 11, 18, X and Y). Furthermore, it would allow the study of the underlying mechanisms of mosaicism and the high rate seen *in vitro* human preimplantation embryos. Optimising conditions and investigating the possibility of different probe

combinations, therefore allowing the detection of all the possible mechanisms that lead to mosaic and chaotic embryos played an intricate role of this study. With the aid of two probes per chromosome in sequential rounds of FISH, problems of the FISH technique such as FISH artefacts e.g. monosomies, probe overlappings and failure of probe hybridisations were overcome. Such artificial findings would be detected and thus separated from the true findings enabling the study of the mechanisms of mosaicism in human blastomeres. This type of methodology has never been attempted before and would allow a innovative approach of investigating the phenomenon of mosaicism. Furthermore, the effects of the IVF culture media were examined. Interphase FISH was applied to spare/untransferred embryos derived from clinical treatment IVF or ICSI cycles

During the last part of this thesis, the use of CGH was investigated as an alternative to FISH for investigating the prevalence of mosaicism in day 3 and day 5 embryos. This involved assessing the efficiency of CGH, improving the protocol for optimised use on single cells, and its application as a research tool on human preimplantation embryos. The aim of this study was to assess the full chromosomal status of 1-2 blastomeres biopsied at day 3, and then confirm whether the abnormality persists until day 5 using FISH for the chromosome(s) involved. This study would allow the detection of mosaicism in a full karyotype at the cleavage stage and further enquire if the mechanisms causing mosaicism continue to exist until day 5.

CHAPTER 2

Materials and Methods

2.1 Materials

General laboratory chemicals and reagents were obtained from BDH Chemicals UK, Sigma UK and were of Analar or biochemical grade unless otherwise stated. The materials and stocks that have been used for this study are separated into '*FISH Materials'*, '*PCR Materials' and 'CGH Materials'* and are listed below:

2.1.1 FISH Materials

2.1.1.1 Materials for Lymphocyte culture

Cell culture flasks, glass pipettes and microscope slides were obtained from BDH whilst all microcapillaries (internal diameters 75-200 μ m) for embryo, oocyte and single cell handling were from Laser (UK). Nunc Nucleon 50x9mm Petri dishes were used for single cell isolation and purchased from Gibco BRL (UK). Reagents used for media preparation were listed below:

- GPS (Glutamine 200mM, penicillin 300mg/ml, streptomycin 500mg/ml; Gibco, UK)
- Iscoves modified Dulbeccos medium (Sigma, UK)
- Phytohaemogglutinin (PHA; Gibco, UK)
- Thymidine (Sigma, UK)
- Deoxycytosine (Sigma, UK)
- Colcemid (Gibco, UK)

2.1.1.2 Materials for Embryo Spreading

- Poly-l-lysine (Sigma, UK)
- Tween-20 (Sigma, UK)

2.1.1.3 Materials for Nick Translation (commercial kit)

- Nick translation enzyme mix (Vysis, UK)
- 10x nick translation buffer (Vysis, UK)
- dATP, dCTP, dGTP and dTTP (0.3mM each; Vysis, UK)
- Nuclease-free water (Vysis, UK)

• Fluorescent-labelled dUTP's used with the kit were SpectrumAqua-dUTP from

2.1.1.4 Materials for the FISH procedure

- Probes (commercial and non-commercial; Vysis, UK)
- Pepsin (Sigma, UK)
- Hydrochloric Acid (0.01N; Sigma, UK)
- Paraformaldehyde [37% formaldehyde (Sigma, UK) was saturated with NAHCO₃ (BDH, UK) and stored in the dark
- Ethanol (99.7%; Sigma, UK)
- Formamide (BDH, UK)
- 20xSSC (Vysis, UK)
- NP-40 (Vysis, UK)
- Vectashield (Vector laboratories, USA)
- 4',6-diamino-2-phenylindole (DAPI; Sigma, UK)

2.1.2 PCR Materials

2.1.2.1 Materials for Single-cell isolation

- PCR Lysis Buffer [125μg/ml Proteinase K (Sigma, UK), 17μM Sodium dodecyl sulphate (SDS; BDH chemicals, UK), Nuclease-Free Water (H₂O; Promega, UK)]
- Phosphate Buffer Saline (PBS; Sigma, UK)
- Polyvinyl Alcohol (PVA; Sigma, UK)
- Mineral oil (Sigma, UK)

2.1.2.2 Materials for DNA extraction

- TKM1 (low concentration salt buffer): 10mM Tris-HCl pH 8.0, 10mM KCl, 10mM MgCl₂, 2mM EDTA (BDH, UK)
- TKM2 (high concentration salt buffer): 10mM Tris-HCl pH 8.0, 10mM KCl, 10mM MgCl₂, 2mM EDTA, 0.4M NaCl (BDH, UK)
- **10xTE buffer**: 10mM Tris-HCl pH 8.0, 1M EDTA (BDH, UK)
- Igepal CA-630 (Sigma, USA)

- SDS (Sigma, UK)
- 6M NaCl (BDH, UK)
- 70% Ethanol (BDH, UK)

2.1.2.3 Materials for PCR procedure

- Primers (Oswel, UK) See Table 2.2
- AmpliTaq Gold Polymerase (Applied Biosystems, UK)
- AmpliTaq Buffer (10x; 1.5mg MgCl₂; Applied Biosystems, UK)
- dNTP mix (10mM; Promega, UK)
- Nuclease-Free Water (H₂O; Promega, UK)

2.1.2.4 Materials for gel electrophoresis

- Agarose (Sigma, UK)
- Ethidium Bromide (500mg/ml; Sigma, UK)\
- 10xTBE (0.89M Tris Base, 0.89M Boric acid, 2.5M EDTA; Sigma, UK)
- Loading buffer (10x; 40% sucrose, 0.025% w/v bromophenol blue, 0.025% w/v xylene cyanol) (Supplied by Bioline, UK)
- 1 Kb DNA ladder (Bioline, UK)

2.1.2.5 Materials for ABI PrismTM 310 and 3100

- Deionised Formamide (Sigma, UK)
- Size Standard (Genescan 500-TAMRA; Applied Biosystems, UK
- Size Standard (Genescan 500-ROX; Applied Biosystems, UK)
- 0.5ml Sample Tubes (Applied Biosystems, UK)
- Septa for 0.5ml Samples Tubes (Applied Biosystems, UK)
- Plate for 3100 (Applied Biosystems, UK)
- Capillary on array for 310 and 3100 ABI Prism (Applied Biosystems, UK)
- Buffer for 310 and 3100 ABI Prism (Applied Biosystems, UK)
- Polymer for 310 and 3100 ABI Prism (Applied Biosystems, UK)

2.1.3 CGH Materials

Single buccal cell, single blastomere as well as clump isolation was performed using the same materials as in 2.1.5. Furthermore, all genomic DNA from whole blood or fibroblasts was carried out using the same material as in 2.1.6.

2.1.3.1 Materials for DNA Extraction

- Hank's medium (Sigma, UK)
- DNA extraction lysis buffer (In 100ml 1.21g Tris, 0.19g/100ml EDTA, 0.2gSDS, 1.17g NaCl., 10mg/ml of proteinase K added after autoclaving)
- Isopropanol (BDH, UK)

2.1.3.2 DOP-PCR Materials

- Primer (Oswel, UK) See Table 5.3
- SuperTaq Plus (HT Biotechnology, UK)
- SuperTaq Buffer (10x; 1mg MgCL₂; HT Biotechnology, UK)
- dNTP mix (10mM; Promega, UK)
- Nuclease-Free Water (H₂O; Promega, UK)

2.1.3.3 Nick-Translation Materials

- Nick Translation kit (Vysis, UK)
- 0.2mM Spectrum Green or Spectrum Red dUTP's (Vysis, UK)
- Human Cot-1 DNA (GibCo BRL, UK)
- NaAc (3M; Sigma, UK)

2.1.3.4 CGH Procedure

- Hybridization mix (50% Formamide, 2xSSC, 10%Dextran Sulphate; Sigma, UK)
- CGH Metaphase Target Slides (Vysis, UK)
- Triton-X (Sigma, UK)
- Formamide (BDH, UK)
- 20xSSC (Vysis, UK)

2.1.4 Embryo Materials

2.1.4.1 Ethical Approval

The work on surplus embryos and the clinical application of PGD (Chapter 3) were approved by the Research Ethics Committees of the University College London Hospital Trusts, and carried out under licence from the Human Fertilisation and Embryology Authority (HFEA). The surplus embryos for Chapters 4 and 5 were donated from normal patients undergoing routine IVF at the London Fertility Clinic which were authorised by the HFEA. Informed written consent was obtained from patients for surplus embryos to be used for research purposes

2.1.4.2 Grading Criteria for the Embryos

Preimplantation cleavage stage embryos were graded according to Bolton et al,

(1989) as follows;

- *Grade 1* Embryo at the correct stage of *in vitro* development with perfect symmetrical and even-sized blastomeres with no fragmentation.
- *Grade 1*⁻ Embryo at the correct stage of *in vitro* development with perfect symmetrical and even-sized blastomeres with less than 10% fragmentation.
- Grade 2⁺ Development with unequally sized blastomeres with less than 20% fragmentation
- *Grade 2* Retarded development with unequally sized blastomeres with 25%-50% fragmentation.
- *Grade 3* Retarded development with unequally sized blastomeres with more than 50% fragmentation.

2.1.4.3 Categorisation of Chromosomal Abnormalities in Embryos.

Embryos analysed for the preliminary work and during PGD cycles were categorised after Delhanty *et al*, (1997) into four groups; normal, uniformly abnormal, mosaic (diploid mosaic or aneuploid mosaic) and chaotic (Table 1.1). Embryos were allocated where possible to each group on the basis of the chromosome constitution of the majority of cells present.

2.1.4.4 Embryo culture

Oocytes were retrieved using Flushing Medium (supplemented with sodium pyruvate, HAS, heparin 10IU/ml, penicillin 50,000IU/l, streptomycin 50mg/l, and HEPES; Medicult UK Ltd), incubated in 6%CO₂ in air, at 37°C, inseminated and cultured in 500µl of IVF Medium (Bicarbonate buffered medium containing human serum albumin, penicillin and sodium pyruvate; Vitrolife, Scandinavia). On day 1, oocytes were assessed for the number of pronuclei and transferred into 25µl microdroplets of Cleavage Medium (Bicarbonate buffered medium containing human serum albumin, penicillin-G, EDTA, glucose, inorganic salts and amino acids; G-1, Vitrolife, Scandinavia). On day 3, the best embryos were selected for transfer and suitable spare embryos were cryopreserved.

The embryos were divided into two groups depending on the culture medium. Group I embryos were cultured in standard IVF medium (6.1 Vitrolife, Scandinavia) from day 0 to day 5 and Group II embryos were cultured in standard IVF medium from day 0-3 and then in blastocyst medium (6.2 Vitrolife, Scandinavia) from day 3-5. Only embryos that arose from a bipronucleate zygote were included in the study.

2.1.4.5 Embryo Freezing

Good quality Day-1 embryos were considered for freezing when two pronuclei and two polar bodies were visible 16-18 hours after the oocytes were subjected to sperm either by IVF insemination or ICSI. Good quality Day-3 embryos were considered when they comprised 6-8 cells and were grade 2 and above.

All embryos (either day-1 or day-3 cell stage) were placed for a five-minute wash in Cryo-PBS medium (Vitrolife; Scandinavia) and then taken through consecutive washes of embryo freezing solutions 1 and 2 (EFS1 and EFS2; Vitrolife, Scandinavia) for ten minutes each at room temperature. The embryos were loaded into the freezing straw (maximum two embryos per straw) and placed into a cryobath. The cryobath is linked to a computer program which controls temperature until it reached -180°C..

2.1.4.6 Embryo Thawing

All thawed embryos (either day-1 or day 3-cell stage) were taken through consecutive washes of embryo thaw solutions 1, 2, and 3 (ETS1, ETS2 and ETS3; Vitrolife, Scandinivia) for five, five and ten minutes respectively at room temperature. A final wash was performed through Cryo-PBS medium for five minutes at room temperature followed by five minutes on a heated stage (38°C) (Nikon SMZ-U microscope stage).

Pronuclei (PN) stage embryos, after thaw, were initially cultured in G1 medium until day 3 (section 4.2.5) and subsequently in G2 medium until blastocyst stage, while day 3 thawed-embryos were directly cultured in G2 medium.

2.1.4.7 Embryo Spreading

Embryos were spread as described previously (Harper *et al*, 1994; Ruangvutilert *et al*, 2000a; section 2.2.1.2).

2.1.4.8 Embryo Classification and Scoring of Embryos

The criteria used for classifying the embryos were adopted from Delhanty *et al* (1997) and can be seen in Table 1.1. Whilst classifying an embryo all major cell lines should be mentioned e.g. if an embryo has 20 diploid cells, 4 tetraploid cells, 2 trisomy 18 cells, 1 monosomy 1 cell and 1 triploid cell, it should be classified as diploid/polyploid/aneuploid mosaic. Furthermore, if a blastomere displayed contradictory results in different rounds of FISH for the same chromosome e.g. one signal for 1p (1st round) but 2 signals for 1het (2nd round), this was considered as an "*inconsistent result*". When scoring the fluorescent signals (see section 2.2.1.5.5), signals must be a minimum of a signal's width apart in order to be scored as 2 individual signals (Hopman *et al*, 1991).

2.1.4.9 Mosaicism and Events

Diploid mosaic embryos with aneuploid cells were considered to have arisen through three different mechanisms: (a) when the embryo contained cells with monosomies, then the mechanism was classed as "chromosome loss" (CL), (b) when the embryo contained cells with trisomies, then the mechanism was classed as "chromosome gain" (CG) and (c) when the embryo had monosomies and trisomies of the same chromosome(s) in different cells, this was classified as mitotic non-disjunction (MND). Nuclei with multiple abnormalities affecting at least three chromosomes were classed as chaotic and not included in the analysis. This included nullisomies and tetrasomies. During the last chapter chromosome breakage and partial mosaicism mechanisms were revealed, hence all three mechanisms (chromosome loss, chromosome gain and mitotic non-disjunction) can occur either partially (p) or in the whole (w) chromosome.

2.1.5 Probes and Primers

2.1.5.1 DNA Probes.

Details of DNA probes used in this study along with their sources are summarised in Table 2.1. The plasmid DNA clones for chromosome 1 was obtained from resource centres as agar stabs. Maxiprep of plasmid and cosmid DNA was carried out with Wizard maxiprep kit from Promega. Probe DNA was labelled via nick translation supplied in kit form (Nick Translation Kit Vysis UK). Commercially obtained labelled α -satellite and locus-specific probes were supplied by Vysis UK. All were stored at -20°C and protected from light.

Probe Label Type Source Vysis, UK 1 pter Spectrum Green 1p Spectrum Orange Vysis, UK 1 qter 1q Satellite-II/III Spectrum Aqua Lab-prepared 1het 2qter Spectrum Orange Vysis, UK 2q α -satellite Spectrum Aqua Lab-prepared 2cep α -satellite 3 Vysis, UK Spectrum Orange 3cep α -satellite 4 Spectrum Aqua Vysis, UK 4cep 5pter Spectrum Green Vysis, UK 5p 5p15.2 / 5q31 Vysis, UK 5p/5q (Cri-du-Chat Spectrum Green /Orange microdeletion probe) α -satellite 6 Spectrum Green Vysis, UK 6cep 7p31 / 7q11.23 Spectrum Vysis, UK 7p/7q (Willams microdeletion probe) Green/ Orange Spectrum Orange Vysis, UK 8qter 8q Vysis, UK α -satellite 8 Spectrum Orange 8cep Vysis, UK α -satellite 9 Spectrum Orange 9cep α -satellite 10 Vysis, UK Spectrum Green 10cep

Table 2.1. List of FISH DNA probes used in the FISH and CGH/FISH study

| Chapter | 2 – | Materials | and | Methods |
|---------|-----|-----------|-----|---------|
|---------|-----|-----------|-----|---------|

| 11q | 11qter | Spectrum Orange | Vysis, UK |
|------------|------------------------|------------------|-----------|
| 11cep | α-satellite 11 | Spectrum Green | Vysis, UK |
| 11cep | α-satellite 11 | Spectrum Aqua | Vysis, UK |
| 13LSI | 13q11 | Spectrum Green | Vysis, UK |
| 14q | 14qter | Spectrum Orange | Vysis, UK |
| 16р | 16pter | Spectrum Green | Vysis, UK |
| 16qter | 16qter | Spectrum Orange | Vysis, UK |
| 16cep | α-satellite 16 | Spectrum Aqua | Vysis, UK |
| 18q | 18qter | Spectrum Orange | Vysis, UK |
| 18cep | α-satellite 18 | Spectrum Aqua | Vysis, UK |
| 22LSI | 22q11.2 | Spectrum Green | Vysis, UK |
| Хсер | α-satellite X | Spectrum Green | Vysis, UK |
| Усер | Satellite II/III | Spectrum Aqua | Vysis, UK |
| X / Y / 18 | Probe-cocktail of | Spectrum Green / | Vysis, UK |
| | heterochromatic probes | Orange / Aqua | |

2.1.5.2 PCR Primers

Several STR markers were used in different preliminary work carried out for couples undergoing PGD treatment for myotonic dystrophy. These markers were either linked or unlinked and varied in their size and fluorescent dyes. All primers for the myotonic dystrophy workup can be seen in Table 2.2 below

Table 2.2. List of PCR primers used in the PCR study

| | Marker | Sequence of primers | Chromosome | Size of the | Label of primer | Reference |
|-------------------|--------------|-----------------------------------|-------------|-------------|-----------------|----------------------|
| Type of Marker | (F=forward) | (5' 3') | location of | PCR product | (at 5' end of | |
| | (R=reverse) | (3-3) | primers | (bp) | forward primer) | |
| STR | DM (F) | 5'-cttcccaggcctgcagtttgcccatc-3' | 19q13.3 | 128-203 | 5'-FAM DYE | Brook et al, 1992 |
| (Dinucleotide) | DM (R) | 5'-gaacggggctcgaagggtccttgtagc-3' | | ľ | | |
| STR | APOC2 (F) | 5'-ggctacatagcgagactccatctcc-3' | 19q12 – | 134-170 | 5'-HEX DYE | - |
| (Dinucleotide) | APOC2 (R) | 5'-gggagagggcaaagatcgataaagc-3' | 19q13.2 | | - | |
| STR | D19S207 (F) | 5'-tgcggtgtttgaaccctcgctg-3' | | 118-160 | 5'-HEX DYE | • • |
| (Dinucleotide) | D19S207 (R) | 5'-actgcactgcagcctgagtgac-3' | | | - | |
| STR | D19S112 (F) | 5'-ctgaaagacacgtcacactggt-3' | | 115-140 | 5'-HEX DYE | Jansen et al, 1992 |
| (Dinucleotide) | D19S112 (R) | 5'-gccagccattcagtcatttgaag-3' | | | - | |
| STR | D19S393 (F) | 5'-gcaatgagccgagatagaa-3' | | | 5'-HEX DYE | - |
| (Tetranuclaotide) | D19S393 (R) | 5'-tggctagcccattactcta-3' | | | = | |
| STR | D21S11 (F) | 5'-tatgtgagtcaattccccaagtga-3' | 21q21 | 200-260 | 5'HEX DYE | Sharma and Litt, |
| (Tetranucleotide) | D21S11 (R) | 5'-gttgtattagtcaatgttctccag-3' | - | | - | 1992 |
| STR | D21S1414 (F) | 5'-aaattagtgtctggcacccagta-3' | 21q21 | 330-370 | 5'-HEX DYE | Sherlock et al, 1998 |
| (Tetranucleotide) | D21S1414 (R) | 5'-caattccccaagtgaattgccttc-3' | | | - | |
| STR | D18S535 (F) | 5'-cagcaaacttcatgtgacaaaagc-3' | 18q12.2 – | 455-500 | 5'-HEX DYE | Lareu et al, 1998 |
| (Tetranucleotide) | D18S535 (R) | 5'-caatggtaacctactatttacgtc-3' | 18q12.3 | | - | - |
| STR | D13S305 (F) | 5'-gcctgtttgaggacctgtcgtta-3' | 13q12.1 – | 430-465 | 5'-TET DYE | - |
| (Tetranucleotide) | D13S305 (R) | 5'-tggttatagagcagttaaggcac-3' | 12q14.1 | | - | |
| DOP | | | | | | Telenius et al, 1992 |
| (WGA) | DOP-50 | 5'-ccgactcgagnnnnnatgtgg-3' | Random | - | - | |

2.1.6 Equipment

2.1.6.1 FISH and CGH Equipment

Dissecting microscopes from Nikon and inverted microscopes from Olympus were used for embryo and oocyte handling as well as slide preparation. Fluorescence microscopy was carried out with the following microscope systems; Reichert Jung Polyvar microscope with single filters for TRITC, FITC and DAPI, Nikon optiphot microscope with Omega dual band-pass TRITC/FITC filter and Zeiss Axioskop microscope with Chroma multi-band pass TRITC/FITC/DAPI filter and single SpectrumAqua filter. Image capture and analysis was carried out using a Zeiss Axioskop microscope equipped with a Photometrics KAF 1400 cooled CCD (charged coupled device) camera controlled by Smartcapture software from Vysis, UK.

2.1.6.2 PCR and CGH Equipment

Three models of thermal cyclers were used for PCR amplification:

- 1. Hybaid Omnigene was manufactured by Hybaid Middlessex, UK.
- 2. ABI-9700 PCR System was manufactured and serviced by Applied Biosystems, UK.
- 3. Mastercycler Gradient was from Eppendorf-Netheler-Hinz GmbH, Cambridge, UK.

The analysis of fluorescent PCR products was performed on the ABI Prism 310 and 3100 using GeneScan analysis software (version 2.0.2). A Nikon dissecting microscope used for single cell isolation and oocyte spreading and a Nikon phase contrast microscope was employed to check slide preparations.

2.2 METHODS

2.2.1 FISH Methods

FISH was used in the mosaicism study (Chapters 4 and 5). FISH consists of:

- Sample (embryo) or control lymphocyte cell suspension
- Probe preparation which consists of:
 - o DNA preparation
 - o Nick Translation or use of commercial probes
- ➢ FISH procedure and analysis

- o Slide preparation and pre-treatment
- Probe preparation
- Denaturation conditions
- Post-hybridisation washes
- Analysis (capturing and scoring)
- Sequential FISH procedure

2.2.1.1 Control Lymphocyte Preparation

Male lymphocyte cells were used as control samples to record the efficiency of the probe combination employed for each FISH procedure. A sample of blood was provided by a male donor and was cultured according to the following procedure. The blood sample was collected in a lithium heparin tube. Iscoves modified Dulbeccos medium (Sigma, UK), and Fetal calf serum-heat inactivated (FCS) (Gibco, UK), were warmed to 37° C. 2 ml of GPS {Glutamin (200 mM), Penicillin (300 mg/ml), Streptomycin (500 mg/ml)}, which was stored at -20° C, was added to the medium in order to ensure antibiotic resistance.

Under aseptic conditions, the following were added to a 50 ml culture flask: 17ml Iscoves, 2 ml FCS, 200 μ l of Phytohaemagglutinin (PHA) (Gibco, UK) which stimulates the mitotic process, and 1 ml of blood. These were mixed and incubated at 37° C for 48 hours or 72 hours. The flasks were gently shaken twice a day to resuspend the cells. On day 3 at 4pm, 200 μ l of thymidine (30 mg/ml stock) (Sigma, UK) was added to achieve synchronisation of mitosis. On day 4 at 10 am, 200 μ l of deoxycytosine (0.227 mg/ml stock) (Sigma, UK) were added which enforced the effect of thymidine on the cell culture in terms of synchronisation of mitosis. Finally, at 2 pm, on day 4, 200 μ l of colcemid (10 μ g/ml stock) (Gibco, UK) was added in order to arrest the cells at metaphase prior to harvesting.

2.2.1.1.1 Harvesting

The flasks were shaken, their contents emptied into two 10 ml tubes and centrifuged at 1000rpm for 5 minutes. The supernatant was discarded leaving only a small amount for pellet resuspension. An aliquot of 10 ml of 0.075 KCl was added slowly to each tube and left at room temperature for 15-20 minutes, in order for the KCl to cause the

cells to swell. The tubes were again centrifuged at 1000rpm for 5 minutes and the supernatant was discarded, leaving a small amount to re-suspend the pellet.

Approximately 1-2 drops of the fixative solution (3:1 methanol/ acetic acid), which was made fresh, were added to re-suspend the pellet by tapping the tube sharply until its contents turned brown and frothy. A few more drops of fix were added slowly, whilst the solution was agitated and this was repeated. When the solution stopped frothing, a few mls of fix were added at a time until the tube was full (approximately 10ml). The tubes were centrifuged at 1000 rpm for 5 minutes and the supernatant was discarded, leaving a small amount for pellet re-suspension. The fixation process was repeated as described above but fix was added a few mls at a time. After the final centrifugation the remaining pellet had a pale whitish colour. The lymphocyte suspension was stored at -20° C and left overnight before being used for control slide preparation.

2.2.1.1.2 Preparation of control lymphocytes

Whenever the FISH procedure was performed, it was of great importance to process a control slide as a reference to ensure the efficiency of the procedure. Fixative solution (3:1 methanol: acetic acid) and 70% acetic acid were prepared each time. The standard cytogenetic preparations of male control lymphocytes which were stored in 3:1 methanol: acetic acid at -20° C, were centrifuged at 400 g for 5 minutes and resuspended in fresh fixative (the volume depended on the pellet size, so as to ensure a good concentration of lymphocytes). A microscope slide was cleaned with a tissue and breathed on. A small drop of lymphocyte solution was placed on the centre of the slide, which was immediately warmed on the back of the hand, to ensure the nuclei were spread out. The slide was left to totally dry, while other slides were prepared. Subsequently, the slides were flooded with fixative for 10 seconds, which was poured off and the slides were left to dry, then they were flooded again with 70% acetic acid for 10 seconds. Once the slides were totally dry, they were checked under a phase microscope to ensure the presence of nuclei. Finally, they were dehydrated successively in 70%, 90%, and 100% ethanol at room temperature for 5 minutes each.
2.2.1.2 Day-5 Embryo Spreading

Poly-l-lysine coated slides were used to enhance the fixation of nuclei on slides. They were prepared by placing glass slides in a mixture of 50ml of 100% methanol and 0.5ml of 1N HCl for 3-5min. The slides were left to air-dry and subsequently rinsed in Poly-l-lysine (Sigma, UK) for 5min. They were left at room temperature overnight and stored at 4°C until required.

The embryos were spread as described previously by Harper *et al* (1994). The embryo was washed in a drop of PBS to remove excess culture medium. Then, the embryo was transferred to a small drop of spreading solution (0.01N HCl, 0.1% Tween 20) on a poly-l-lysine slide. The embryo was constantly observed under an inverted microscope. Most of the spreading solution was removed and replaced with fresh spreading solution in order to dissolve the zona pellucida and excess cytoplasm. The nuclei were further washed by gentle agitation of the spreading solution until clear of cytoplasm.

The slides were left to dry completely and washed for 5min in PBS and dehydrated through an ethanol series for 3min (70%, 90% and 100% respectively). The nuclei were located and mapped using an England Finder (Graticules Ltd, UK) under a phase contrast microscope. The slides were left at RT for subsequent FISH. Most of the embryos were spread by embryologists in the London Fertility Centre (LFC) after being trained by the author on spare IVF human embryos.

2.2.1.3 Plasmid Probe Preparation

The DNA for the laboratory-prepared (lab-prepared) probe for the α -satellite of chromosome 1 was isolated from an *E. coli* strain carrying the pZ20 vector. Initially, *E. coli* cells from stabs were inoculated into 4 ml of 2xTY medium (16g/l Bactoryptone, 10g/l Bacto-yeast extract, 5g/l NaCl), also containing 4 μ l of ampicillin (100mg/ml). The culture was left overnight in a shaking incubator at 37°C. The following day this culture was re-inoculated into 200 ml of 2xTY medium and 200 μ l ampicillin (100 mg/ml), and was again incubated overnight in a shaking incubator at 37°C. The DNA was extracted using a commercial maxiprep kit (Promega, UK). The procedure included the following: First the cells were pelleted by

centrifugation at 4,000 rpm for 10 min. The supernatant was discarded and the cells re-suspended into 15 ml of cell re-suspension solution (50 mM Tris-HCL pH 7.5, 10 mM EDTA, 100 μ g/ml, RNase A). Approximately 15ml of cell lysis solution (0.2 NaOH, 1% SDS) were added and mixed gently but thoroughly, by stirring or inverting. When the cellular mixture became clear, 15 ml of neutralization solution (1.32M potassium acetate pH 4.8) were added, and immediately mixed by gently inverting the centrifuge bottle several times. The suspension was centrifuged at 9,000 rpm for 15 minutes at 22-25°C in a room temperature rotor.

The supernatant was filtered through blotting paper and transferred into a 100 ml graduated cylinder. After its volume was measured, the supernatant was transferred to a new centrifuge bottle. Half a volume of isopropanol was added to this supernatant, which was mixed by inversion. The suspension was centrifuged at 9,000 rpm for 15 minutes as above. The supernatant was discarded and the DNA pellet was resuspended in 2 ml TE buffer (10 mM Tris-HCL, 1mM EDTA pH 7.5). The DNA purification was achieved with the use of the Wizard resin and vacuum pump. Finally, the concentration of the DNA was measured by a fluorometer, and the latter was stored at -20°C.

2.2.1.4 Probe Labelling

The plasmid probes were labelled using a nick translation kit (Vysis, UK). However, the labels that this kit contains (spectrum green or spectrum orange direct-labelled dUTP) were not used. The label used was diethylaminocoumarin (DEAC)-5-dUTP (Perkin Elmer, USA), which emits aqua fluorescence and incorporates into loci-specific identifier DNA probes for its use in FISH. Half the amount of the dTTP was substituted with labelled dUTP. The latter diluted the label incorporation, and increased the DNA Polymerase I efficiency. Hence, the procedure enabled the incorporation of about 20% of the fluorescent-labelled nucleotide into the DNA, generating in this way a clear bright signal during hybridization. Ethanol precipitation removed the unincorporated nucleotides.

2.2.1.4.1 Nick Translation

Initially a DEAC-5-dUTP (Perkin Elmer, USA) 50nmol concentration was prepared at 0.2 mM by adding 10 μ l of 0.1 mM dUTP to 40 μ l nuclease-free water. dTTP was prepared in a concentration of 0.1 mM with the addition of 10 μ l of 0.3 mM dTTP to 20 μ l nuclease-free water. A concentration of 0.1 mM dNTP mix was achieved by mixing together 10 μ l each of 0.3 mM dATP, 0.3 mM dCTP, and 0.3 mM dGTP (all three from Vysis, UK).

The nick translation reaction contained the following: 17.5-x nuclease free water mixed with 1 μ g of DNA (volume x, varied depending on DNA concentration), 2.5 μ l of 0.2 mM spectrum green and spectrum orange, 5 μ l of 0.1 mM dTTP, 10 μ l of dNTP mix, 5 μ l of 10X nick translation buffer, and, 10 μ l of nick translation enzyme. The latter consisted of DNA polymerase I, DNase I in 50% glycerol, 50 mM Tris-HCl, pH 7.2, 10 mM MgSO₄, 0.1 DTT, and 0.5 mg/ml nuclease free BSA.

The Eppendorf tube was briefly centrifuged and vortexed to mix the reaction components and incubated for 2 hours at 15° C. This temperature was crucial for probe efficiency and the success of the labelling method. The nick translation reaction was stopped by the addition of 5 µl of 0.5 mm EDTA pH 8 (BDH, UK). Then, the following components were added to the tube: 5 µl of herring sperm DNA (Sigma, UK), 6 µl of 3M sodium acetate, and 1 ml of 100% ice cold ethanol. The probe was incubated at -70°C for 1 hour, and centrifuged for 10 min at 14,000 rpm. The supernatant was removed, and the pellet was air-dried by leaving the tubes open in the dark. Finally the probes were re-suspended in 100 µl of hybridisation buffer (2x SSC, 60% deionized formamide, 10% dextran sulphate), and stored in the dark at -4°C.

2.2.1.5 FISH Procedure

The probes used in this study can be viewed in Table 2.1. All the probes were commercial (Vysis, UK) apart from one (1cep in spectrum Aqua), which was labprepared. The probe combinations during the sequential rounds of FISH for Chapter 4 are summarised in Table 2.3

2.2.1.5.1 Slide Pre-treatment

The method of FISH performed was described previously by Harper *et al* (1994). The slides were incubated in 0.01N HCl containing 10mg/ml pepsin at 37^{0} C for 20 minutes, in order for any remaining protein to be removed and to make the nuclei accessible to the probes. The slides were briefly washed in bidistilled (FISH) water and PBS and incubated for 10 minutes at 4^{0} C in 1% paraformaldehyde in PBS to refix the nuclei. The slides were then washed in PBS and a further two washes in water. The slides were finally dehydrated through an ethanol series (70%, 90% and 100% for three minutes each) and left to dry.

2.2.1.5.2 Probe Preparation

A 5µl probe mix was prepared according to the source and the type of the probe and added to the nuclei under a 13mm diameter coverslip. The probe mix consisted of different volumes of probes and CEP hybridisation buffer.

2.2.1.5.3 Separate Denaturation

The FISH protocol and embryo analysis was more efficient whilst carrying out separate denaturation. Separate denaturation was performed for all three rounds of FISH. During separate denaturation both the slide and the probes were treated separately. The slide pre-treatment was followed as described above (Section 2.2.1.5.1) until the last dehydration. After the washes with PBS and twice with water, 100 μ l of denaturation solution (70% formamide in 2xSSC) was added to the slide. The slide was immediately denatured at 75°C for 5 min and incubated in 50ml of 70% ice-cold ethanol for 5 min to stop the denaturation. Subsequently, the slide was dehydrated through an ethanol series (70%, 90% and 100% for 3 min each) and left to dry. All probe combinations were sealed with fixo-gum (Qbiogene, UK) and incubated overnight in a moist chamber at 37^{0} C.

2.2.1.5.4 Post-Hybridisation Washes

The stringency of probe binding was controlled by the formamide concentration, salt concentration and temperature during the post-hybridisation washes. Therefore conditions were dependent on probe type. All washes were carried out in 50ml volume coplin jars with those containing formamide restricted to a laminar flow

cabinet and slides were protected from light at all stages. After hybridisation any rubber cement was discarded and coverslips were gently removed by immersing briefly in the first wash solution. For combinations of probes including locus-specific probes, slides were treated at 45° C with 3 x 3 minutes washes in 50% formamide in 2xSSC and then 3 x 3 minutes washes in 2xSSC followed by a 10 minute wash in SSCT [4xSSC; 0.05% Tween 20 (polyoxyethylene sorbitan monolaurate), Sigma, UK] at room temperature. When only repetitive probes were used the stringency was raised by increasing the formamide concentration to 60% in the 2xSSC solution. All other washes were as above.

According to the 'long washes' protocol from Vysis (UK) any unbound probe was removed by washing the slides for 10 minutes at 45° C in 70% formamide/2xSSC pH 5.3 (three times), 10 minutes in 2xSSC pH 7.0 at 45° C and then 5 minutes in 0.05% 2xSSC/ 0.1% NP40 at 45° C.

2.2.1.5.5 Detection, Capturing and Scoring of the Probes

The examination and scoring of the slides was carried out using a fluorescent Olympus (BX-40) microscope, which was fitted with a photometrics cooled CCD camera utilising pathVysion software (Digital Scientific, Cambridge, UK). Two hundred interphase nuclei were scored in order to calculate probe efficiency as part of the FISH study. All embryonic blastomeres were located using an England Finder. FISH probe signals in interphase nuclei were scored following Hopman *et al.*, (1988) such that two signals closer than a signals diameter apart were considered a single split signal and those further apart were considered as two separate signals.

2.2.1.5.6 Sequential FISH Procedure

After the detection and capturing of the 1^{st} (or 2^{nd}) round probe combination (Table 2.4), co-ordinates of three interphase nuclei and three metaphase spreads were recorded in the control lymphocyte slide using an England Finder in order to assess the efficiency of the probes in future rounds. The coverslip was removed with care so as not to disturb the nuclei. The slide was washed twice in 4xSSC/0.05% Tween 20 (while shaking) for 10 min each time. The slide was incubated in PBS, washed again

for 10 min and then dehydrated through a 70, 90, and 100% ethanol series. The probe mix was prepared for the second round of FISH.

The probes were separately denatured from the slide (section 2.2.4.3) for 5 min in 75° C and the slide incubated at 37° C overnight. Any unbound probe was removed by washing the slides for 10 minutes at 45° C in 70% formamide/2xSSC pH 5.3 (three times), 10 minutes in 2xSSC pH 7.0 at 45° C and then 5 minutes in 0.05% 2xSSC/ 0.1% NP40 while at 45° C. The slides were left to dry in the dark and mounted in Vectarsheild (Vector Laboratories, USA) containing 5µl of 0.2mg/ml 4',6'-diaminidino-2-phenyolindole (DAPI). The DAPI volume was lowered to reduce its intensity and thus avoid bleaching the aqua probe signals. The FISH signals were analysed as described in section 2.2.1.5.5

2.2.1.6 FISH Protocol for Chapter 4

Initially the study was carried out on control lymphocytes to select and optimise the FISH protocol. Subsequently, the study was performed on embryos that were not transferred and not suitable for freezing and were donated with written consent from patients undergoing routine IVF treatment at the London Fertility Centre.

2.2.1.6.1 FISH Protocol Selection and Optimisation

The FISH protocol was carried out as described by Harper *et al* (1994) above. However, some of the steps were altered in order to obtain high efficiency in all three sequential rounds of FISH. Several probe combinations were tested during the initial stages of this study (Table 2.3).

| Chromosome combinations | Probes used in the 1 st round | Probes used in the 2 nd round |
|---|--|--|
| 1 dual / 8 dual | 1het / 1p | 8cep / 8q |
| 4 / 16 dual & 4 / 11 dual | 4cep / 16p / 16q | 4cep / 11cep / 11q |
| 7 dual [#] / 18 & 11 dual / 18 | 7q / 7q / 18cep | 11cep / 11q / 18cep |
| 2/4/5 | 2q / 4cep / 5p | 5p&5q */ 2cep |
| 1 / 11 / 18 | 1p / 11q / 18cep | 1het / 11cep / 18q |
| X / Y / 18 (cocktail) | Хсер / Үсер / 18сер | Хсер / Үсер / 18сер |

Table 2.3. Multi-colour FISH-probe combinations

The probe is the 7 (7q11.23 and 7p31) Willams microdeletion probe

* The probe is the 5 (5q31 and 5p15.2) Cri-du-Chat microdeletion probe

All FISH protocols were tested on lymphocytes and their efficiency was measured on 200 interphase nuclei for each probe. Once each probe by itself was efficient, all the probes were combined and adjustments to salt concentrations, temperature, formamide concentration were carried out in various FISH experiments to achieve increased efficiency values and clear, sharp, large signals for each probe. The final three different combinations of 3-colour FISH in the three rounds were optimised. Probe combinations as well as stringency conditions were altered and tailored to produce the best possible fluorescent signals for all the chromosomes tested.

2.2.1.6.2 Final FISH Protocol

FISH was performed in three sequential rounds (section 2.2.1.5). The 1st and 2nd rounds were performed with probes for chromosomes 1, 11 and 18, whilst the 3rd round used probes for the sex chromosomes and chromosome 18 (Table 2.2). All probes except one (the probe 1stall/III in spectrum aqua) were obtained from Vysis (UK) Ltd. The first round included the following probes: 1p SpectrumGreen (telomere CEB108/T7), which hybridises to the sub-telomere region of the short arm of chromosome 1; 11q SpectrumOrange (telomere VIJyRM2072), which hybridises to the sub-telomere region of the long arm of chromosome 11; CEP18 SpectrumAqua (alpha satellite D18Z1), which hybridises to the centromere region of the chromosome 18. The second round included the following probes: 1het (satellite II/III), which hybridises to the heterochromatic region of chromosome 1 (laboratory-prepared); CEP11 SpectrumGreen (satellite D11Z1), which hybridises to the centromere region of the chromosome 11; 18q SpectrumOrange (telomere VIJyRM2050), which hybridises to the sub-telomere region of the long arm of chromosome 18. The third round included the following probes: repeated use of 18 SpectrumAqua (as an internal control); CEPX SpectrumGreen (alpha satellite DXZ1), which hybridises to the centromere region of chromosome X; and CEPY SpectrumOrange (alpha satellite DYZ3), which hybridizes to the centromere region of chromosome Y (Table 2.2).

2.2.1.6.3 Stringency Conditions

Each probe combination reacts differently with respect to stringency conditions. The telomeric (labelled with a "p" or a "q") and locus specific probes (labelled *LSI*)

required an LSI buffer solution to hybridise more efficiently onto the DNA. However, centromeric probes (labelled "*CEP*" or "*sat11/111*" or "*het*") required a CEP hybridisation buffer solution. During the post-hybridisation washes (see section 2.2.5.4) telomeric and locus specific probes needed a lower formamide concentration of 50% whereas centromeric probes need a 60% formamide concentration. In all three rounds of FISH, CEP buffer and 50% formamide was used. The denaturation time/temperature factor was modified since all the telomeric and LSI probes require 5min at 73°C, whereas CEP probes need 3min at 75°C. Hence, in all three rounds the probe combinations were denatured for 5min at 75°C. Also, further optimisation of the protocol was carried out whilst working on human embryos. It was found that if the denaturation step was carried out in a 75°C waterbath, rather than a 75°C oven, higher efficiencies were obtained. The final stringency conditions can be seen in Table 2.4. During this study, all the post hybridization washes were carried out according to the '*long washes*' protocol from Vysis (UK).

| Round of FISH | Probe combination | Volume of probe (µl) | Denaturation temperature/ time | Hybridisation time |
|-----------------------|----------------------|-------------------------|-----------------------------------|-----------------------|
| | 1p | 0.6 | | |
| 1 st Round | 11q | 0.6 | 75°C / 5min | Overnight |
| | 18cep | 0.5 | | |
| | CEP buffer | 3.3 | | |
| | | | | |
| | 1het | 1.0 | | |
| 2 nd Round | 11cep | 0.5 | 75°C / 5min | Overnight |
| | 18q | 0.6 | | |
| | CEP buffer | 2.8 | | |
| | | | | |
| 3 rd Round | X / Y / 18 | 2.0 | 75°C / 5min | 2 hours |
| | CEP buffer | 2.0 | | |

Table 2.4. Probe combinations and conditions for all three rounds of sequential FISH

cep = centromeric probe, het = heterochromatic probe

p = sub-telomeric probe for the small arm, q = sub-telomeric probe for the big arm

2.2.2 PCR Methods

PCR was carried out in the CGH (Chapter 5) as well as the PCR-PGD (Chapter 3) study. Blood from normal individuals was collected and prepared for PCR to be used as control. Specimens for each case (Chapter 3) consisted of maternal and paternal blood (as well as relative blood when applicable) collected in ethylene-di-amine-tetra-

acetic acid (EDTA) blood tubes. Both maternal and paternal bloods were prepared for PCR.

2.2.2.1 DNA Extraction from Blood

The method described by Lahiri and Nurnberger (1991) was employed for DNA extraction from blood.

Blood samples of 5ml were collected into Falcon centrifuge tubes (Falcon, UK) containing 400mM EDTA. Whole blood was transferred into centrifuge tubes and 5ml of low salt buffer TKM1 and 125µl of Igepal CA-630 (Sigma, UK) were added to lyse the red blood cells. The mixture was mixed well by inversion and shaking and centrifuged at 2,200rpm for 10 minutes. The supernatant was slowly removed and the pellet was washed in 5ml of TKM1 and 125µl of Igepal CA-630 as previously. The washing and spinning were repeated until redness of the pellet was diminished. The pellet was then re-suspended in 100µl TKM1. A total of 800µl of TKM2 and 50µl of 10% (w/v) sodium dodecyl sulphate (SDS) were added to the suspended pellet to lyse the white blood cells. The solution was mixed thoroughly by pipetting up and down. The tube was placed in a 55° C water-bath for >30min until the lumps disappeared completely. 300µl of 6M NaCl were added to ensure that all cells had lysed, and the tube was centrifuged at 10,000rpm for 10 minutes. The supernatant was transferred to a new centrifuge tube and the precipitated protein pellets were discarded. Two volumes of 100% ice-cold ethanol were added to the supernatant and the tube was inverted until the DNA strands precipitated. The sample was spun at 10,000rpm for 5 minutes and the supernatant was poured off. The remaining precipitated DNA was washed in 1ml of ice-cold 70% ethanol and spun at 10,000rpm for 5 minutes. The supernatant was discarded and the DNA pellet was left to dry for 5 minutes. The DNA pellet was dissolved in 300µl of 1xTE and stored at -20° C until further use.

2.2.2.2 Single Cell Isolation

2.2.2.1 Preparation for Single-cell Isolation

All single cell isolations and single cell PCR procedures were carried out in a dedicated room (termed single cell room) which was free of PCR amplified samples. The 'single cell' room was a room with positive pressure which was fitted with a

Class II Laminar flow hood were all preparations took place which was cleaned before and after each use with 100% ethanol. Dedicated bench microfuge, a set of Eppendorf pipettes, sterile Eppendorf tips (Eppendorf, UK), sterile gloves and laboratory coats were all separate from the main lab. Approximately 3μ l lysis buffer consisting of 2μ l of 125μ g/ml Proteinase K (PK) and 1μ l of 17mM sodium dodecyl sulphate (SDS) was pipetted into each of the microcentrifuge tubes before adding the single buccal cells, single human blastomeres and cell clumps (El-Hashemite and Delhanty, 1997). Once the single cell was transferred into the lysis buffer, the mixture was covered with a drop of light mineral oil to prevent contamination and evaporation before closing the lid. The lysis buffer was activated at 37° C for 1h, and inactivated by incubating at 99° C for 15 min. After lysis, the DNA from the single cells or cell clumps was ready for PCR or storage at -80° C.

2.2.2.2 Single Buccal Cells

 50μ l of the concentrated cell suspension was transferred to a 5cm petridish in a laminar flow cabinet. Approximately ten 30µl drops of PBS containing 0.1% PVA were spotted onto the petridish. A drop of cell suspension (about 500µl) was transferred to an adjacent drop of PBS using a pulled glass micropipette, while visualising under a dissecting microscope. The transfer of the cells in different PBS/0.1% PVA drops was repeated 2-3 times to dilute the cell concentration until the isolation of a single cell was achieved. The single cells, once isolated, were transferred in and out of at least three fresh PBS drops to wash away any contaminants. Then the single cells were transferred into an individual thin-wall microcentrifuge tubes (0.2µl Eppendorf tubes, Eppendorf, UK) containing lysis buffer. A clump of 3-5 cells was taken as a positive control for each PCR. Also, 2µl of the last drop was taken as a blank (negative control) for each single cell.

2.2.2.3 Isolation and Tubing of Single Human Blastomeres

Blastomeres were used as part of the PGD procedure carried out in Chapter 3 in a clinical setting. Furthermore, surplus blastomeres were used for Chapters 5 and 3 for research purposes.

Human blastomeres were provided following informed patient consent, from day 3 or day 4 donated spare embryos from standard IVF/ICSI cycles. The blastomeres biopsied were selected based on good morphological appearance. The single blastomeres derived from biopsy were subsequently washed through several drops of PBS/0.1% PVA, to remove excess contaminants such as cumulus cells or sperm, before they were transferred into individual thin-wall microcentrifuge tubes (0.2µl Eppendorf tubes, Eppendorf, UK) containing 3μ l PCR lysis buffer containing proteinase K (2.1.2.1.1). Each tube was checked under the microscope in order to visualise the cell within the tube. 2μ l of the last wash drop was taken as a blank for each single blastomere. The lysed cells were stored at -70°C.

2.2.2.3 PCR Procedure

2.2.2.3.1 DOP-PCR

The procedure for the DOP-PCR was carried out according to Wells *et al* (1999) (Chapters 3 and 5). The DOP50 PCR mixture consisted of 0.2mM deoxynucleoside triphosphates (dNTP's: dATP, dTTP, dCTP and dGTP), 2μ M DOP primer (Table 5.2) and 2 units of SuperTaq® (HT Biotechnology, UK) DNA polymerase with 10xSuperTaq buffer and was made up to a total volume of 50 μ l with nuclease-free, distilled, deionised sterile water (Promega, UK). The reaction mix was added to 200-300ng of genomic DNA or extracted DNA from single cells; one extra tube was taken as PCR-mix-only negative control for each reaction or blanks from each single cell. 25 μ l of lightweight mineral oil was added to prevent contamination and evaporation. The PCR was set up on ice, in a laminar flow cabinet using dedicated pipettes and sterile filtered tips.

The thermal cycling conditions were as follows: 94°C for 9 min, 30°C for 1 min, 72°C for 3 min (eight cycles); 94°C for 1 min, 62°C for 1 min and 72°C for 1.5 min (thirty five cycles); 72°C for 8 min.

2.2.2.3.2 Fluorescent PCR

Full details of the oligonucleotides used as primers for the fluorescent PCR can be seen in Table 2.2. The annealing (melting) temperature for the oligonucleotide primers was estimated by the following 'the-rule-of –thumb' calculation:

$$Tm = 2x(A+T) + 4x(G+C)$$

Where A, T, C and G are the nucleotides adenine, thymine, cytosine and guanine respectively. The working optimal annealing temperatures were determined empirically by experiments using the Mastercycler Gradient® thermal cycler. A temperature gradient of $\pm 5^{\circ}$ C from the calculated temperature across the block was carried out. The temperatures giving the most intense amplified products were considered as the optimal annealing temperature.

The PCR procedure was carried out as previously described by Piyamongkol *et al* (2001). The PCR mixture consisted of 0.2 μ M of each primer, 200 μ M deoxynucleoside triphosphates (dNTP's: dATP, dTTP, dCTP and dGTP) and 1 unit of SuperTaq® (Cambridge Technologies, UK) DNA polymerase with 1xSuperTaq buffer and was made up to a total volume of 25 μ l with nuclease-free, distilled, deionised sterile water (Promega, UK). The reaction was added to the genomic DNA or extracted DNA from single cells; one extra tube was taken as PCR-mix-only negative control for each reaction. 25 μ l of lightweight mineral oil was added to prevent contamination and evaporation. The PCR was set up on ice, in a laminar flow cabinet using dedicated pipettes and sterile filtered tips. For single cell PCR, 1.5 units of the polymerase enzyme were used in the reaction mixture. When AmpliTaq GoldTM polymerase with 1xGeneAmp® buffer were used instead of SuperTaq®, the primary denaturation step was set to 12 minutes to activate the enzyme. Table 2.5 reveals the conditions of the thermal cycler for all the cases.

| Enzyme Polymerase | SuperTaq® | AmpliTaq | Number of cycles |
|----------------------|------------|--------------------|--------------------|
| PCR steps | Plus | Gold TM | |
| Primary denaturation | 94°C 4min | 94°C 12min | 1 |
| Denaturation | 94°C 30sec | 94°C 45sec | 30 for genomic DNA |
| Annealing | 60°C 30sec | 60°C 45sec | 40 for single cell |
| Extension | 72°C 45sec | 72°C 1min | |
| Final Extension | 72°C 10min | 72°C 10min | 1 |

Table 2.5 Conditions for the thermal cycler using different polymerase enzymes

2.2.2.4 PCR Analysis

2.2.2.4.1 Gel Electrophoresis

This type of analysis was only used during the early stages where it was required to test the DOP-PCR product as well as the Fluorescent-PCR (F-PCR) product to ensure amplification.

A 2% agarose gel containing 0.1 μ l/ml ethidium bromide was prepared by mixing agarose in 1xTBE. The mixture was heated (approx 2min) gently in a microwave until the agarose dissolved completely. Ethidium bromide (5 μ l) was mixed into the agarose mixture and the mixture was poured into a mini-gel tank with an 8 or 16 well-forming-comb and left to set at room temperature (approx 20 min).

The comb was removed and the gel was immersed in 50ml of 1xTBE. Each PCR (8μ) product was mixed with a one tenth volume of the loading buffer and was loaded into each well. One Kb of ladder was used as a reference. Electrophoresis was performed for 30-60 min after which the gel was viewed under ultra-violet transillumination.

2.2.2.4.2 Automated laser DNA analyser (ABI PrismTM 310 and 3100)

Fluorescent PCR products were separated and analysed using an automated laser DNA analyser (ABI PrismTM 310 or 3100) and the appropriate software (GeneScan version 2.0.2; Applied Biosystems, UK). Each F-PCR product (1µl) was mixed with 12µl of deionised formamide and 0.5µl of the size standard (Genescan 500-TAMRA or Genescan 500-ROX, Applied Biosystem, UK). The mixture was run through a capillary (12kVolts, 26 min at 60°C) after 5 min denaturation at 95°C. Each PCR product was sized and distinguished illustrated as a peak using the GeneScan 2.0.2 version analysis software (Applied Biosystems, UK). Each peak represented the allele size of the product and the peak height the amount of the product.

2.2.2.4.3 Optimisation of PCR protocol

Each PCR protocol was optimised so the final product showed sharp, high peaks when analysed using the GeneScan 2.0.2 version analysis software (Applied Biosystems, UK). Efficiency of single cell PCR protocols was carried out by calculating the number of single cells which showed amplification of at least one correct size allele (either homozygote or heterozygote samples). The allele dropout (ADO) rate was calculated only for heterozygote samples. The ADO rate was determined from the cells that did show amplification. e.g. if 9/10 cells showed amplification, then the ADO rate would be calculated from the 9 cells.

2.2.2.5 PGD

2.2.2.5.1 PGD Consultation

Patients were referred to the UCL centre for PGD. In most cases referral was from clinical genetics departments and the patients had already had genetic counselling. Prior to commencing treatment all patients were fully informed regarding the limitations of PGD. The requirement for IVF treatment, the risk of misdiagnosis, the problems caused by mosaicism, the expected implantation and pregnancy rates, were all outlined during two thorough IVF/PGD consultations. Before commencement of treatment, patients had full gynaecological investigations and the male partners sperm was assessed and a specific single cell diagnoses was developed for each couple.

2.2.2.5.2 IVF Treatment and Manipulation of Embryos.

The patients underwent routine IVF procedures as described previously (Ranieri *et al*, 2001). Following ovarian stimulation follicles were aspirated and fertilisation was evaluated 24h after insemination. Oocytes and embryos were cultured in IVF medium (Cook, Australia).

2.2.2.5.3 Embryo Biopsy (UCL)

PGD was performed for two patients 'G' and 'H'. On day 3 embryos were biopsied by the senior embryologist in $Ca^{2+}Mg^{2+}$ -free embryo biopsy medium (Medicult, UK), using Research Instrument micromanipulators. Zona drilling was performed using acid Tyrode's solution as described previously (Piyamongkol *et al*, 2001a). One or two blastomeres were aspirated according to the developmental stage and morphology of the embryos. In general two cells were biopsied from embryos consisting of 6 or more cells. Biopsied cells were washed and tubed for analysis (section 2.2.2.2.3). The untransferred (spare) embryos were donated for research with the patient consent for confirmatory diagnosis. All the cells from the spare embryos were biopsied and were tubed for molecular analysis (section 2.2.2.2.3).

2.2.2.5.4 Embryo Biopsy (LFC)

The embryo biopsy method carried out for this study was different to that used in chapter 3 carried out in the UCL Assisted Conception Unit (section 2.2.2.5.3). The embryos were incubated for a 1-2 minutes in Ca^{+2}/Mg^{+2} -free biopsy medium. The zona pellucida surrounding the embryo was drilled with a laser system as described by Veiga *et al* (1997). 1-2 cells were removed gently from the embryo and the embryo was subsequently washed in IVF medium. The embryos were then placed in blastocyst medium (6.2 Vitrolife, Scandinavia) from day 3-5.

2.2.2.6 PGD Strategy for DM Using F-PCR

The diagnosis of a normal embryo was based on the observation of two normal DM alleles, one from each parent. The mutant DM expanded allele cannot be detected since fragment sizes above 500bp are refractory to PCR, hence an affected embryo would only show one normal allele. Different polymorphic markers (STR's) linked and unlinked, were tried for each family to test whether they were informative (Table 5.2). A linked marker, such as the D19S112 which is located approximately 0.3cM upstream to the DMPK, would provide further information about the genetic state of the embryo i.e. if the embryo carries the mutation or not. An unlinked marker, such as the D21S1414, would only allow the detection of contamination (Piyamongkol *et al*, 2001a).

2.2.2.6.1 Genomic DNA Workup

Each family was tested for a variety of STR's from genomic DNA extracted from their blood (section 2.2.2.1), which were classed as informative depending on their allele size. For example, a tetranucleotide STR would be rendered informative if all four alleles from the two parents were of different size and at least one base pair apart from each other i.e. 120/130 for the mother and 126/142 for the father. If a family was informative for a linked marker, another affected member was asked to produce a DNA sample in order to obtain the 'phase'. Table 2.2 summarises the different STR's tested for all the couples' extracted genomic DNA (see Tables 3.1-3.10 for results)

2.2.2.6.2 Single Cell Workup

If a couple had at least one informative linked or unlinked polymorphic marker including the DM mutation marker, testing on single cells was initiated for each couple individually. Fifty single buccal cells along with 10-20 clumps of buccal cells were tested for each multiplex reaction on buccal cells from a normal heterozygous individual to test the protocol at the single cell level.

The protocol for family G was the multiplex reaction of DM+D21S1414 (Protocol 1), whereas for family H was the multiplex reaction of DM+D19S112 (Protocol 2). Subsequently, both protocols were performed on at least 100 single cells isolated from each partner of each family as well as spare single human blastomeres for assessing the amplification efficiency and ADO rates. The ADO rates were calculated from the number of single cells that were amplified, when only one allele was visible as a peak.

For the optimal protocol in both cases, the PCR mixture consisted of 0.1μ M of the DM primer and 2.5 μ M for the D21S1414 primer (Protocol 1) and 3.0 μ M for the D19S112 primer (Protocol 2), 200 μ M dNTP's, 10xGeneAmp® Buffer and 1.5U AmpliTaq GoldTM and was made up to a final volume of 25 μ l with double-distilled, nuclease-free, de-ionised water. The amplifications were performed with the conditions 94°C, 45sec (96°C for the first ten cycles), annealing at 60°C, 45sec for family G (Protocol 1) and 61°C 45sec for family H (Protocol 2) and extension of 72°C, 1min for 42 cycles. These were preceded by denaturation at 94°C for 12 min to activate the AmpliTaq GoldTM enzyme.

Each marker was analysed on the 3100ABI PrismTM as wells as backup analysis on the 310 ABI PrismTM. Since each marker was fluorescently labelled differently (see Table 5.2), analysis was possible in different dyes.

2.2.2.6.3 Clinical DM PGD Cases

Following ICSI treatment, one PGD cycle for each family (G and H) was performed. The single biopsied blastomeres were analysed using the optimal PGD protocols 1 and 2. Normal embryos (if any) were chosen for transfer on day 4 post-fertilisation. All the cells from the untransferred embryos were biopsied and analysed using the same protocols for each family for confirmatory results.

2.2.2.7 PGD Strategy for DM Using WGA

A universal protocol for PGD for DM was attempted using DOP-PCR as the method of whole genome amplification and then subsequent amplification with five different markers. The markers that were selected were DM (mutation marker), D19S112 (linked polymorphic marker) as well as D13S305, D18S535 and D21S1414 (unlinked polymorphic markers) (Table 5.2). These unlinked polymorphic markers were chosen as they were on different chromosomes in order to observe the coverage of the genome when amplified using DOP-PCR.

2.2.2.7.1 Procedure on Genomic DNA

The DOP-PCR procedure was carried out as described by Wells *et al* (1999) (section 2.2.2.3.1) initially on genomic DNA. The F-PCR procedure was carried out as previously described by Piyamongkol *et al* (2001) (section 2.2.2.3.2). However, not all the primers produced efficient yields when analysed on the automated laser DNA analyser (ABI PrismTM 3100). Therefore, the F-PCR protocols were adjusted, at the genomic DNA level, in order to generate sufficient quantities of PCR product when carried out at the single cell level. Several annealing temperatures were tested using a Gradient Thermal Cycler as well as different primer concentrations, to produce an efficient PCR protocol. The final annealing temperatures and concentrations of the primers can be seen in Table 2.6.

Table 2.6. Annealing temperatures and primer concentrations for the singleplex reactions for each polymorphic marker

| Type of Marker | Marker | Annealing Temperature (°C) | Primer Concentration |
|-----------------------|----------|----------------------------------|-------------------------|
| STR (Dinucleotide) | DM | 60 °C | 0.1 μM |
| STR (Dinucleotide) | D19S112 | 61 °C | 0.6 µM |
| STR (Tetranucleotide) | D21S1414 | 62 °C | 1.0 µM |
| STR (Tetranucleotide) | D18S535 | 60 °C | 0.6 μΜ |
| STR (Tetranucleotide) | D13S305 | 56 °C | 0.8 μΜ |

2.2.2.7.2 Procedure on Single Cells

The single cell DOP-PCR procedure was optimised during the CGH study (section 4.2.6). Once the singleplex F-PCR protocols were optimised on single cells, two individuals were chosen ('X' and 'Z') from which ten single cells and two clumps from each were amplified using DOP-PCR. Subsequently, a 1.0µl aliquot from each single cell DOP-PCR product was used as a template for each singleplex reaction with each of the markers (Table 5.2). The final product was run on the 3100ABI PrismTM for analysis. Genomic DNA from each individual was amplified to act as positive control.

2.2.2.7.3 CGH on DOP-PCR Amplified Single Cells

From a 50µl DOP-PCR reaction, only 10µl were used. The rest of the DOP-PCR product was utilised for CGH analysis. The full CGH protocol can be seen in section 2.2.3.4, which was performed as described by Wells *et al* (2002). The optimisation of the CGH protocol was performed as part of the study in Chapter 5.

2.2.3 CGH Methods

CGH was used for the study summarised in Chapter 5 (as well as Chapter 3). Some sections of the CGH Methods are identical to PCR; hence similar methods will be referred to the appropriate sections.

2.2.3.1 DNA Extraction from Frozen Fibroblasts

Several cell pellets (trisomy 13, 18, 21, 22, XXX and normal) were subjected to DNA extraction to be subsequently used as positive controls for preliminary studies to assess efficiency of the genomic CGH protocol. Cell suspensions were centrifuged at 1000rpm for 5 minutes, and the supernatant containing glycerol was removed. The cells were washed with Hank's medium (Sigma, UK) and centrifuged at 6,000rpm for 5 minutes. The supernatant was removed and 2.5ml of DNA extraction lysis buffer (section 2.1.3.1.2) warmed at 37°C was added and the cells were left at 37°C for 30 minutes. An equal volume of isopropanol was added in order to precipitate the DNA which was "hooked out" and dissolved in water.

For DNA Extraction from Whole Blood refer to section 2.2.2.1.

2.2.3.2 Single Cell isolation

Refer to section 2.2.2.2

2.2.3.3 DOP-PCR

Refer to previous section 2.2.2.3.1

2.2.3.4 CGH Procedure

2.2.3.4.1 Labelling of Probe DNA

The DNA product of DOP-50 was ethanol precipitated, redissolved in water and labelled by nick translation. The 50 μ l reaction contained 5 μ l of 10X reaction buffer mix, 10 μ l 0.1mM dNTP Mix, 5 μ l 0.1 mM dTTP mix, 2.5 μ l of 0.2mM Hapten/fluorochrome-dUTP, 10 μ l DNA Polymerase (10U/ μ l) and 17.5 μ l of Nuclease-free water (Promega, UK). The time for the nick translation was between 60-120 minutes at 15°C. The reaction was stopped with a 10 minute incubation at 72°C

2.2.3.4.2 Preparation of Labelled DNA

A 1% agarose gel was run to check the fragment sizes (see section 2.2.2.4.1). Optimal sizes for CGH were estimated between 300-2000bp. The reference DNA was labelled with Spectrum Red (Vysis, UK) and the test DNA was labelled with spectrum Green (Vysis, UK). The appropriate DNA to be hybridised against the single blastomere was a group of 3 buccal cells that had undergone the same treatment as the single blastomeres and were amplified during the same DOP50-PCR experiment. Labelled reference and test DNA were mixed with 30µg Cot-1 DNA and ethanol precipitated and washed with 70% ethanol and air-dried. The pellet was resuspended in 6µl Hybridisation mix (50% formamide; 2xSSC; 10% dextran dulphate) and dissolved by 20 minutes incubation at 37°C. The probe was denatured at 75°C for 10 minutes and cooled at 37°C by incubation for 2 minutes at room temperature in the dark before being applied to the denatured normal chromosome spreads as described below.

2.2.3.4.3 Denaturation of Metaphase Chromosome Spreads and Probe Hybridisation

Denaturation of male lymphocyte slides was performed by a 5 minute incubation of the slides in a coplin jar with denaturation solution (70% formamide; 2xSSC pH 7.5) pre-warmed at 73°C. Immediately after denaturation the slides were put through an ice-cold (chilled at -20°C) ethanol series (70%, 85%, 100%) and dried. The probe was applied to the slide and covered with a coverslip sealed with rubber cement and hybridised in a moist chamber at 37°C for 72 hours.

2.2.3.4.4 Post-Hybridisation Washing

The post hybridisation washes consisted of 5 minutes in 2x SSC at 72°C, 5 minutes in 4xSSC at 37°C, 5 minutes in 4xSSC + 0.1% Triton-X (Sigma, UK) at 37°C, 5 minutes in 4xSSC at 37°C, and 5 minutes in 2xSSC at room temperature followed by dipping of the slides in double-distilled water. The slides were put through an ethanol series 3 minutes each, air-dried and mounted in anti-fade medium (Vector Labs, Peterborough, UK) containing diamidinophenylindole (DAPI) to counterstain the chromosomes and nuclei.

2.2.3.4.5 Microscopy and Image Analysis

Metaphase chromosome preparations were captured using a Zeiss Axioscope microscope equipped with a Photometrics KAFF 1400 cooled CCD camera, and SmartCapture software (Vysis Richmond, UK). Image analysis was performed using Vysis Quips CGH software. Green:Red fluorescence ratios of >1.2:1 indicated gain of genetic material, while ratios of <0.8:1 was indicative of deletions.

2.2.3.5 CGH Procedure for Chapter 5

This study was performed on frozen embryos which were subsequently thawed (frozen-thawed embryos) generated from patients undergoing IVF and ICSI at the London Fertility Centre. These embryos were suitable for freezing, hence were considered good quality embryos when assessed morphologically. All embryos were donated with written consent from patients and the study was licensed by the Human Fertilisation and Embryology Authority.

Quality control was performed for every step of this study. Prior to commencing this study a preliminary study was performed to ensure the optimal method for analysing single blastomeres by CGH. Single cell isolation of 100 buccal cells was carried out. Subsequent tubing of the buccal cells was followed by DOP–PCR. The final amplified product was run on a 2% agarose gel electrophoresis carried out at 50V for 30 minutes after which gels were viewed via ultra-violet trans-illumination to assess the efficiency of the tubing technique.

2.2.3.5.1 WGA Protocol

DOP-PCR was employed to amplify the whole genome of the cell uniformly as described by Wells *et al* (2002) using the Eppendorf PCR machine (See section 2.2.3.3)

2.2.3.5.2 Optimisation of WGA Protocol

For the WGA technique, namely DOP-PCR, gel electrophoresis provided the only means of assessing whether the PCR conditions were optimal. A non-intense smear should be visible for each sample; genomic DNA, clump of buccal cells or single cell. Furthermore, in each experiment a negative control was added for the DOP-PCR amplification to assess each reaction for contamination. All negative controls were tested by gel electrophoresis and by carrying out CGH experiments where the negative controls acted as '*test*' samples.

Two different thermal cyclers were tested: the Eppendorf and the Hybaid Omnigene (see section 2.1.5.2) in order to examine which provided the best amplification results. The DOP-PCR products were analysed on a 2% agarose gel

2.2.3.5.3 CGH Protocol

The full CGH protocol is detailed in section 2.2.3.4, which was performed as described by Wells *et al* (2002). Each Nick-Translation kit (Vysis, UK) was tested before use, since each aliquot of the Nick-translation enzyme (contained in each kit) behaved differently and required slightly different times of incubation. Hence, a trial Nick-Translation CGH experiment was set up after purchase of a new kit, and it was assessed by carrying out DOP on single buccal cells, labelling the sample and testing the fragment sizes by gel electrophoresis. The optimal sizes for CGH were estimated to be between 300-2000bp (Wells *et al*, 1999).

Six to twelve metaphases were captured using a Zeiss Axioscope microscope equipped with a Photometrics KAFF 1400 CCD camera. Each metaphase was karyotyped and assessed using Vysis Quips CGH software.

2.2.3.5.4 Optimisation of CGH Protocol

The optimisation of the CGH protocol was achieved by testing initially with high quality genomic DNA, which had been extracted 'in house', from 'healthy' individuals. Furthermore, genomic DNA from frozen fibroblast cell lines (cultured inhouse) with a defined chromosomal abnormality such as trisomy 13, trisomy 18, trisomy 21, trisomy 22 and triploidy 69 XXX was employed as a positive control in assessing genomic CGH protocol efficiency. Single cell isolation was performed for buccal cells and fibroblasts to provide practice, and the positive controls in assessing single cell CGH protocol accuracy. A set of embryos were thawed with patient consent and were biopsied in order to establish the biopsy technique as well as the CGH technique applicable on single blastomeres. This set of embryos were used as a learning curve for both biopsy practitioner and the author.

2.2.3.5.5 FISH Protocol

The FISH protocol was carried out as described by Harper *et al* (1994). However, some of the steps were altered in order to obtain high efficiency in all the sequential rounds of FISH. All day 5 embryos were subjected to two sequential rounds of FISH (section 2.2.1.5). The first round of FISH was always carried out using the probe cocktail for chromosomes X, Y and 18, which acted partly as a control for the CGH experiment since the sex of the biopsied cells should match the sex of the rest of the

embryo (unless mosaic). The second round of FISH was a combination of 9cep/16cep/22LSI if the CGH showed euploid results for the biopsied blastomeres, or a combination of probes matching the abnormalities observed by CGH in as many chromosomes as possible. Hence, there were a variety of FISH protocols carried out for this study. The different combinations can be viewed in Table 2.7.

| Chromosomes | Probe | Stringency Conditions | | | | | | |
|-------------|------------------|-----------------------|--|-------------------------|--|--|--|--|
| | Combination | Probe Vol (μl) | Denaturation | P-H ^a washes | | | | |
| X, Y, 18 | Xcep/Ycep/18cep | 2µl ^b | Co [°] , 75 [°] for 5min | 60% FA | | | | |
| 9, 16, 22 | 9cep/16cep/22LSI | 0.5µl/0.5µl/0.7µl | Sep ^d , 75° for 5min | Vysis ^e | | | | |
| 3, 11, 13 | 3cep/11cep/13LSI | 0.5µl/0.5µl/0.7µl | Sep, 75° for 5min | Vysis ^e | | | | |
| 10, 14 | 10cep/14q | 0.5µl/0.6µl | Co, 75° for 5min | 60% FA | | | | |
| 1, 16 | 1p/1q/16cep | 0.6µl/0.5µl/0.5µl | Sep, 75° for 5min | 50% FA | | | | |
| 3, 6, 18 | 3cep/6cep/18q | 0.5µl/0.5µl/0.6µl | Sep, 73° for 5min | 60% FA | | | | |
| Y, 4 | Ycep/4cep | 0.5µl/0.5µl | Co, 75° for 5min | 60% FA | | | | |
| X, Y, 16 | Xcep/Ycep/16q | 0.6µl/0.5µl/0.6µl | Co, 75° for 3min | 50% FA | | | | |

Table 2.7. FISH probe combinations and stringency conditions.

^aPost-Hybridisation washes

^bThis was a probe cocktail, where $2\mu l$ from the cocktail were added into the probe mixture ^c Co-denaturation

^dSeparate Denaturation (section 2.2.1.5.3)

^eThe post-hybridisation washes termed 'long washes' proposed by Vysis (UK) (section 2.2.1.5.3)

For probe information refer to section 2.1.5.1 in Table 2.1. Most of the probe combinations required co-denaturation with the slide. However, there were also probe mixtures where diffused signals and cross-hybridisation was observed, therefore separate denaturation of the probes and the slide was performed (see section 2.2.1.5.3).

2.2.3.5.6 Optimisation of FISH Protocol

During each FISH experiment on the embryos a control slide of normal male lymphocytes was also included to assess the efficiency of the FISH protocol. 50-100 interphase nuclei were scored and the efficiency of each probe combination was examined.

2.2.3.6 Patient Details for Chapter 3

A total of ten couples were initially referred to our centre for PGD of DM. PGD was performed for two couples: G and H (Table 2.8). The rest are awaiting PGD for various reasons.

| Family | Patient with | Maternal | Reproductive History |
|--------|--------------|----------|--|
| | DM expansion | Age | |
| Α | Mother | 42 | 6 pregnancies, 2 affected sons, 4 spontaneous |
| | | | miscarriages |
| В | Mother | 39 | Infertile (Gave birth to healthy female after PGD) |
| С | Father | 38 | 2 pregnancies, 1 normal birth, 1 affected (TOP) |
| D | Mother | 40 | 4 pregnancies (x2 with IVF), 3 TOP, 1 affected |
| | | | son, 1 PGD cycle where all were affected |
| Е | Mother | 36 | 1 pregnancy, 1 affected female which died at |
| | | | weeks old |
| F | Mother | 39 | 4 pregnancies, 1 normal son, 3 TOP |
| G | Mother | 36 | 1 affected son |
| | | | 4 affected natural pregnancies |
| Н | Mother | 41 | 4 pregnancies, 4 TOP's |
| Ι | Mother | 40 | 2 pregnancies, 1 affected son, 1 TOP |
| J | Father | 36 | 3 pregnancies, 1 normal female, 1 affected male, |
| | | | 1TOP |

Table 2.8. Patient details for couples undergoing PGD for DM

2.2.3.7 Patient Details for Chapter 4

All patients were healthy individuals that were referred to the London Fertility Centre for fertility problems. Patients were either IVF or ICSI patients. Table 2.9 reveals the background information of all the patients and their embryos. All patients gave consent for their embryos to be used for research. The first three patients' embryos (n=10) were used for preliminary optimisation of the protocol (data not shown).

| Patient | Age of | Date of | IVF or | No. of eggs | No. of embryos | Embryo |
|---------|--------------------|-----------|--------|-------------|-----------------|-------------|
| No. | patient | treatment | ICSI | collected | (no. of nuclei) | Туре |
| 4 | 40 | 5/5/02 | ICSI | 8 | 2 | Arrested |
| | | | | | (4) | |
| 5 | 35 | 17/5/02 | IVF | 20 | 5 | Arrested |
| | | | | | (65) | |
| 6 | 47 | 17/5/02 | ODOR/ | 13 | 1 | Arrested |
| | (35-O.D.) | | IVF | | (22) | |
| 7 | 36 | 28/6/02 | ODOR/ | 17 | 2 | Arrested |
| | (33-O.D.) | | ICSI | | (35) | |
| 8 | 41 | 05/07/02 | ICSI | 9 | 1 | Arrested |
| | | | | | (6) | |
| 9 | 27 | 05/07/02 | IVF | 11 | 1 | Arrested |
| | | | | | (8) | |
| 10 | 41 | 08/07/02 | ICSI | 12 | 1 | Arrested |
| | | | | | (5) | |
| 11 | 34 | 09/07/02 | IVF | 37 | 1 | Lost |
| | | | | | (1) | |
| 12 | 35 | 09/07/02 | IVF | 34 | 2 | Arrested |
| | | | | | (31) | |
| 13 | 37 | 30/08/02 | ICSI | 12 | 3 | Blastocysts |
| | | | | | (110) | |
| 14 | 19 | 30/08/02 | IVF | 13 | 3 | Blastocysts |
| | | | | | (90) | |
| 15 | 26 | 03/10/02 | ICSI | 19 | 4 | Expanded |
| | | | | | (236) | Blastocysts |
| 16 | 33 | 04/10/02 | ICSI | 19 | 1 | Expanded |
| | | | | | (58) | Blastocyst |
| 17 | 31 | 09/01/03 | IVF | 11 | 2 | Arrested |
| | | | | | (58) | |
| 18 | 38 | 21/01/03 | IVF | 7 | 1 | Hatched |
| | | | | | (55) | Blastocyst |
| 19 | 30 | 31/01/03 | ICSI | 14 | 2 | Expanded |
| | | | | | (175) | Blastocysts |
| 20 | 32 | 03/02/03 | ICSI | 16 | 1 | Expanded |
| | | | | | (43) | Blastocyst |
| 21 | 39 | 04/02/03 | IVF | 8 | 2 | Arrested |
| | | | | | (57) | |
| 22 | 39 | 25/03/03 | IVF | 21 | 2 | Expanded |
| | | | | | (143) | Blastocysts |
| 23 | 49 | 28/03/03 | ODOR/ | 23 | 1 | Hatched |
| | (20 - O.D.) | | IVF | | (100) | Blastocyst |
| 24 | 41 | 04/04/03 | ICSI | 6 | 1 | Expanded |
| | | | | | (63) | Blastocyst |
| 25 | 36 | 07/04/03 | IVF | 11 | 2 | Expanded |
| | | | | | (94) | Blastocysts |
| 26 | 34 | 14/05/03 | IVF | 9 | 2 | Expanded |
| | | | | | (61) | Blastocysts |

Table 2.9. Patient details for Chapter 4

| 27 | 46 | 04/06/03 | ODOR/ | 11 | 2 | Arrested |
|----|-----------|----------|-------|----|------|----------|
| | (33-O.D.) | | IVF | | (59) | |

IVF = In vitro fertilisation, ICSI = Intracytoplasmic sperm injection, ODOR = oocyte donation cycle, O.D. = oocyte donor

2.2.3.8 Patient Details for Chapter 5

All patients were healthy individuals that were referred to the London Fertility Centre for fertility problems. Patients were either IVF or ICSI patients. Table 2.10 below reveals the background information of all the frozen embryos. All patients gave consent for their embryos to be thawed and used for research.

| D 4 4 4 | The second down beganning the second second and the second s | | | | | | | | | |
|---------|--|------------|------|---------------|-------------|--------|----------------------------|-------|-------|--|
| Patient | Age of | Date of | IVF | NO. OI | Frozen | Embry | Embryo Type (time of thaw) | | | |
| No. | patient | Egg | or | eggs | embryos | (Grade | (Grade) | | | |
| | | Collection | ICSI | collected | donated for | | | | | |
| | | | | | research | 1 | 2 | 3 | 4 | |
| 1 | 38 | 19/03/03 | IVF | 21 | 3 | 7c | 6c | 6c | | |
| | | | | | | (2-3) | (2-3) | (1-2) | | |
| 2 | 29 | 20/04/03 | ICSI | 30 | 3 | 7c | 7c | 5c | | |
| | | | | | | (3) | (2) | (2) | | |
| 3 | 35 | 05/02/97 | IVF | 16 | 1 | 6c | - ` ź | | | |
| | | | | | | (2) | | | | |
| 4 | 35 | 23/02/99 | ICSI | 16 | 4 | 5c | 8c | 7c | 6c | |
| | | | | | | (3) | (2) | (2-3) | (2-3) | |
| 6 | 30 | 18/05/02 | IVF | 18 | 3 | 5c | 8c | 9c | | |
| | | | | | | (2-3) | (1-2) | (1-2) | | |
| 7 | 39 | 06/09/01 | IVF | 14 | 3 | 5c | 10c | 7c | | |
| | | | | | | (2) | (1-2) | (2) | | |
| 8 | 31 | 02/03/03 | IVF | 19 | 2 | 6c | 5c | | | |
| | | | | | | (2-3) | (2) | | | |
| 9 | 33 | 14/05/02 | IVF | 22 | 4 | 6c | 8c | 5c | 6c | |
| | (OD) | | | | | (2-3) | (1-2) | (3) | (2-3) | |
| 9 | 33 | 14/05/02 | IVF | 22 | 4 | 5c | 6c | 7c | 8c | |
| | (OD) | | | | | (2-3) | (2) | (2-3) | 1-2) | |
| 10 | 32 | 23/02/01 | IVF | 12 | 1 | 6c | | | | |
| | | | | | | (1-2) | | | | |
| 11 | 35 | 7/10/02 | IVF | 15 | 1 | 5c | | | | |
| | | | | | | (2) | | | | |
| 12 | 30 | 19/03/99 | IVF | 14 | 3 | 5c | 8c | 7c | | |
| | | | | | | (2) | (1-2) | (2-3) | | |
| 13 | 36 | 12/10/00 | ICSI | 24 | 3 | 5c | 8c | 7c | [| |
| | | | | | | (1-2) | (1-2) | (1-2) | | |
| 14 | 28 | 12/06/02 | IVF | 18 | 3 | 6c | 6c | 5c | | |
| | (OD) | | | | | (1) | (1) | (1-2) | | |

Table 2.10. Patient details regarding their IVF/ICSI cycle and their embryo details

IVF = In vitro fertilisation, ICSI = Intracytoplasmic sperm injection, OD = oocyte donor, OR = oocyte recipient

CHAPTER 3

Novel approaches for the preimplantation genetic diagnosis of Myotonic Dystrophy (DM1)

3.1 Aims

There were two parts to this study. The first was to develop a novel protocol for PGD for DM1 for a number of patients using a single, multiplex PCR and the second was to determine if WGA followed by a number of singleplex PCRs could be used for PGD of DM1 with the potential to also carry out CGH.

PGD for DM1 requires inclusion of highly polymorphic markers, the primary function of which is to confirm the affected embryos and reveal when contamination has occurred. Several polymorphic markers, linked and unlinked, were tested for a variety of families. Two couples underwent PGD for DM1 employing two different protocols. The first protocol included the mutation marker and a contamination marker whereas the second novel protocol included the mutation marker and a linked marker which was able to provide information regarding the mutation as well as presence of contamination.

Due to the complexity of developing specific protocols depending on the patients being informative for different markers and optimising the multiplex PCR, the second part of the project was attempted. A whole genome amplification method, namely DOP-PCR, was employed initially on genomic DNA and subsequently on single cells to investigate whether specific regions of the genome were amplified. Five different markers were tested to see if they were adequately amplified, a mutation marker (DM1), a linked marker (D19S112) and three highly polymorphic markers on different chromosomes (D21S1414, D18S535 and D13S305). The combination of WGA and F-PCR would provide a universal-like protocol able to tackle the time-consuming predicament of devising patient-specific single cell multiplex protocols, since after DOP-PCR of a single cell, sufficient amounts of DNA would be available for many singleplex reactions. Furthermore, the ability to carry out other assays from one single cell such as CGH could be performed.

3.2 Results

3.2.1 Preliminary Work-up of Families

Each family was referred to our centre (UCL Centre for Preimplantation Genetic Diagnosis) where bloods were taken from each prospective parent or DNA aliquots were sent from cytogenenetic laboratories where original investigations were carried out. The extracted DNA (section 2.2.2.1) from both partners was tested with a variety of markers to determine for which markers they are informative in order to devise a multiplex F-PCR protocol (Tables 3.1-3.10). If a linked marker was informative for each patient (i.e. parents share no allele in common), if possible, an affected family member was asked to provide a DNA sample in order to obtain the 'phase' (Underlined allele sizes in Tables 3.1-3.10). Although for some families the same F-PCR protocols can be devised and applied e.g. the DM/APOC2 duplex protocol for families B, E, F and I; each family was informative for different combinations of markers. Furthermore, there were some families where the DM1 primers were semiinformative since they share one allele (B, E, F and I), hence the diagnosis could only be carried out by multiplexing with a linked marker (Tables 3.2, 3.5, 3.6 and 3.9). For family C although there were four informative markers (APOC2, D19S112, D21S1414 and D21S11) because their allele sizes were very close, stutter peaks can create problems during diagnosis (Table 3.3).

| Table 3.1. | List | of | polymorphic | markers | tested | on | genomic | DNA | for | family | A | and |
|--------------|-------|----|-------------|---------|--------|----|---------|-----|-----|--------|---|-----|
| their allele | sizes | | | | | | | | | | | |

| Primer | Marker | Types | Maternal allele sizes | Paternal allele sizes | Comments |
|----------|----------|--------|--------------------------|--------------------------|-----------------|
| | | | Am | Af | |
| DM1 | | | - /146 | 122 (hm) | Informative |
| APOC2 | Linked | di- | 149/151 | 126/149 | Not informative |
| D19S207 | Linked | di- | 145/145 | 140/143 | Not informative |
| D19S112 | Linked | di- | 130/132 | 130/136 | Not informative |
| D19S393 | Linked | Tetra- | 285/285 | 285/285 | Not informative |
| D21S1414 | Unlinked | tetra- | 349/349 | 345/349 | Not informative |
| D21S11 | Unlinked | tetra- | 227/227) | 223/227 | Not informative |
| D18S535 | Unlinked | tetra- | 485/485 | 481/485 | Not informative |

Patient in *italics* indicates the affected partner

Table 3.2. List of polymorphic markers tested on genomic DNA for family B and their allele sizes

| Primer | Marker | Types | Maternal allele sizes | Paternal allele sizcs | Mother's father allele sizes | Comments |
|----------|----------|--------|-----------------------------|-----------------------------|------------------------------------|------------------------------|
| | | | Bm | Bf | Bfm | |
| DM1 | _ | | 120/- | 120/138 | 170/- | Semi-informative |
| APOC2 | Linked | di- | 126/ <u>153</u> | 142/157 | 150/ <u>153</u> | Informative |
| D19S207 | Linked | di- | <u>125</u> /141 | 144/144 | <u>125</u> /125 | Informative ^{&} |
| D19S112 | Linked | di- | <u>128/130</u> | 116/123 | <u>128/130</u> | Informative |
| D19S393 | Linked | tetra- | 285/ <u>296</u> | 296/296 | 289/ <u>296</u> | Not informative |
| D21S1414 | Unlinked | tetra- | 343/348 | 339/343 | 335/348 | Not informative |
| D21S11 | Unlinked | tetra- | 223/228 | 223/229 | 225/227 | Not informative |
| D18S535 | Unlinked | tetra- | 478/486 | 478/482 | 477/490 | Not informative |

[&]Practically difficult since close allele sizes would cause difficulties in interpretation due to stutter peaks

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles would be classed as affected

Patient in *italics* indicates the affected partner

- indicates the affected allele that is refractory to PCR

| Table 3.3. | List o | of polymorphic | e markers | tested | on | genomic | DNA | for | family | С | and |
|--------------|--------|----------------|-----------|--------|----|---------|-----|-----|--------|---|-----|
| their allele | sizes | | | | | | | | | | |

| Primer | Marker | Types | Paternal allele sizes | Maternal allele sizes | Comments |
|----------|----------|--------|--------------------------|--------------------------|------------------------------|
| | | | Cf | Cm | |
| DM1 | | | 146/- | 140/149 | Informative |
| APOC2 | Linked | di- | 134/153 | 149/155 | Informative ^{&} |
| D19S207 | Linked | di- | 141/145 | 144/144 | Not informative |
| D19S112 | Linked | di- | 130/132 | 124/134 | Informative ^{&} |
| D21S1414 | Unlinked | tetra- | 337/355 | 341/349 | Informative |
| D21S11 | Unlinked | tetra- | 215/233 | 219/237 | Informative |
| D18S535 | Unlinked | tetra- | 485/485) | 477/482 | Not informative |

[&]Practically difficult since close allele sizes would cause difficulties in interpretation due to stutter peaks

Patient in *italics* indicates the affected partner

| Primer | Marker | Types | Maternal allele sizes | Paternal allele sizes | Mother's sister allele sizes | Comments |
|----------|----------|--------|-----------------------------|-----------------------------|------------------------------------|-----------------|
| | | | Dm | Df | Dms | |
| DM1 | | | 139/- | 146/184 | 139/- | Informative |
| APOC2 | Linked | di- | 128/ <u>150</u> | 128/152 | 135/ <u>150</u> | Not informative |
| D19S207 | Linked | di- | 125/143 | 142/142 | 125/143 | Not informative |
| D19S112 | Linked | di- | 123/128 | 128/132 | 124/128 | Not informative |
| D19S393 | Linked | tetra- | 281/289 | 285/289 | 285/289 | Not informative |
| D21S1414 | Unlinked | tetra- | 337/337 | 343/348 | 338/361 | Not informative |
| D18S535 | Unlinked | tetra- | 482/486 | 478/490 | 486/490 | Informative |

Table 3.4. List of polymorphic markers tested on genomic DNA for family D and their allele sizes

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles would be classed as affected Patient in *italics* indicates the affected partner

- indicates the affected allele that is refractory to PCR

| Table 3.5. List of polymorphic | markers tested on genor | nic DNA for family E and their |
|--------------------------------|-------------------------|--------------------------------|
| allele sizes | | |

| Primer | Marker | Туре | Maternal allele sizes | Paternal allele sizes | Mother's brother allele sizes | Comments |
|----------|----------|--------|-----------------------------|-----------------------------|-------------------------------------|------------------|
| | | | Em | Ef | Emb | |
| DM1 | | | 122/- | 122/149 | 122/- | Semi-Informative |
| APOC2 | Linked | di- | 142/ <u>156</u> | 126/152 | 152/ <u>156</u> | Informative |
| D19S207 | Linked | di- | 125/ <u>143</u> | 142 (hm) | <i>127/<u>143</u></i> | Not Informative |
| D19S112 | Linked | di- | <u>132</u> /136 | 130/136 | 128/ <u>132</u> | Not Informative |
| D19S393 | Linked | tetra- | 289/297 | 290/298 | - | Not Informative |
| D21S1414 | Unlinked | tetra- | 344/348 | 344/360 | 344/369 | Not Informative |
| D21S11 | Unlinked | tetra- | 224/228 | 224/240 | 224/228 | Not Informative |
| D18S535 | Unlinked | tetra- | 466/478 | 482/486 | 478/490 | Informative |
| D13S305 | Unlinked | Tetra- | 452/452 | 446/456 | - | Not Informative |

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles would be classed as affected

Patient in *italics* indicates the affected partner

| Primer | Marker | Туре | Maternal allele sizes | Paternal allele sizes | Comments |
|----------|----------|--------|--------------------------|--------------------------|------------------|
| | | | Fm | Ff | |
| DM11 | | | 139/- | 122/139 | Semi-Informative |
| APOC2 | Linked | di- | 148/158 | 150/152 | Informative |
| D19S112 | Linked | di- | 129/129 | 127/127 | Not Informative |
| D19S207 | Linked | di- | 127/144 | 123/141 | Informative |
| D19S393 | Linked | tetra- | 289/293 | 293/301 | Not Informative |
| D21S1414 | Unlinked | tetra- | 349/363 | 345/351 | Informative |
| D21S11 | Unlinked | tetra- | 227/242 | 223/230 | Informative |
| D18S535 | Unlinked | tetra- | 466/486 | 489/489 | Not Informative |
| D13S305 | Unlinked | tetra- | 443/455 | 443/455 | Not Informative |

Table 3.6. List of polymorphic markers tested on genomic DNA for family F and their allele sizes

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles would be classed as affected Patient in *italics* indicates the affected partner

- indicates the affected allele that is refractory to PCR

| Table 3.7. | List | of | polymorphic | markers | tested | on | genomic | DNA | for | family | G | and |
|--------------|-------|----|-------------|---------|--------|----|---------|-----|-----|--------|---|-----|
| their allele | sizes | ; | | | | | | | | | | |

| Primer | Marker | Types | Maternal allele sizes | Paternal allele sizes | Affected son allele sizes | Comments |
|----------|----------|--------|-----------------------------|--------------------------|---------------------------------|------------------------------|
| | | | Gm | Gf | Gas | |
| DM11 | | | 142/- | 122/122 | 122/- | Informative |
| APOC2 | Linked | di- | 152/ <u>154</u> | 135/150 | 150/ <u>154</u> | Informative ^{&} |
| D19S207 | Linked | di- | <u>142</u> /142 | 141/141 | <u>142</u> /142 | Not informative |
| D19S112 | Linked | di- | 128/130 | 128/130 | 128/130 | Not informative |
| D21S1414 | Unlinked | tetra- | 346/355 | 335/350 | 350/354 | Informative |

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles would be classed as affected Patient in *italics* indicates the affected partner

[&]Practically difficult since close allele sizes would cause difficulties in interpretation due to stutter peaks

| Primer | Marker | Types | Maternal allele sizes | Paternal allele sizes | Mother's sister allele sizes | Comments |
|----------|----------|--------|-----------------------------|-----------------------------|------------------------------------|-----------------|
| | | | Hm | Hf | Hms | |
| DM1 | | | 146/ - | 122/143 | - /146 | Informative |
| APOC2 | Linked | di- | <u>155</u> /155 | 149/155 | 132/ <u>155</u> | Not informative |
| D19S207 | Linked | di- | 127/142 | 144/146 | 127/142 | Not informative |
| D19S112 | Linked | di- | <u>130</u> /132 | 117/117 | <u>130</u> /134 | Informative |
| D21S1414 | Unlinked | tetra- | 359/359 | 345/349 | 340/355 | Not informative |
| D21S11 | Unlinked | tetra- | 244/249 | 259/259 | - | Not Informative |
| D18S535 | Unlinked | tetra- | 478/482 | 481/481 | - | Not informative |
| D13S305 | Unlinked | tetra- | 440/448 | 438/448 | _ | Not informative |

Table 3.8. List of polymorphic markers tested on genomic DNA for family H and their allele sizes

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles would be classed as affected Patient in *italics* indicates the affected partner

- indicates the affected allele that is refractory to PCR

| Table 3.9. List of polymorphic marker | s tested on genomic Dl | NA for family I and their |
|---------------------------------------|------------------------|---------------------------|
| allele sizes | | |

| Primer | Marker | Туре | Maternal allele sizes | Paternal allele sizes | Mother's sister allele sizes | Comments |
|----------|----------|--------|-----------------------------|-----------------------------|------------------------------------|------------------|
| * | | | Im | If | Imf | |
| DM1 | | | 122/- | 122/145 | 122/- | Semi-informative |
| APOC2 | Linked | di- | 136/150 | 127/147 | 136/150 | Informative |
| D19S207 | Linked | di- | <u>144</u> /146 | 141/145 | 142/ <u>144</u> | Not informative |
| D19S393 | Linked | tetra- | 285/297 | 289/293 | - | Informative |
| D19S112 | Linked | di- | 117/128 | 117/134 | 128/132 | Not informative |
| D21S1414 | Unlinked | tetra- | 359/359 | 345/349 | 359/359 | Not informative |
| D21S11 | Unlinked | tetra- | 238/359 | 224/228 | 238/359 | Not informative |

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles would be classed as affected Patient in *italics* indicates the affected partner

| Primer | Marker | Туре | Paternal allele sizes | Maternal allele sizes | Affected fetus allele sizes | Comments |
|----------|----------|--------|-----------------------------|-----------------------------|-----------------------------------|-----------------|
| | | | Jf | Jm | Jaf | |
| DM1 | | | 141/- | 154/177 | -/178 | Informative |
| APOC2 | Linked | di- | <u>152</u> /154 | 126/143 | 143/ <u>152</u> | Informative |
| D19S207 | Linked | di- | 142/142 | 125/143 | 125/143 | Not informative |
| D19S112 | Linked | di- | 128/130 | 126/128 | 128/130 | Not informative |
| D19S393 | Linked | tetra- | 293/293 | 289/293 | - | Not informative |
| D21S1414 | Unlinked | tetra- | 335/352 | 339/347 | 335/339 | Informative |
| D21S11 | Unlinked | tetra- | 220/227 | 216/232 | 216/227 | Informative |
| D18S535 | Unlinked | tetra- | 481/481 | 490/490 | 481/490 | Not informative |

Table 3.10. List of polymorphic markers tested on genomic DNA for family J and their allele sizes

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles would be classed as affected Patient in *italics* indicates the affected partner

[&]Practically difficult since close allele sizes would cause difficulties in interpretation due to stutter peaks

- indicates the affected allele that is refractory to PCR

3.2.2 PGD Workup

Workup for a total of ten families was performed during this study (Tables 3.1-3.10). PGD was carried out for two families only (families G and H). For family A apart for the DM1 marker, no other marker was found to be informative hence further investigations of different markers should be carried out. For families B, E, F and I the DM1 mutation marker was found to be 'Semi-informative' and at least two other markers, linked or unlinked, were informative for each couple. The DM1 marker was termed 'Semi-informative' when the affected partner's allele that could be sized was the same as one of the alleles from the other e.g. for family B, for 'Bm' it was (120/-), which was the same for 'Bf' (120/138). The DM marker could be semi-informative in the case where the embryo would show the 120/138 genotype, meaning that the 120 was inherited from the mother and the 138 from the father. However, it is always necessary to perform a multiplex protocol for these families with a linked marker since the 120/138 genotype can also arise from paternal contamination. Families C and J were awaiting PGD to be performed due to personal reasons, since the protocol for both families was ready. For family D, the DM and the D18S535 markers were informative, which required optimisation at the single cell DNA level.

Two families opted to come through for a PGD cycle (families G and H). Two different multiplex F-PCR protocols for the detection of DM1 affected embryos were devised for this study for the two families. In addition to amplifying the triplet repeat region within the DMPK gene, the PGD protocols for DM1 were designed to incorporate either an unlinked polymorphic marker on chromosome 21 (Protocol 1) or a linked polymorphic marker flanking the DM1 gene (Protocol 2). The CTG repeat region of DM1 is also very polymorphic; therefore it was usually possible to distinguish both alleles of normal heterozygote subjects on an automated sequencer.

For both protocols the markers could be simultaneously analysed on the 3100ABI PrismTM using GeneScan[®] analysis software. The peak area of the F-PCR analysis was indicative of the amount of amplified product, which was similar to the intensity of a band when analysing a sample after gel electrophoresis. However, F-PCR allows for accurate sizing of fragments and precise quantification of the amount of the PCR product. In some instances, the DNA samples revealed large differences between their allele sizes (for all three markers) e.g. for Ef (father of family E) the DM1 allele sizes were 122/149 indicating a 27bp difference. In those cases, there was significant preferential amplification of the smaller allele and also increased incidence ADO of the larger allele during single cell amplification. However, this was partly overcome by decreasing the temperature by 1-2°C of the annealing temperature. The incidence of stutter peaks was also encountered, especially at the single cell level (Figure 3.1). Stutter peaks were mostly observed when dinucleotide repeat regions where amplified and did not cause diagnostic error since they were always one repeat unit smaller than the true allele and relatively under-amplified. However, they were taken into account in individuals where their allele sizes differed in size by just one repeat and were regarded as practically difficult.



Figure 3.1. Illustration of preferential amplification, stutter peaks and the +A artefact on a dinucleotide repeat region

The GeneScan® analysis software shows the amplification fragment for each marker in different colours (here the APOC2 is in blue). Below the blue graph there is a table which shows the allele size (heading 'Size') and the quantification of each fragment (heading 'Peak Height'). Samples 4B and 5B represent single cells from the father and mother of family E respectively. Sample 6B is the negative control. In both parents the smaller allele shows preferential amplification over the larger allele. For example in sample 4B the smaller allele (126) has a peak height of 1344, whereas the larger allele (153) has a peak height of 282.

The red circle represents the stutter peak that is caused due to polymerase slippage during the PCR extension step. The **black square** represents the +A effect

Furthermore, the Plus-A (+A) artefact was also encountered during the optimisation process of single cell F-PCR. This was due to the addition of a nucleotide, usually adenosine, from the DNA polymerase enzyme to the 3'ends of the amplified fragments, which lead to PCR products one bp longer than the expected fragment size (Figure 3.1). The +A effect would not cause misdiagnosis, however, it may confuse the peak area calculation. However, after subsequent optimisation experiments, this problem was reduced by omitting the final 10min extension step typically used in PCR protocols.
3.2.2.1 Analysis of Control Single Cells

Clumps of buccal cells (2-4 cells) were isolated from the author (section 2.2.2.2) to test whether the two multiplex protocols were working with a lower starting amount of DNA. Protocol 1 was tested on 30 clumps and showed 98% amplification rate for both loci with 3% and 5% ADO rate for the DM1 and D21S1414 primers respectively (Table 3.11). Protocol 2 was tested on 30 clumps revealing 3% amplification failure and ADO rates of 4% and 5% for the DM1 and D19S112 primers respectively. Single cell analysis was performed initially on 50 single buccal cells from normal heterozygote subjects. Amplification of the DM1 and D21S1414 primers (protocol 1) showed 96% and 93% amplification efficiency respectively and 9.1% and 10.2% ADO rate for the DM1 and D21S1414 respectively in heterozygote individuals. Protocol 2 (DM1 and D19S112) revealed similar amplification efficiency of 95% for the DM primers and 91% for the D19S112 primers. The ADO rates were 8.8% for the DM 12.9% and for the D19S112 primers in heterozygote individuals.

| Table 3.11. Results | of amplification | efficiency | and ADO | in control | clumps | and single |
|---------------------|------------------|------------|---------|------------|--------|------------|
| cells | | | | | | |

| Type of | Rates | Pro | otocol 1 | Pro | tocol 2 |
|-----------------|-----------------------------|-------------|----------|------|---------|
| cells | | DM 1 | D21S1414 | DM | D19S112 |
| Clumps | Amplification efficiency | 98% | 98% | 98% | 97% |
| (n=30) | Allele Dropout | 3% | 5% | 4% | 5% |
| Single | Amplification efficiency | 96% | 93% | 95% | 91% |
| cells (n=50) | Allele Dropout | 9.1% | 10.2% | 8.8% | 12.9% |

3.2.2.2 Analysis of Patients Single Cells

The single buccal cells used in optimising and testing the PGD protocol were derived from members of both families, including 80 cells from each father ('Gf' and 'Hf') and 80 cells from each mother ('Gm' and 'Hm') and 50 cells from 'Hs' (mother's sister; Table 3.7). The data from all 160 single buccal cells of the members of family G displayed an amplification rate of 90.9% and 85.7% and an ADO rate of 14.1% for the D21S1414 primer (Table 3.12).

| Results | 'Gf' | 'Gm' | Total |
|-------------------------|--------|--------|---------|
| | (n=80) | (n=80) | (n=160) |
| | | | |
| DM1 | | | |
| 4 Amplification rate | 89.8% | 92% | 90.9% |
| Amplification failure | 10.2% | 8% | 9.1% |
| ADO | | | - |
| 4 Contamination | 1.2% | - | 0.6% |
| | | | |
| | | | |
| D21S1414 | | | |
| Amplification rate | 84.4% | 87% | 85.7% |
| 4 Amplification failure | 15.6% | 13% | 14.3% |
| 🔺 ADO | 15% | 13.2% | 14.1% |
| 4 Contamination | | 1.2.2 | - |
| | | | |

Table 3.12. DM1 and D21S1414 amplification results on single buccal cells from family G (Protocol 1)

'Gf' = father of G family and 'Gm' = mother of G family

The total amplification rate from 210 single buccal cells for family H was 91.3% and 83.7% for the DM1 and D19S112 respectively, whilst the ADO rate was 7.9% and 15.6% for the DM1 (from 80 cells for Hf) and D19S112 primers respectively (Table 3.13 The ADO rates for 'Hms' single buccal cells were markedly higher compared to those of 'Hm' for the D19S112 primer. The underlying reason might be the prolonged time from the sample collection until the time of single cell isolation, since the cells were sent at room temperature during transit, hence, many cells might have been dead or degenerating.

| Results | 'Hf' (n=80) | 'Hm' (n=80) | 'Hms' (n=50) | Total (n=210) |
|---------------------------|----------------|----------------|-----------------|------------------|
| DM1 Amplification rate | 92% | 92% | 90% | 91.3% |
| 4 Amplification failure | 8.1% | 8% | 10% | 8.7% |
| 🕹 ADO | 7.9% | - | | 7.9% |
| 4 Contamination | 2.5% | - | 4% | 2.9% |
| D19S112 | | | | |
| Amplification rate | 85.7% | 87% | 78.3% | 83.7% |
| 4 Amplification failure | 14.3% | 13% | 21.7% | 16.3% |
| 🔸 ADO | _ Call * | 12.8% | 23.2% | 15.6% |
| Contamination | | 5% | 4% | 3.8% |
| | | | | |

Table 3.13. DM1 and D19S112 amplification results on single buccal cells from family H (Protocol 2)

'Hf'= father of H family, 'Hm' = mother of H family and 'Hms' = mother's sister of H family

A negative control was included in every ten single buccal cells from the last wash drop for both protocols. No contamination was observed in any of the final wash drop blanks, lysis-buffer only negative controls or PCR mixture only negative controls. During blastomere analysis a wash drop blank was used for every single blastomere. All personnel involved in the diagnosis were genotyped to allow detection of contamination. No contamination was found by one or more of the personnel.

3.2.3 Clinical PGD Results

The patients underwent routine IVF procedures and ICSI was employed for both families to avoid sperm contamination. The embryos were biopsied on day 3 and the single blastomeres were prepared for the diagnosis (section 2.2.2.3).

3.2.3.1 Family G

During the PGD cycle for family 'G', 7 oocytes were collected from which 5 underwent ICSI. Five embryos were of sufficient quality for biopsy on day 3 post-fertilisation; however overall embryonic morphology was not good since from all embryos only one cell was biopsied (Table 3.14). Molecular analysis revealed 2 embryos to be affected, 2 with incomplete results (one of which was suspected to be abnormal) and one with no results for either marker. No contamination was detected and no allele dropout.

| Embryo No. | No. of cells (time of biopsy) | Grade | Cell No. | Notes | DM1 | D21S1414 | Result |
|---------------|-------------------------------------|-------|-------------|----------------------------|-------|---------------|------------|
| G1 | 5 | 2 | а | 800 - 100 (100 AND 100 | 122/- | 335/346 | Affected |
| G2 | 5-6 | 2+ | a | Cell lysing | AF | 335/355 | Incomplete |
| G3 | 5 | 2+ | a | Degenerating embryo | 122/- | Inconsistent* | Affected |
| G4 | 5 | 2- | а | seen | AF | AF | No result |
| G5 | 6 | 2 | а | - 11 | 122/- | 346/350 | Affected |
| +ve ctrl | | - | - | - | 142/- | 346/355 | - |
| +ve ctrl | | - | | | 122/- | 335/350 | |

Table 3.14. PGD results of family 'G'

DM=myotonic dystrophy mutation marker; D21S1414=contamination STR marker; AF=amplification failure

*Allele sizes that did not correspond to any of the parents nor the people involved in the diagnosis

Full information was achieved for embryo G1a and embryo G5a (Figure 3.2). In two instances (G2a and G4a) the DM1 primer failed to give any kind of results. However, cell G2a was lysing during biopsy and some material might have been lost during transfer since the contamination marker was able to give a result. Furthermore, in cell G4a no nucleus could be seen during biopsy and washing procedures.

| Embryo No. | Cell No. | DM | D21S1414 | Result |
|---------------|-------------|--------|----------|-----------|
| Gs1 | А | 122/ - | 346/351 | Affected |
| | В | AF | AF | No Result |
| Gs2 | А | 122/ - | 335/346 | Affected |
| | А | 122/ - | AF | Affected |
| Gs4 | b | 122/ - | 351/355 | Affected |
| | с | AF | AF | No Result |
| | а | 122/ - | AF | Affected |
| | b | 122/ - | /351 ADO | Affected |
| Gs5 | с | 122/ - | 346/351 | Affected |
| | d | 122/ - | 346/351 | Affected |
| | e | 122/ - | /351 ADO | Affected |

Table 3.15. Confirmatory results from spare embryos after PGD for family 'G'

DM=myotonic dystrophy mutation marker; D21S1414=contamination STR marker; AF=amplification failure; ADO=allele dropout

Overall, from 16 blastomeres the DM1 locus revealed a 75% amplification rate whereas the D21S1414 showed a 68.75% amplification rate and 12.5% allele dropout.

3.2.3.2 Family H

During the PGD cycle for family 'H', 9 oocytes were collected from which 8 underwent ICSI. Five embryos were of sufficient quality for biopsy on day 3 post-fertilisation and from all embryos two cells were biopsied (Table 3.16). One embryo (embryo H1) was normal and was chosen for embryo transfer. Two embryos were considered to be affected and two revealed inconclusive results.

| Embryo No. | No. of cells (time of biopsy) | Grade | Cell No. | Notes | DM1 | D19S112 | Result |
|---------------|-------------------------------------|-------|-------------|-----------------|----------------------|-----------------|--------------------------|
| H1 | 8 | 1- | 1a | | 122/146 | 117/132 | Normal |
| | | | 1b | - | -* ¹ /146 | 117/ ADO | TTOTTIA |
| H2 | 7 | 2 | 2a | - No | AF | 117/130 | Affected (incomplete) |
| | | | 2b | nucleus seen | AF | AF | |
| H3 | 6 | 2+ | 3a | - | 122/-*2 | ADO /132 | Inconclusive |
| | | | 3b | | 122/-*2 | AF | |
| H4 | 8 | 1- | 4 a | - | 122/- | 117/130 | Affected |
| | | | 4b | - | 122/- | 117/ ADO | |
| Н5 | 8 | 1- | 5a | Cell lysing | AF | AF | Affected |
| tua atel | | | 5b | - | 143/- | 117/130 | |
| 4 | - | - | - | - | 146/- | <u>130</u> /132 | - |
| +ve ctrl | - | - | - | - | 122/143 | 117/- (hm) | |

Table 3.16. PGD results of family 'H'

*¹Extreme cases of preferential amplification of the large allele (v.v low amplification of the 122 allele), which at the time of diagnosis were considered as allele dropout to avoid misdiagnosis.

 $*^{2}$ Extreme cases of preferential amplification of the small allele (v.v low amplification of the 146 allele), which at the time of diagnosis were considered as allele dropout to avoid misdiagnosis

Underlined numbers indicate the phase i.e. those alleles are linked to the mutation hence if the embryo carries these alleles it would be classed as affected

DM=myotonic dystrophy mutation marker; D19S112=linked STR marker; AF=amplification failure; ADO=allele dropout; mh=homozygous

In embryo H1b, there was extreme preferential amplification of the larger allele (122) for the DM1 marker (Figure 3.3). However, since the larger allele was the mother's unaffected allele, the whole embryo was considered normal (Figure 3.3). Embryo H2 was thought to be affected since only the linked marker (D19S112) amplified in one of two cells (117/130) containing the phase allele (130). In embryo H3 the linked marker showed inconclusive results and the mutation marker (DM) showed extreme

preferential amplification of the 146 allele inherited from the mother but it was probably normal (Table 3.16). However, to reduce the chance of misdiagnosis it was considered as affected at the time of diagnosis.



Figure 3.3. Analysis results for embryo H1 (blastomeres H1a and H1b) and the parents Hm and Hf

The DM1 (Blue) and D19S112 (Black) genotypes of blastomeres H1a and H1b from embryos H1 can be seen during the clinical case. Furthermore, the prospective couples' DM1 and D19S112 genotypes (Hm=mother and Hf=father) can be seen which were used as positive controls. Cell H1a showed conclusive results indicating a normal cell. Cell H1b revealed extreme preferential amplification of the smaller allele of the DM1 genotype (122) and ADO of the larger allele for the D19S112 marker.

However, since the mother's normal allele was inherited, it was concluded that the embryo was normal.

The rest of the embryos were donated for confirmation diagnosis. The results from the spare embryos revealed that embryo H3 (Hs3 in Table 3.17) was actually normal and could have been transferred (Table 3.17). Embryos H2 and H5 (Hs2 and Hs5 in Table 3.17) were confirmed as affected. In embryo H4 (Hs4 in Table 3.17) no cells could be biopsied since it was degenerating and when the whole embryo was collected, maternal contamination was detected (Table 3.17). Embryos Hs6, Hs7 and Hs8 were not suitable for biopsy during the diagnosis. Embryo Hs6 was degenerating and no cells could be biopsied, thus the whole embryo was lysed and amplified, showing multiple allele sizes probably due to mosaicism or maternal contamination. For embryo Hs8 only one cell could be biopsied and amplified which was normal.

| | | | | ····· |
|---------------|--------------|---------|---------------------------------------|-----------------------|
| Embryo No. | Cell No. | DM1 | D19S112 | Result |
| | a | 143/- | 117/130 | |
| Hs2 | b | 143/- | 117/130 | Affected |
| | с | 143/- | 117/130 | |
| | a | 122/146 | 117/132 | |
| Hs3 | b | 122/146 | 117/132 | Normal |
| | с | 122/146 | 117/132 | |
| Hs4 | Whole embryo | 122/146 | 117/130/132 | Multiple allele sizes |
| | a | 122/- | 117/130 | |
| Hs5 | b | AF | 117/ ADO | Affected |
| | с | 122/- | 117/130 | |
| Hs6 | Whole embryo | 122/146 | 117/130/132 | Multiple allele sizes |
| H8 | a | 122/146 | 117/132 | Normal |
| | | | · · · · · · · · · · · · · · · · · · · | |

Table 3.17. Confirmatory results from spare embryos after PGD for family 'H'

DM=myotonic dystrophy mutation marker; D19S112=linked STR marker; AF=amplification failure; ADO=allele dropout

Overall, from 20 blastomeres the DM1 locus revealed an 80% amplification rate and no ADO, 15% of cases showed extreme preferential amplification. The D19S112 showed an 85% amplification rate and a 15% allele dropout.

3.2.4 DOP-PCR on Genomic DNA

Initially, DOP-PCR was carried out on genomic DNA which was subsequently used as template for singleplex reactions of each primer from the mother and father of family E as well as individuals X and Z (Table 2.8). However, due to lack of single cell material from family E, single cell results were only obtained from individuals X and Z. All the F-PCR primers where successfully amplified when genomic DNA (100-300ng approximately) was used as template. The amplification rate was 98% for all primers except for the D13S305 which was 96%. There was no incidence of allele dropout, which was expected due to the nature of the starting template. The similarity of amplifications when a singleplex reaction was carried out from genomic DNA compared to a singleplex reaction when it was carried out using a DOP-PCR product as template is depicted in Figures 3.4, 3.5, 3.6 and 3.7. The peaks in all lanes (except the negative control lanes) are high indicating sufficient PCR product, with no ADO and little or no stutter effects. However, more importantly the genotypes for both primers can be easily interpreted in the amplifications which used the DOP-PCR product as starting template (Figures 3.4B, 3.5B, 3.6B and 3.7B).

| A | 90 | 120 | 150 | 180 |
|------------------------|--------------------|-----|----------------------------------|--------------------|
| 480 240 0 | 122 | In | 149 | 1 |
| | Father of family E | | | |
| 460 240 0 | | 122 | 10 | 2 |
| | Mother of family E | | | |
| 480 240 0 | | | | 3 |
| | 3G :-VE DM / | | | |
| | 90 | 120 | 150 | |
| 840 760 880 0 | | 122 | 149 | 4 |
| | Father of family E | | | The Art and |
| 640 760 880 0 | | 122 | ~ | 5 |
| | Mother of family E | | | |
| 640 760 880 | | | | 6 |
| | SG :-VE - DM / DM | | Contraction of the second second | Contraction of the |

Figure 3.4. Analysis results for the DM1 primer for family E using genomic DNA

- A. Results for the DM1 mutation marker when amplified directly from genomic DNA. Lanes 1 and 2 show the genotypes of the father and the mother of family E respectively (Table 3.5) and lane 3 is the negative control.
- B. Results for the DM1 mutation marker when amplified from a DOP-PCR product. Lanes 4 and 5 show the genotypes of the father and the mother of family E respectively and lane 6 is the negative control.

Each peak represents the alleles of the individual. In both sets of results the peak were easily distinguished with no +A artefacts and little or no stutter effects. However, there was a slight difference appearance of 'B' compared to 'A', since the latter showed more distinct and sharper peaks.

| A | 320 | 360 | 406 |
|-------------------|-------------------------|------------------------------|-----------------|
| 3600 1800 0 | 34 | 14 1360 | 1 |
| | Father of family E | 2151414 | |
| 3600) 1800 | 34 | 44 348 | 2 |
| | Mother of family E | 2151414 | |
| 3600 1800 | Michael & Banky b | | 3 |
| | 228 :-VE - D2151414 / 1 | 021\$1414 | |
| B | 300 | 330 360 | 390 |
| 1480 740 0 | | 344 360 | 4 |
| | Father of family E | and the second second second | ter freshiet |
| 1480 740 0 | Packer of Density P | 344 348 | 5 |
| | Mother of family E | | CONSTRUCTION OF |
| 1480 740 0 | Moher ef Karse F | | 6 |
| | 98 :-VE D2151414 / | and the second second | |

Figure 3.5. Analysis results for the D21S1414 primer for family E using gen. DNA

- A. Results for the D21S1414 mutation marker when amplified directly from genomic DNA. Lanes 1 and 2 show the genotypes of the father and the mother of family E respectively (Table 3.5) and lane 3 is the negative control.
- B. Results for the D21S1414 mutation marker when amplified from a DOP-PCR product. Lanes 4 and 5 show the genotypes of the father and the mother of family E respectively and lane 6 is the negative control.

Each peak represents the alleles of the individual. In both sets of results the peak was easily distinguished with no +A artefacts and no stutter effects. However, there was a slight difference in appearance of 'B' compared to 'A', since the latter showed increased peak heights of more than 3600 in comparison to 'B' which showed peak heights of around 1500. Such results indicated that amplification products in 'B' were half the amount of 'A'.

| A 42 | 450 480 | 510 |
|------------------------------|--------------------|-----|
| 5760] 3840] 1920] 0 | 482 486 | 1 |
| | Father of family E | |
| 5760] 3840] 1920] | 466 478 | 2 |
| | Mother of family E | |
| 5760 3840 1920 | | 3 |
| B | | |
| 2320) 1740 1180 580 | 482 486 | 4 |
| | Father of family E | |
| 2920 1740 1160 580 | 466 478 | 5 |
| | Mother of family E | |
| 2320 1740 1160 580 | | 6 |
| | | |

Figure 3.6 Analysis results for the D18S535 primer for family E using genomic DNA

- A. Results for the D18S535 unlinked marker when amplified directly from genomic DNA. Lanes 1 and 2 show the genotypes of the father and the mother of family E respectively (Table 3.5) and lane 3 is the negative control.
- B. Results for the D18S535 unlinked marker when amplified from a DOP-PCR product. Lanes 4 and 5 show the genotypes of the father and the mother of family E respectively and lane 6 is the negative control.

Each peak represents the alleles of the individual. In both sets of results the peak were easily distinguished with no +A artefacts and little or no stutter effects. However, in 'B' the peak heights were nearly half the size compared to 'A'. Such results indicated that amplification products in 'B' were half the amount of 'A'

| A 390 | 420 | 450 | 480 |
|-------------------|--------------------------|---|-------------------------------|
| 2360 1180 0 | | 446 456 | many sea for the ¹ |
| | Father of family E | | stream all the market |
| H120 | COL State of planet. Our | 452 | 2 |
| | Mother of family F | ~ | |
| 360 180 | | an elimente contreren le 100 geomie: DiVA rang | 3 |
| | 186:06/ | a Acquiring products of | that is appendix the |
| 390 | 420 | 450 | 480 |
| 200 | DOB-PCR employeeteer | 446 456 | 4 |
| | Father of family E | | A THE AVER AND |
| 2400 | | 452 | 5 |
| | Mother of family E | | |
| 2360 | | | 6 |
| | | | C. Marker Street |

Figure 3.7 Analysis results for the D13S305 primer for family E using genomic DNA

- A. Results for the D13S305 unlinked marker when amplified directly from genomic DNA. Lanes 1 and 2 show the genotypes of the father and the mother of family E respectively (Table 3.5) and lane 3 is the negative control.
- B. Results for the D13S305 unlinked marker when amplified from a DOP-PCR product. Lanes 4 and 5 show the genotypes of the father and the mother of family E respectively and lane 6 is the negative control.

Each peak represents the alleles of the individual. In both sets of results the peak were easily distinguished with no +A artefacts and little or no stutter effects. There as however, slightly increased stutter band effect in 'B' which could be seen by comparing lanes 1 and 4.

3.2.5 DOP-PCR on Single Cells

The encouraging results on genomic DNA prompted the amplification on single cells. The two individuals that were chosen for their heterozygote genotypes for the DM1 mutation marker in order to detect ADO rates in the most important marker were X and Z. All ten single cells from each individual (100%) were successfully amplified with DOP-PCR. Each single cell (including the positive control) were analysed with a 1% agarose gel (Figures 3.8 and 3.9). The single cells from both individuals were fresh and were amplified the same day of collection and isolation to avoid freeze-thaw of samples. The amplification of Z was enhanced compared to X and that can be seen if the results are evaluated against the genomic DNA amplification. Interestingly minor bands were seen only in Z's amplified products which usually represent mitochondrial DNA.



Figure 3.8. DOP-PCR amplification results run on 1% agarose gel of individual X

The first lane displays the 1Kb ladder. Lines X1-X10 depict each single cell whereas XCL1 and XCL2 represent the positive controls from clumps and X+ve DOP shows the positive control from genomic DNA. All single cells showed a smear of results with no distinct bands except for C4 where the smear was faint and reduced in size. The smear from the genomic DNA was more intense and increased in size which suggested better amplification of the genome.



Figure 3.9. DOP-PCR amplification results run on 1% agarose gel of individual Z

The first lane displays the ladder. Lines Z1-Z10 depict each single cell whereas ZCL1 and ZCL2 represent the positive controls from clumps and Z+ve DOP shows the positive control from genomic DNA. All single cells showed a smear of results with no distinct bands except for Z10 where the smear was reduced in size, however it was very intense.

From the 20 single cells (10 from each individual) the amplification rate was 85%, 55%, 65%, 60% and 65% for DM, D19S112, D21S1414, D18S535 and D13S305 respectively (Table 3.18).

| Marker | Individual Z | | Individual X | | | Total | | |
|----------|--------------|-----|--------------|--------|-----|-------|-----|-----|
| | (n=10) | | | (n=10) | | | | |
| | AR | ADO | IG | AR | ADO | IG | AR | ADO |
| DM1 | 90% | 0% | 0% | 80% | 20% | 0% | 85% | 10% |
| D19S112 | 70% | 50% | 10% | 40% | 30% | 20% | 55% | 40% |
| D21S1414 | 50% | 20% | 10% | 80% | 40% | 10% | 65% | 30% |
| D18S535 | 60% | 60% | 10% | 60% | | 10% | 60% | 60% |
| D13S305 | 70% | | 10% | 60% | 40% | 0% | 65% | 40% |
| | | | | | | | | |

AR = amplification rate, ADO = allele dropout, IG = incorrect genotype.

Overall, there was a high incidence of ADO for all STR's ranging from 10-60%. The highest allele dropout rates were observed for the D18S535 primers, showing 60% ADO (Table 3.18 and 3.9), whilst the DM1 locus displayed the lowest rate of allele

dropout. This was probably due to the size of the products that the D18S535 primers amplify, which is >430bp, whereas the DM1 are considerably lower (Table 3.2 From Materials and Methods). Moreover, there was occurrence of incorrect genotype in all the markers except for the DM1 (Table 3.18, termed IG). It was rarely observed and the differences in allele sizes were of 1-2 base pairs. The D18S535 also revealed high ADO rates (60%), however, it showed that ADO affected equally the large and the small allele (Figure 3.10).



Figure 3.10. Illustration of allele dropout of the D18S535 locus

A. Allele dropout of the large allele (482) in a single cell. **B.** DOP-PCR positive control showed amplification of both alleles. **C.** F-PCR positive control showed amplification of both alleles

The D19S112 and D21S1414 showed increased stutter effect as well as random peaks outside the region of the amplified products (Figure 3.11 and 3.12). Furthermore, the DM marker showed high +A artefact incidence compared to when they were amplified as singleplex reactions from single cell DNA (Figure 3.13).



Figure 3.11. Illustration of allele dropout of the D19S112 locus

A. Allele dropout of the large allele (134) in a single cell 1. **B.** Allele dropout of the small allele (117) in a single cell 2. **C.** F-PCR positive control showed amplification of both alleles



Figure 3.12. Increased occurrence of 'stutter effect' for the D21S1414 locus

All three single cells showed increased stutter effect of both alleles. However, in single cell 2 there was increased stutter effect. Such results were only observed on single cells and never at genomic DNA level.



Figure 3.13. Illustration of the +A artefact affecting the DM1 locus

All five single buccal cells reveal cases of significant +A effect. This phenomenon was especially apparent in single cells 3 and 5 whereby the peak on the right (circled in red), which is the +A artefact, was almost a distinct peak and could be difficult to interpret which was the true peak representing the allele. In normal single cell F-PCR the +A effect was solved by eliminating the final 10min extension step.

3.2.6 CGH Results on DOP-PCR Products

Each single cell amplified using DOP-PCR yielded 50μ l of product. Around 10μ l were used for the F-PCR reactions. The rest of the product was subjected to CGH analysis. All twenty single cells were able to hybridise into target metaphase chromosomes; however, 17/20 (85%) produced analysable results (Figure 3.14). All

analysable cells showed normal chromosome results and the sex was confirmed in all samples.

Figure 3.14. CGH results from DOP-PCR amplified single buccal cell of individual X against control male DNA.



For this CGH experiment 11 metaphases were captured, karyotyped, analysed individually and then interpreted cumulatively. 1. Capturing of metaphase spread. 2. Karyotype of the captured metaphase 3. This shows the cumulative analysis of eleven metaphases, which was the basis of the interpretation. There was a shift in fluorescence only towards the green in chromosome X and towards the red in chromosome Y showing that the test sample was a normal female. Overall, there was dynamic hybridisation with little background fluorescence demonstrating that the DOP-PCR amplified single cell can be used for PGD of DM as well as further chromosomal analysis.

3.3 Discussion

This study aimed to develop two multiplex PGD F-PCR protocols for DM1 and furthermore to determine if WGA could be used for PGD of DM. The PGD protocols were developed by exploring informative STR markers for the parents, multiplexing them on genomic DNA and finally testing and perfecting the procedure on single cells. Two multiplex F-PCR protocols were devised and carried out for clinical PGD cases of two families (G and H). In the first protocol a contamination marker was used (similar to that used by Piyamongkol et al, 2004). No embryo was transferred. The second protocol entailed a linked marker never used before, which enabled both ADO and contamination to be tackled. However, this laborious and patient-specific process prompted the development of a universal protocol for PGD of DM. Whole genome amplification was employed using DOP-PCR, initially on genomic DNA and subsequently on single cell DNA. For each DOP-PCR product a 1µl aliquot was taken in order to carry out a separate singleplex F-PCR procedure able to amplify the corresponding locus. Thus, five different polymorphic markers were tested and their single cell efficiency and reproducibility was determined. Finally, CGH was performed utilising the DOP-PCR product to assess whether chromosomal analysis was also feasible along with single gene analysis of DM.

3.3.1 Strategy for PGD

In total 10 families were referred to our centre for PGD for DM. For all families at least six polymorphic markers plus the DM1 mutation marker were genotyped for each parent. This was carried out in order to find an informative linked marker (STR) for the mutation as well as an unlinked marker (STR) for detection of possible contamination. All this work was carried out using genomic DNA extracted from whole blood from each parent and some relatives. All primers used were fluorescently labelled and analysed in an automated sequencer (ABI PrismTM 3100). Once a strategy was devised, the protocol was tested and optimised on single buccal cells and single blastomeres.

Initially accurate molecular analysis of the DM1 repeat expansion was (and is still) performed by Southern Blotting (Brook *et al*, 1992). However, this method requires a high copy number of the DNA template and thus cannot be applied to PGD. The use

of PCR can provide a fast and efficient means for the molecular diagnosis of DM1. However, the interpretation is based on the exclusion principle as the amplification of the fragment size larger than 500 bases is inefficient, hence only normal alleles can be examined. The first report of PGD for DM DM1 was from Sermon and colleagues (1997) who performed traditional PCR and analysis by gel electrophoresis. The DM DM1 mutation marker is polymorphic and provides the advantage of identifying the presence of both normal alleles in the normal sample (unless homozygous) and in some cases contamination. However, in the case of an affected cell or abnormal cell with ADO where only one allele is present, the occurrence of contamination may be concealed and may lead to misdiagnosis. The first PGD case using F-PCR was also performed by the same group (Sermon et al, 1998a), however, a misdiagnosis occurred which was attributed to maternal DNA contamination (Vandervors et al, 2000). This provoked Piyamongkol et al (2001a) to develop a multiplex F-PCR protocol able to tackle contamination, which is one of the most significant an be a problems affecting single cell PCR. Hence, the DM DM1 marker was amplified along with another polymorphic marker. An addition of a fluorescent polymorphic marker, such as the D21S1414, in the F-PCR protocol would eliminate the need for nested PCR since the F-PCR technique is sensitive without the need of carrying out two2 sets of PCR amplifications and the addition of a contamination marker wouldand would provide information on whether a sample is contaminated. The D21S1414 locus is unlinked to the DMPK gene, thus providing no information on inheritance of the DM DM1 mutation, although it is able to serves as a very basic form of DNA fingerprint. The compound heterozygoscity of the D21S1414 locus is 0.88 (Sherlock et al, 1998). During this study, a similar protocol to Piyamongkol et al (2001a) was employed (Protocol 1) for family G. For family H a linked marker was used, namely D19S112, in addition to the DM DM1 marker (Protocol 2). The polymorphic characteristic of the D19S112 locus offers solution to both contamination and ADO problems encountered during single cell PCR. In family H the father was homozygous (117) and the mother was heterozygous (130/132) for the D19S112 locus. Hence, the parents had different allele sizes between them and were quite far apart rendering the marker informative. The mother's sister gave a samples of blood sample so that and the phase was distinguished. The phase is the allele of the linked locus (here the D19S112) which is transmitted with the disease. If that allele (in this case the phase is the 130 allele) is present in an embryo it will be considered to be affected. In total, the

linkage information of the D19S112 marker can predict the presence of the mutant DM1 expanded allele in the embryo, which cannot be obtained by direct PCR analysis of the DM DM1 gene (due to the over-expanded allele). This allows differentiation between the affected embryo and normal embryo whose results are complicated by ADO. Therefore, complete diagnosis can be drawn from more normal embryos, giving rise to a large number of embryos for transfer and a better chance of pregnancy. Moreover, using the DM1/D19S112 multiplex F-PCR protocol reduces the case of misdiagnosis since the D19S112 marker acts aslike a safety net for the DM DM1 mutation marker (Ao *et al*, 1998).

The analysis was performed using the automated laser sequencers ABI PrismTM 3100 and ABI PrismTM 310. The use of F-PCR in comparison to traditional analyses provides protocols with further sensitivity since even tiny amounts of amplified fluorescent product can be detected as a small peak. The accuracy of the sequencers even allows differentiation of single base pair differences making the protocol more specific than those using traditional gel analysis (Sermon *et al*, 1998a). In the context of PGD, the most striking advantage of using the automated laser sequencers is time. Both the ABI PrismTM 310 and the ABI PrismTM 3100 can analyse up to three genes immediately and simultaneously after one multiplex PCR. However, the added advantage of the ABI PrismTM 3100 is that it can analyse 4 samples per hour (in comparison to one sample per hour for the ABI PrismTM 310), reducing the time of the analysis. During the PGD case the ABI PrismTM 310 was used as a backup analysis for all the samples.

During optimisation of the PCR protocols the phenomenon of stutter bands and the +A artefact was encountered (Figure 5.1). Stutter bands usually cause problems during allele scoring of heterozygote individuals whose alleles are close in size especially whilst amplifying mono- and dinucleotide repeats. This could be observed in many of the families when the allele sizes were different for the parents e.g. for family C the APOC2 marker showed 149/155 for the mother and 134/153 for the father, however, some of the alleles are only one two base pair apart (155 from the mother and 153 from the father) (Table 5.63). Stutter bands are usually one repeat unit length shorter than the main allele (Ellegren, 2004), hence during analysis of the APOC2 locus for family C the stutter band for the 155 allele would coincide with the

153 allele. The solution to stutter bands was to choose in our multiplex protocols STR's which displayed alleles between the parents of more than one base pair apart.

The +A artefact was caused by the addition of an extra nucleotide, usually adenosine, to the 3' ends of the amplified fragment during the amplification reaction. The +A products are mostly depicted as an extra peak or as a split peak one base pair longer than the expected PCR products. Such a problem would confuse the interpretation of the results but would not cause misdiagnosis. The split peak pattern can be minimised by either encouraging or suppressing the nucleotide addition. During the whole study it was decided to minimise the +A artefact by omitting the final extension step.

Both protocols showed high amplification efficiencies in clumps of 98% for both DM DM1 and D21S1414 and 97% for D19S112 (Table 5.1411). The ADO rates in clumps were also below 5% for both protocols. In the control single cells the ADO rates increased as expected to 8.8%-9.1% for the DM1, 10.2% for the D21S1414 and 12.9% for the D19S112 respectively. These results are within range of previously reported rates in studies of different genes (Ray and Handyside, 1996; Piyamongkol *et al*, 2001a). The optimised protocols 1 and 2 were tested in single buccal cells from the members of both families (Tables 5.15 12 and 5.1613). The amplification results slightly decreased which was expected due to the fact that the samples were not fresh and some were frozen. This was exaggerated in the case of 'Hms' (mother's sister of H family) which was due to the prolonged time from the sample collection until the time of single cell isolation causing the cells to die or degenerate (Table 5.1613). However, the amplification efficiencies were high with ADO rates remaining low.

3.3.2 Clinical DM PGD Cases

One cycle for each of the two families was carried out during this study. For family G five embryos were suitable for biopsy however, the embryos were of poor quality. Hence, only one blastomere was biopsied from each embryo. This was possibly due to the advanced maternal age (42 years of age) at the time of egg collection. Three embryos were affected and two embryos had either incomplete or no results, therefore no embryos were transferred and all were donated for confirmation of diagnosis (Table 5.1714). There were no cases of contamination observed in any of the negative

control samples. The low quality of the embryos was also reflected in the number of blastomeres that were biopsied for confirmation of the diagnosis. Embryo no. 3 had completely degenerated and no cells could be retrieved. During confirmation analysis, all embryos were found to be affected (Table 5.1815). In total the amplification rates for the DM DM1 and D21S1414 markers were 75% and 68.75%. The reduced amplification rate for both markers when compared to the workup single cell rate has been widely reported. Sermon *et al* (1997) revealed 100% amplification during single cell workup which was reduced to 78% whilst performing the clinical cases. In a study by Dean and co-workers (2001) a decrease in amplification was also found when tested on single lymphocytes (92%) and single blastomeres (84%). No problems occurred regarding stutter bands or the +A effect in any of the blastomeres. However, in embryo Gs2 there were peaks, outside the region which the D21S1414 product produces a peak, which were disregarded and were thought to be artefacts.

For family H the DM/D19S112 (Protocol 2) was also used on five embryos. Three embryos were affected, one was normal and one showed incomplete results (Table 5.1916). The normal embryo was transferred but no pregnancy was achieved. In embryo H2 the DM DM1 locus failed to amplify in either of the cells biopsied and the D19S112 showed results from one blastomere which was considered affected. However, in the case of embryo H3 the results displayed a normal embryo, but, due to extreme cases of preferential amplification for both blastomeres regarding the DM DM1 allele and ADO of the D19S112 locus it was not considered for transfer to avoid misdiagnosis. Carrying out confirmation of the diagnosis in the untransferred (spare) embryos revealed that embryo H3 was found to be normal and embryo H2 was affected (Table 5.2017). The phenomenon of mosaicism or maternal contamination in the spare embryos Hs4 and Hs6 was due to more than one cell being present in the tube since some embryos were compacted and could not be either biopsied or disaggregated therefore some cumulus cells attached to the ZP might have given such results. During the clinical case no incidence of maternal contamination was found in any of the 16 blanks (one final wash drop blank for each blastomere; three for lysisbuffer-only blanks; and three PCR reaction-mixture-only negative controls). Overall, from 20 blastomeres the DM DM1 locus revealed an 80% amplification, no ADO but 15% of extreme preferential amplification, whereas the D19S112 showed an 85% amplification rate and a 15% allele dropout.

ADO is a major concern of single cell PCR and although no single protocol has managed to eliminate ADO completely, the use of F-PCR used in this study can increase 1000-fold the sensitivity of PCR (Hattori et al, 1992). Findlay and coworkers (1995) were able to demonstrate the advantages of F-PCR over conventional PCR and concluded that a number of cases of ADO are in fact preferential amplification of one of the alleles and that the other weakly amplified product is not detected by conventional analysis. Such results were seen during the clinical case for family G H when the DM DM1 locus presented in three blastomeres (15%) extreme cases of preferential amplification, which at the time of diagnosis were considered as ADO. However, upon reanalysis of the data it was observed that the larger allele (146) was not able to amplify in the blastomeres biopsied on the day of diagnosis. Normal samples with markedly preferential amplification of the father's DM DM1 allele can resemble an affected sample. Unless the mother's unaffected allele was present at an analysable level above the base line (peak height = 150) it was not considered safe to transfer the embryo. This was the case for embryo H3. It has been noted during our experience with single cell F-PCR analysis that it is more common for the large allele to be affected by preferential amplification or allele dropout. Similar findings were reported from Piyamongkol et al (2003) whilst trying to identify the reasons behind allele dropout.

The use of the STR markers on chromosome 21 (D21S1414) and 19 (D19S112) could also have the added benefit of providing copy number information for these chromosomes in the cells sampled. Sherlock *et al* (1997) utilised QF-PCR to detect small numbers of trisomic fetal cells isolated from the transcervical canal. However, when the same group was able to test the D21S1414 on single cells the method produced precise results for 75% of amplifications (Sherlock *et al*, 1998). This technique was based on the fact that the alleles of a heterozygous STR locus in a normal individual should amplify equally as long as the PCR is in the exponential phase of amplification (Mansfield, 1993). At an STR locus a trisomic subject has three alleles each amplified to the same extent (i.e. three different copies of the trisomic chromosome) or alternatively two alleles with one amplified twice as much as the other (i.e. two identical copies of the trisomic chromosome and one nonidentical copy). Dean and colleagues (2001) were able to distinguish between an

embryo being monosomic (or mosaic) for chromosome 19 from an embryo being affected from ADO, by including an unlinked STR. During analysis the group observed only the affected parent's normal allele being amplified for the DM DM1 marker for both biopsied blastomeres, whilst the unlinked STR showed both alleles amplified, which was later confirmed from the spare embryos (Dean *et al*, 2001). Therefore, additional STR's during multiplex analysis can provide further genetic information and maximise the number of embryos in which a genotype can be assigned.

For both clinical cases the genotypes matched the expected genotypes that were known from the workup of control DNA and patient single cells. There was rarely a base pair difference (123 instead of 122) in the allele sizes. Apart from the unexpected fragments seen in embryo Gs2, there was no other unexplained amplification indicating that the results obtained were true genotypes of the embryos and DNA contamination was unlikely. Similar observations were observed from Dean *et al* (2000), however, their analysis was performed using an ALF automated sequencer (Pharmacia Biotech). Maximising the number of diagnosed embryos is particularly important in the case of dominant diseases since only 50% of all generated embryos will be expected to be normal and thus available for transfer.

It has been postulated that two cells per embryo should be biopsied in order to reduce misdiagnosis (Sermon *et al*, 1998; Ao *et al*, 1998; Ray *et al*, 1998). During this study it was appreciated that biopsy of two cells per embryo is a necessity in order to avoid misdiagnosis. Two cells are biopsied (in the UCL centre for PGD) from embryos with ≥ 6 cells. However, for family G only one blastomere was biopsied from each embryo due to reduced embryo development. This lead to attaining a complete result from only 60% of the embryos which were biopsied. An embryo that does not reach the 7-cell stage by 72h of development is less likely to progress to the expanded blastocyst stage (8.1% for ≤ 6 cells compared to 43.3% for ≥ 7 cells) (Shapiro *et al*, 2000). Although, no misdiagnosis occurred, the results for family H were enhanced due to the addition of the second cell.

In conclusion a comprehensive PGD protocol for DM DM1 has been developed using single step multiplex analysis of the normal DM DM1 triplet region and either a

polymorphic linked marker (D19S112) or a polymorphic unlinked marker (D21S1414). The linked marker will provide backup linkage analysis as well as contamination identification and when the linked marker is partially informative for a family, the unlinked D21S1414 marker can be used for the detection of contamination. In an ideal world triplex analysis would be performed for a PGD protocol by employing the mutation marker, a linked marker and an unlinked marker, in order to have backup linkage analysis as well as backup contamination exposure.

3.3.3 Universal PGD Protocol for DM

During the second part of this study the development of a PGD protocol for DM1 was attempted using a whole genome amplification method, namely DOP-PCR. DOP-PCR amplified genomic and single cell DNA was tested for genome coverage through amplification of the DM1 mutation marker along with another four polymorphic linked (D19S112) and unlinked markers (D21S1414, D18S535 and D13S305). Thus, the DOP-PCR coverage would be tested in at least four chromosomes (13, 18, 19 and 21). Such a protocol would eliminate the time consuming need for multiplexing and optimising different F-PCR protocols in single cells for different patients.

3.3.3.1 DOP-PCR on Genomic DNA

To be useful, a whole genome amplification procedure should have the following features: generation of long DNA fragments, successful amplification of starting DNA template, high amplification fidelity and yield, and good coverage of the genome. Due to the increased amount of information gathered from the workup of all the families (section 3.2.1) initially genomic DNA was amplified using DOP-PCR from family E (mother and father). The genomic DNA was approximately 100-300ng and was able to produce very distinct smears when run on agarose gel. All microsatellite markers showed similar amplification results when they were amplified from genomic DNA or from genomic DNA amplified using DOP-PCR (Figures 3.2, 3.3, 3.4 and 3.5). When the DOP-amplified DNA was used as a template for the further amplification of the five polymorphic markers, there was a slight increase of stutter bands especially for the DM1 marker and the D19S112 marker. This was not unexpected as mono- and dinucleotide repeats are well known to produce prominent stutter bands which might complicate the genotyping analysis (Ellegren, 2004).

Furthermore, all markers amplified using the DOP-PCR as template showed reduced amount of product, especially for the D21S1414 and D18S535 (Figures 3.3 and 3.4 respectively). However, all DOP-PCR amplified genomic DNA displayed the correct genotypes which was in accordance with the findings of Cheung and Nelson (1996) which analysed microsatellite repeats of DOP-PCR amplified genomic DNA (1ng concentration). There was no ADO present, although there were instances of mild preferential amplification of the smaller allele. Similar results have been reported from Cheung and Nelson (1996) who achieved 100% amplification of the markers and correct assignments of genotypes, though it was noticed that there was some preferential amplification of the shorter allele. Recently, Struan and colleagues (2002) managed to use DOP-PCR to amplify genomic DNA (1-40ng) and subsequently carry out single nucleotide polymorphism (SNP) genotyping. In the latter study DOP-PCR yielded satisfactory results but, displayed loss in accuracy and quality of the genotype assignments. Furthermore, a new method of DOP-PCR (LL-DOP-PCR) has been reported which is able to generate long fragments from pg quantities of genomic DNA (Kittler et al, 2002). It has been shown that by using LL-DOP-PCR, products of up to 10kb can be produced from less than one ng template genomic DNA, thus providing better coverage for microsatellite and unique sequences compared to the conventional DOP-PCR method. During this study, the overall amplification rate was 98% for all primers except for the D13S305 which was 96%.

3.3.3.2 DOP-PCR on Single Cells

Two individuals (X and Z) were chosen to donate 10 single cells each as well as genomic DNA. This was due to the fact that they were found to have heterozygous alleles for almost all the markers and especially for the DM DM1 mutation marker. The underlying reason was to be able to detect any allele dropout present in the most important polymorphic markers in this study, since a diagnosis would certainly involve this marker as well as the availability of fresh single cells. 100%Hundred percent amplification (20/20) was achieved for the DOP-PCR part of the study (Figures 5.6 and 5.7). Moreover, the single cells from individual G Z showed better amplification results compared to individual X. However, all single cells from both individuals were collected, isolated and amplified the same day in order to avoid freezing and thawing of the samples which might lead to dead or degenerating cells

and increase the levels of ADO (Piyamongkol *et al*, 2003). Only one cell from each individual (X4 and Z10) displayed reduced amplification with a smaller and fainter smear. This was expected since those particular buccal cells might have been in the process of degeneration or the lysis protocol was unable to completely lyse them. Furthermore, only in individual G Z the DOP amplified product showed some distinct bands in the 450bp region, which has previously been observed from Voullaire *et al*, (2000). For each individual two clumps of buccal cells (3-4 cells) and one sample of genomic DNA was also added to observe the differences between the amounts of starting DNA and act as a positive control. The negative controls for all single cells and of the DOP reaction displayed no amplification and were free of contaminants.

During the DOP-PCR amplified single cell DNA all microsatellite loci revealed decreased amplification, increased ADO and the incidence of incorrect genotype of about 10% (Table 5.2118). As seen in the DOP results, the genotyping results for the microsatellite loci regarding individual G Z showed better overall results. The combined amplification efficiency for all markers was 66% which was lower compared to the previously reported 85% amplification rate achieved from DOP-PCR amplified single cell DNA (Wells et al, 1999). The DM DM1 locus exhibited the highest amplification rate of 85% and the most reduced ADO rate of 10%. Overall, during this entire study it had been observed that the primer for the DM DM1 locus achieves high amplification rates, with minimal ADO and was not affected by the presence of other primers (e.g. during multiplex PCR). It was postulated during the study by Wells et al (1999), that unique sequences (such as the DM DM1 mutation marker) have a higher fidelity for replication compared to microsatellite loci (linked and unlinked STR's), which might account for the higher amplification results. However, it was noticed in this study that during amplification of the DM DM1 locus using DOP-PCR amplified single cell DNA, there was a greater increase of the +A artefact (section 5.4.1; Figure 5.10). The complete elimination of the final ten minute extension step and the reduction of the extension step during each cycle of amplification from one minute to 30 seconds were not found to improve the fidelity of the primer. However, these artefacts would not cause misdiagnosis since the true allele was easily distinguishable.

and increase the levels of ADO (Piyamongkol *et al*, 2003). Only one cell from each individual (X4 and Z10) displayed reduced amplification with a smaller and fainter smear. This was expected since those particular buccal cells might have been in the process of degeneration or the lysis protocol was unable to completely lyse them. Furthermore, only in individual G Z the DOP amplified product showed some distinct bands in the 450bp region, which has previously been observed from Voullaire *et al*, (2000). For each individual two clumps of buccal cells (3-4 cells) and one sample of genomic DNA was also added to observe the differences between the amounts of starting DNA and act as a positive control. The negative controls for all single cells and of the DOP reaction displayed no amplification and were free of contaminants.

During the DOP-PCR amplified single cell DNA all microsatellite loci revealed decreased amplification, increased ADO and the incidence of incorrect genotype of about 10% (Table 5.2118). As seen in the DOP results, the genotyping results for the microsatellite loci regarding individual G Z showed better overall results. The combined amplification efficiency for all markers was 66% which was lower compared to the previously reported 85% amplification rate achieved from DOP-PCR amplified single cell DNA (Wells et al, 1999). The DM DM1 locus exhibited the highest amplification rate of 85% and the most reduced ADO rate of 10%. Overall, during this entire study it had been observed that the primer for the DM DM1 locus achieves high amplification rates, with minimal ADO and was not affected by the presence of other primers (e.g. during multiplex PCR). It was postulated during the study by Wells et al (1999), that unique sequences (such as the DM DM1 mutation marker) have a higher fidelity for replication compared to microsatellite loci (linked and unlinked STR's), which might account for the higher amplification results. However, it was noticed in this study that during amplification of the DM DM1 locus using DOP-PCR amplified single cell DNA, there was a greater increase of the +A artefact (section 5.4.1; Figure 5.10). The complete elimination of the final ten minute extension step and the reduction of the extension step during each cycle of amplification from one minute to 30 seconds were not found to improve the fidelity of the primer. However, these artefacts would not cause misdiagnosis since the true allele was easily distinguishable.

The linked marker to the DM DM1 mutation, the D19S112, showed the loweeast amplification rate (55%) with one of the highest ADO rates (40%). However, this result was partly due to the low amplification seen in individual CX. Such reduced amplification was in contrast to the amplification seen for the D19S112 marker when amplified directly from single cell DNA (78.3%; Table 5.1613). Furthermore, the ADO rates nearly tripled compared to direct amplification (15%; Figure 5.8). It has been observed in previous studies that microsatellite loci show less reliability for replication (Wells et al, 1999) which is in accordance with this study. The D21S1414 microsatellite locus showed good amplification (65%), but, lower compared with direct amplification. A significant number of PCR artefacts were seen to affect the D21S1414 and D19S112 repetitive markers particularly small deletions and insertions. Allele sizes were increased or decreased by a number of base pairs equivalent to one repeat length (Figure 5.9). This phenomenon was locus specific but was shown to be present in all the markers though with decreased incidence. Similar findings were observed from Wells et al (1999), which suggested that this locus specificity is due to variation in factors such as chromatin structure, GC content and whether the repeat is perfect or disrupted by other sequences. Focault et al (1996) also reported such results when amplifying (CA)_n repeats from less than ten single cells and concluded that this phenomena are due to replication slippage.

The highest incidence of allele dropout was exhibited from the D18S535 locus (50%; Table 5.2118). Allele dropout and preferential amplification was found to affect 60-83% of the heterozygous samples when amplified with repetitive microsatellite loci (Wells *et al*, 1999). ADO however, was rarely observed (3-5%) in unique sequences such as CFTR Δ F508 carrier cells in the same study. Preferential amplification was not often seen in this study however, it was shown to mainly affect the D18S535 and D13S305 loci (data not shown). This might be due to the fact that both microsatellite loci produce large products of more than 430bp. Piyamongkol and colleagues (2003) have noticed that the larger the amplified product the more prone to preferential amplification and ADO it will be. However, both loci showed the best results regarding sharp peaks, limited stutter peaks and +A artefact. Such results are in accordance with tetranucleotide microsatellite alleles being amplified directly from single cells.

During the analysis of the five markers after whole genome amplification of single cells, the incidence of incorrect genotyping was observed (Table 5.2118). Although a small number of single cells were tested, it was found that around 10-20% of the loci showed incorrect genotypes i.e. 1-2bp difference from true allele size and rarely 3-4bp difference e.g. for the D18S535 marker the correct genotype was 478/482, but, the analysis revealed 473/477. However, more single cells need to be tested and more loci investigated.

The poor fidelity of amplification seems to be attributable to the WGA reaction, namely the DOP-PCR. Firstly, whilst devising a PGD protocol for DM DM1 using F-PCR in the first part of this study, it was found that ADO was present in less than 20% when directly genotyping microsatellite markers from single cell DNA. This is in accordance with other studies that have amplified single cells (Sherlock *et al*, 1998; Sermon *et al*, 1998). When DOP-PCR amplified single cell DNA was genotyped with STR's, during this study, the overall ADO rate was 35%, suggesting that probably the DOP-PCR part was not reliable. Furthermore, different aliquots from the same DOP-PCR amplified product showed similar PCR artefacts and the increased failure of amplification of the five markers was random during this study. Wells *et al* (1999) noticed similar behaviour. However, when genomic DNA was amplified using DOP-PCR, the results were excellent and there were no increase in ADO and PCR artefacts. Hence, the success of the DOP-PCR reaction depends on the starting template DNA and if it is limited, the DOP-PCR enhances the problems of single cell PCR.

3.3.3.3 CGH and DM

In the final part of this study, the remaining DOP-PCR product was subjected to CGH analysis. It has been shown (Chapter 4) that DOP-PCR provides reliable and reproducible results for CGH analysis after amplification of single cell DNA. A total of 17/20 (85%) DOP-PCR amplified single cell DNA produced dynamic fluorescence and specific hybridisation, thus yielding analysable results. Both individuals (C and G) were female, hence, their amplified single cells were hybridised against control male DNA. In all 17 cells the sex status of the test samples was confirmed (Figure 3.11). Wells *et al* (1999) was also able to show reproducible CGH results when using as template DOP-PCR amplified single cell DNA. This additional test from the DOP-

PCR amplified single cells provides the opportunity in the context of PGD of investigating a specific single gene disorder as well as the chromosomal status of the embryos. Therefore, the phenomenon of mosaicism (Chapter 3) can be explored even in single gene analysis and a greater amount of information can be attained from a single cell. During a PGD case of myotonic dystrophy the affected embryos as well as the overall chromosomal status of the embryo can be distinguished, which will lead to better chances of a normal embryo being transferred.

3.3.4 Conclusions and Future Work

It was shown during this study that the development of multiplex F-PCR protocols for clinical cases of PGD are efficient and can provide accurate diagnosis. Two protocols were used for two families. Protocol 1 has been previously reported (Piyamongkol *et al*, 2001a) however it was found that certain modifications needed to be carried out to increase the fidelity of the amplifications. Unfortunately, only 5 embryos were biopsied and all were found to be unsuitable for transfer. The multiplex reaction for protocol 2 was devised specifically for family H and included a linked marker (D19S112), which could detect contamination and ADO. One embryo was found to be unaffected and was transferred but no pregnancy was achieved.

An alternative protocol was tested which would allow the detection of DM affected embryos as well as any whole chromosome abnormalities with the use of WGA. DOP-PCR amplified genomic and single cell DNA were used as templates for subsequent F-PCR reactions using five different markers. This would alleviate the problem of multiplexing and optimising protocols as well as creating patient-specific protocols. DOP-PCR amplified genomic DNA showed excellent results of amplifications from four individuals (mother and father of family E and individuals X and Z). DOP-PCR amplified single cell DNA showed reduced amplification efficiencies and increased ADO rates compared to direct amplification of the five markers from single cells. Although it was shown that in addition to single gene analysis, whole chromosome analysis using CGH can be carried out.

Future work should include analysis of more single cells (>1000) from more individuals, in order to assess the fidelity of amplification. Even though, DOP-PCR

amplified single cell DNA showed efficient CGH analysis, it did not provide acceptable coverage of the genome. Hence, the testing of novel WGA techniques such as multiple displacement amplification should be performed (sections 1.3.3.1.3 and 1.5.3.3).

Recently, the scope of PGD of single gene disorders using micro-arrays has been explored. Cystic fibrosis (CF), and specifically the Δ F508 mutation, has been used as a model disease to prove the diagnostic capability of micro-arrays for PGD (Salvado *et al*, 2004). A single micro-array platform was constructed using oligonucleotide probes for both the normal and Δ F508 disease alleles and the target DNA was fluorescently labelled which was hybridised to the array and the Δ F508 genotypes assigned from the fluorescence bound to ach allelic probe (Salvado *et al*, 2004). In a mix of samples (homozygous normal, homozygous affected, heterozygous, samples from previous PGD case for CF) strong binding of the target DNA to the probes was observed and all samples were correctly genotyped (Salvado *et al*, 2004).

CHAPTER 4

Detailed FISH analysis of day 5 human embryos reveals the mechanisms leading to mosaic aneuploidy Chapter 4 – Detailed FISH analysis of day 5 human embryos reveals the mechanisms leading to mosaic aneuploidy

4.1 Aims

In this study the level of mosaicism was explored using sequential rounds of FISH in two groups of day 5 human embryos (mainly blastocysts). The chromosomes studied were 1, 11 and 18, X and Y. Three rounds of FISH were performed. The aim of this study was to develop a FISH protocol that can accurately and precisely determine the true level of abnormalities and mosaicism. This unique protocol employed two probes for each chromosome at different loci, which were analysed in sequential rounds of FISH. Such protocol technique would allow the direct distinction of FISH artefacts, failure of hybridisation that can occur during FISH analysis on human embryos. Any contradictive results between the two probes for the same chromosome were excluded from the analysis therefore revealing the true level of chromosomal abnormalities and specifically aneuploidies. So far previous studies have utilised only one probe per chromosome and have been hurdled with FISH failure and unexplained findings. However, with this novel protocol the FISH procedure is more robust and can aid in understanding the extent of mosaicism and most significantly the mechanisms leading to mosaicism.

4.2 RESULTS

4.2.1 Preliminary Work

Initial work included developing three FISH probe combinations that would yield high efficiencies whilst performing 3-round sequential FISH procedures. The various combinations that were attempted and their efficiencies are shown in Table 4.1. Most probes were commercially obtained however, probes e.g. 1het and 2cep were laboratory grown (Table 2.1). From the combinations, the 1/11/18 chromosome combination was picked since it produced high efficiencies and was able to test 3 chromosomes of different size. A 3rd round of FISH was included using the probecocktail for chromosome 18 was included in all rounds acting as a control and the presence of gonosome abnormalities can be investigated in blastocyst and their involvement in mosaic patterns.
| Table 4.1. Various probe combinations | that y | were | attempted | in | order | to | find | the | most |
|---|--------|------|-----------|----|-------|----|------|-----|------|
| efficient 3-colour FISH in both rounds. | | | | | | | | | |

| Chromosome | Probes used in | Probe efficiency | Probes used in 2 nd | Probe efficiency |
|----------------------------|-----------------------|----------------------|---------------------------------|--|
| combinations | 1 st round | | round | 2 Mar. 1 |
| 1 dual / 8 dual | 1het / 1p | 95% / 93% | 8cep / 8q | 94% / 87% |
| 4 / 16 dual & | 4cep / 16p / 16q | 97% / 95% / 97% | 4cep / 11cep / 11q | 89% / 94% / 90% |
| 4 / 11 dual | - at a | 5 3. Const | | i de la companya de la |
| 7 dual [#] / 18 & | 7q / 7q / 18cep | 91%/92%/95% | 11cep / 11q / 18cep | 91% / 89% / 94% |
| 11 dual / 18 | | | | |
| 2/4/5 | 2q / 4cep / 5p | 95% / 98% / 94% | 5p&5q */ | 80%&78% / |
| a hat a | | | 2cep | 67%&61% |
| 1 / 11 / 18 | 1p / 11q / 18cep | 94% / 97% / 96% | 1het / 11cep / 18q | 88% / 89% / 90% |
| | | | | |
| X / Y / 18 | Xcep / Ycep / | 96% / 95% / 98% | Хсер / Үсер / 18сер | 94% / 90% / 93% |
| (cocktail) | 18cep | te het Guite I. Spak | (3 rd Round results) | (3 rd Round |
| contra to assig | n efferne a slave | a somether weit at | to acquestal contrils. Ov | results) |

The probe was the 7 (7q11.23 and 7p31) Willams microdeletion probe

* The probe was the 5 (5q31 and 5p15.2) Cri-du-Chat microdeletion probe

The colours show the corresponding fluorescent labels i.e. orange-spectrum red, green-spectrum green, blue-spectrum aqua. cep = centromeric or heterochromatic probe, p = sub-telomeric probe for the small arm, q = sub-telomeric probe for the big arm

The chosen combination was then applied to 10 arrested day 5 human embryos (118 blastomeres) donated from 3 patients (data not shown). This was performed to assess the spreading technique in order to reduce loss of cells and more significantly to evaluate the FISH protocol. The preliminary results showed high FISH efficiencies with bright and sharp signals in all three rounds of FISH and minimal cell loss. Overall, 101 blastomeres were FISHed for all sequential rounds out of 118 blastomeres counted before spreading (86%). 98/101 blastomeres (97%) produced visible and foremost interpretable results for all FISH rounds. The embryos were analysed in two separate FISH procedures (five embryos each time). In each of the two procedures a control male lymphocyte slide was also included in the FISH

procedure and 200 interphase nuclei were counted to calculate the probes' efficiency. The yields were 97%, 91% and 88% for the 1^{st} , 2^{nd} and 3^{rd} round respectively.

4.2.2 FISH Study

Twenty-seven couples donated a total of 47 embryos for this research project (Table 2.9). The overall mean maternal age was 33.7 (range 19-41) years. The mean maternal age for Group I embryos was 35.2 (range 27-41) and for Group II was 32.1 (range 19-41) years. Five arrested embryos were not included, since the signals were not analysable due to loss of most nuclei during sequential rounds of FISH and poor quality of the nuclei. Embryos from both groups included nuclei which showed contradicting information for the two probes used for each autosome in the different rounds of FISH. These cells were classed as *inconsistent results* and excluded from the analysis.

4.2.2.1 FISH Analysis of Controls

Each FISH experiment included a control male lymphocyte slide with mapped nuclei in order to assess efficiency of probe hybridisation in the sequential rounds. Overall, 87.1% (range 78-96) of the control nuclei showed normal signals for all 8 probes used. Sub-telomeric probes for chromosomes 1p, 11q and 18q, showed a higher incidence of one signal per chromosome per nucleus, 7.8% (range 4-11%), 3.1% (range 1.2-4.5%) and 3.9% (range 2-5%) respectively. Heterochromatic region or centromeric probes, 1het, 11CEP and 18CEP (in both 1st and 3rd rounds) displayed a lower incidence of one signal per chromosome per nucleus, 6.2% (range 4.4-7.5%), 2.4% (range 0.9-3.3%) and 2.2% (1.5-3.4%) respectively (Table 4.2). Nuclei with one signal for the X chromosome and no signal for chromosome Y and nuclei with three or more signals for autosomes comprised less than 1%. Furthermore, all three subtelomeric probes, 1p, 11q and 18q showed the occurrence of split signals (i.e. replicated DNA) in 2.2%, 1.4% and 1.9% of nuclei respectively, which was considered normal due to the position of the probes. The difference in the efficiency of the 18cep probe (used in the 1st and 3rd round of FISH) was not statistically significant (p<0.05) and can be attributed to the fact that FISH efficiency decreases with sequential rounds of hybridisation due to the DNA being over-processed.

| Round | Probe | % of missi | ing a signal | Split s | signals |
|-----------------|-------|------------|--------------|---------|---------|
| of FISH | | Mean | Range | Mean | Range |
| | 1p | 7.8 | 4-11 | 2.2 | 1.1-2.8 |
| 1 st | 11q | 3.1 | 1.2-4.5 | 1.4 | 0.4-1.9 |
| | 18cep | 2.2 | 1.5-3.4 | 0.3 | 0.1-0.4 |
| | 1het | 6.2 | 4.4-7.5 | 0.4 | 0.3-0.7 |
| 2 nd | 11cep | 2.4 | 0.9-3.3 | - | - |
| | 18q | 3.9 | 2-5 | 1.9 | 0.8-2.4 |
| | Хсер | 0.9 | 0.5-1.1 | - | - |
| 3 rd | Усер | 0.5 | 0.1-0.9 | - | - |
| | 18cep | 3.4 | 2.1-4.3 | 0.5 | 0.4-0.8 |

Table 4.2. Probe efficiencies scored in 200 interphase nuclei of each control slide whilst carrying sequential 3-round FISH.

The results for the probes were calculated during sequential rounds of FISH, hence each control slide was subjected to 3 rounds of FISH. The table lists only the percentages where a signal was missing, hence 1 signal was seen for the autosomes and no signal for the gonosomes. The incidence of extra and complete lack of signals was minute (<0.4%) for all the probes tested.

4.2.2.2 FISH Analysis of Embryos

Twenty-one embryos were analysed for each group. FISH results for Group I and Group II embryos are shown in Tables 4.3 and 4.6 respectively. In Group I the total number of nuclei analysed was 401/410 (9 lost during sequential rounds of FISH). Thus, the mean number of nuclei analysed was 20 (range 4-45). In all, 72% (\pm 20 SD) of the nuclei analysed were diploid for the chromosomes tested (Figure 4.1) however only a single embryo was uniformly diploid for the tested chromosomes (embryo 9.1). One embryo was aneuploid mosaic, and 14 were diploid mosaics (4.3). Three embryos contained chaotic complements (embryos 8.1, 12.2 and 17.1) and thus a mechanism of mosaicism could not be suggested for those embryos. Since 19/21 embryos were considered arrested for Group I (<30 cells), four embryos contained polyploidy cells and predominantly tetraploidy cells, however, 17.2 had hexaploid and octaploid cells. One embryo only (20.1) contained a haploid chromosome complement for all five chromosomes tested. Although, this is quite low compared to other studies, haploid cells have been characterised as less viable and less actively dividing than tetraploid cells (Ruangvutilert *et al.*, 2000a)

| Embryo | Cells | Sex | Chromosome constitution of cells (number) | Diploid | Classification |
|--------|----------|-----|--|------------|------------------------------------|
| No. | analysed | | | (%) | |
| 4.1 | 4 | F | -18 (3) / +18 (1) | 0 | Mosaic Aneuploid |
| 5.1 | 9 | M | Dip (7) / +X, +Y (1) / *(1) | 77.8 | Mosaic Diploid/Aneuploid |
| 5.2 | 10 | F | Dip (6) / tet (3) / -1 (1) | 60 | Mosaic Diploid/Polyploid/Aneuploid |
| 5.3 | 20 | F | Dip (15) / -1 (1) /-X (1) / *(3) | 75 | Mosaic Diploid/Aneuploid |
| 5.4 | 22 | F | Dip (18) / -1 (2) /+18 (1) / *(1) | 81.8 | Mosaic Diploid/Aneuploid |
| 5.5 | 4 | F | Dip (3) / -1 (1) | 75 | Mosaic Diploid/Aneuploid |
| 6.1 | 22 | M | Dip (17) / tet (2) / *(3) | 77.2 | Mosaic Diploid/Polyploid |
| 7.1 | 26 | F | Dip (24) / -1 (1) / -11 (1) | 92 | Mosaic Diploid/Aneuploid |
| 7.2 | 9 | M | Dip (7) / -1 (1) / -1, +X (1) | 77.8 | Mosaic Diploid/Aneuploid |
| 8.1 | 6 | M | Dip (4) / +X, +Y (1) / chaotic (1) | 66.7 | Mosaic Diploid/Aneuploid/Chaotic |
| 9.1 | 8 | F | Dip (7)/ *(1) | 87.5 | Uniformly Diploid |
| 10.1 | 5 | F | Dip (4) / -11 (1) | 80 | Mosaic Diploid/Aneuploid |
| 12.1 | 13 | M | Dip (6) / -18, -Y (3) / -Y (1) / -18, +Y (1) / -18 (1) / -1 (1) / *(1) | 46.2 | Mosaic Diploid/Aneuploid |
| 12.2 | 30 | F | Dip (22) / -11 (1) / -18 (1) / -X (1) / -1, -18 (1) / | 73.3 | Mosaic Diploid/Aneuploid/Chaotic |
| | | | -11, -18 (1) / chaotic (3)/ *(1) | | |
| 17.1 | 45 | M | Dip (40) / +Y (2) / chaotic (2) / *(1) | 88.9 | Mosaic Diploid/Aneuploid/Chaotic |
| 17.2 | 12 | F | Dip (9) /tet (3) | 75 | Mosaic Diploid/Polyploid |
| 20.1 | 43 | F | Dip (34) / -18 (4) / +11 (4) / hap (1) / *(3) | 79 | Mosaic Diploid/Aneuploid/Haploid |
| 21.1 | 27 | M | Dip (22) / hex (3) / oct (1) / $*(1)$ | 81.5 | Mosaic Diploid/Polyploid |
| 21.2 | 29 | F | Dip(25)/+11(1)/+1,+11(1)/*(2) | 86.2 | Mosaic Diploid/Aneuploid |
| 27.1 | 24 | M | Dip (16) / +Y (2) / +18 (1) / +X (1) / *(4) | 66.6 | Mosaic Diploid/Aneuploid |
| 27.2 | 33 | M | Dip $(28) / +Y(1) / *(4)$ | 84.8 | Mosaic Diploid/Aneuploid |

Table 4.3. Results of FISH analysis of Group I embryos after three sequential rounds of hybridisation with probes for chromosomes 1, 11, 18, X and Y

-

-

* Inconsistent results that were observed during scoring. dip = diploid, tet = tetraploid, trip = triploidy, hex = hexaploidy, oct = octaploidy.

The - indicates loss of chromosome and + indicates gain of chromosome e.g. +18 is trisomy 18 or -1 is monosomy 1.

I ----

Figure 4.1. Results of three sequential rounds of FISH analysis on a single female embryonic nucleus from embryo 9.1.



a) Using probes 1p, 11q and 18cep, b) using probes 1het, 11cep and 18q and c) using probes for Xcep, Ycep and 18cep. In all three rounds the cell appears to be diploid. The probe 18CEP is present in rounds 1 and 3 and shows identical results.

In Group I there were 32 post-zygotic errors (16 chromosome losses, 14 chromosome gains and 2 instances of mitotic non-disjunction) (Table 4.4). There was no statistical significance (p<0.05) between chromosome loss and chromosome gain. For the chaotic blastomeres the mechanism causing mosaicism was not included due to the nature of chaotic cells. Embryo 12.2 showed the highest incidence of aneuploid mosaicism involving 4 chromosomes (chaotic cells not included). The two instances of MND were for chromosomes 18 and Y in embryos 4.1 and 12.1.

(a) a 4.5 standards for themselves? In the control Direct-state 1, was mustly Proted by chromosome long characteristic tractory and the chromosome should the ophest measures of electronyme one cost 1, chromosomer did not display any structurence for characteristic attraction prenty by chromosomerical gen and structurence for characteristic attraction prenty by chromosomerical gen and Diright of its

| Embryo | Type of | Classification | Events | Chromosome |
|--------|------------|-----------------------------|--------|--------------|
| No. | embryo | | | |
| 4.1 | Arrested | Mosaic Aneuploid | 1 MND | 18 |
| 5.1 | Arrested | Mosaic Diploid/Aneuploid | 2 CG | Χ, Υ |
| 5.2 | Arrested | Mosaic | 1 CL | 1 |
| | | Diploid/Polyploid/Aneuploid | | |
| 5.3 | Arrested | Mosaic Diploid/Aneuploid | 2 CL | 1, X |
| 5.4 | Arrested | Mosaic Diploid/Aneuploid | 1 CL | 1 |
| | | | 1 CG | 18 |
| 5.5 | Arrested | Mosaic Diploid/Aneuploid | 1 CL | 1 |
| 6.1 | Arrested | Mosaic Diploid/Polyploid | - | - |
| 7.1 | Arrested | Mosaic Diploid/Aneuploid | 2 CL | 1, 11 |
| 7.2 | Arrested | Mosaic Diploid/Aneuploid | 1 CL | 1 |
| | | | 1 CG | X |
| 8.1 | Arrested | Mosaic | 2 CG | X, Y |
| | | Diploid/Aneuploid/Chaotic | | |
| 9.1 | Arrested | Uniformly Diploid | None | - |
| 10.1 | Arrested | Mosaic Diploid/Aneuploid | 1 CL | 11 |
| 12.1 | Arrested | Mosaic Diploid/Aneuploid | 2 CL | 1, 18 |
| | | | 1 MND | Y |
| 12.2 | Arrested | Mosaic Diploid/Aneuploid | 4CL | 1, 11, 18, X |
| 17.1 | Arrested | Mosaic | 1 CG | Y |
| | | Diploid/Chaotic/Aneuploid | | |
| 17.2 | Arrested | Mosaic Diploid/Polyploid | - | - |
| 20.1 | Expanded | Mosaic | 1 CL | 18 |
| | Blastocyst | Diploid/Aneuploid/Haploid | 1 CG | 11 |
| 21.1 | Arrested | Mosaic Diploid/Polyploid | - | - |
| 21.2 | Arrested | Mosaic Diploid/Aneuploid | 2 CG | 1, 11 |
| 27.1 | Arrested | Mosaic Diploid/Aneuploid | 3 CG | 18, X, Y |
| 27.2 | Arrested | Mosaic Diploid/Aneuploid | 1 CG | Y |
| Total | | - | 16 CL | |
| 21 | | | (50%) | |
| | | | 14 CG | 1 |
| | | | (44%) | |
| | | | 2 MND | |
| | | | (6%) | |

Table 4.4. Aneuploidy mosaicism mechanisms for Group I embryos

CL = chromosome loss, CG = chromosome gain, MND = mitotic non-disjunction

Table 4.5 summaries the breakdown by chromosome. Chromosome 1 was mostly affected by chromosome loss (7 events) whereas the sex chromosomes showed the highest incidence of chromosome gain. The Y chromosome did not display any chromosome loss events, although it was affected greatly by chromosome gain and

mitotic non-disjunction. Inconsistent results affected 26/401 (6.5%) nuclei and were considered to be FISH artefacts

| Event Chromosome | Chromosome Loss | Chromosome Gain | Mitotic Non- Disjunction |
|---------------------|-----------------|-----------------|-----------------------------|
| 1 | 7 | 1 | 0 |
| 11 | 3 | 2 | 0 |
| 18 | 4 | 2 | 1 |
| X | 2 | 4 | 0 |
| Y | 0 | 5 | 1 |
| | | | |
| Total | 16 | 14 | 2 |

Table 4.5. Mechanisms of mosaicism by chromosome for Group I embryos

Table 4.6 shows the results of Group II embryos. The total number of nuclei analysed was 1143/1171 (28 nuclei lost during sequential rounds of FISH). The mean number of nuclei per embryo was 55, ranging from 20-100. 78% (±15 SD) of the nuclei analysed were found to be diploid, although only two embryos were found to be uniformly diploid (16.1 and 23.1) as tested. The majority (16/21) of embryos were found to have some tetraploid cells (confirmed in all rounds), which totalled approximately 7% of the abnormal cells analysed (Figure 4.2). Six out of twenty-one embryos included a chaotic complement. Those embryos included at least one cell, which had more than three abnormalities per cell, including nullisomies and tetrasomies. Furthermore, there was one embryo (15.4), which was a diploid/aneuploid/haploid/chaotic mosaic. This embryo contained only one cell which was haploid, with the results being consistent in all rounds, and three aneuploid cells affecting chromosomes 18 and X (Figure 4.3). 20/21 embryos in Group II were dividing embryos and only one was arrested (13.1). Ten embryos were male and eleven embryos were female, hence no difference was observed in abnormalities regarding the sex, although both uniformly diploid (for the chromosomes tested) embryos were male.

Table 4.6. Results of FISH analysis of Group II embryos after three sequential rounds of hybridisation with probes for chromosomes 1, 11, 18, X and Y

_

| Embryo | Cells | Sex | Chromosome constitution of cells (number) | Diploid | |
|--------|----------|-----|---|---------|--|
| No. | analysed | | | (%) | Classification |
| 13.1 | 29 | F | Dip(13) / trip(7) / tet(5) / -18(1) / +1, +11(1) / chaotic (1) / *(1) | 44 | Mosaic Diploid/Polyploid/Aneuploid/Chaotic |
| 13.2 | 38 | F | Dip(18) / tet(8) / trip(4) / hex(1) / enn(1) / +1, +11(1) / -1(1) / | 47.4 | Mosaic Diploid/Polyploid/Aneuploid/Chaotic |
| | | | chaotic (2) / *(1) | | |
| 13.3 | 43 | F | Dip(33) / tet(6) / trip(1) / chaotic (1) / *(2) | 76.7 | Mosaic Diploid /Polyploid/Chaotic |
| 14.2 | 20 | F | Dip(12) / -X(3) / +X(2) / +11(1) / tet(1) / chaotic (1) | 60 | Mosaic Diploid/Aneuploid/Polyploid/Chaotic |
| 14.3 | 43 | Μ | Dip(36) / tet(2) / chaotic (3) / *(2) | 85.7 | Mosaic Diploid/Polyploid/Chaotic |
| 15.1 | 34 | F | Dip (31) / tet (3) | 91.7 | Mosaic Diploid/Polyploid |
| 15.2 | 100 | M | Dip(67) / tet(22) / trip(5) / -18(1) / -11, -18(1) / *(4) | 67 | Mosaic Diploid /Polyploid/Aneuploid |
| 15.3 | 62 | F | Dip(57) / tet(1) / trip(1) / +11(1) / *(2) | 91.9 | Mosaic Diploid /Polyploid/Aneuploid |
| 15.4 | 33 | F | Dip(26) / -X(2) / hap(1) / -18(1) / chaotic (1) / *(2) | 72.2 | Mosaic Diploid/Aneuploid/Haploid/Chaotic |
| 16.1 | 57 | M | Diploid (57) | 100 | Uniformly Normal |
| 18.1 | 57 | M | Dip (43) / +X(2) / +Y(2) / -11(2) / -X(1) / -Y(1) / *(6) | 75.4 | Mosaic Diploid/Aneuploid |
| 19.1 | 76 | F | Dip(66) / -X(3) / -18(2) / tet(1) / -11(1) / *(3) | 86.8 | Mosaic Diploid/Aneuploid/Polyploid |
| 19.2 | 94 | Μ | Dip(81) / tet(5) / trip(1) / +18(2) / -Y(1) / +X, +Y(1) / *(3) | 86.1 | Mosaic Diploid/Polyploid/Aneuploid |
| 22.1 | 75 | F | Dip(67) / +X(2) / tet(1) / -18(1) / -X(1) / *(3) | 89.3 | Mosaic Diploid/Aneuploid/Polyploid |
| 22.2 | 67 | F | Dip(55) / tet(3) / -X(3) / +1, +X(2) / +11, -18(1) / +X(1) / *(2) | 83.3 | Mosaic Diploid/Aneuploid/Polyploid |
| 23.1 | 100 | Μ | Diploid(94) / *(6) | 94 | Uniformly Normal |
| 24.1 | 63 | Μ | Dip(55) / tet(4) / *(4) | 87.3 | Mosaic Diploid/Polyploid |
| 25.1 | 62 | F | Dip(53) / tet(3) / -18(3) / *(3) | 85.5 | Mosaic Diploid/Polyploid/Aneuploid |
| 25.2 | 31 | М | Dip(24) / +X(2) / tet(1) / -18(1) / -18, -X(1) / *(2) | 77.4 | Mosaic Diploid/Aneuploid/Polyploid |
| 26.1 | 30 | М | Dip(20) / +1, +11(2) / -1(1) / +X(1) / +X, +Y(1) / chaotic (2) / *(3) | 66.6 | Mosaic Diploid/Aneuploid/Chaotic |
| 26.2 | 31 | Μ | Dip(19) / tet(5) / trip(1) / -11, -18(1) / +X(1) / chaotic (2) / *(2) | 61.3 | Mosaic Diploid/Polyploid/Aneuploid/Chaotic |

* Inconsistent results that were observed during scoring. dip = diploid, tet = tetraploid, trip = triploidy, hex = hexaploidy, oct = octaploidy, enn

= enniaploidy. The - indicates loss of chromosome and + indicates gain of chromosome e.g. +18 is trisomy 18 or -1 is monosomy 1.

Figure 4.2. Results of three sequential rounds of FISH analysis on a single male embryonic nucleus from embryo 13.2.



a) Using probes 1p, 11q and 18cep, b) using probes 1het, 11cep and 18q and c) using probes for Xcep, Ycep and 18cep. This nucleus displayed a tetraploid chromosome complement in all three rounds of FISH. In the 3rd round there are 2 signals for X and 2 signals for Y.

Figure 4.3. Results of three sequential rounds of FISH analysis on a single female embryonic nucleus from embryo 15.4.



a) Using probes 1p, 11q and 18cep, b) using probes 1het, 11cep and 18q and c) using probes for Xcep, Ycep and 18cep. This nucleus displayed monosomy 18 in all three rounds and XO in the final round of FISH. Therefore, two events of chromosome loss were reported for this nucleus.

In Group II 37 post-zygotic errors took place, which gave rise to 14 mosaic aneuploid embryos (Table 4.7). Some of those embryos also contained polyploid, chaotic and haploid cells.

Embryos 22.2 and 26.1 displayed the most chromosomal events affecting four chromosomes, sharing between them four chromosome losses, two chromosome gains and two mitotic non-disjunctions as mechanisms leading to aneuploid mosaicism.

Table 4.7. Aneuploidy mosaicism mechanisms for Group II embryo

```
CL = chromosome loss, CG = chromosome gain, MND = mitotic non-disjunction
```

| Embryo No. | Classification | Number of Events | Chromosome |
|------------|--|------------------|------------|
| 13.1 | Mosaic | 1 CL | 18 |
| | Diploid/Polyploid/Aneuploid/Chaotic | 2 CG | 1, 11 |
| 13.2 | Mosaic Diploid/Polyploid/Chaotic | 1 CG | 11 |
| | | 1 MND | 1 |
| 13.3 | Mosaic | - | - |
| | Diploid /Polyploid/Chaotic | | |
| 14.2 | Mosaic Diploid/Aneuploid/Polyploid | 1 CG | 11 |
| | | 1 MND | X |
| 14.3 | Mosaic Diploid/Polyploid/Chaotic | - | - |
| 15.1 | Mosaic Diploid/Polyploid | - | - |
| 15.2 | Mosaic Diploid /Polyploid/Aneuploid | 2 CL | 11, 18 |
| 15.3 | Mosaic Diploid /Polyploid/Aneuploid | 1 CG | 11 |
| 15.4 | Mosaic Diploid/Aneuploid/Haploid/Chaotic | 2 CL | 18, X |
| 16.1 | Uniformly Normal | None | - |
| 18.1 | Mosaic Diploid/Aneuploid | 1 CL | 11 |
| | | 2 MND | X, Y |
| 19.1 | Mosaic Diploid/Aneuploid/Polyploid | 3 CL | 11, 18, X |
| 19.2 | Mosaic Diploid/Polyploid/Aneuploid | 2 CG | 18, X |
| | | 1 MND | Ý |
| 22.1 | Mosaic Diploid/Aneuploid/Polyploid | 1 CL | 18 |
| | | 1 MND | X |
| 22.2 | Mosaic Diploid/Aneuploid/Polyploid | 1 CL | 18 |
| | | 2 CG | 1, 11 |
| | | 1 MND | X |
| 23.1 | Uniformly Normal | None | - |
| 24.1 | Mosaic Diploid/Polyploid | - | - |
| 25.1 | Mosaic Diploid/Polyploid/Aneuploid | 1 CL | 18 |
| 25.2 | Mosaic Diploid/Aneuploid/Polyploid | 1 CL | 18 |
| | | 1 MND | X |
| 26.1 | Mosaic | 3 CG | 11, X, Y |
| | Diploid/Aneuploid/Chaotic | 1 MND | 1 |
| 26.2 | Mosaic | 2 CL | 11, 18 |
| | Diploid/Polyploid/Aneuploid/Chaotic | 1 CG | X |
| TOTAL | - | 15 CL (40.5%) | |
| 21 embryos | | 13 CG (35.1%) | |
| | | 9 MND (24.4%) | |

Moreover, in Group II there were nine MND events, mostly affecting chromosome X, occurring in comparison to two in Group I embryos (Table 4.5). Once again, chromosome loss was the predominant mechanism leading to mosaic aneuploidy (Table 4.8), but in this group it involved mainly chromosome 18, while an additional copy of chromosome 11 was the most frequent gain. FISH artefacts (inconsistent results) affected 50/1143 (4.4%) of nuclei.

| Event Chromosome | Chromosome Loss | Chromosome Gain | Mitotic Non- Disjunction |
|---------------------|-----------------|-----------------|-----------------------------|
| 1 | 0 | 2 | 2 |
| 11 | 4 | 6 | 0 |
| 18 | 9 | 1 | 0 |
| X | 2 | 3 | 5 |
| Y | 0 | 1 | 2 |
| Total | 15 | 13 | 9 |

Table 4.8. Mechanisms of mosaicism by chromosome for Group II embryos

Of the nuclei showing inconsistent results, 34/76 (45%) were due to failure of hybridisation of the sub-telomeric probe for chromosome 18 (18q). The probe for the satellite II/III region of chromosome 1, exhibited the highest failure rate of the centromeric probes (14.5%) (Table 4.9). The underlying reason for the high rate of artefacts for the heterochromatic probe for chromosome 1 can be attributed to the fact that it was the only laboratory prepared probe. Group II embryos showed increased number of FISH artefacts (67%), termed '*inconsistent results*' (highlighted in red colour in Tables 4.3 and 4.6), compared to Group I (33%). This finding, however, might due to the difference in the number of cells analysed for each group.

| Table 4.9. Failure rates for each probe in the embryonic nuclei for bo | both Groups |
|--|-------------|
|--|-------------|

| Probe | lp | 11q | 18q | 1HET | 11CEP | 18CEP | Total |
|-----------------------|-------|------|-----|-------|-------|-------|-------|
| No. of FISH artefacts | 12 | 6 | 34 | 11 | 7 | 6 | 76 |
| Percentages | 15.8% | 7.9% | 45% | 14.5% | 9% | 7.9% | |

4.3 Discussion

During this study the levels and mechanisms leading to aneuploid mosaicism in human embryos were investigated. This was achieved by developing a triple colour FISH protocol carried out in three sequential rounds of hybridisation. The use of two probes per autosome and the dual use of a chromosome 18 centromere probe allowed the distinction between true single cell anomalies and artefacts, which showed up as inconsistent results. This study was able to accurately define the true extent of chromosomal mosaicism in human embryos without the use of statistical aids like previous studies have done. Of the 42 day 5 embryos, three only were uniformly diploid for the tested chromosomes, a single embryo was the product of a meiotic error and the remainder were mosaic. Furthermore, it was found that chromosome loss was the main mechanism leading to mosaic embryos.

4.3.1 Preliminary Work Assessment

Early work to develop interphase cytogenetics using FISH was hampered in part by the lack of access to a reliable panel of locus-specific DNA probes and as such was limited in scope (Griffin et al, 1991; Griffin et al, 1992). Methods of screening DNA clones for use as FISH probes can be labour intensive and time-consuming as many will be chimaeric (large BAC/YAC clones etc), map to another location (crosshybridize) or produce weak FISH signals and so need to be discarded. The plasmid clone for the 1het centromeric probe used in this study was found to be suitable, producing discrete, easily scored FISH signals in lymphocyte interphase nuclei. Fortunately, the increasing availability of a wide range of commercial probes has made the method of producing 'laboratory prepared' diagnostic probes largely redundant. These commercial probes have revolutionised the use of interphase cytogenetics particularly in clinical laboratories where FISH is now used routinely to complement standard karyotyping for a wide range of pre- and postnatal applications (Blennow et al, 1995; Knight and Flint, 2000; Quilter et al, 2001). In addition the introduction of multi-probe cocktails specifically designed for polar bodies and blastomere analysis have contributed to the increase in the number of groups carrying out PGD and related research. Finally, the introduction and subsequent general accessibility to subtelomeric probes for all chromosomes has greatly simplified the

strategies for FISH protocols. The majority of the probes used in this study were commercially available.

Whilst carrying out FISH in order to analyse chromosome constitutions several difficulties emerge such as scoring errors, cross-hybridisation, background fluorescence as well as low quality fluorescent signals (Griffin et al, 1994). Therefore, during this study several FISH protocols were carried out using different probe combinations and changing the stringency conditions (salt and formamide concentration, temperature and pH), on control male blood lymphocytes. The most efficient protocol was found to be a sequential FISH protocol involving chromosomes 1, 11, 18, X and Y. It has been postulated from Bielanska et al (2002b) that similar rates of mosaic imbalances can be detected from different probe combinations either for three or five chromosomes. The first three chromosomes were examined in both the first two rounds, using probes for different loci for each chromosome. Similar double loci analysis for just one chromosome was concluded as a preferable method of avoiding false monosomies whilst carrying out PGD for an euploidy (Magli et al, 2001). Furthermore, the use of two differentially labelled probes to detect a single chromosome increased the accuracy of detection, by reducing scoring errors and confirming that the mosaicism observed in previous preimplantation diagnosis cycles is not a FISH artefact owing to hybridisation failure or overlapping signals, even when just one cell differs and no cell resulting from a reciprocal non-disjunction event is found (Conn et al, 1999). This study was able to detect any failure of hybridisation or FISH artefacts since the three major chromosomes studied (1, 11 and 18) were analysed in two sequential rounds using differentially labelled probes at different loci. Where differences occurred they were defined as "inconsistent results". These inconsistent results have not been included in the interpretation of the results. Other studies have compared levels of aneuploidy present in embryos and only if that proportion was higher than the value obtained from the control lymphocytes, its presence was considered true (Ruangvutilert et al, 2000b). However, in this study such comparison was not needed and the genuine level of aneuploidy and mosaicism could be detected due to the safety net provided by the two probes per chromosome analysis.

As a three sequential round of FISH protocol was used in this study it was possible to assess the efficiency of each FISH experiment in each round. A control slide from male blood lymphocytes was used that was FISHed at the same time as the embryos and scored. 200 interphase nuclei were scored for all probes in all the rounds. This step enabled observation of whether the FISH technique worked efficiently without the presence of cross-hybridisation and background fluorescence and assessed the efficiency of a third-round of hybridisation. All three rounds showed high efficiencies when the same slide was FISHed sequentially (Table 4.2). However, the efficiency of the 3rd hybridisation decreased by 9% (to 88%) in comparison with the 1st round of FISH. This seems to be due to the decrease in the quality of the nuclei DNA and has been confirmed by other studies (Ruangvutilert et al, 2000a). Also, the image quality of the nuclei and fluorescent signals decreased in intensity during sequential hybridisations. Hence, during the denaturation step the slides were incubated in a 75°C waterbath containing 70% formamide solution instead of incubation on a hot plate in an oven. This step showed an increase the intensity of the signals and nuclei quality (data not shown).

4.3.2 Distinction between Artefacts and True Results

The use of two probes per autosome allowed this study to detect an error rate of 5% per nucleus and to exclude those artefacts from the analysed results. For both Groups a true chromosomal error in a total of 76 nuclei would have been missed if only one probe per chromosome had been used. Therefore, by adopting our chosen strategy we were able to detect the true levels of mosaicism for the three autosomes studied. The probe that showed the highest rate of failure of hybridisation on embryonic material was the sub-telomere probe for chromosome 18 (Table 4.9). Sub-telomeric probes generally have a lower hybridisation efficiency than probes that detect repeat sequences, but in this case the situation was possibly exacerbated because the probe was used in the second treatment round. However, in the lymphocyte control nuclei, the probes for chromosome 1, both for the sub-telomeric (used in the first round) and heterochromatic regions, showed the highest failure rates.

Many of the aneuploidies detected in our study were of a single chromosome and confined to a single cell. Other studies would have dismissed these aneuploidies as FISH artefacts due to their lack of probe backup. Low level aneuploidy is rarely considered to be a mosaicism phenomenon and it is usually classed as normal (Group I embryos in Table 1.2 from Munne and Weier, 1996). Bielanska and co-workers (2002a) postulated that minor aneuploid cell lines indicated that the aneuploid cells did not persist from early cleavage, but were formed at, or shortly before, blastulation, and suggest that most cleavage stage embryos with a high degree of mosaic aneuploidy do not complete preimplantation development. It is clear from CGH data on full analysis of individual blastomeres that single cell anomalies affecting one chromosome are a common feature at cleavage stages (Voullaire et al, 2000; Wells and Delhanty 2000). In these two small series, nine of the 24 embryos fully analysed showed such single cell anomalies. Furthermore, unlike that which has been observed in diploid-tetraploid mouse aggregation chimeras (James and West, 1994), recent studies of human blastocysts did not show evidence of preferential allocation of aneuploid cells into the trophectoderm lineage (Evsikov and Verlinsky, 1998; Magli et al, 2000)

4.3.3 Group I Embryos

The embryos that were left to grow in the basic media showed a decreased rate of growth. From the 21 embryos spread, the mean number of blastomeres present was 20, which was low considering that normal developing day 5 embryos should have more than 30-40 cells on day 5 of development. Hence, all embryos, apart from two, from Group I were considered arrested (less than 30 cells). The fact that almost all embryos were arrested in this group was due to the media they were grown in. Group I embryos were left into the medium which contains lactate and pyruvate which are considered to be adverse for healthy embryonic development after genome activation (Gardner and Sutherland, 1996; section 1.1.3).

The FISH results showed a high prevalence of mosaicism in this group of embryos (95%, 20/21 embryos). Veiga *et al* (1999) observed a lower rate of mosaicism (62.5%) in arrested embryos however, most mosaic embryos included a chaotic complement. The prevalent type of mosaicism was diploid/aneuploid mosaics (76%)

and the predominant mechanism that lead to mosaicism was chromosome loss (50%) which is in contrast with other studies (Munne, 2002) that showed that mitotic nondisjunction was the most common type of mosaicism. This is due to the fact that most studies rely on one probe per chromosome in order to analyse the highest number of chromosomes simultaneously, thus sacrificing the reliability of their results. Monosomies, when carrying out FISH using one probe per chromosome, are considered false based on the fact that it was failure of hybridisation or a FISH artefact (if they are found at low percentages). However, this diminishes the ability to determine the true values from the false ones. This study was able to confirm whether a monosomy was false or true and classified as an inconsistent result or as true monosomy respectively.

Some mosaic embryos containing aneuploid cell lines arose due to chromosome loss and gain such as embryo 5.4, which had a cell monosomic for chromosome 1 and a cell trisomic for chromosome 18 (Table 4.4). Therefore, two separate mechanisms lead to the occurrence of such mosaic embryos. Diploid/aneuploid mosaic embryos resulting from mitotic non-disjunction were only found in two embryos "4.1" and "12.1" for chromosomes 18 and X respectively (Table 4.4). In mosaic diploid/aneuploid embryos arising from MND, there was a reciprocal loss and gain of the same chromosome in different embryonic nuclei. Three embryos (14%) were classified as diploid/aneuploid/chaotic (8.1, 12.2 and 17.1). There was no apparent mechanism that can identify the reason of the chaotic lines. Although, in these embryos there was a high percentage of diploid nuclei (67-89%), the remaining cells were either aneuploid or chaotic. Veiga et al (1999) described an increased rate of chaotic complement of 37.5% in arrested embryos in comparison to the 0% observed in blastocysts. This can be partly explained by the ability of chaotic complements in embryos to block further development. Sandalinas and co-workers (2001) found that some chaotic mosaic embryos developed further but never more than 50 cells and were considered to be developmentally compromised. Moreover, it has been suggested that there is a strong developmental block at compaction of chromosomally abnormal embryos compared to normal embryos (difference was statistically significant) (Sandalinas et al, 2001).

4.3.4 Group II Embryos

None of Group II embryos were considered arrested since they contained more than thirty cells per embryo at the time of spreading. 90% (19/21) of the blastocysts analysed were mosaic and only 2 embryos were uniformly diploid for the chromosomes tested (Table 3.7). Studies where blastocysts were karyotyped also found increased levels of mosaicism (Clouston *et al*, 1997, 2002). Veiga *et al* (1999) also found 7/8 (87.5%) blastocysts analysed to be mosaic and Ruangvutilert *et al* (2000a) reported similar figures of mosaicism to this study in blastocysts (89.5%). Evsikov and Verlinsky (1998) also found high levels of mosaicism whilst performing FISH on blastocysts (86%), although they did not distinguish between normal and diploid mosaics. Ten out of 21 embryos (47%) were diploid/aneuploid mosaics (including haploid or polyploidy cell lines). Embryo "14.2" was a diploid/aneuploid/polyploid mosaic embryo, which was generated due to three different mechanisms, chromosome loss for chromosomes 1 and 18, chromosome gain for chromosome 11 and mitotic non-disjunction for chromosome X (Table 4.5).

Group II embryos showed a larger variety of cell lines within each embryo compared to Group I, which might have been attributed to the increased number of cells analysed (1123 over 401 cells in Group I) or the marked presence of polyploidy cell lines. 80% of the embryos analysed contained relatively high levels of tetraploidy ranging from 2-22%. This has been also been reported by Verlinsky and Evsikov (1998), Veiga *et al* (1999) and Ruangvutilert *et al* (2000a). Ruangvutilert *et al* (2000a) suggested that mosaic embryos with a tetraploid cell line might have more viability that those with other non-diploid cell lines. The increased number of cells present in Group II embryos was expected since embryos in this group were transferred to blastocyst medium. The blastocyst medium contains the required nutrients that an embryo needs after day 3 of development i.e. glucose.

Compared to studies carried out thus far, a relatively high percentage of triploid cells have been found during this study. For 4% (19/458) of the cells scored were found to

be triploid for the five chromosomes tested. Ruangvutilert *et al* (2000a) found only 0.3% (4/1272) of triploid cells present in the blastocyst group. However, similar findings of haploid cells were observed between this study and other studies of human blastocysts, since only 0.2% (1/458) of the cells were haploid. Triploid and haploid nuclei have been found more frequently in cleavage stage embryos.

4.3.5 Types of Mosaicism

Aneuploid mosaics arise from post-zygotic mitotic errors. In Group I almost all mosaic embryos were diploid/aneuploid mosaics. These embryos arose mostly due to post-zygotic chromosome loss, followed by chromosome gain, with only a few examples of mitotic non-disjunction. Other studies that found aneuploid mosaic embryos suggested that the main mechanism leading to mosaicism was mitotic nondisjunction (Munne et al, 1998c; 2003a). The basis of this observation lies in the fact that those studies did not include two probes (at a different locus) for each chromosome, thus accounting monosomy findings as failure of hybridisation or signals overlapping, hence, reducing the incidence of aneuploid mosaics due to chromosome loss. Although most studies are able to test a higher number of chromosomes, they are not able to be positive whether their findings are 100% accurate. By using two probes per chromosome at different loci in different rounds of FISH can accurately determine the chromosome abnormalities present and suggest a mechanism. Magli et al (2001) also came to a similar conclusion, when two probes were applied for chromosome 21 on a PGS case so that the misdiagnosis of normal embryos as monosomies were reduced. It has been hypothesised that the transition from the morula to the blastocyst stage is critical in terms of starting a negative selection against aneuploid cells (Evsikov and Verlinsky, 1998), since a high degree of mosaicism was observed up until the morula stage in comparison to that of blastocysts. However, this hypothesis was not shared with Sandalinas et al (2001), which showed that extended culture to blastocyst stage is not a reliable selection tool to screen against those chromosomally abnormal embryos that may survive after implantation. Monosomies were observed for both groups for all the chromosomes tested. Although studies by Sandalinas et al (2001) and Clouston et al (2002) retrieved low numbers or were unable to find monosomies respectively, in this study monosomies were present, especially in embryos grown in the basic medium. 15%

(29/188) of the cells scored in Group I were monosomic for all chromosomes tested (1, 11, 18 and X), hence, the increased incidence of chromosome loss was a mechanism for the aneuploid mosaics. But, in the blastocyst medium group only 3% (14/458) of the cells scored included monosomies.

The embryos grown in the blastocyst medium (Group II) showed an increased rate of tetraploid cells. It has been suggested that tetraploidy is a normal feature of the trophectoderm (Angell et al, 1987) and may be associated with invasion of the maternal deciduas (Drury et al, 1998). The significant increase of polyploidy cells, and especially tetraploidy, from day 4 to day 6 of development has been suggested to be, in human embryos as in other mammalian species, a hallmark of trophoblast differentiation (Bielanska et al, 2002a). Harper et al (1995) suggested that the underlying mechanism leading to tetraploid cells might be failure of cytokinesis after the chromosomes divide. Furthermore, endoreduplication might cause cells to become tetraploid by doubling of the chromosomes and failure to divide. Also, tetraploid cells might originate from cell fusion (Benkhalifa et al, 1993). Fusion of cells of different ploidies, such as 2N plus 4N and 4N plus 4N, may also explain the origins of 6N and 8N complements found in both Group I and II embryos. Similar findings of mixoploidy have been identified in 109 blastocysts analysed with FISH by Bielanska et al (2002c). It has been postulated that not all types of mosaicism and the proportion of abnormal cells have the same impact on embryo development (Sandalinas et al, 2001), and a high proportion of tetraploid cells may be detrimental in embryo development. In the study by Sandalinas et al (2001) 2n/4n mosaics with <38% abnormal cells developed 78% of the time to blastocyst stage compared with only 33% of those with >38% abnormal cells (the difference was shown to be statistically significant).

Slightly higher levels of triploid cells (in comparison to similar studies carried out so far) were interestingly observed in the Group II embryos. Seven out of the twenty one embryos analysed contained at least one triploid cell line. The origin of diploid/triploid cells is not clear, however the presence of an extra haploid set might be derived from an extra gamete, such as a second sperm (in a dispermy event where the second sperm remains unincorporated into the formation of the zygote) or a polar body (Mueller *et al*, 1993). Triploid cells were more often seen in cleavage-stage

embryos (Delhanty *et al*, 1997), nevertheless, in some instances triploid cells may persist until later in development as in cases of mosaic diploid/triploid have been reported postnatally (Edwards *et al*, 1994). Haploid cells were rarely found and it seems that they are less viable or less actively dividing in comparison to tetraploid cells. Delhanty *et al* (1997) suggested that they maybe associated with binucleate cell production with a meiotic type of segregation. Staessen and co-workers (1999) hypothesised that haploid cells might be due to incorporation of a polar body into the embryo. In this study only one embryo from each Group revealed a haploid cell line displaying an incidence of 0.1%. It has been suggested that 1N cells have a proliferative disadvantage among 2N cells (Harper *et al*, 1995; Bahce *et al*, 1999; Bielanska *et al*, 2002a). In contrast to diploid/triploid mosaics, diploid/haploid mosaicism has not been documented in fetal tissues, hence such embryos must become eliminated at, or shortly after, implantation.

Chaotic embryos were first described in cleavage-stage embryos (Harper et al, 1995; Delhanty et al, 1997). Three embryos were chaotic in Group I and eight in Group II; however, all 11/42 embryos included diploid cells as the major cell line. Evsikov and Verlinsky (1998) found three chaotic embryos out of 91 in their series, Ruangvutilert et al (2000a) observed five chaotic embryos out 40 after FISH analysis whereas Baart et al (2004), found only one chaotic embryo out of 22 analysed. Delhanty and Handyside (1995) proposed that the absence of cell cycle checkpoints might cause such chaotic chromosome constitutions. Cell cycle checkpoints were first identified in yeast and would normally protect cells from genetic damage by ensuring that each cycle phase is completed before the initiation of the next (Hartwell and Weinert, 1989). In mammalian embryos the absence of cell cycle checkpoints may relate more generally to the absence of embryonic gene transcription and dependence on maternal products inherited in the oocyte (Delhanty and Handyside, 1995). Hardy et al (1993) suggested that the presence of cells with nuclear abnormalities or highly abnormal chromosome complements may reflect both lack of co-ordination of the different processes of the cell cycle. Other studies have proposed that chaotic cell lines arise from a group of events such as a chromosome misalignment on a disorganised spindle in combination with a non-functional metaphase/anaphase checkpoint control (LeMarie-Adkins et al, 1997; Harrison et al, 2000). The abnormalities of the mitotic

spindle could be related to suboptimal *in vitro* culture environment (Pickering *et al*, 1990). Munne *et al* (2004) has suggested that chaotic mosaicism might occur due to differences in hormonal stimulation as well as imperfect culture conditions.

4.3.6 Mechanisms of Aneuploidy Mosaicism

Compared to other species, humans display a low fecundity. Of all human conceptions, only ~30% progress successfully to delivery (Hassold *et al*, 1986). To a large extent, embryonic death is caused by chromosomal abnormalities that are primarily the result of chromosomal errors during female gamete formation. Some errors will arise at the time of fertilisation but most are the recently discovered mitotic errors which take place during early cleavage division. These errors in cell divisions, namely non-disjunction and anaphase lagging lead to mosaicism and chaotic imbalances. Kalousek (2000) highlighted the significance of the chromosome centromere in the process of cell division. It was suggested that the centromere is involved in: i) sister chromatid pairing, ii) mitotic and meiotic spindle attachment, iii) chromosome movement, iv) cell cycle control i.e. cell cycle checkpoint control and v) marshalling of passenger proteins. This was taken into account in our study and five out of the nine probes used were centromeric probes.

In the current study chromosome loss was the predominant mechanism leading to mosaicism in both groups of embryos, being responsible for 50% of an euploid cells in Group I and 40.5% in Group II. Chromosome gain followed with 44%, in Group I and 35.1% in Group II. However only 6% of an euploid cells occurred due to mitotic non-disjunction in Group I, while four times this percentage arose by this mechanism in Group II. The differential involvement of chromosomes in MND (predominantly chromosome X in Group II which was not involved at all in Group I) is interesting but may well be due to chance.

Similarly, it is of interest that in Group I embryos, in which growth had slowed or arrested, chromosome 1 showed a high incidence of loss but in Group II embryos, which had continued dividing, chromosome 1 was not affected by loss at all. This might possibly be due to the fact that the presence of cells with monosomies of such a large chromosome would have a detrimental effect on development and had been selected against in the more rapidly dividing Group II embryos. Trisomy 18 caused by

chromosome gain was seldom found in either Group, which is a similar finding to that reported recently by Coonen *et al* (2004). The occurrence of monosomy 18 was shown to be high (especially in Group II embryos), indicating that chromosome 18 might be more prone to chromosome loss compared to the other autosomes and gonosomes tested.

The current study shows that chromosome loss is the most common mechanism that leads to mosaicism detected in human day 5 embryos. This reinforces data on day 3 embryos obtained earlier by our group using dual locus specific YAC and plasmid probe combinations for various autosomes (C. Conn & JDA Delhanty, unpublished observations). Chromosome loss is presumed to occur via anaphase lag, in this case during mitosis. Coonen et al (2004) concluded in their study on a much larger number of blastocysts that anaphase lagging is the major cause of chromosomal mosaicism. However, since they were using a single probe for each chromosome they were only able to count as valid abnormalities affecting at least two cells. In relation to the findings in these two studies, it is of considerable interest that an euploidy screening of cleavage stage embryos has shown that chromosome loss is more common than chromosome gain as a cause of constitutional aneuploidy arising during meiosis (Munné et al, 2004). This is in contrast with all previous findings from the same group which supported that mitotic non-disjunction was the most common type of mosaicism mechanism (Munne et al, 1994; 1995a; 1998c; Munne and Cohen 1998, Munne, 2002). However, all the studies above were limited to one probe per chromosome and were attributing abnormalities confined to low number of cells as signal overlap, probe failures, whereas in the recent study by Munne et al (2004) all samples were re-analysed with probes binding to a different locus.

In contrast to the chromosomal abnormalities seen in cleavage stage embryos, mosaic chromosome patterns observed in blastocysts are derived from mitotic division errors. A meiotic division error would render all embryonic cells chromosomally aneuploid. In this study only one embryo (from Group I) out of 42 rose due to a meiotic error. Embryo 4.1 was aneuploid mosaic and must have originated due to a non-disjunction event during meiosis. Coonen and co-workers (2004) also reported only one out of 299 blastocysts analysed by FISH had occurred from meiotic division errors. Other

studies have shown no or only a few blastocysts presenting with aneuploid cells only (Magli *et al*, 2000b; Ruangvutilert *et al*, 2000a; Bielanska *et al*, 2002a; Baart *et al*, 2004)

No difference in embryonic sex ratio between Group I and Group II was observed in this series. In Group I there were 12 female and 9 male embryos whereas in Group II there were 11 female and 10 male embryos (Tables 4.3 and 4.6). It has been shown in a recent study that more male than female babies have been born after blastocyst transfer (Menezzo *et al*, 1999), which might reflect a greater viability of male embryos after implantation. However, no other study has confirmed these results.

During this study the average maternal age was 34 years but still there was a high prevalence of mosaicism. The three embryos that were uniformly diploid for the tested chromosomes were donated from three women less than 30 years of age. Whereas six women whose age was above 38 donated nine embryos, which were either diploid/aneuploid mosaic or diploid/aneuploid/chaotic mosaic. While these observations may indicate a trend, clearly the numbers in our study are too small to enable analysis of a maternal age effect. However, a report of a study of large number of cleavage stage embryos from women in an older age group showed a significant association of mitotic aneuploidy with advanced maternal age (Munné et al, 2002). The association was particularly marked for the MND category. In that study chromosome loss was only a third as frequent as MND and consequently mitotic anaphase lag failed to show a significant increase with maternal age. The increased frequency of MND may be due to the age of the group studied but since only a single probe per chromosome was used a cut off point of 10% was used to avoid 'FISH error'. That is, embryos with fewer than 10% abnormal cells were considered 'normal', almost certainly leading to an underestimate of anomalies affecting a single cell. Possibly a study of a large number of embryos using two probes per chromosome might show a significant association of mitotic anaphase lag with increasing maternal age.

4.3.7 Conclusions and Future Work

In conclusion, during this project it was revealed that three rounds of sequential FISH can provide efficient results in human preimplantation embryos in order to study the effect of chromosomal mosaicism. High proportion of mosaicism was found during this study in both groups of embryos which were mostly affected by aneuploidy. The double-loci analysis of each chromosome in different rounds of FISH enabled the determination of true level of mosaicism in day 5 human embryos, by detecting the FISH artefacts. The major mechanism behind aneuploidy mosaicism for both groups of embryos was found to be chromosome loss, which was postulated to be due to anaphase lagging. These findings were found to be in agreement with similar studies which used different protocols (Coonen *et al*, 2004; Baart *et al*, 2004; Munner *et al*, 2004). Furthermore, the effects of the two different media used were observed and it was found that the nutrients in IVF media are very important to the development of human blastocysts.

Examination of a larger cohort of embryos would enable extensive information of the chromosomal mosaicism and would further allow statistical analysis to be performed in order to find statistical significance, if any, between the mechanisms of mosaicism i.e. chromosome loss or gain or mitotic non-disjunction. Analysis of all the chromosomes of all the blastomeres would reveal the true extent of mosaicism in day 5 human embryos. Such task could be carried out using CGH (Wells *et al*, 1999; Voullaire *et al*, 1999) or using micro-arrays (Schaeffer *et al*, 2004; Shaffer *et al*, 2004). CGH analysis of all the blastomeres on normally developing day 5 embryos or blastocysts has not been carried out and might demonstrate results that might explain whether mosaicism affects blastocysts more than day 3 embryos. Micro-arrays can provide comprehensive (genome-wide), high resolution, amenable to automation, rapid and sensitive detection. Moreover, molecular analysis of genes responsible for embryo genome activation could be performed on embryos.

CHAPTER 5

Analysis of the development of chromosome abnormalities in human embryos from day 3 to day 5 using CGH and FISH

5.1 Aims of the Study

The aim of this study was to assess the level of chromosomal abnormalities in human embryos at different times of development. A novel protocol was devised were the chromosomal status of the embryos was investigated at the cleavage stage and furthermore at the blastocyst stage. Frozen-thawed embryos were biopsied on day 3 and 1 or 2 cells were extracted for CGH analysis in order to examine the full chromosomal status of those blastomeres. The rest of the embryo was left to grow until the blastocyst stage (day 5) were it was spread on the whole to carry out sequential FISH analysis. The first combination of probes included the X/Y/18 which would allow confirmation of the sex when compared to the CGH results from the biopsies blastomeres as well as any abnormalities for the corresponding chromosomes. However, the second round involved probe combinations according to the CGH results i.e. any abnormalities seen in the CGH results would be tested on the spread blastomeres using FISH. This would allow the monitoring of abnormalities during the different stages of development.

5.2 Results

5.2.1 WGA Optimisation

The efficiency of the tubing technique was found to be 97% (section 2.2.2.2.3). Efficient tubing was confirmed by successful amplification, which in turn was revealed by a smear on the gel. Bands indicating amplification were observed corresponding to a range of fragment sizes of DNA of approximately 1550, 1200, 600 and 450 bp in length, while the average fragment size was approximately 600 bp in length (Figure 5.1). Positive and negative controls were always included in the tubing process, to ensure reliability of results. The frequency of amplification in negative controls was <5%, while amplification in positive controls was successful with a frequency of 98%. During optimisation of the WGA protocol two thermal cyclers were tested and the results were run on a 2% agarose gel (Figure 5.2). As shown in Figure 5.2B the Omnigene Thermal cycler proved to be superior compared to the Eppendorf since the smears from the samples were more intense and more similar to the genomic DNA samples revealing a better coverage of the genome. Furthermore,

the size of the smears was definitely improved when using the Omnigene, since the smears were expanded over the 1Kb Ladder.

Figure 5.1. DOP-PCR amplification results from single cells, clumps of 3-4 buccal cells and genomic DNA on an agarose gel.



In lane L the 1kb Ladder can be seen. Lanes 1-10 were amplified products from single buccal cells. Lanes 11-15 a clump of 3-4 buccal cells were run and in lanes 16-23 single blastomeres were amplified and analysed. Finally, in lanes 24, 25, 26, 27 and 28 DNA from frozen fibroblasts of known abnormality (trisomy 13, 18, 21 and 22 and triploidy XXX respectively) was run. Lanes 29 and 30 represent amplification results from normal genomic DNA.





A. In lanes 2-5 and 7 the amplified products from single buccal cells were run. In Lane 6 the amplified product from genomic DNA was analysed and in lane 1 the negative control can be viewed. B. Lanes 1, 2, 3 and 5 show the final amplified product from single buccal cells. In lanes 4 and 6 the negative control and genomic DNA was run. In both Figures, A and B, the lane L represents the 1kb ladder.

5.2.2 CGH Optimisation

CGH analysis of both control lymphocytes and trisomic cells was carried out to optimise the protocol. All control and embryo CGH experiments were carried out on control male lymphocyte slides. Figure 5.3 shows an initial CGH experiment carried out on DOP-PCR amplified genomic DNA (section 2.2.2.1) between a normal male and female individual. The interpretation of 8-12 metaphases was performed for each sample. In Figure 4.3 the CGH software was able to distinguish between the male and female status of the control (red) and test (green) samples. Five different chromosomally abnormal fibroblast cell cultures (trisomy 13, 18, 21, and 22 and triploidy XXX) were made available for this study acting as positive controls in the identification of an euploidy in single cells. Figure 5.1 illustrates the DOP-PCR results from the amplification of the trisomic and triploidy samples. Aneuploidy was correctly identified in single cells with a success rate of 96%, while no false positives were recorded. Figures 5.4, 5.5, 5.6, 5.7 and 5.8 show the CGH results which acted as positive controls for trisomy 13, 18, 21, 22 and triploidy XXX respectively once the CGH protocol was optimised. In all positive control experiments, the CGH interpretation was able to distinguish the type of abnormality as well as the sex of each trisomic line. However, when the triploid cell line was tested against a single male buccal cell, the CGH was not able to distinguish the ploidy status (Figure 5.8). This result was expected, since it is known that the CGH technique is not able to detect differences in the ploidy status of a tissue (Kallioniemi et al, 1994).

Thirty single cells (including buccal cells and blastomeres) were analysed by CGH once the protocol was optimised. 93.3% (28/30) gave analysable results after CGH and in 83.3% (25/30) the difference in the sex chromosomes was distinguished. In 10% (2/30) the fluorescence was of low intensity and with excess of background fluorescence indicating that the single cell was either anucleate or degenerating during isolation.

Figure 5.3. Illustration of a control CGH experiment between male genomic DNA (red) acting as control against female genomic DNA (green) acting as test



For each CGH experiment 8-12 metaphases were captured, karyotyped, analysed individually and then interpreted cumulatively. A. Capturing of metaphase spread. B. In the karyotype of the captured metaphase the X chromosome has a distinct green fluorescence indicating excess in the test sample and the Y chromosome shows red fluorescence indicating deficiency in the test sample. C. This shows the cumulative analysis of ten metaphases, which was the basis of the interpretation. There was a shift in fluorescence only towards the green (excess) in chromosome X and towards the red (deficiency) in chromosome Y showing that the test sample was a normal female

245

Figure 5.4. Illustration of a positive control CGH experiment between male single cell (red) acting as control against DNA trisomic for 13 (green) acting as test





For this CGH experiment 9 metaphases were captured, karyotyped, analysed individually and then interpreted cumulatively. A. Capturing of a metaphase spread. B. In the karyotype of the captured metaphase chromosome 13 has a distinct green fluorescence indicating excess in the test sample. C. This shows the cumulative analysis of the nine metaphases, which was the basis of the interpretation. There was a shift in fluorescence only towards the green in chromosome 13 showing that the test sample was male, trisomic for chromosome 13.

Figure 5.5. Illustration of a positive control CGH experiment between male single cell (red) acting as control against DNA trisomic for 18 (green) acting as test





For this CGH experiment 11 metaphases were captured, karyotyped, analysed individually and then interpreted cumulatively. A. Capturing of a metaphase spread. B. In the karyotype of the captured metaphase chromosomes 18 and X have a distinct green fluorescence indicating excess and red fluorescence in the Y chromosome indicating deficiency in the test sample. C. The CGH interpretation successfully detected that the test sample was female, trisomic for chromosome 18.

Figure 5.6. Illustration of a positive control CGH experiment between female single cell (red) acting as control against DNA trisomic for 21 (green) acting as test



For this CGH experiment 11 metaphases were captured, karyotyped, analysed individually and then interpreted cumulatively. A. Capturing of a metaphase spread. B. In the karyotype of the captured metaphase chromosomes 21 and Y have a distinct green fluorescence indicating excess and red fluorescence in the X chromosome indicating deficiency in the test sample. C. The CGH interpretation successfully detected that the test sample was male, trisomic for chromosome 21.

248

Figure 5.7. Illustration of a positive control CGH experiment between male single cell (red) acting as control against DNA trisomic for 22 (green) acting as test



For this CGH experiment 10 metaphases were captured, karyotyped, analysed individually and then interpreted cumulatively. A. Capturing of a metaphase spread. B. In the karyotype of the captured metaphase chromosomes 22 has a distinct green fluorescence indicating excess. C. The CGH interpretation successfully detected that the test sample was male, trisomic for chromosome 22.

5,2,3 Passa Optimization

Each FIGH expansions we have a cover it main type phony in above when empred matching In order to present efficiency of works to be finition in the minimatol mangle (bash).

Figure 5.8. Illustration of a positive control CGH experiment between male single cell DNA (red) acting as control against triploid DNA (green) acting as test



For this CGH experiment 12 metaphases were captured, karyotyped, analysed individually and then interpreted cumulatively. A. Capturing of a metaphase spread. B. In the karyotype of the captured metaphase chromosome X has a distinct green fluorescence indicating excess and red fluorescence on chromosome Y indicating deficiency. C. The CGH interpretation successfully detected that the test sample is male, however, was unable to detect the triploidy status.

5.2.3 FISH Optimisation

Each FISH experiment included a control male lymphocyte slide with mapped nuclei in order to assess efficiency of probe hybridisation in the sequential rounds. Overall,
90% (range 87-97) of the control nuclei showed normal signals for all 14 probes used (Table 5.1). Sub-telomeric probes for chromosomes 1p and 16q showed the lowest hybridisation efficiencies of 89% and 88% respectively; whilst the α -satellite probes for chromosomes 16cep and Ycep demonstrated the highest hybridisation efficiencies of 94% and 93% respectively. The difference in the efficiency of the Xcep and Ycep probe (used in the 1st and 2nd round of FISH) was not statistically significant (p<0.05) and can be attributed to the fact that FISH efficiency decreases with sequential rounds of hybridisation due to the degeneration of the DNA. Furthermore, the \geq 5% difference between the first round overall efficiency (95%) and the various second round efficiencies has been well documented (Conn *et al*, 1998; Munne *et al*, 1998; Ruanvutilert *et al*, 2000a and b).

Table 5.1. Probe efficiencies scored in 200 interphase nuclei of each control slide whilst carrying 2-round FISH

| Probe Combinations | Efficiency per probe (%) | Overall Efficiency (%) |
|--------------------|--------------------------|------------------------|
| Xcep/Ycep/18cep | 97% / 96% / 97% | 95% |
| 9cep/16cep/22LSI* | 92% / 94% / 91% | 90% |
| 10cep/14q* | 90% / 90% | 88% |
| 3cep/11cep/13LSI | 92% / 92% / 89% | 90% |
| 1p/1q/16cep* | 89% / 90% / 93% | 87% |
| 3cep/6cep/18q* | 95% / 91% / 90% | 90% |
| Ycep/4cep* | 94% / 92% | 90% |
| Xcep/Ycep/16q* | 92% / 93% / 88% | 87% |

* Carried out in the second round

5.2.4 CGH Analysis of Embryos

Thirteen couples donated a total of 37 embryos for this research project (Appendix Table 8.1). The overall mean maternal age was 33.1 (range 28-39) years.





One embryo was not included since it failed to provide any results during either CGH or FISH, which was not unexpected due to its low morphology grade (Graph 5.1). A total of fifty four blastomeres were biopsied on day 3 for CGH analysis and fortyeight (88.9%) provided an interpretable result. There are a variety of factors which may account for failure to provide a result. Blastomere 3.1b failed to give hybridisation from the test DNA, while the control DNA hybridised successfully. This could indicate either an anucleate blastomere (as visibility of the nucleus was not possible for all the blastomeres biopsied), or that the cell was mistaken for an anuclear fragment. Moreover, a poor slide preparation or premature lysis of the blastomere can be responsible. Apart from the non-specific fluorescence and the granular hybridisation effect (meaning that although hybridisation was successful it presents with a granular effect due to poor chromosome quality, failing to provide analysable fluorescence), the strength of the counterstain banding, essential for the identification of chromosomes, would also be negatively affected by this factor. These problems accounted for lack of analysable results for blastomeres 1.1a, 1.2b, 2.1b, 2.2b and 7.2b (as well as 3.1b mentioned above).

A total of 48 blastomeres obtained for this study provided good analysable results. All embryos were labelled in green and the control male DNA was labelled in red. Hybridisation that presented with no granulation, dynamic fluorescence (strength and consistency of fluorescent signals) was considered to be of high-quality and good chromosome morphology allowed strong counterstain banding essential for the analysis. If the sex chromosomes could be determined confidently this provided an

internal positive control regarding the efficiency and reliability of the CGH technique. All biopsied cells from each embryo were consistent with regard to the sex chromosome pattern as confirmed by FISH analysis. With regard to the autosomes, the literature suggests the exclusion for an analysis of certain regions of the karyotype as they have been proven to show variation in the profile, (Kallioniemi *et al*, 1994) and this was also taken into consideration during this study. Moreover, deletions or amplifications concerning small subtelomeric regions were found extremely difficult to detect and interpret, as reported by other groups (Kallionemi *et al*, 1994). Consequently abnormalities involving distal breakpoints could not be confidently detected.

Six out of 36 embryos failed to grow further to day 5 and degenerated, thus no cells were available for spreading (4.1, 4.3, 4.4, 13.1, 14.1, 14.3). However, all of these embryos showed, in at least one biopsied cell, analysable CGH results (Table 5.2).

Table 5.2. CGH results from six embryos that did not have blastomeres available on day 5 for FISH analysis. The CGH results are presented as excess of GR (green) or RD (red) fluorescence in part of the chromosome or in the whole chromosome.

| Embryo No. | No. of cells prior to biopsy | No. of cells biopsied | CGH Result | Interpretation |
|---------------|------------------------------------|-----------------------------|--|---|
| 4.1 | 5 cells | 1 | Normal | rev ish XY |
| 4.3 | 7 cells | $1 (+1)^{a}$ | Normal | rev ish XY |
| 4.4 | 6 cells | 1 (+1) ^a | Excess GR in whole of X | rev ish XY, enh(X) |
| 13.1 | 5 cells | 1 | Excess GR: 5p15.3-q11.2 & 5q14-q34 | rev ish XY, enh(5) |
| 14.1 | 6 cells | 2 | Excess GR: whole of X, 19q12-q13.4 ^b Excess RD: whole of Y | rev ish XX |
| | | | Excess GR: whole of 2, 4, X, 9q (whole arm) Excess RD: whole of 1, 16, 21 and Y | rev ish XX, enh(2,4, 9qter), dim(1, 16, 21) |
| 14.3 | 5 cells | 1 | Excess GR: whole of X Excess RD: whole of Y | rev ish XX |

^aThe other cell lysed during the biopsy procedure

^b Excluded from analysis since it was considered to be an artefact

5.2.5 FISH Analysis of Embryos

FISH was able to achieve higher efficiencies of 93% overall (out of the thirty embryos), since 343/359 blastomeres from day 5 were analysable. The 7% of blastomeres that failed to return a result included blastomeres lost whilst spreading and blastomeres which could not provide an analysable result either on the first or second round of FISH. The first round of FISH was carried out using the probe cocktail for sex determination. This was performed in order to act as an internal control between the two techniques. FISH and CGH were in agreement regarding the sex in all of the embryos.

5.2.6 Interpretation of CGH and FISH Results

A total of thirty embryos (41 biopsied blastomeres) provided results from both techniques (Graph 5.2). Only three embryos (10%) showed normal results from at least one cell after CGH and normal results for all the blastomeres analysed by FISH for the chromosomes tested. The rest of the embryos demonstrated various levels of abnormality and mosaicism. Table 5.3 displays the embryos with completely normal CGH results whilst Table 5.4 shows the embryos where CGH revealed at least one cell with an abnormal karyotype.



Graph 5.2. Cumulative analysis results from 30 embryos (41 blastomeres)

CGH showed normal chromosome complements in thirteen embryos from which 18 blastomeres were biopsied (termed Group 1) (Table 5.3; Figure 5.9).

| Embryo | Cell | CGH Result on day 3 | | FISH Result on day 5 ^b | Interpretation |
|--------|------|---|-------|--|--|
| No. | | | Cells | Results (no. of cells) | |
| 1.1 | a | No result | 13 | Dip (7) / +X,+Y (2) / +22(2) / -18(2) | Mosaic Diploid/Aneuploid |
| | b | rev ish XY | | | |
| 1.3 | a | rev ish XY | 2 | Dip (1) / +X,+Y(1) | Mosaic Diploid/Aneuploid |
| 2.1 | a | rev ish XY | 4 | Dip (4) | Uniformly normal |
| | b | No result | | | |
| 6.2 | a | rev ish XY | 5 | Dip(3) / -22(2) | Mosaic Diploid/Aneuploid |
| | b | rev ish XY | | | |
| 7.1 | a | rev ish XX | 8 | Dip (3) / -18(2) / +18(1) / -22(1) / -16,-18(1) | Mosaic Aneuploid/Diploid |
| 9.1 | a | rev ish XY, enh(Xp11.2-q22) ^a | 7 | Dip (7) | Uniformly normal |
| 9.2 | a | rev ish XY, dim(Yq12) ^a | 31 | Dip (25)/-18(1)/-X(1)/-16,+22(1)/chaotic(3) | Mosaic Diploid/Aneuploid/Chaotic |
| 9.4 | a | rev ish XX, dim(11q23-q25) ^a | 6 | Dip (3) / tet(2) / chaotic(1) | Mosaic Diploid/Polyploid/Chaotic |
| 9.5 | a | rev ish XY | 11 | Dip(7) / -X(3) / chaotic(1) | Mosaic Diploid/Aneuploid/Chaotic |
| 11.1 | a | rev ish XY, enh(Yp11.3-11.2) ^a | 10 | Dip (10) | Uniformly normal |
| 12.1 | a | rev ish XX | 3 | Dip(1) / chaotic(2) | Mosaic Diploid/Chaotic |
| 13.2 | a | rev ish XX, dim(19p13.3-p13.2) ^a | 15 | Dip(7) / tet(1) / trip(1) / -9,+22(1) / chaotic(5) | Mosaic Diploid/Polyploid/Aneuploid/Chaotic |
| | b | rev ish XX | | | |
| 13.3 | a | rev ish XX | 35 | Dip(28) / trip(2) / +X(2) / chaotic(3) | Mosaic Diploid/ Polyploid/Aneuploid/ Chaotic |
| | b | rev ish XX | | | |

Table 5.3. Results from embryos showing normal CGH findings (Group 1) followed by sequential FISH analysis

^aThese findings were considered as artefacts. dip = diploid, tet = tetraploid, trip = triploidy. - indicates loss of chromosome and + indicates gain of chromosome e.g. +18 is trisomy 18 or -1 is monosomy 1.

^bThe 1st round of FISH was always performed with the X/Y/18 probe cocktail. The 2nd round was performed with the 9/16/22 probe combination

Two blastomeres (11%) failed to give interpretable results (1.1a and 2.1b). In the case where two cells were available (embryos 6.2, 13.1 and 13.3) for analysis, the sex was in agreement between the two cells. Furthermore, in 5 blastomeres (9.1a, 9.2a, 9.4a, 11.1a and 13.2a) there were some cases of excess of green or red fluorescence in small subtelomeric regions, which were considered as artefacts due to their position, which is an area characterised by repetitive DNA sequences and has been described as difficult to interpret via CGH (Kallioniemi et al, 1994). In three embryos (2.1, 9.1 and 11.1) the CGH and FISH results were normal and diploid respectively for the chromosomes tested (Figure 5.9). Ten embryos were mosaic with various levels of abnormalities. The probe combination for all of these embryos was Xcep/Ycep/18cep for the first round and 9cep/16cep/22LSI for the second round. FISH results regarding the gonosomes were concordant with the CGH results. Only two embryos were considered to be blastocysts (>30 cells) on day 5, whilst the rest were arrested. For the embryos that showed a normal karyotype on day 3, on day 5 in total 150 cells were spread and 106 (70.6%) were diploid for the chromosomes tested. However, six out of the ten mosaic embryos (60%) contained cells with chaotic chromosome complements including nullisomies, tetrasomies and cells with more that three abnormalities present in different chromosomes. Embryos 9.4, 13.2 and 13.3 contained polyploid cells (4% overall), either triploidy or tetraploidy.

Figure 5.9. Illustration of a control CGH experiment between a male single buccal cell (red) acting as control against blastomere 7.1a (in green)



For each single cell CGH experiment 10 metaphases were captured, karyotyped, analysed individually and then interpreted cumulatively. A. Captured metaphase spread. B. In the karyotype of the captured metaphase the X chromosome has a distinct green fluorescence indicating excess in the test sample and the Y chromosome shows red fluorescence indicating deficiency in the test sample. C. This shows the cumulative analysis of ten metaphases, which is the basis of interpretation. There was a shift in fluorescence only towards the green in chromosome X and towards the red in chromosome Y showing that a the test sample (blastomere 7.1a) was a normal female

E1511 and prix up dry 1 wednesses, in entiry prix 2, 5 (for a unitypes show of our left XI and prix the Pinis meeting dynamic only wells were related in the prix and the pinis dynamic of the second prix and pairs how ever, where the set is although the solution of pinis dynamics. Table 3 ¹¹ is analyzed 2.3 the COPI distinction.

Table 5.4 displays the embryos which showed an abnormal finding in at least one biopsied cell (Group 2). Therefore, the FISH combinations were tailored according to the abnormalities the CGH revealed. From these 17 embryos 28 blastomeres were biopsied and 3 (10.7%) were not analysable. In four embryos one cell displayed a normal karyotype whereas the other was abnormal (4.2a, 7.3b, 8.1a and 9.6b). All chromosomes where shown to be affected either by partial or whole duplication or deletion. Chromosomes 1(x4), 16(x5) and 22(x4) demonstrated the highest incidence of chromosome deletion (whole or partial). Chromosomes 1(x4), 2(x3) and Y(x5)revealed a higher rate of chromosome duplication (whole or partial). Chromosomes 13 and 15 showed three whole chromosome deletions each, however, no duplication, partial or whole, were observed. Interestingly, chromosome 1 was involved in three instances of partial chromosome duplication or deletion at the locus of 1p36.1 (Figure 5.10 - bold letters in legends). This might be due to a fragile site at that specific location. Similarly, on chromosome 2 a possible fragile site between the region 2q21q31 was found (Figures 5.10, 4.11 and 4.12 – bold letters in legends). All embryos were mosaic or completely chaotic at the time of spreading on day 5 and only two were considered to be blastocysts (4.2 and 7.3). Furthermore, from 183 cells that were spread on day 5, only 68 (37.1%) were diploid for the chromosomes tested. In total, 12/17 (70.5%) embryos contained at least one cell with a chaotic complement, of which four embryos (23.5%) were completely chaotic (Table 5.4). Three embryos (17.6%) completely lacked cells diploid for the chromosomes tested and all were mosaic aneuploid or mosaic aneuploid/chaotic.

In four embryos it was demonstrated that one of two biopsied cells was abnormal whereas the other blastomere may carry a normal chromosome complement (4.2, 7.3, 8.1 and 9.6; Figures 5.12 and 5.13). The FISH results for such embryos varied with two embryos showing that diploid cells were the predominant cell line (7.3 and 9.6) and two embryos having principally aneuploid cell lines (4.2 and 8.1). In 15 out of the 17 embryos the abnormality seen in a biopsied cell after CGH was confirmed during FISH analysis on day 5 embryos. In embryo 1.2, CGH analysis showed *rev ish XY dim(22)* and the FISH results showed cells with tetrasomy 22 and monosomy 22 indicating reciprocal loss and gain however, was classed as chaotic due to additional complex abnormalities by FISH (Appendix Table 8.1). In embryo 2.2 the CGH result

Chapter 5 – Analysis of the development of chromosome abnormalities in human embryos from day 3 to day 5 using CGH and FISH

was rev ish XY, dim(18) and the FISH results also revealed chromosome loss since monosomy and nullisomy 18 were observed. Interestingly, in embryo 8.2 from one biopsied cell the karyotype rev ish XX, enh(9 and 16) was seen, whereas during FISH analysis the chaotic cells were monosomic and/or nullisomic for chromosome 16 indicative of reciprocal loss and gain in the embryo (Appendix Table 8.1). In an attempt to see whether the phenomenon of chromosome breakage can be observed by FISH later in development, in embryo 9.7 where cell 9.7a displayed the following karyotype: rev ish XY, enh(1p36.3-q21), dim(1q31-q44), the chosen FISH probe combination was 1p/1q/16cep (Table 5.4). Two cells showed whole chromosome 1 loss, one cell revealed partial loss of the 1p arm and gain for the 1q arm and the one other cell showed partial loss of the *lq* arm and gain for the *lp* arm. It can be argued that the numbers are small and FISH artefacts can be affecting the results, however, both probes are sub-telomeric locus specific probes and not repetitive. In embryo 12.2 the CGH results hinted of chromosome breakage at the short arm at chromosome 4 and FISH was able to detect the partial chromosome loss in 2/12 cells which were also classed as chaotic.

| Embryo | Cell | CGH Result on day 3 | FISH Result on day 5 ^b | | Interpretation |
|--------|------|---|-----------------------------------|--|---------------------------|
| No. | | | Cells | Results (no. of cells) | |
| 1.2 | a | rev ish XY, dim(22) | 2 | Chaotic(2) | Chaotic |
| 2.2 | a | rev ish XY, dim(18) | 3 | Chaotic(3) | Chaotic |
| | b | No result | | | |
| 3.1 | a | rev ish XY, enh(5pter, 9qter, 17), dim(4,19) | 2 | Chaotic(2) | Chaotic |
| | b | No result | | | |
| 4.2 | a | rev ish XX | 33 | Dip(9) / -X(8) / -18(5) / -18,-22(2) / -X,-18(2) / | Mosaic Aneuploid/Diploid/ |
| | b | rev ish XX, enh(1, 22), dim(16pter, 18) | | +18,+16(1) / trip(1) / chaotic(5) | Polyploid/Chaotic |
| 6.3 | a | rev ish XY, enh(6p25-p21.1) ^a | 8 | Dip(2) / +X(2) / -16,-X(1) / -X,+22(1) / | Mosaic Aneuploid/Diploid |
| | b | rev ish XY, enh(Y) | | -X(1) / -22(1) | |
| 7.2 | a | rev ish XYY, enh(1p36.2-36.1, 2q31-p25, 5, | 3 | 3cep/11cep/13LSI ^b | Chaotic |
| | | 7, 8, 18, 19, 20, 21 and Y), dim(1p31-q44, 2q32-q37, 3, 6, 11, 13, 14, 15) | | Chaotic(3) | |
| | b | No result | 1 | | |

Table 5.4. Results from embryos showing abnormal CGH findings (Group 2) followed by sequential FISH analysis.

^aThese findings were considered as artefacts.

^bThe 1st round of FISH was always performed with the X/Y/18 probe cocktail. The 2nd round was performed with the 9/16/22 probe combination unless stated in bold and underlined combinations.

dip = diploid, **tet** = tetraploid, **trip** = triploidy. The - indicates loss of chromosome and + indicates gain of chromosome e.g. +18 is trisomy 18 or -1 is monosomy 1.

| Embryo | Cell | CGH Result on day 3 | | FISH Result on day 5 ^b | Interpretation |
|--------|------|--|-------|---|--|
| No. | | | Cells | Results (no. of cells) | |
| 7.3 | a | rev ish XY, dim(22q11.1-q13) | 30 | Dip(13) / +22(5) / -16, -22(2) / -9(1) / tet(1) / | Mosaic Diploid/Aneuploid/ |
| | b | rev ish XY | | chaotic (8) | Polyploid/Chaotic |
| 8.1 | a | rev ish XX | 4 | <u>10cep/14q^b</u> | Mosaic Aneuploid/Chaotic |
| | b | rev ish XX, enh(4, 6qter, 12p11.2-q24.3, 14q21-q32) dim(2q31-q37, 10) | | -10(2) / -10,+14qter(1) / chaotic(1) | |
| 8.2 | a | rev ish XX, enh(9 and 16) | 10 | Dip(6) / chaotic(4) | Mosaic Diploid/Chaotic |
| 9.3 | а | rev ish XY, dim(15q15-q26, 16) | 4 | Dip(1) / -16, -16(2) / -16,+22(2) | Mosaic Aneuploid/Diploid |
| 9.6 | a | rev ish XX, enh(2q21-q33, 3q11.1-q25), dim(1p36.1-p31, 16, 19 and 22) | 23 | Dip(13) / -16,-22(3) / tet(2) / -22(2) / -18(1) / +X(1) / chaotic(1) | Mosaic Diploid/Aneuploid/ Polyploid/Chaotic |
| | b | rev ish XX | | | |
| 9.7 | a | rev ish XY, enh(1p36.3-q21), dim(1q31-q44) | 5 | <u>1p/1q/16cep^b</u> | Mosaic Aneuploid/Chaotic |
| | b | rev ish XY, dim(1) | | -1qter,-1pter,+X(1) / +1qter,-1pter,+Y(1) / | |
| | | | | +1qter,-1pter(1) / -16,-X(1) / chaotic(1) | |

^aThese findings were considered as artefacts.

^bThe 1st round of FISH was always performed with the X/Y/18 probe cocktail. The 2nd round was performed with the 9/16/22 probe combination unless stated in bold and underlined combinations.

dip = diploid, **tet** = tetraploid, **trip** = triploidy. The - indicates loss of chromosome and + indicates gain of chromosome e.g. +18 is trisomy 18 or -1 is monosomy 1.

| Embryo | Cell | CGH Result | | FISH Result (day 5) ^b | Interpretation |
|--------|------|---|-------|--|---------------------------------------|
| No. | | (day 3) | Cells | Results | |
| 9.8 | a | rev ish XY, enh(17p13-q11, 18p11.3-q11.1), dim(3p26-p14) | 17 | <u>3cep/6cep/18cep^b</u> Dip(11) / +6(3) / -6(1) / -18,+Y(1) / -X,+Y(1) | Mosaic Diploid/Aneuploid |
| | b | rev ish XY, enh(6) | | | |
| 10.1 | а | rev ish XX, enh(20) | 19 | Dip(6) / +X,+X(6) / +X(3) / -X(1) /-22(1) / tet(1) / trip(1) / | Mosaic Aneuploid/Diploid/Polyploid |
| 12.2 | а | rev ish XY, enh(Y), dim(4pter) | 12 | <u>Ycep/4cep^b</u> Dip(6) / -18(2) / -Y(2) / chaotic(2) | Mosaic Diploid/Aneuploid/Chaotic |
| 12.3 | a | rev ish XY, enh(Yq11.1-q12), dim(16q21-q24) | 2 | <u>Xcep/Ycep/16q^b</u> -16,-18(1) / -16(1) | Mosaic Aneuploid |
| 14.2 | a | rev ish XX, enh(1, 10, 16qter), dim(8, 13, 21, 22) | 6 | Dip(1) / -16(1) / -16,-22(1) / +16,+22(1) / chaotic(2) | Mosaic Aneuploid/Diploid/Chaotic |
| | b | rev ish XX, enh(2,6), dim(9qter,13, 15, 16qter, 17) | | | |

^aThese findings were considered as artefacts.

^bThe 1st round of FISH was always performed with the X/Y/18 probe cocktail. The 2nd round was performed with the 9/16/22 probe combination unless stated in bold and underlined combinations.

dip = diploid, **tet** = tetraploid, **trip** = triploidy. The - indicates loss of chromosome and + indicates gain of chromosome e.g. +18 is trisomy 18 or -1 is monosomy 1.

Chapter 5 – Analysis of the development of chromosome abnormalities in human embryos from day 3 to day 5 using CGH and FISH



Figure 5.10. CGH and FISH analysis results from embryo 7.2

A. Interpretation of CGH experiment showing the cumulative analysis of ten metaphases from blastomere 7.2a. The test sample (blastomere 7.2a) is rev ish XYY, enh(1p36.2-36.1, **2p25-q31**, 5, 7, 8, 18, 19, 20, 21 and Y), dim(**1p31-q44**, **2q32-q37**, 3, 6, 11, 13, 14, 15). The blue arrow indicates the possible fragile site for chromosome 1 and the red arrow the fragile site for chromosome 2.

B. Results from the two sequential rounds of FISH in a cell from embryo 7.2. In B1 the cell was subjected to the X(green) / Y(red) / 18(aqua) probe cocktail where it showed female (XX) and monosomy 18. In B2 the cell was subjected to the 3(orange) / 11(aqua) / 13(green) probe combination which showed monosomy 3, nullisomy 11 and trisomy 13

Overall, the FISH results confirmed the chromosome loss events for chromosomes 3 and 11 and revealed reciprocal loss and gain (MND event) for chromosomes 13, 18 and Y.



Figure 5.10. CGH and FISH analysis results from embryo 7.2

A. Interpretation of CGH experiment showing the cumulative analysis of ten metaphases from blastomere 7.2a. The test sample (blastomere 7.2a) is rev ish XYY, enh(1p36.2-36.1, <u>2p25-q31</u>, 5, 7, 8, 18, 19, 20, 21 and Y), dim(<u>1p31-q44</u>, <u>2q32-q37</u>, 3, 6, 11, 13, 14, 15). The blue arrow indicates the possible fragile site for chromosome 1 and the red arrow the fragile site for chromosome 2.

B. Results from the two sequential rounds of FISH in a cell from embryo 7.2. In B1 the cell was subjected to the X(green) / Y(red) / 18(aqua) probe cocktail where it showed female (XX) and monosomy 18. In B2 the cell was subjected to the 3(orange) / 11(aqua) / 13(green) probe combination which showed monosomy 3, nullisomy 11 and trisomy 13

Overall, the FISH results confirmed the chromosome loss events for chromosomes 3 and 11 and revealed reciprocal loss and gain (MND event) for chromosomes 13, 18 and Y.



Figure 5.11. CGH and FISH analysis results from embryo 8.1

A. Interpretation of CGH experiment showing the cumulative analysis of ten metaphases from blastomere 8.1b. The test sample (blastomere 8.1b) was rev ish XX, enh(4, 6qter, 12p11.2-q24.3, 14q21-q32) dim(2q31-q37, 10). The red arrow indicates the possible fragile site for chromosome 2.

B. Results from the two sequential rounds of FISH in a cell from embryo 8.1. In B1 the cell was subjected to the X(green) / Y(red) / 18(aqua) probe cocktail where it showed female (XX) and tetrasomy 18. In B2 the cell was subjected to the 10(green) / 14(orange) probe combination which showed monosomy 10 and trisomy 14.

Overall, the FISH results confirmed the female status of the embryo, the whole chromosome loss event for chromosome 10 and the partial chromosome gain for the telomere of chromosome 14 (14qter).





A. Interpretation of CGH experiment showing the cumulative analysis of eight metaphases. There is a shift in fluorescence towards the green in chromosome X and towards the red in chromosome Y showing that a the test sample (blastomere 9.6a) was rev ish XX, enh(2q31-q33, 3p25-p21), dim(1p36.1-p31, 16, 19 and 22). The red arrow indicates the possible fragile site for chromosome 2.

B. Results from the two sequential rounds of FISH in a cell from embryo 9.6. In B1 the cell was subjected to the X(green) / Y(red) / 18(aqua) probe cocktail where it showed female (XX) and diploid for chromosome 18. In B2 the cell was subjected to the 9(green) / 16(aqua) / 22(orange) probe combination which showed monosomy for chromosomes 16 and 22 and diploid for chromosome 9.

Overall, FISH analysis confirmed the sex status of the embryo as well as the presence of monosomy for chromosomes 16 and 22.



Figure 5.13. CGH and FISH analysis results from embryo 9.6 (blastomere b)

A. Interpretation of CGH experiment showing the cumulative analysis of twelve metaphases. There is a shift in fluorescence towards the green in chromosome X and towards the red in chromosome Y showing that a the test sample (blastomere 9.6b) was rev ish XX. The test sample (blastomere 9.6b) was female with a normal chromosome complement.

B. Results from the two sequential rounds of FISH in a cell from embryo 9.6. In B1 the cell was subjected to the X(green) / Y(red) / 18(aqua) probe cocktail where it showed female (XX) and disomy 18. In B2 the cell was subjected to the 9(green) / 16(aqua) / 22(orange) probe combination which showed disomy 9, 16 and 22.

Overall, the FISH results confirmed the sex status of the embryos and the presence of normal (diploid) cells in embryo 9.6

5.2.6.1 Cytogenetic Events Leading to Mosaicism

In total, only 10% (3/30) of the embryos were uniformly normal. The rest were either mosaic 76.7% (23/30) or completely chaotic 13.3% (4/30). The mechanisms were either CL, CG or MND. All events involved whole chromosomes for the embryos where the CGH results showed normal karyotypes (Group 1). The main mechanism leading to mosaicism was whole chromosome loss (50%) (Table 5.5). However, whole chromosome gain had also a high incidence with 44.5% (8/18), with MND of whole chromosomes in only 5.5% of cases (Table 5.5).

| Embryo | Classification | Event | Chromosome |
|--------|--|-------------|------------|
| 1.1 | Mosaic Diploid/Aneuploid | 1CL(w) | 18 |
| | | 3 CG(w) | 22, X, Y |
| 1.3 | Mosaic Diploid/Aneuploid | 2CG(w) | Х, Ү |
| 2.1 | Uniformly normal | - | - |
| 6.2 | Mosaic Diploid/Aneuploid | 1CL(w) | 22 |
| 7.1 | Mosaic Aneuploid/Diploid | 2CL(w) | 16, 22 |
| | | 1MND(w) | 18 |
| 9.1 | Uniformly normal | - | - |
| 9.2 | Mosaic Diploid/Aneuploid/Chaotic | 3CL(w) | 16, 18, X |
| | | 1CG(w) | 22 |
| 9.4 | Mosaic Diploid/Polyploid/Chaotic | - | - |
| 9.5 | Mosaic Diploid/Aneuploid/Chaotic | 1CL(w) | X |
| 11.1 | Uniformly normal | - | - |
| 12.1 | Mosaic Diploid/Chaotic | - | - |
| 13.2 | Mosaic Diploid/Polyploid/Aneuploid/Chaotic | 1CL(w) | 9 |
| | | 1CG(w) | 22 |
| 13.3 | Mosaic Diploid/ Polyploid/Aneuploid/ Chaotic | 1CG(w) | X |
| | Total | 9CL (50%) | |
| | 13 | 8CG (44.5%) | |
| | | 1MND (5.5%) | |

Table 5.5. Mechanisms leading to aneuploid mosaicism detected by FISH analysis for the embryos which revealed cells with normal karyotypes after CGH (Group 1).

'w' stands for whole chromosome event (either loss, gain or MND)

'p' stands for partial chromosome event (either loss, gain or MND)

It must be noted, that taking into account the previous study (Chapter 3), some losses can be considered as FISH artefacts when they are confined to just one cell.

In the embryos where there was at least one abnormal karyotype (Group 2) revealed during CGH interpretation from 57 events of aneuploid mosaicism, the main mechanism was whole CL (37%) followed by whole CG (19%) and whole MND (16%) (Table 5.6). In this group of results the incidence of events leading to mosaicism involving partial chromosomes was also considerable especially for partial chromosome loss which accounted for 14%. Chromosomes involved in partial chromosome events included chromosomes 1, 2, 3, 4, 9, 14, 15, 16, 17 and 18. All these chromosomes showed breakage in the short (p) or long (q) arms, however, chromosome 18 showed partial CG in the region of 18p11.3-q11.1 during CGH and partial CL in the same region in one cell during FISH, thus revealing reciprocal aneuploidy due to partial MND for the that region of chromosome 18 (Table 5.4 and Table 5.6)

| Embryo | Classification | Event | Chromosome |
|--------|-------------------------------------|-----------------|----------------|
| 1.2 | Chaotic | - | - |
| 2.2 | Chaotic | - | - |
| 3.1 | Chaotic | - | - |
| 4.2 | Mosaic | 2CL(w) | 22, X |
| | Aneuploid/Diploid/Polyploid/Chaotic | 1CG(w) | 16 |
| | | 1MND(w) | 18 |
| 6.3 | Mosaic Aneuploid/Diploid | 1CL(w) | 16 |
| | | 1CG(w) | Y |
| | | 2MND(w) | 22, X |
| 7.2 | Chaotic | - | - |
| 7.3 | Mosaic | 2CL(w) | 9, 16 |
| | Diploid/Aneuploid/Polyploid/Chaotic | 1MND(w) | 22 |
| 8.1 | Mosaic Aneuploid/Chaotic | 1CL(p) / 1CL(w) | 2qter / 10 |
| | | 1CG(p) / 2CG(w) | 14qter / 4, 12 |

Table 5.6. Mechanisms leading to aneuploid mosaicism for the embryos, which revealed a least one cell with abnormal karyotype after CGH (Group 2).

Chapter 5 – Analysis of the development of chromosome abnormalities in human embryos from day 3 to day 5 using CGH and FISH

| 8.2 | Mosaic Diploid/Chaotic | - | - |
|------|-------------------------------------|----------------------|---------------------------|
| 9.3 | Mosaic Aneuploid/Chaotic | 1CL(p) / 1CL(w) | 15qter / 16 |
| | | 1CG(w) | 22 |
| 9.6 | Mosaic | 1CL(p) / 4CL(w) | 3pter / 16,18,19, 22 |
| | Diploid/Aneuploid/Polyploid/Chaotic | 2CG(p) / 1CG(w) | 2qter, 3qter / X |
| 9.7 | Mosaic Aneuploid/Chaotic | 3CL(w) | 1, 16, X |
| | | 1CG(w) | Y |
| | | 2MND(p) | lpter, lqter |
| 9.8 | Mosaic Diploid/Aneuploid | 1CL(p) / 1CL(w) | 3pter / X |
| | | 1CG(p) / 1CG(w) | 17 <i>pter/</i> Y |
| | | 1MND(p) / 1MND(w) | 18 <i>cep</i> / 6 |
| 10.1 | Mosaic Aneuploid/Diploid/Polyploid | 1CL(w) | 22 |
| | | 1CG(w) | 20 |
| | | 1MND(w) | Х |
| 12.2 | Mosaic Diploid/Aneuploid/Chaotic | 1CL(p) / 1CL(w) | 4 <i>pter</i> / 18 |
| | | 1MND(w) | Y |
| 12.3 | Mosaic Aneuploid | 1CL(p) / 1CL(w) | 16qter / 18 |
| | | 1CG(w) | Y |
| 14.2 | Mosaic Aneuploid/Diploid/Chaotic | 2CL(p) / 3CL(w) | 9qter, 16qter /13, 15, 17 |
| | | 2CG(w) | 2, 6 |
| | | 2MND(w) | 16, 22 |
| - | Total | 8CL(p) (14%) | |
| | 17 | 21CL(w) (37%) | |
| | | 5CG(p) (9%) | |
| | | 11CG(w) (19%) | |
| | | 3MND(p) (5%) | |
| | i | 9MND(w) (16%) | |

'w' stands for whole chromosome event (either loss, gain or MND)

'p' stands for partial chromosome event (either loss, gain or MND)

5.3 Discussion

This study was aimed at assessing the CGH technique and examining its suitability as a research tool on human preimplantation embryos. Each step of the technique was optimised for single cell use. Furthermore, CGH was tested against FISH in order to investigate whether by obtaining a full karyotype on 1-2 cells on day 3 of embryo development can reveal any surplus information about the mechanisms of mosaicism on day 5 of embryo development when the whole embryo was analysed by FISH.

5.3.1 WGA Optimisation

Initial work included assessing the handling of single buccal cells and tubing and amplifying each single cell. Amplification efficiency reached 97% (from 100 single cells) after DOP-PCR. DOP was selected as the preferred method of whole genome amplification since it has been postulated that it produces sufficient quantities of amplified product and provides sufficient coverage of the genome from a single cell (Wells et al, 1999; Voullaire et al, 1999). All DOP-PCR products were analysed using gel electrophoresis where contamination in the negative controls was present in less than 5% (Figure 5.1). All single buccal cells and blastomeres yielded a smear that contained a single distinct band at 450bp (Figure 5.1). Such bands have previously been suggested to be mitochondrial DNA, which is amplified by DOP-PCR although this does not interfere with the CGH profile, as it does not hybridise to the template chromosomes (Voullaire et al, 2000). It was essential to maintain stringent precautions against contamination throughout single cell isolation, lysis and amplification. The incidence of contamination was assessed regularly using numerous control blanks containing PCR reaction mixture but no DNA. A non-intensive smear in the single cell amplified products was visible thus displaying sufficient coverage of the whole genome. Such heterogeneous mixture of fragments generated by DOP-PCR and visualised as a smear has also been reported by Wells et al (1999). During amplification of single cells using DOP-PCR, two different thermal cyclers were assessed, the OmnigeneTM and Eppendorf Master Gradient[®]. In each thermal cycler 20 single cells were amplified and their results were analysed on 1-2% agarose gel stained with ethidium bromide (Figure 5.2). During analysis it was observed that the OmnigeneTM thermal cycler displayed a larger smear thus showing enhanced amplification of the genome. This considerable difference in smears between thermal cyclers was later reduced by adjusting the 'ramp' speed of the Eppendorf Master

Gradient[®], however, the OmnigeneTM remained the preferred choice of thermal cyclers during DOP-PCR amplification.

5.3.2 CGH Optimisation

Preliminary work on genomic DNA extracted 'in house' was utilised to optimise the CGH technique in order to be able to produce reliable results in single cells. Normal male genomic DNA (labelled in red which acted as the 'control' sample) against normal female DNA (labelled in green which acted as the 'test' sample) was amplified and analysed using CGH. CGH was able to detect the sex chromosomes in the samples with shift of fluorescence towards the green for the X chromosome indicating excess and towards the red for the Y chromosome indicating deficiency in the 'test' sample (Figure 5.3). The determination of the sex chromosome CGH pattern was a criterion for the reliability of the procedure. However, it must be taken into consideration that the euchromatic region of the Y chromosome is at the limit of size resolution for which aneuploidy can be detected by CGH (Voullaire et al, 1999). Subsequently CGH was performed on clumps of buccal cells from normal male and female individuals, which was successful. Thus, the remaining single cell DOP-PCR products (section 5.2.1) were put through CGH to assess its efficiency at the single cell level. The CGH efficiency on control single cell samples was 93.3%, with only two cells producing non-interpretable results. Furthermore, in a study of single fibroblasts the DOP-PCR amplification allowed reliable detection of trisomies 13, 18, 21 and 22 (Figures 5.4-5.7). The hybridisation of DOP-PCR products to normal metaphase chromosomes produced strong even signals with no obvious sites of amplification deficiency or excess. The fluorescence shifts on the trisomic chromosomes were later used as reference for shifts displayed by day 3 blastomeres. Similar results on single fibroblasts of trisomic samples have been shown by Wells et al (1999). Furthermore, the fluorescence intensity observed on the Y chromosome of the female 'test' samples revealed the extent of background fluorescence could be used 'roughly' as a reference to distinguish between monosomic and nullisomic samples. During the whole of this study homo-hybridisation (the hybridisation of samples containing similar amounts of DNA, each amplified and labelled using the same methods) was performed for all blastomeres, since it has been suggested that

hetero-hybridisation of DOP-PCR CGH displays high background, uneven hybridisation and is also associated with false deletions or amplifications (Huang *et al*, 2000)

CGH sensitivity is usually considered to be low in terms of the minimum size of detectable over/underrepresented chromosome fragments. Studies using CGH in a variety of samples have reported a resolution limit of 10-40Mbp (Kallioniemi et al, 1994; Daniely et al, 1998; Voullaire et al, 1999; Lestou et al, 2000; Malmgrem et al, 2002). However, Kirchhoff et al (1999; 2000; 2001) managed to develop the technique further to increase the sensitivity as well as the specificity in order to detect chromosomal aberrations as small as 3Mbp. Their modification was termed high resolution CGH (HR-CGH). During this study the CGH technique was not designed to detect small deletions such as the study which aimed and accomplished the detection of Prader Will/Angelman deletions (Kirchhoff et al, 1998), which are thought to be approximately 4Mbp (Christian et al, 1998) and thus are likely to be undetected by normal G-banding. However, during CGH interpretation of the captured images in some cases the presence of hybridisation artefacts was observed for the heterochromatic regions, and both the short and long arm telomeres of certain chromosomes, including 1, 9, 16, and Y and the satellite regions of the acrocentric chromosomes. These were caused due to the extreme suppression of these regions by the Cot-1 DNA and any low level fluorescence at these sites was attributable to background. Consequently, these regions were not considered during interpretation. In addition, in cases where abnormal results were obtained for chromosomes 19 and 22, they were interpreted with caution, due to the fact that these chromosomes are also known to be prone to labelling artefacts. Other similar observations regarding such artefacts on these chromosome regions have been reported from various studies in the literature (Tabet et al, 2001; Wilton et al, 2001).

5.3.3 CGH and FISH on Human Embryos

CGH is a technique with several difficulties due to the complexities of its nature. All cells from each embryo were consistent for the determination of the sex chromosomes and all were in agreement with the respective FISH results. However, it should be

noted that a possible case of mosaicism for a sex chromosome abnormality would negate this suggestion of an 'internal control'. In addition, the chromosomes present in a normal diploid copy number can be used as an internal negative control for assessment of efficiency of hybridisation (Malmgren *et al*, 2002). It has been reported that small deletions or amplifications of the telomeric regions are difficult to interpret by CGH analysis and might be missed (Malmgren *et al*, 2002).

A total of 54 blastomeres were biopsied on day 3 and 48 (88.9%) produced interpretable results. Failure of the CGH could indicate either an anucleate blastomere (as visibility of the nucleus was not possible for all the blastomeres biopsied), or that the cell was mistaken for an anuclear fragment which is a common finding in day three embryos (Voullaire *et al*, 2000). It has been shown that 5% of good quality embryos and 12% of poor quality embryos contain anucleated blastomeres (Hardy *et al*, 1993). Another explanation accounting for failure of results could be premature cell lysis or loss of a cell during transfer to the PCR tube. A cytoplasmic metaphase preparation of poor chromosome morphology would present a major obstacle to the successful hybridisation of both control and test DNA.

The efficiency of single blastomere CGH for this study was 88.9%, which is exactly the same as the single buccal cell rates. Wells and Delhanty, (2000) reported a CGH efficiency of 88% whilst Voullaire *et al*, (2000) revealed a slightly higher efficiency of 89%. Similarly to this study, both studies employed IVF embryos of good quality. However, during this study the embryos were frozen-thawed and not fresh which may explain the minor difference in efficiency. The efficiency of the study performed by Malmgren *et al*, (2002) using spare embryos from PGD cases was 70%, considerably lower than the one reported in this project. The fact that Malmgren *et al*, (2000) used a different type of WGA than the one used by Voullaire *et al*, (2000), Wells and Delhanty (2000) and this study could account for the lower efficiency seen in their study. The protocol used, based on linker-adapter mediated PCR, might be less reliable or amplify DNA less efficiently leading to their poorer results (D. Wells, personal communication). The FISH efficiency on embryos was higher than CGH with 343/354 (93%) producing analysable results. The FISH probe efficiency was

88.6% (range 81-97) for the 14 probes used, which was calculated on 200 interphase nuclei from a normal male lymphocyte slide.

Thirteen couples donated a total of 37 embryos for this research project with a mean maternal age of 33.1 (range 28-39) years. One embryo was not included since it failed to provide any kind of results during either CGH or FISH, which was expected due to their low morphology grade. A further six embryos failed to grow to day 5 and degenerated thus no cells could be spread. However, from all these embryos at least one cell was biopsied and the CGH results showed 3/7 blastomeres being of a normal karyotype, 2/7 were aneuploid and one was chaotic. Overall, from the thirty embryos where analysis was possible with both techniques the mean number of nuclei investigated on day 5 was 12. The low number of nuclei which was available for FISH analysis on day 5, where the embryos should have been blastocysts (at least 30 cells per embryo) can be attributed to several factors. Firstly, the culture media at the time might have been problematic hence, blastocyst formation was not promoted. Furthermore, the biopsy procedure was performed with a laser and might have had an adverse effect on the embryos. However, similar pregnancy rates have been obtained from studies comparing the laser biopsy technique to the widely used acid Tyrodes technique (De Vos and Van Steirteghem, 2001). It has been postulated that blastomere excision on day 3 could risk embryonic death by reducing the number of cells available for differentiation or by the excision of blastomeres essential to form a particular cell line (De Vos and Van Steirteghem, 2001). Technical expertise is required for a successful biopsy procedure and the person performing the technique was training and so was not adequately experienced. It has been suggested that the biopsy procedure can cause a strain on the embryo by disturbing tight junctions either by incubating the embryos in Ca^{+2}/Mg^{+2} -free medium prior to operating or by forceful aspiration (Munne and Cohen, 1998). Dumoulin and co-workers (1998) evaluated the embryo viability and implantation after exposing embryos for 45min to Ca⁺²/Mg⁺²free medium and concluded that subsequent development to the blastocyst was not affected. Overall it has been shown from untransferred biopsied embryos in PGD cycles for sexing or aneuploidy screening that less than 30% show blastocyst formation after cleavage stage biopsy (Veiga et al, 1999; Magli et al, 2000; Sandalinas et al, 2000). A more recent study by Baart et al (2004) showed that around 50% of the embryos formed into blastocysts after biopsy of two blastomeres on day 3. Another factor that may have affected the blastocyst formation is the freeze-thaw

procedure, which might have rendered the embryos more fragile. It has been broadly suggested that the formation to blastocyst stage and consequent implantation potential of frozen-thawed embryos is probably compromised (Levran *et al*, 1990; Van Steirteghem *et al*, 1992; Van der Elst *et al*, 1995). Therefore, addition of the biopsy procedure on frozen-thawed embryos might have influenced the low numbers of blastomeres present during day 5 of spreading.

A total of thirty embryos provided results from both techniques. Only three embryos (10%) showed normal results both after CGH and for all the blastomeres analysed by FISH for the chromosomes tested. Compared to other studies the normality rate for this study is quite low since Wells and Delhanty (2000), Voullaire et al (2000) and Trussler et al (2004) showed rates of 25%, 25% and 42.5% respectively. In all three studies good quality fresh embryos were analysed of a grade 3-4/4 with grade 4 being the best quality embryos. Whereas in the present study though the embryos were considered of good quality, they had been frozen-thawed thus they might have been compromised. Laverge et al (1998) after analysing 63 frozen-thawed embryos by FISH concluded that embryos which do not grow further after freezing and thawing carry chromosomal abnormalities. Malmgrem et al (2002) found 0% normal embryos however the embryos used in that study were previously diagnosed as unbalanced and were considered not fit for transfer hence the higher degree of mosaicism. IVF culture conditions could also be responsible for the high frequency of mosaicism in this study. An example is a sudden decrease in temperature that could in turn affect cytokinesis, leading to the generation of diploid/polyploid embryos (Munne and Cohen, 1998). In addition, it has been suggested that embryos produced by different stimulation protocols and cultured under different conditions have very diverse mosaicism rates (Munne et al, 1997).

5.3.3.1 Group 1 Embryos

CGH showed a normal chromosome complement in thirteen embryos (Group 1). FISH analysis confirmed the sex of the embryo in all thirteen embryos. However, in five blastomeres (Table 5.3) there were sites on the sub-telomeric regions of chromosomes 19, X and Y that showed enhanced or diminished fluorescence. This set

of results was considered as artefacts. The underlying reason was that certain regions of the karyotype such as centromeres and sub-telomeric segments show variation in the profile and these regions are normally excluded from CGH (Kallionemi et al, 1994). Moreover, certain chromosomes (1p, 17, 19, 22 and Y) are prone to show frequent enhancement of the test signal and are also excluded from the analysis (Moore et al, 1997). Other chromosomes that have been reported to be prone to artefactual results in placental tissues include chromosomes 4, 13, and 18 (Lestou et al, 2000) however, these chromosomes did not present similar results in the current study. Embryo 13.2 displayed the only blastomere which was affected by an abnormality involving chromosome 9 after FISH analysis, whereas all other embryos had mixed aneuploid cells of monosomies or trisomies for chromosomes 16, 18, 22, X and Y, with an euploidies for chromosomes 22 and X showing the highest incidence. Trisomy of the small autosomes and monosomy for the X chromosome, such as embryo 9.5 (Table 5.3) are the most common abnormalities detected in human pregnancy (Boue et al, 1985). Furthermore, embryos 13.2 and 13.3 displayed triploid cells the origin of which is unclear in the literature. The underlying mechanism that could lead to diploid/triploid mosaics may be due to an incorporation of another gamete or its genome into one of the daughter cells derived after the first mitotic division or later. The extra gamete might be a polar body (Mueller et al, 1993). It has also been suggested that diploid/triploid mosaicism could result from fusion of a diploid zygotic nucleus with an extra sperm nucleus or the extrusion and degeneration of a haploid nucleus to produce a diploid cell line in a triploid embryo (Kuo et al, 1998). Embryo 13.3 displayed mosaicism involving the sex chromosomes as well as triploidy. It has been postulated that delayed IVF might also cause triploid cells (Plachot et al, 1988) and more interestingly the embryo came from a couple which had ICSI treatment. Although this is an isolated phenomenon an increase of sex chromosome abnormality has been reported in children conceived using ICSI affecting at least 1% (Martin, 1996). Similar observations have been reported in a CGH study of human day 3 embryos where 2/12 embryos displayed mosaicism involving only sex chromosomes which were thought to have arisen from 47, XXX zygotes after ICSI treatment (Wells and Delhanty, 2000). Chaotic complements were also present in this group of embryos, which revealed normal CGH karyotypes from the biopsied blastomeres, in 6 embryos. Chaos has been reported in almost all the studies in embryos using either FISH or CGH.

5.3.3.2 Group 2 Embryos

In 17 embryos at least one cell showed an abnormal karyotype when analysed by CGH. Ultimately, all 17 embryos were classified as mosaic or chaotic (Group 2). Four embryos (23.5%) were completely chaotic containing cells with three or more abnormalities in different chromosomes. All four embryos when analysed by FISH on day 5 were arrested consisting of only 2-3 blastomeres each, which is consistent with suggestions that chaos is associated with impaired development (Delhanty and Handyside, 1995; Delhanty *et al*, 1997). All four embryos displayed abnormalities such as nullisomies, tetrasomies and completely abnormal gonosome karyotypes.

Except for embryo 4.2 all embryos were arrested (<30 blastomeres) and all embryos had a decreased number of blastomeres with a diploid cell line compared to Group 1. It has been implied from blastocyst studies that embryos with a lower proportion of diploid cell line (therefore higher degree of mosaicism) have a low developmental potential (Ruangvutilert *et al*, 2000a). However, this selection against embryos with a high degree of mosaicism does not operate perfectly.

Most of the evidence for different chromosome susceptibilities to aneuploidy has been obtained from clinically recognised pregnancies. Major studies reported that although trisomy 16 accounts for 20-35% of all trisomies, acrocentrics and chromosome 2 account for 5-10% each (Hassold et al, 1984; Warburton et al, 1986). In Group 2 embryos all chromosomes were involved in abnormal karyotypes either due to duplications or deletions (partial or whole). Chromosome 1 was shown to be most affected with four incidences of chromosome duplication (whole or partial) and four incidences of chromosome deletion (whole or partial). Similar data were obtained during our FISH study on day 5 embryos for the group of embryos (Group I), which were arrested (Chapter 3). However, in the case of these set of results it must be highlighted that some confined chromosome losses can be attributed to FISH artefacts since there was not two-loci per chromosome analysis during the current study. Therefore, FISH failure could not be detected and excluded from the results. Wells and Delhanty (2000) also found an increased incidence of chromosome 1 deletion however chromosome 2 was shown to be affected by chromosome deletion (partial or whole) on six different occasions. Baart et al (2004) also found chromosome 1

affecting 9/29 (31%) embryos when analysed by FISH. In our study chromosome 2 was mostly involved in chromosome duplications (partial or whole). In a recent CGH study nine instances of trisomy 22 were found (Trussler *et al*, 2004), which is in accordance with a FISH study investigating differences in chromosome susceptibility to aneuploidy that concluded that trisomy 22 is the most common aneuploidy (Munne *et al*, 2003b). In this study aneuploidy of chromosome 22 was also increased however, chromosome 22 deletions were more prominent than duplications.

In two embryos, 7.2 and 14.2, a total of 16 and 13 chromosomes were shown to be affected respectively. Embryo 7.2 displayed a complete breakdown of normal chromosome complement (Figure 5.10). The embryo was classed as chaotic and the FISH results were able to confirm the CGH abnormalities in six chromosomes. Different aneuploidies were seen including nullisomy for chromosomes 11, 13, 14 and 15, trisomy for chromosomes 3, 4, 5, 17, 18 and 19 as well as abnormalities in the sex chromosomes revealing karyotypes of XXYY, XXXO and XXXXXYY. Similar complete aneuploidy and embryos with random allocation of chromosomes to daughter cells have been previously observed by Wells and Delhanty (2000) in 2/12 embryos and by Trussler *et al* (2004) in 3/40 embryos. There is no pattern to the aneuploidy seen; chromosome losses and gains occur with similar frequency and there is no evidence of that any particular chromosome is involved more often than any other (Wells and Delhanty, 2000). Evsikov and Verlinsky (1998) postulated that embryos with chaotic chromosome segregation do survive to the blastocyst stage but will not progress further and would fail to implant.

FISH analysis was able to confirm the sex status of all seventeen embryos shown by CGH. Analysis of the CGH and FISH data demonstrated results 'in agreement' in 15/17 embryos (in the other two embryos there were no FISH probes available for the specific chromosome). The 'agreeing results' showed a similar abnormality on the chromosome in question or revealed a reciprocal abnormality for that chromosome. For example in embryo 8.1 CGH analysis revealed deletion of chromosome 10 and duplication of the 14qter. On day 5, the FISH protocol showed 4/4 cells with monosomy 10 and 2/4 cells with trisomy for the 14qter chromosome segment.

Furthermore, the CGH results in embryo 4.2 displayed duplication of chromosome 22 and deletion in chromosomes 16pter and 18. FISH analyses demonstrated two blastomeres with monosomy 22, nine blastomeres with monosomy 18 and one blastomere with trisomy 16 and 18 therefore revealing, overall, reciprocal loss and gain for chromosomes 16, 18 and 22.

The use of CGH to study human preimplantation embryos has enabled the detection of a specific type of error involving structural alteration of chromosomes, namely chromosome breakage. Embryos 7.2, 8.1, 9.6 and 9.7 displayed possible breakage in chromosomes 1 and 2. In embryo 9.7 the FISH protocol was chosen so that it can detect whether the breakage was an experimental artefact by having one sub-telomeric probe for each arm ('p' and 'q'). FISH analysis revealed 2/5 blastomeres showing reciprocal products of the breakage and 2/5 blastomeres displaying similar products of the breakage. In embryos 7.2 and 8.1 chromosome breakage was noted at the region of 2q31-2q37 where the acentric fragment was lost. Interestingly, in embryo 7.2 the remaining centric chromosome fragment (2p25-q31) was shown to be duplicated. Partial aneuploidy due to chromosome breakage is likely to result in an unstable karyotype through the formation of acentric and dicentric chromosomes (Voullaire *et al*, 2002).

5.3.4 Mosaicism and Chaos

Overall, 23/30 (76.6%) of the embryos analysed were mosaic, from which 14/23 and 9/23 were diploid as the major cell line and the aneuploid line respectively. In Group 1 embryos FISH results on 150 blastomeres showed 70.6% being diploid whereas from 183 cells analysed by FISH in Group 2 embryos only 37.1% were diploid for the chromosomes tested. The significant drop of the diploid cells was expected since Group 1 embryos would have been diagnosed as normal in a PGD setting due to their normal CGH results. All errors leading to aneuploid mosaic embryos were postzygotic with no meiotic error present. A meiotic division error would render all embryonic cells chromosomally aneuploid. This is an interesting finding which is in accordance with a similar study performed by Baart *et al* (2004) which analysed two blastomeres by FISH on day 3 and re-analysed the whole embryos by FISH, using the

same probe combination on day 5. Their aim was to clarify whether it is more informative and less prone to misdiagnosis to carry out PGD for an euploidy screening on day 3 or on day 5. However, in all embryo CGH studies a low percentage of meiotic errors (7.5-8.3%) has been reported (Wells and Delhanty, 2000; Voullaire *et al*, 2000; Trussler *et al*, 2004).

Furthermore, it was disturbing that in 18/30 (60%) embryos at least one cell was chaotic in its chromosome complement. Chaotic embryos are thought to arise due to absence of cell cycle checkpoints. A study performed on mouse oocytes demonstrated the absence of the metaphase-anaphase checkpoint (Le Maire-Adkins *et al*, 1997). This checkpoint is responsible for the correct alignment of chromosomes onto the mitotic spindle, and the situation could be similar for human oocytes. The abnormalities of the mitotic spindle could be related to the sub-optimal *in vitro* culture environment (Pickering *et al*, 1990). Moreover, maternal genome support could result in the survival of embryos with multiple aneuploidies up until the blastocyst stage. Elimination of maternal mRNAs could lead to the arrest that is frequently observed prior to blastocyst formation for highly abnormal embryos (Wells and Delhanty, 2000). Wilding and co-workers (2003) recently published data stating that if poor vascularization of follicles is the underlying cause for the loss of mitochondrial activity in maturing oocytes this would lead to chaotic development.

5.3.5 Mechanisms of Aneuploidy Mosaicism

Embryonic death is caused by chromosomal abnormalities that are primarily the result of chromosomal errors during female gamete formation. However, there are some errors that will arise during early cleavage divisions as a consequence of mitotic errors. These errors in cell divisions, namely mitotic non-disjunction and anaphase lag lead to mosaicism and chaotic imbalances.

In Group 1 embryos only whole chromosome mosaicism mechanisms were seen from which the predominant was chromosome loss (CL) (50%), followed by chromosome gain (CG) (44.5%) and just one instance of mitotic non-disjunction (MND) (5.5%).

These results are very similar to the 'Group I embryos' in our FISH study on day 5 embryos (Chapter 4 Table 4.3). All errors were post-zygotic for Group 1 embryos. In the group of embryos that the CGH analysis displayed at least one cell with an abnormal karyotype (Group 2) there was an increased incidence of partial chromosome errors (28% in total). The predominant mechanism of mosaicism was whole chromosome loss (37%). Whole CG gave rise to 19% of the aneuploid mosaic embryos and whole MND to 16%. In comparison to Group 1, in Group 2 embryos the incidence of whole CG was halved whereas the rate of whole MND was tripled. Chromosome 22 was mostly affected by reciprocal loss and gain of material. Anaphase lagging is the causative agent behind whole CL and CG. Anaphase lagging leads to mosaicism and chaotic chromosome distribution, most probably reflecting asynchrony between karyokinesis and cytokinesis. Coonen et al (2004) recently reported that anaphase lag is the main cause chromosomal mosaicism in embryos. This was in accordance with our findings in the FISH study (Chapter 3). Similarly by employing CGH and FISH it has been shown that chromosome loss and gain are the main causes of mosaicism which occur probably due to anaphase lag.

In the partial chromosome mechanisms, again, partial CL was found to be 14%, followed by partial CG (9%) and partial MND (5%). This is the first study that recognises partial chromosome mechanisms as an underlying reason causing chromosomal aneuploidy. It has been postulated that chromosome breakage and whole chromosome aneuploidy could be caused by different factors (Wells and Delhanty, 2000). In a recent study, 6% of the embryos analysed by CGH were found to be affected solely by partial aneuploidy (Voullaire et al, 2002). Wells and Delhanty (2000) have proposed that acentric fragments would not be stably transmitted to daughter cells and the resultant loss of material would leave the embryo with a potentially lethal monosomy for that chromosome region. Furthermore, both initial studies of embryos using CGH observed chromosome breakage at the point of 2q31 (Wells and Delhanty, 2000; Voullaire et al, 2000) which is similar to this study (Figures 5.10, 5.11 and 5.12) along with the breakpoint in chromosome 1 (1p31). Both sets of breakpoints map to defined chromosomal fragile sites (Sutherland, 2003), which are prone to chromosome breakage. Fragile sites are non-randomly located gaps or breaks in chromosomes that are induced to appear by specific culture

conditions (Sutherland, 1979) and are frequently involved in de-novo chromosome rearrangements (Warburton, 1991). They are classified according to their chemistry and of induction and by their frequency in the population (Richards, 2001). Rare (or heritable) fragile sites are found on the chromosomes of less than 5% of individuals, whereas common (or constitute) fragile sites are found in all individuals (Sutherland and Baker, 2000). The most common 'fragile' site in the genome is FRA3B at 3p14.2 which has occurred once in this study in embryo 9.8 where that part of the chromosome 3 region (3p14-p26) was lost. A fragile site is a region of chromatin that fails to compact normally for mitosis (Sutherland, 2003). Therefore, during this study CGH has demonstrated that chromosome breakage in human preimplantation embryos can be considered as a possible mechanism of aneuploidy. FISH can only reveals information about a small area of each chromosome to which they hybridise, thus rearrangements affecting chromosomal regions rather than whole chromosomes, are not detected. During this study, this FISH limitation was used to our advantage in order to establish chromosome breakage rather than regard it as a CGH artefact.

5.3.6 Conclusions

Throughout this study all three techniques used i.e. whole genome amplification, FISH and CGH were optimised through a series of control experiments in order to achieve high amplification rates and hybridisation rates respectively. After FISH and CGH analysis it was found that only 10% of the day 5 embryos were uniformly diploid, while the rest were mosaic or completely chaotic. The major mechanism of mosaicism leading to aneuploid cells was found to be chromosome loss from Group 1 and Group 2 embryos. However, the difference between chromosome loss with chromosome gain and mitotic non-disjunction was not statistically significant. Similar results were obtained in Chapter 4, adding to the conclusion that chromosome loss, which probably arises due to anaphase lag, is mostly responsible for the high level of mosaicism present in embryos.

5.3.7 Future Work

CGH and FISH were employed during this study and assessed as techniques. Each technique offered information on the process of understanding the mechanisms that lead to chromosomal mosaicism and chaos in human preimplantation embryos.

Initially the CGH technique should be made less laborious and more reliable in order to be used in a clinical setting e.g. for PGD of chromosomal abnormalities. This can be achieved by reducing the time of hybridisation from 72 hours to less than 24. Wells and co-workers (2002) were able to reduce the time to 30 hours during a PGD case for aneuploidy screening, however, the CGH was applied to polar bodies.

The use of DNA microarrays to act as hybridisation templates, instead of the target metaphase chromosomes that have been used so far could further reduce the period of hybridisation, and improve the analysis of the obtained results, increasing the simplicity of interpretation by avoiding the need to karyotype metaphase chromosomes. During CGH-array analysis, patient DNA and normal reference DNA are hybridised to arrays of genomic clones in order to detect unbalanced gains or losses of genetic material across the genome (Shaffer et al, 2004). In a recent study by Schaffer et al (2004) where production-of-conception samples were analysed with Gbanding and CGH-array technology. It was found that the array technology detected all abnormalities as reported by G-banding analysis and revealed ~10% new abnormalities. Bermudez et al (2004) carried out microarray CGH to analyse 5 human oocytes and were able to identify 1361 transcripts expressed in oocytes. Chan et al (2002) designed a DNA disc chip array, based on comparative genomic hybridisation in order to study sperm and concluded that the technique was reliable however it was still prone to manual processing involving the fluorescent microscope and computer. In recent study carried out on single lymphocytes, Hu et al (2004), described the application of array-CGH for normal and trisomic (13, 15, 18) single cell lymphocytes. During that study the slides were arrayed with chromosome-specific DNA libraries and the expected karyotypes of all cells was analysed and confirmed whilst hybridisation took place for 30hrs (Hu et al, 2004). These results prompted the author to suggest the use of this technique for PGS (Hu et al, 2004).

The whole genome amplification procedure could be replaced and use MDA as an alternative method of WGA. This method uses a $\varphi 29$ DNA polymerase and random exonuclease-resistant primers, whilst DNA amplification takes place at 30°C (Dean *et al.*, 2002). It has been shown that MDA provides a highly uniform representation of the genome, with the amplification bias being less than 3-fold among eight chromosomal loci, compared to the 4-6-fold observed with the application of the DOP-PCR (Dean *et al*, 2002). Spits *et al* (2006) developed a protocol using MDA and were able to amplify 22 loci in 60 single cells and concluded that it can be used for PGD for single gene disorders.

Chapter 6

Conclusion
6.1 Final Thoughts

The constant evolution of molecular, cytogenetic and combinatorial techniques aims to facilitate and enrich the study of human preimplantation embryos and the subsequent application of PGD. During this study new methods and protocols for different techniques were devised and tested.

The phenomenon of mosaicism was explored in human preimplantation embryos with a novel protocol using multi-colour FISH. Mosaicism can be explained by the presence of two or more cell line present in one embryo. Mosaicism has been the 'culprit' causing misdiagnosis during PGD using either FISH or PCR. Therefore, the need of biopsying two cells per embryo whilst carrying out PGD has been put forward. The decision as to whether one or two cells should be removed is controversial; removing two cells reduces the cellular mass and could potentially reduce its developmental capacity (Braude et al, 2002). Early studies to measure the effect of biopsy on human embryonic development showed that two cells could be removed from 8-cell embryos on day three post-insemination without reducing the number blastulating or disturbing cleavage rates (Hardy et al, 1990). Furthermore, two-cell biopsy procedures should only be carried out on day three post-insemination at the 6-8 cell stage, when up to a quarter of the blastomeres can be removed without disturbing subsequent development (Handyside et al, 1989; Handyside, 1991; Hardy et al, 1990), as the biopsied cells are still undifferentiated (Harper et al, 1996). The debate over whether taking two cells rather than one is detrimental to the embryo, is ongoing (De Vos and Van Steirteghem, 2001). However, it has been shown that the accuracy of the diagnosis is likely to be enhanced if embryos are replaced when results from both cells are concordant (Van de Velde et al, 2000).

Therefore, it is necessary to determine the level of mosaicism. Many studies have been carried out in search of the levels of mosaicism using FISH both in cleavage stage (Delhanty *et al*, 1993; Munne *et al*, 1994b; Harper *et al*, 1995; Delhanty *et al*, 1997; Laverge *et al*, 1997; Iwarsson *et al*, 1999; Magli *et al*, 2000; Coonen *et al*, 2004) as well as blastocysts (Evsikov and Verlinksy, 1998; Veiga *et al*, 1999; Ruangvutilert *et al*, 2000a). All of these studies tried to assess the level of mosaicism using probes for as many chromosomes as possible. However, all previous studies conclude that the FISH efficiency ranges from 60-95% depending on the type of probe, the number of probes used simultaneously and the number of sequential rounds. Hence, up to 40% of abnormalities can be missed whilst studying mosaicism and its effects. The protocol devised in this study was based on the detection of mosaicism of three main chromosomes (1, 11 and 18) as well as the degree of mosaicism in the sex chromosomes (due to the near perfect efficiency results of the probes). The three main chromosomes were investigated during two sequential rounds of FISH by employing a different probe and at a different site for each chromosome e.g. a centromere and a sub-telomere for each chromosome. This allowed the detection of any FISH artefacts and any FISH failure that might occur, providing the true level of abnormality for the chromosomes investigated. A total of 76 abnormal results would have been missed unless the two probes per chromosome protocol was carried out.

One of the main aims of this study was to explore the mechanisms that give rise to aneuploid mosaic embryos in order to understand the high levels in *in vitro* human preimplantation embryos. It has been suggested that the high level of embryonic death is mainly due to chromosomal abnormalities (Wells and Delhanty, 2000). Chromosomal abnormalities and especially aneuploidy has been found to be extremely high after as >60% of spontaneous abortions at <12 weeks are aneuploid (Hassold *et al*, 1980). In our study chromosome loss was found to be the mechanism affecting most aneuploid lines. Chromosome loss probably occurs due to anaphase lagging. Other studies have found that mitotic non-disjunction was the main mechanism affecting mosaicism. This may be due to the fact that the FISH protocols used by other groups regard low level monosomy findings as FISH failure or artefacts since they are using one probe per chromosome. In a very recent study, it was suggested that anaphase lagging is the major mechanism behind aneuploid mosaic embryos (Coonen *et al*, 2004).

The two different groups of embryos analysed were distinguished based on the type of IVF medium that they were grown. Group I embryos were cultured until day 5 in nonsequential medium, which is considered sub-optimal (Jones *et al*, 1998), whereas Group II embryos were grown in sequential medium. Their main difference was the number of nuclei available on days which was ~20 cells per embryo for Group I and \sim 55 cells per embryo for Group II signifying the limited growth in Group I. Both Groups revealed chromosomal loss as the main mechanism behind aneuploid mosaic embryos, however, in Group II there was an increased number of tetraploid cells present compared to Group I. This can be explained by the findings of multiple studies, which have proposed that probably tetraploid cells may be a normal feature in the development of the trophectoderm. This would add more support to the presumption of limited growth for Group I embryos.

In order to achieve definitive proof of the underlying mechanisms of mosaicism more embryos need to be analysed as well as analysis of all the chromosomes. Although, the current study was performed on spare, not fit for transfer embryos, therefore morphologically the embryos were compromised, still the data obtained were useful. A larger cohort of embryos which are morphologically sound should be analysed with means that can provide information regarding all the chromosomes. Novel techniques such as interphase conversion allowing the entire karyotype to be screened in one step have been developed in the recent years and have found clinical application. Their principle involves fusion of polar bodies or blastomeres with enucleated human or bovine oocytes to induce mitosis (Verlinsky and Evsikov, 1999a; Willadsen et al, 1999). Interphase conversion methods for inducing metaphase in biopsied blastomeres have been successfully applied in a number of translocation cases to date with subsequent analysis by standard chromosome painting (Verlinsky and Evsikov, 1999a; Evsikov et al, 2000) or SKY FISH (Willadsen et al, 1999). For both these methods however there remains the difficulty inherent in working with single metaphase spreads of limiting artefactual chromosome loss.

Therefore, in an attempt to assess the level of mosaicism in all the chromosomes, by avoiding the problems encountered by FISH i.e. FISH failure and artefacts, CGH was used in the second part of this study. One or two cells were biopsied on day 3 in order to carry out CGH and the rest of the embryo was left to grow until day 5, when it was spread and FISHed. CGH has been found to be able to diagnose 25-38% more abnormalities during PGS in comparison to five- and nine-probe FISH protocols (Voullaire *et al*, 2002). Therefore, by employing CGH in embryos to investigate the whole genome, the entire level of mosaicism would be revealed. The set of embryos used for this study were frozen-thawed embryos of good quality, which at the time of

freezing were considered to be fit for transfer. Thus, good quality embryos (morphologically assessed) would be analysed using CGH. However, CGH has been considered to be laborious and not able to detect small deletions or insertions (Mamgrem et al, 2002), as well as responsible for artefacts regarding chromosomes 19, 22 and Y (Kallioniemi et al, 1994; Tabet et al, 2001; Wilton et al, 2001). Hence, FISH was carried out on day 5 to act as a safety net for CGH as well as provide confirmation results regarding the abnormalities present on day 3 and whether these abnormalities persist until day 5. CGH and FISH results were found to be in agreement by scoring the sex chromosomes, where it was found that both techniques showed the sex chromosome complement. Overall, CGH was found to be an efficient technique whilst analysing single embryonic blastomeres (83.3%). During CGH analysis it was found that on day 3 when 1-2 cells were biopsied, 13 embryos were found to be normal, however, when left to grow and spread until day 5 and FISHed with probes for six chromosomes (9, 16, 18, 22, X and Y), only three were uniformly normal. Overall, only 10% (3/30) of the embryos analysed were normal, whilst the rest were either mosaic or chaotic. This high level of mosaicism was also observed in the 1st study and has been reported from other CGH and FISH studies (Tables 3.3 and 4.1). Sandalinas and colleagues (2001) found similar percentages of chromosomal mosaicism (85%) during FISH analysis of human blastocysts. However, Baart and coworkers (2004) analysed embryos using FISH on day 3 (by 2-cell biopsy) and day 5 (whole embryos spreading) and found significantly decreased levels of abnormality and mosaicism (55%). However, the latter study was carried out as part of the PGS programme and was aiming to distinguish whether day 3 analysis of two biopsied cells is able to provide a reliable estimate of chromosomal mosaicism (Baart et al, 2004).

Similar to the 1st study, chromosome loss was found to be the most common mechanism of mosaicism leading to mosaic aneuploid cell lines. However, due to the CGH involvement, partial aneuploid events were also exposed with an incidence of 28% of events producing partial aneuploidy results. Partial chromosome loss was found to be the predominant mechanism leading to partial aneuploidy (14%). The phenomenon of partial aneuploidy has not been well documented during studies carried out on human preimplantation embryos. This is partly due to the fact that FISH is unable to distinguish between whole and partial aneuploidy since each FISH

probe span is relatively small and can only cover a limited stretch of the chromosome. For example, when one or three signals are observed during a FISH study employing the locus-specific probe for chromosome 21, whole monosomy or trisomy respectively is assumed rather than loss or gain of that specific part. However, studies on embryos employing CGH (Wells and Delhanty, 2000; Voullaire et al, 2000; Voullaire et al, 2002; Malmgrem et al, 2002; Trussler et al, 2004), have all documented chromosome breakage. It has been postulated that chromosome breakage and whole chromosome aneuploidy are caused by different factors (Wells and Delhanty, 2000). In our study, chromosome breakage was observed and documented and two chromosome sites were mostly affected, namely 1p31 and 2q31. Both locations have been found to be fragile sites within the chromosomes (Richards, 2001). Where possible the FISH protocol was devised in order to detect and confirm chromosome breakage by employing two probes per chromosome (embryo 9.7 Table 4.7). Therefore, during this study CGH has demonstrated that chromosome breakage in human preimplantation embryos can be considered as a possible mechanism of aneuploidy and mosaicism.

During the two studies on preimplantation embryos and especially at the blastocyst stage more than 80 embryos and >2200 cells were analysed either by CGH or FISH for at least five chromosomes. Overall, it was shown that chromosome loss, either whole or partial was the main mechanism leading to mosaic aneuploid cell lines. Both studies were developed to produce accurate and reliable results, either by employing a second probe for the same chromosome to confirm the status of the cell (Chapter 3) or by utilising two techniques to confirm the results. It has been shown that CGH can provide reliable analysis on single cells as well as blastomeres. However, it still remains laborious and time-consuming and will probably not be used for clinical purposes in this form. Initially the CGH technique should be made less laborious and more reliable in order to be used in a clinical setting e.g. for PGD of chromosomal abnormalities. This can be achieved by reducing the time of hybridisation from 72 hours to less than 24. Wells and co-workers (2002) were able to reduce the time to 30 hours during a PGD case for aneuploidy screening, however, the CGH was applied to polar bodies. The use of DNA microarrays to act as hybridisation templates, instead of the target metaphase chromosomes that have been used so far could further reduce the period of hybridisation, and improve the analysis of the obtained results,

increasing the simplicity of interpretation by avoiding the need to karyotype metaphase chromosomes. DNA microarray or microarray CGH (M-CGH) analysis is a rapid evolving method of molecular analysis that could find several potential uses in PGD (Maughan et al, 2001; Clarke et al, 2001). Following in the steps of CGH, microarrays could replace the metaphase spreads that are used now to assess chromosomal imbalance in CGH. It is likely that their versatility will make them an attractive option for PGD. However, at present, technical imitations such as paucity of material available for hybridisation, sensitivity and reliability of the data, and the cost of producing appropriate microarrays are likely to hinder their application in PGD for some time (Braude et al, 2002). Chan et al (2002) designed a DNA disc chip array, based on comparative genomic hybridisation in order to study sperm and concluded that the technique was reliable but it, was still prone to manual processing involving the fluorescent microscope and computer. Furthermore, the whole genome amplification procedure could be replaced and MDA can be tested as an alternative method of WGA. This method uses a φ 29 DNA polymerase and random exonucleaseresistant primers, whilst DNA amplification takes place at 30°C (Dean et al, 2002). It has been shown that MDA provides a highly uniform representation of the genome, with the amplification bias being less than 3-fold among eight chromosomal loci, compared to the 4-6-fold observed with the application of DOP-PCR (Dean et al, 2002).

In the last part, two protocols were devised for carrying out PGD for a single gene disorder, namely myotonic dystrophy. These protocols were tested and carried out during clinical cases. Furthermore, a novel universal protocol was developed and tested on single cells for future application of clinical PGD cases for myotonic dystrophy. During the whole PGD protocol workup fluorescent PCR was utilised to provide the most sensitive and accurate results. An array of fluorescent polymorphic markers was tested initially for 10 different families at the genomic DNA level to observe which markers were informative for each family. Two different F-PCR protocols were tested and analysis on more than 700 single cells in order to optimise the protocols. Protocol 1 had been recently published for its efficiency (Piyamongkol *et al*, 2001a), because it employs a duplex method of detection is the most significant

problem during single cell PCR. Contamination can be caused by cumulus cells of maternal origin, sperm, culture media and the PCR products present in the laboratory environment (Delhanty, 1998; section 1.6.3.4.1). Protocol 2 was devised for family H and was clinically applied for the first time during this study. The novel side of Protocol 2 entails the use of a single-step multiplex F-PCR protocol including the DM mutation marker and the polymorphic marker D19S112. This protocol was able to detect the presence of contamination as well as the presence of ADO. ADO is the amplification of only one of the two parental alleles present in the single cell and can lead to misdiagnosis especially during PGD of a dominant genetic disease (Handyside et al, 1997; section 16.3.4.2). The D19S112 marker has a polymorphic nature therefore contamination can be detected in a sample specific fashion. An additional aspect of the D19S112 marker is that it is linked to the DM locus, and so can provide back-up diagnostic information for example when ADO has been detected for the DM locus. The use of linked markers for the adenomatous polyposis disease during clinical PGD has been reported previously (Ao et al, 1998). However, clinical application of a duplex F-PCR protocol for PGD of DM using the D19S112 linked marker was carried out only during this study. Unfortunately, no pregnancy was achieved in either of the PGD cycles performed.

Due to the extensive work in devising a patient-specific PGD protocol for the diagnosis of DM, the development of a universal-like protocol for the diagnosis of DM was tested initially in genomic DNA and subsequently in single cell DNA. The protocol was based on our previous experience with whole genome amplification procedures such as DOP-PCR and the use of fluorescent PCR markers. The protocol entailed amplification of DNA using WGA and then subsequent amplification of the DOP-PCR amplified DNA with the DM mutation marker (DM) as well as linked (D19S112) and unlinked polymorphic markers (D13S305, D18S535, D21S1414) for the DM disease. Initial testing a the genomic DNA level produced excellent results of amplification of the mutation and polymorphic markers (96-98%). Such amplification results prompted testing at the single cell DNA level. Twenty single cells from two heterozygote individuals were amplified using DOP-PCR and at least ten subsequent reactions (2 reactions per marker) were performed for each single cell. A decreased amplification rate and increased ADO rate was observed for the amplification (even during testing on various single cell DNA where results were not shown) of single

cells. The DM marker only showed acceptable amplification and an ADO rate of 85% and 10%. Respectively, CGH analysis was performed on the DOP-PCR amplified single cells and proved successful (85%). Hence, it can be concluded that the idea of a universal-like protocol for the PGD of DM is feasible however, requires refinement of the whole genome amplification protocol which may be achieved by changing the protocol to MDA (above) which offers a more reliable representation of genome. Further work should include purification fidelity of the polymorphic markers. Overall, it was shown that this protocol can be used for low quantities of DNA, however, certain alterations and further optimisation should be carried out to improve the reliability in the minute quantities present in single cells.

6.2 Ethical Considerations towards PGD

The wide media coverage regarding advances in ART is the source of most of the ethical concerns raised today concerning what could become possible in the future. Cloning, prenatal diagnosis, and gene therapy (Fletcher, 1978; Fiddler and Pergament, 1995; Fiddler *et al*, 1999) are research fields mainly linked with fear of excessive embryo and fetal experimentation. Consequently most countries where this technology exists now have in place, or are in the process of defining, ethical guidelines or legislation to regulate human embryo research.

The philosophy of PGD is to provide couples at risk of transmitting an inherited disorder with the option and assurance of selectively having an unaffected child, in cases of an unacceptably high risk for a defective child. The great advantage from the ethical point of view that PGD provides is that it avoids implantation of defective embryos, and this process of selection eliminates the need for future TOP. Couples at high risk are offered the opportunity to overcome the worrisome burden of a possible abortion, as affected embryos are detected *in vitro* and only healthy embryos are implanted (Raeburn, 1995). Furthermore, PGD is able to prevent disabling inherited disorders prior to embryonic development (Viville *et al*, 1998). By applying PGD in countries where abortions are illegal, the number of terminations due to genetic factors can be reduced (Viville *et al*, 1998). PGD avoids the heated debate of selective abortion in society and in individual cases (Fasouliotis and Schenker, 1998). The goal

of avoiding the birth of offspring with severe genetic handicaps is part of the procreative liberty and parental decision (Kanavakis and Synodinos, 2002). However, the selection of embryos on genetic grounds is not yet ethically acceptable by all countries.

Attitudes to embryo research vary widely from country to country, which has raised great debate towards the diagnosis of embryos at the preimplantation stage. Most concerns have been pointed to the fact that PGD can and will be used for eugenic purposes i.e. the possibility that embryos could be analysed for characteristics not related to health issues (Wells and Delhanty, 2001). PGD has been referred to as a vehicle of eugenics more powerful than any of its predecessors (Fasouliotis and Schenker, 1998). Therefore, in the UK centres that offer PGD are controlled by the Human Fertilisation and Embryology Authority (HFEA) (Harper and Delhanty, 2000). The Human Fertilisation and Embryology Act 1990 permits embryo research due to the following reasons: (1) promoting advances in the treatment of infertility, (2) Increasing knowledge about the causes of congenital disease, (3) Increasing knowledge about the causes of miscarriage, (4) Developing more effective techniques of contraception and (5) Developing methods for detecting the presence of gene or chromosome abnormalities in embryos before implantation. However, prohibited are reproductive cloning, inter-species fertilisation or transfer of human embryos into other species and gene therapy in the pre-embryo.

"Designer babies" and PGD raise real ethical dilemmas in certain unusual cases (Braude *et al*, 1998; Braude *et al*, 2001). A dilemma might occur with inherited deafness. In a recent case a non-hearing child was deliberately conceived to be deaf like its lesbian parents (Savulescu *et al*, 2002). Sex selection referred to as "family balancing" remains controversial (Gleincher and Karande, 2002) and many consider it not to be a legitimate use of PGD (Robertson, 2002). Finally, the attempt to save the life of a sibling by having another child provide a suitable tissue match can be rationalised and commended however, this process has met with great controversy (Boyle and Savulescu, 2002). In addition, further problems that PGD faces is the possibility of misdiagnosis (Harper *et al*, 1999; Harper and Delhanty, 2000; Wells, 2004).

CHAPTER 7

References

Abdelhadi I., Colls P., Sandalinas M., Escudero T., Munné S. (2003). Preimplantation genetic diagnosis of numerical abnormalities for 13 chromosomes. *Reprod. Biomed. Online*, **6**, 226-31.

Abou-Sleiman P.M., Apessos A., Harper J.C., Serhal P., Winston R.M. and Delhanty J.D. (2002). First application of preimplantation genetic diagnosis to neurofibromatosis type 2 (NF2). *Prenat Diagn.* **22**, 519-24

Acevedo N, Smith GD. (2005). Oocyte-specific gene signaling and its regulation of mammalian reproductive potential. *Front Biosci.*, **10**, 2335-45.

Almeida P.A. and Bolton V.N. (1995). The effect of temperature fluctuations on the cytoskeletal organisation and chromosomal constitution of the human oocyte. *Zygote*, **3**, 357-65.

Almeida P.A. and Bolton V.N. (1996). The relationship between chromosomal abnormality in the human preimplantation embryo and development in vitro. *Reprod Fertil Dev*, **8**, 235-41

Amack JD, Paguio AP, Mahadevan MS. (1999). Cis and trans effects of the myotonic dystrophy (DM) mutation in a cell culture model. *Hum Mol Genet*, **8**, 1975-84.

Amack JD and Mahadevan MS (2004). Myogenic defects in myotonic dystrophy. *Dev Biol*, **265**, 294-301

Amicucci P., Gennrelli M., Novelli G., Dallapiccola B. (2000). Prenatal diagnosis of DM using fetal DNA obtained from maternal plasma. *Clin Chem*, **46**, 301-302

Angell RR, Aitken RJ, van Look PF, Lumsden MA, Templeton AA. (1983) Chromosome abnormalities in human embryos after in vitro fertilization. *Nature*, **26**, 336-8.

Angell RR, Templeton AA and Aitken RJ (1986) Chromosome studies in human in vitro fertilization. *Hum Genet*, **72**, 333-339

Angell R.R., Sumner A.T., West J.D., Thatcher S.S. *et al* (1987). Post-fertilisation polyploidy in human preimplantation embryos fertilized in-vitro. *Hum Reprod*, **2**, 721-7

Angell R.R., Xian J., Keith J., Ledger W., Baird D.T. (1994a). First meiotic division abnormalities in human oocytes: Mechanism of trisomy formation. *Cytogenet Cell Genet*, **65**, 194-202

Angell RR (1994b) Possible pitfalls in preimplantation diagnosis of chromosomal disorders based on polar body analysis. *Hum Reprod*, **9**, 181-183.

Angell RR (1997). First-meiotic-division nondisjunction in human oocytes. Am J Hum Genet, 61, 23-32.

Antonarakis S.E. (1998). 10 Years of genomics, chromosome 21 and Down syndrome. *Genomics*, **51**, 1-16

Antonarakis SE, Petersen MB, McInnis MG, Adelsberger PA, Schinzel AA, Binkert F Pangalos C, Raoul O, Slaugenhaupt SA, Hafez M (1992) The meiotic stage of nondisjunction in trisomy 21: determination by using DNA polymorphism. Am J *Hum Genet*, **50**, 544-550

Antonarakis S.E., Lewis J.G., Adelsberg P.A., Schinzel A.A., Binkert F. (1991). Parental origin of the extra chromosome in trisomy 21 revisited: DNA polymorphism analysis suggests maternal origin in 95% of cases. *N Engl J Med*, **324**, 872-6

Antonarakis SE, Avramopoulos D, Blouin J-L, Talbot CC, Schinzel AA (1993) Mitotic errors in somatic cells cause trisomy 21 in about 4.5% of cases and are not associated with advanced maternal age. *Nat Genet*, **3**, 146-150

Anvret M., Ahlberg G., Grandell U., Hedberg B., Johnson K., Edstrom L. (1003). Larger expansions of the CTG repeat in muscle compared to lymphocytes from patients with myotonic dystrophy. *Hum Mol Genet*, **2**, 1397-1400

Ao A, Wells D, Handyside AH, Winston RM, Delhanty JD. (1998). Preimplantation genetic diagnosis of inherited cancer: familial adenomatous polyposis coli. *J Assist Reprod Genet*, **15**, 140-4

Apessos A, Abou-Sleiman PM, Harper JC, Delhanty JD. (2001). Preimplantation genetic diagnosis of the fragile X syndrome by use of linked polymorphic markers. *Prenat Diagn*, **21**, 504-11

Armstrong S.J., Goldman A.S., Speed R.M. (2000). Meiotic studies of a human male carrier of the common translocation, t(11;22), suggests postzygotic selection rather than preferential 3:1 MI segregation as the cause of liveborn offspring with an unbalanced translocation. *Am J Hum Genet*, **67**, 601-9

Baart E.B., Van Opstal D., Los F.J., Fauser B.C..M.J., Martini E. (2004). Fluorescence in situ hybridisation analysis of two blastomeres from day 3 frozen-thawed embryos followed by analysis of the remaining embryo on day 5. *Hum Reprod*, **19**, 685-693

Bahçe M, Cohen J, Munné S (1999) Preimplantation genetic diagnosis of aneuploidy: were we looking at the wrong chromosomes? *J Assist Reprod Genet*, **16**, 176-181.

Bahçe M, Escudero T, Sandalinas M, Morrison L, Legator M, Munné S (2000) Improvements of preimplantation diagnosis of aneuploidy by using microwave hybridization, cell recycling and monocolour labelling of probes. *Mol Hum Reprod*, **6**, 849-854.

Baker TG (1963) A quantitative and cytological study of germ cells in human ovaries. Proc R Soc Lond B 158: 417-433.

Balakier H., Cabaca O., Bouman D. *et al* (2000). Spontaneous blastomere fusion after freezing and thawing of early human embryos leads to polyploidy and chromosomal mosaicism. *Hum Reprod*, **15**, 2404-10

Barrett IJ, Lomax BL, Loukianova T, Tang SS, Lestou VS, Kalousek DK. (2001). Comparative genomic hybridization: a new tool for reproductive pathology. Arch Pathol Lab Med, 125, 81-4

Battaglia A and Carey JC. (1998) Wolf-Hirschhorn syndrome and Pitt-Rogers-Danks syndrome. Am J Med Genet. 1998 Feb 17;75(5):541. *Am J Med Genet*, **17**, 541.

Battaglia A, Carey JC, Cederholm P, Viskochil DH, Brothman AR, Galasso C. (1999) Natural history of Wolf-Hirschhorn syndrome: experience with 15 cases. *Pediatrics*, **103**, 830-6.

Battaglia DE, Goodwin P, Klein NA, Soules (1996) Influence of maternal age on meiotic spindle assembly in oocytes from naturally cycling women. *Hum Reprod*, **11**, 2217-2222.

Bean C.J., Hunt P.A., Millie E.A., Hassold T.J. (2001). Analysis of a malsegregating mouse Y chromosome: evidence that the earliest cleavage divisions of the mammalian embryo are non-disjunction-prone. *Hum Mol Genet*, **10**, 963-972

Bean C.J., Hassold T.J., Judis L., Hunt P.A. (2002). Fertilization in vitro increases non-disjunction during early cleavage divisions in a mouse model system. *Hum Reprod*, **17**, 2362-7

Becher R, Korn WM, Prescher G. (1997) Use of fluorescence in situ hybridization and comparative genomic hybridization in the cytogenetic analysis of testicular germ cell tumors and uveal melanomas. *Cancer Genet Cytogenet*, **93**, 22-8.

Begleiter M.L. and Ishmael H.A. (2000). FISH in prenatal diagnosis. *Prenat Diagn*, **8**, 686

Bennett R, Harris J (1997) The ethics of human embryo studies. In: Strachan T, Lindsay S, Wilson DI (eds) Molecular Genetics of Early Human Development. BIOS Scientific Publishers Ltd, Oxford, pp51-61.

Benkhalifa M, Janny L, Vye P, Malet P, Boucher D, Menezo Y. (1993). Assessment of polyploidy in human morulae and blastocysts using co-culture and fluorescent insitu hybridization. *Hum Reprod*, **8**, 895-902

Benkhalifia M, Ménézo Y, Janny L, Rouly JL, Qumsiyeh MB (1996) Cytogenetics of uncleaved oocytes and arrested zygotes in IVF programs. *J Assist Reprod Genet*, **13**, 140-148.

Benzaken B., LApierre J-M., Siffroi J.P., Chavlon A., Tacdjian G. (1998). Identification and characterisation of a de novo partial trisomy 10p by CGH. *Clin Genet*, **54**, 334-340

Bermudez M.G., Piyamongkol W., Tomaz S., Dudman E., Sherlock J.K., Wells D. (2003). Single-cell sequencing and mini-sequencing for preimplantation genetic diagnosis. *Prenat Diagn*, **23**, 669-77

Bermudez MG, Wells D, Malter H, Munne S, Cohen J, Steuerwald NM. (2004). Expression profiles of individual human oocytes using microarray technology. *Reprod Biomed Online*, **8**, 325-37.

Betteridge K.J., Hare W.C.D., Singh E.L. (1981). Approaches to sex selection in farm animals. *In New Technologies in Animal Breeding*, Bracket B.G., Seidel G.E., Seidel S.M. (eds). Academic Press: New York: 302-07

Bianchi D.W. (1999). Fetal cells in the maternal circulation: feasibility for prenatal diagnosis. *Br J Haematol*, **105**, 574-583

Bielanska, M., Tan, S.L., Ao, A. (2000). FISH of sex chromosomes in spermatozoa and spare preimplantation embryos of a Klinefelter 46,XY/47,XXY male. *Hum Reprod*, **15**, 440-444

Bielanska, M., Tan, S.L., Ao, A. (2002a). Chromosomal mosaicism throughout human preimplantation development in vitro: incidence, type, and relevance to embryo outcome. *Hum. Reprod*, **17**, 413-19.

Bielanska M., Tan S. L., Ao A. (2002b). Different probe combinations for assessment of postzygotic chromosomal imbalances in human embryos. *J Assis Reprod Genet*, **19**, 177-182

Bielanska M., Tan S. L., Ao A. (2002c). High rate of mixoploidy among human blastocysts cultured in vitro. *Fertil Steril*, **78**, 1248-53

Bielanska M, Tan SL, Ao A. (2003). Chromosomal information derived from single blastomeres isolated from cleavage-stage embryos and cultured in vitro. *Fertil Steril*, **79**, 1304-11.

Billi C., Divane A., Apessos A., Tassis K., Apostolos A., Barbarigos I., Theodoropoulos P., Florentin L. (2002). Prenatal diagnosis of common aneuploidies using quantitative PCR. *Prenat Diagn*, **22**, 360-365

Blanco L., Bernad A. Lazaro J.M., Martin G., Salas M. (1989). Highly efficient DNA synthesis by the phage $\varphi 29$ DNA polymerase. Symmetrical mode of DNA replication. *J Biol Chem*, **264**, 8935-40

Blennow E., Brøndum Nielsen K., Telenius H., Carter N.P., Kristoffersson U., Holmberg E., Gillberg C., Nordensköld M. (1995). Fifty probands with extra structurally abnormal chromosomes characterised by fluorescence in situ hybridisation. *Am J Med Genet*, **55**, 85-94.

Bond D.J., Chandley A.C. (1983). Aneuploidy. Oxford Monographs on medical genetics. Oxford University Press.

Bolton V.N., Hawes S.M., Taylor C.T., Parsons J.H. (1989). Development of spare human preimplantation embryos in vitro: an analysis of the correlations among gross morphology, cleavage rates, and development to the blastocyst. *J In Vitro Fert Embryo Transf*, **6**, 30-35.

Bolton V.N., Wren M.E., Parsons J.H. (1991). Pregnancies after in vitro fertilisation and transfer of human blastocysts. *Fertil Steril*, **55**, 830-832

Bonduelle M, Aytoz A, Van Assche E, Devroey P, Liebaers I, Van Steirteghem A (1998) Incidence of chromosomal aberrations in children born after assisted reproduction through intracytoplasmic sperm injection. *Hum Reprod*, **13**, 781-782.

Bonduelle M, Legein J, Derde MP, Buysse A, Schietecatte J, Wisanto A, Devroey P, Van Steirteghem A, Liebaers I (1995) Comparative follow-up study of 130 children born after intracytoplasmic sperm injection and 130 children born after in vitro fertilisation. *Hum Reprod*, **10**, 3327-3331.

Book JA and Santesson B (1960) Malformation syndrome in man associated with triploidy (69 chromosomes). *Lancet*, I, 858.

Boue A., Galano P. (1984). A collaborative study of the segregation of inherited chromosomal structural rearrangements in 1356 prenatal diagnoses. *Prenat Diagn*, **4**, 45-67

Boue A., Boue J., Gropp A. (1985). Cytogenetics of pregnancy wastage. Adv, Hum Genet, 14, 1-57

Boyle R and Savulescu J. (2002). Ethics of using Preimplantation Genetic Diagnosis to select a stem cell donor for an existing person. *BMJ*, **340**, 1240-1241.

Brandriff B., Gordon L., Ashworth L.K., Littman V., Watchmaker G., Carrano A.V. (1986). Cytogenetics of human sperm: meiotic segregation in two translocation carriers. *Am J Hum Genet*, **38**, 197-208

Braude P, Bolton V, Moore S (1988) Human gene expression first occurs between the four and eight-cell stages of preimplantation development. Nature 332: 459-461.

Braude P. (2001) Preimplantation Genetic Diagnosis and embryo research-human developmental biology in clinical practise. *Int J Dev Biol.*, **45**, 607-611.

Braude, P., Pickering, S., Flinter, F., Ogilvie, C.M. (2002). Preimplantation genetic diagnosis. *Nature*, **3**, 941-53.

Brook J.D., McCurrach, M.E., Harley H.G., Buckler A.J., Church D., Aburatani H., Hunter K., Stanton V.P., Thirion J.P., Hudson T., Sohn R., Snell R.G., Buxton J., Jones C., Juvonen V. *et al* (1992). Molecular basis of Myotonic dystrophy: expansion of a trinucleotide (CTG) repeat at the 3' end of transcript encoding a protein kinase family member. *Cell*, **68**, 799-808

Brugnoni R., Morandi L., Brambati B., Brioscioli V., Cornelio F. Mantegazza R. (1998). A new non-radioactive method for screening and prenatal diagnosis of myotonic dystrophy patients. *J Neurol*, **245**, 289-293

Bryndorf T., Kirchhoff M., Rose H., Maahr J., Gerdes T., Karhu R., Kallioniemi A., Philip J. (1995). CGH in clinical cytogenetics. *Am J hum Genet*, **57**, 1211-1220

Caskey C.T., Pizzuti A., Fu Y-H. et al (1992). Triplet repeat mutations in human disease. Science, 256, 784-8

Bryndorf T. Christensen B., Vad M., Perner J., Carelli M.P., Ward B.E., Phillip J. (1996). Prenatal detection of chromosome abnormalities in uncultured chorionic villus samples by FISH. *Am J Hum Genet*, **59**, 918-926

Bryndorf T. Christensen B., Vad M., Perner J., Brocks V., Phillip J. (1997). Prenatal detection of chromosome aneuploidies by FISH: experience with 2000 uncultured amniotic samples in a prospective clinical trial. *Prenat Diagn*, **17**, 333-341

Butterfield D., Roses A.D., Cooper M.L. (1974). A comparative electron spin resonance study of the erythrocyte membrane in myotonic muscular dystrophy. *Biochemistry*, **13**, 50-78-5082

Cassel M.J., Munne S., Fung J., Weier H.U.G. (1997). Carrier-specific breakpointspanning DNA probes: an approach to preimplantation genetic diagnosis in interphase cells. *Hum Reprod*, **12**, 2019-27

Cassidy SB, Whitworth T, Sanders D, Lorber CA, Engel E (1977) Five month extrauterine survival in a female triploid (69,XXX) child. *Ann Genet*, **20**, 277-279.

Chamberlain J.S., Gibbs R.A., Ranier J.E, Nquyen P.N., Caskey C.T., (1988).

Chadley A.C. (1991). On the parental origin of de novo mutation in man. J Med Genet, 28, 217-223

Chan P.J., Mann S.L., Corselli J.U., Pattom W.C., King A., Jacobson J.D. (2002). A simple DNA disc chip in a microarray design based on modified comparative genomic hybridisation for sperm DNA analysis. *Fertil Steril*, **77**, 1056-59

Deletion screening if the Duchenne muscular dystrophy locus via multiplex DNA amplification. *Nucleic Acids Res*, **15**, 253-257

Chang H.J., Clark R.D. and Bachman H. (1990). The phenotype of 45,X/46,XY mosaicism: an analysis of 92 prenatally diagnosed cases. *Am J Hum Genet*, **46**, 156-167

Chatzimeletiou K, Morrison EE, Panagiotidis Y, Prapas N, Prapas Y, Rutherford AJ, Grudzinskas G, Handyside AH. (2005). Comparison of effects of zona drilling by non-contact infrared laser or acid Tyrode's on the development of human biopsied embryos as revealed by blastomere viability, cytoskeletal analysis and molecular cytogenetics. *Reprod Biomed Online*, **11**, 697-710.

Chen X.Q., Stroun M., Magnenat J.L., Nicod L.P., Kurt A.M. (1996). Microsatellite alterations in plasma DNA of small cell lung cancer patients. *Nat Med*, **2**, 1033-35

Cheung, V. G. & Nelson, S. F. (1996). Whole genome amplification using a degenerate oligonucleotide primer allows hundreds of genotypes to be performed on less than one nanogram of genomic DNA. *Proc. Natl. Acad. Sci. USA* **93**, 14676–14679

Chong S.S., Kristjansson K., Cota J., Handyside A.H., Hughes M.R. (1993) Preimplantation prevention of X-linked disease: reliable and rapid sex determination of single human cells by restriction analysis of simultaneously amplified ZFX and ZFY sequences. *Hum Mol Genet*, **2**, 1187-1191

Christian S.L., Bhatt N.K., Martin S.A (1998). Integrated YAC contig map of the Prader-Willi/Angelman region on chromosome 15q11-q13 with an average STS spacing of 35kb. *Genome Res*, **8**, 146-7

Cieslak J, Ivakhnenko V, Wolf G, Sheleg S, Verlinsky Y (1999) Three-dimensional partial zona dissection for preimplantation genetic diagnosis and assisted hatching. *Fertil Steril*, **71**, 308-313

Cirigliano V., Canadas P., Plaja A., Ordonez E., Mediano, C., Farran I. (2003). Rapid prenatal diagnosis of aneuploidies and zygocity in multiple pregnancies by amniocentesis with single insertion of the needle and QF-PCR. *Prenat Diagn*, **23**, 629-33

Clouston H.J., Fenwick J., Webb A.L. *et al* (1997). Detection of mosaic and nonmosaic chromosome abnormalities in 6- to 8-day old human blastocysts. *Hum Genet*, **101**, 30-6

Clouston H.J, Herbert M., Fenwick J., Murdoch A.P., Wolstenholme J. (2002). Cytogenetic analysis of human blastocysts. *Prenat Diagn*, **12**, 1143-52.

Cohen J., Alikani M., Trowbridge J., Rosenwaks Z. (1992). Implantation enhancement by selective assisted hatching using zona drilling of human embryos with poor prognosis. *Hum Reprod*, 7, 685-91

Collins F.S. (1992).cystic Fibrosis: molecular biology and therapeutic implications. *Science*, **256**, 774-9

Conn C.M., Harper J.C., Winston R.M.L., Delhanty J.D.A. (1998). Infertile couples with Robertsonian translocations: preimplantation genetic analysis of embryos reveals chaotic cleavage divisions. *Hum Genet*, **102**, 117-123

Conn CM, Cozzi J, Harper JC, Winston RM, Delhanty JD. (1999). Preimplantation genetic diagnosis for couples at high risk of Down syndrome pregnancy owing to parental translocation or mosaicism. *J Med Genet*, **36**, 45-50

Coonen E., Derhaag J.G., Dumoulin J.C.M., Lucie C.P., van Wissen, Bras M., Janssen M., Evers J.L.H., Geraedts J.P.M. (2004). Anaphase lagging mainly explains chromosomal mosaicism in human preimplantation embryos. *Hum Reprod*, **19**, 316-

Corenelisse C.J., van Vliet M., Hermans J., Deville P. (1992). Fractional allelic imbalance in human breast cancer increases with tetraploidization and chromosome loss. *Int J Cancer*, **50**, 544-8

Cooke H.J., Hindley J. (1979). Cloning of human satellite III DNA: different components are on different chromosomes. *Nucleic Acids Res*, **6**, 3177-3197.

Cooke H.J., Hargreave T. and Elliot D.J. (1998). Understanding the genes involved in spermatogenesis: a progress report. *Fertil Steril*, **69**, 989-995

Cozzi J., Conn C.M., Harper J., Winston R.M.L., Rindl M., Delhanty J.D.A. (1999). A trisomic germ cell line and precocious chromatid segregation leads to recurrent trisomy 21 conception. *Hum Genet*, **104**, 23-8

Christensen B., Bryndorf T., Philip J. (1993). Prenatal diagnosis by in situ hybridisation on uncultured amniocytes: reduced sensitivity and potential risk of misdiagnosis in blood-stained samples. *Prenat Diagn*, **13**, 581-7

Cronier L., Bastide B., Defamie N., Niger C., Pointis G., Gasc J.M., Malassine A. (2001). Involvement of gap junctional communication and connexion expression in trophoblast differentiation of the human placenta. *Histol Histopathol*, **16**, 285-95

Cummins J.M. and Jequier A.M. (1994). Treating male infertility needs more clinical andrology, not less. *Hum Reprod*, **9**, 1214-1219.

Cummins J. (2000). The role of maternal mitochondria during oogenesis, fertilisation and embryogenesis. *Reprod Biomed Online*, **4**, 176-182

Dailey T., Dall B., Cohen J., Munne S. (1996). Association between nondisjunction and maternal age in meiosis I in human oocytes. *Am J Hum Reprod*, **59**, 176-84

Dale B, Gualtieri R, Talevi R, Tosti E, Santella L, Elder K (1991) Intercellular communication in the early human embryo. *Mol Reprod Dev*, **29**, 22-8.

Daniel A (1981) Structural differences in pericentric inversions. Application to a model of risk of recombinants. *Hum Genet*, **56**, 321-328.

Daniel A., Wu Z., Darmanian A., Collins F., Jackson J. (2003). Three different origins for apparent triploid/diploid mosaics. *Prenat Diagn*, **23**, 529-534

Daniels R, Monk M (1997) Gene expression in human preimplantation embryos. In: Strachan T, Lindsay S, Wilson DI (eds) Molecular Genetics of Early Human Development. BIOS Scientific Publishers Ltd, Oxford, pp155-170.

Daniely M, Aviram-Goldring A, Barkai G, Goldman B (1998) Detection of chromosomal aberration in fetuses arising from recurrent spontaneous abortion by comparative genomic hybridization. *Hum Reprod*, **13**, 805-809.

Daniely M, Barkai G, Goldman B, Aviram-Goldring A (1999) Structural unbalanced chromosome rearrangements resolved by comparative genomic hybridization. *Cytogenet Cell Genet*, **86**, 51-55.

Daphnis DD, Delhanty JD, Jerkovic S, Geyer J, Craft I, Harper JC. (2005). Detailed FISH analysis of day 5 human embryos reveals the mechanisms leading to mosaic aneuploidy. *Hum Reprod.*, **20**, 129-37.

Day T. and Taylor P.D. (1998). Chromosomal drive and the evolution of meiotic nondisjunction and trisomy in humans. *Proc Natl Acad Sci USA*, **95**, 2361-65

Day JW, Ranum LP (2005). RNA pathogenesis of the myotonic dystrophies. *Neuromuscular Disorders*, **15**, 5-16.

Dean N.L., Tan S.L., Ao A. (2001). The development of preimplantation genetic diagnosis for myotonic dystrophy using multiplex fluorescent polymerase chain reaction and its clinical application. *Mol Hum Reprod*, **7**, 895-901

Dean F.B., Nelson J.R., Giesler T.L., Lasken R.S. (2001). Rapid amplification of plasmid and phage DNA using Phi 29 DNA polymerase and multiply-primed rolling circle amplification. *Genome Res*, **11**, 1095-1099

Dean F.B., Hosono S., Fang L., Wu X., Faruqi A.F., Bray-Ward P., Sun Z., Zong Q., Du Y., Du J., Driscoll M., Song W., Kingsmore S.F., Egholm M., Lasken R.S. (2002). Comprehensive human genome amplification using multiple displacement amplification. *Proc Natl Acad Sci U S A*, **99**, 5261-6

de Boer KA, Catt JW, Jansen RP, Leigh D, McArthur S. (2004). Moving to blastocyst biopsy for preimplantation genetic diagnosis and single embryo transfer at Sydney IVF. *Fertil Steril*, **82**, 295-8.

de Die-Smulders C.E., Smeets H.J. Loots W. et al (1997). Paternal transmission of congenital Myotonic dystrophy. J Med Genet, 34, 930-4

Delhanty J.D.A., Griffin D.K. and Handyside A.H. (1993). Detection of aneuploidy and chromosomal mosaicism in human embryos during preimplantation sex determination by FISH. *Hum Mol Genet*, **2**, 1185

Delhanty J.D.A. and Handyside A.H. (1995). The origin of genetic defects in the human and their detection in preimplantation embryo. *Hum Reprod Update*, **1**, 201-15 Delhanty J.D.A. (1997). Chromosome analysis by FISH in human preimplantation genetics. *Hum Reprod*, **12**(11 Suppl), 153-5.

Delhanty J.D.A., Harper J.C., Asangla A., Handyside A.H., Winston R.M.L. (1997). Multicolour FISH detects frequent chromosomal mosaicism and chaotic division in normal preimplantation embryos from fertile patients. *Hum Genet*, **99**. 755-60

Delhanty J.D.A. and Harper J.C. (2000). Pre-implantation genetic diagnosis. Baillieres Best Pract Res Clin Obstet Gynaecol, 14(4), 691-708.

De Temmerman N, Sermon K, Seneca S, De Rycke M, Hilven P, Lissens W, Van Steirteghem A, Liebaers I. (2004). Intergenerational instability of the expanded CTG repeat in the DMPK gene: studies in human gametes and preimplantation embryos. *Am J Hum Genet*, **75**, 325-9

De Vos A, Van Steirteghem A. (2001) Aspects of biopsy procedures prior to preimplantation genetic diagnosis. *Prenat Diagn*, **21**, 767-80.

Devreker F., Winston R.M.L., Hardy K. (1998). Glutamine improves human preimplantation embryo development. *Fertil Steril*, **69**, 293-99

Dokras A., Sargent I.L., Ross C., Gaedner R.L., Barlow D.H. (1990). Trophectoderm biopsy in human embryos. *Hum Reprod*, **5**, 821-825

Dreesen J.C., Bras M., de Die-Smulders C., Dumoulin J.C., Cobben J.M., Evers J.L., Smeets H.J., Geraedts J.P. (1998). Preimplantation genetic diagnosis of spinal muscular atrophy. *Mol Hum Reprod*, **4**, 881-885.

Dreesen JC, Jacobs LJ, Bras M, Herbergs J, Dumoulin JC, Geraedts JP, Evers JL, Smeets HJ. (2000). Multiplex PCR of polymorphic markers flanking the CFTR gene; a general approach for preimplantation genetic diagnosis of cystic fibrosis. *Mol Hum Reprod*, **6**, 391-6

Drury K., Kovalinskaya R., William S. (1998). Polyploidy as a normal function of the trophoblastic development in human preimplantation embryos observed by fluorescent in situ hybridisation analysis. *Fertil Steril*, **70**(Suppl 1), 0-25

Dubel J.R., Armstrong R.M., Perryamn M.B., Epstein H.F., Ashizwa T. (1992). Phenotypic expression of the myotonic dystrophy gene in monozygotic twins. *Neurology*, **42**, 1815-17

Dumoulin J.C., Vanvuchelen R.C., Land J.A., Pieters M.H., Geraedts J.P., Evers J.L. (1995). Effect of oxygen concentration on in vitro fertilization and embryo culture in the human and the mouse. *Fertil Steril*, 63, 115-9.

Dumoulin J.C., Bras M., Coonen E., Dreesen J., Geraedts J.P., Evers J.L. (1998). Effect of Ca^{2+}/Mg^{2+} -free medium on the biopsy procedure for preimplantation genetic diagnosis and further development of human embryos. *Hum Reprod*, **13**, 2880-2883.

Dumoulin J.C.M., Van Wissen L.C.P., Menheree P.P.C.A., Geraedts J.P., Evers J.L. (1997). Taurine acts as an osmolyte on human and mouse oocytes and embryos. *Biol Reprod*, **56**, 739-44

Dumoulin JC, Bras M, Coonen E, Dreesen J, Geraedts JP, Evers JL. (1998). Effect of Ca2+/Mg2+-free medium on the biopsy procedure for preimplantation genetic diagnosis and further development of human embryos. *Hum Reprod*, **13**, 2880-3.

Dunn T.M., Grunfeld L., Kardon N.B. (2001). Trisomy 1 in a clinically recognized IVF pregnancy. Am J Med Genet, 99, 152-3

Durban M., Benet J., Sarquella J., Egozcue J. and Navarro J. (1998). Chromosome studies in first polar bodies from hamster and human oocytes. *Hum Reprod*, **13**, 583-587

Eddy EM, O'Brien DA (1993) The spermatozoan. In: Knobil E, Neil JD (eds) The Physiology of Reproduction (Second Ed) Raven Press, New York, pp29-78.

Edgar DH, Bourne H, Speirs AL, McBain JC (2000) A quantitative analysis of the impact of cryopreservation on the implantation potential of human early cleavage stage embryos. *Hum. Reprod*, **15L**, 175-179

Edwards M.J., Park J.P., Graham Jr J.M. (1994). Mixoploidy in humans: two surviving cases of diploid/tetraploid mixoploidy and comparison with diploid/triploid mixoploidy. *Am J Med Genet*, **52**, 324-30

Eggerding F.A., Iovannisci D.M. *et al* (1995). Fluorescence-based oligonucleotide ligation assay for analysis of cystic fibrosis transmembrane conductance regulator gene mutations.. *Hum Mutat*, **5**, 153-65

Eggermann T, Zerres K, Eggermann K, Moore G., Wollmann HA (2002). Uniparental disomy: clinical indications for testing in growth retardation. *Eur J Pediatr*, **161**, 305-12

Eggermann T, Curtis M, Zerres K, Hughes HE. (2004). Maternal uniparental disomy 16 and genetic counseling: new case and survey of published cases. Genet Couns, 15, 183-90

Egozcue J. (1996). Preimplantation diagnosis in older patients: To biopsy or not to biopsy? Of course, not. *Hum Reprod*, **11**, 2077-2078.

Egozcue J., Santalo J., Gimenez C., Perez N., Vidal F. (2000). Preimplantation genetic diagnosis. *Mol Cellular Endocr*, 77, 21-25

Eiben B., Bartels I., Bahr-Porsch S., Borgmann S., Gatz G., Gellert G., Goebel R., Hammans W., Hentemann M., Osmers R. (1990). Cytogenetic analysis of 750 spontaneous abortions with the direct-preparation method of chorionic villi and its implications for studying genetic causes of pregnancy wastage. *Am J Hum Genet* **47**, 656-663

Eiben B., Trawicki W., Hammans W., Goebel R., Epplen J. (1998). A prospective comparative study on FISH of uncultured amniocytes and standard karyotype analysis. *Prenat Diagn*, **18**, 901-906

Eibling L., Colot M. (1985). Abnormal development and transport and increased sister-chromatid exchange in preimplantation embryos following superovulation in mice. *Mut Res*, **147**, 189-195

Eisenberg B., Wapner R.J. (2002). Clinical Procedures in prenatal diagnosis. Best Pract Res Clin Obstet Gynaecol, 16, 611-627

Eknog R. and Wolfe J. (1998). Advances in fluorescent *in situ* hybridisation. *Curr Op Biotech.* **9**, 19-24

Ellegren H. (2004). Microsatellites: simple sequences with complex evolution. *Nat Rev Genet*, **5**, 435-445

El-Hashemite N. and Delhanty J.D.A (1997). A technique for eliminating allele specific amplification failure during DNA amplification of heterozygous cells for preimplantation diagnosis. *Mol Hum Reprod*, **3**, 975-8

Ellison J, Dean M, Goldman D. (1993) Efficacy of fluorescence-based PCR-SSCP for detection of point mutations. *Biotechniques*. **15**, 684-91

Engel E. (1993). Uniparental disomy revisited: the first twelve years. Am J Med Genet, 46, 670-4.

Engel W., Murphy D., Schmid M. (1996). Are there genetic risks associated with microassisted reproduction?. *Hum Reprod*, **11**, 2359-70

Engel E. (1997). Uniparental disomy (UPD). Genomic imprinting and a case for new genetics (prenatal and clinical implications: the "Likon" concept). *Ann Genet*, **40**, 24-34

ESHRE PGD Consortium Steering Committee (1999). ESHRE Preimplantation Genetic Diagnosis (PGD) Consortium: preliminary assessment of data from January 1997 to September 1998. *Hum Reprod*, **14**, 3138-3148

ESHRE preimplantation genetic diagnosis (PGD) consortium: data collection II (2000). *Hum Reprod*, **15**, 2673-83

ESHRE Preimplantation Genetic Diagnosis Consortium: data collection III (2001). Hum Reprod, 17, 233-46 ESHRE Preimplantation Genetic Diagnosis Consortium: data collection III. (2002). *Hum. Reprod.*, **17**, 233-46.

ESHRE Preimplantation Genetic Diagnosis Consortium: data collection IV (2004). *Hum Reprod*, **20**, 19-34

Ess KC, Kamp CA, Tu BP, Gutmann DH (2005). Developmental origin of subependymal giant cell astrocytoma in tuberous sclerosis complex. Neurology, 64, 1446-9.

Estop A.M., Van Kirk V., Cieply K. (1995). Segregation analysis of four translocations, t(2;18), t(3;15), t(5;7) and t(10;12) by sperm chromosome studies and a review of the literature. *Cytogen Cell Genet*, **70**, 60-67

Estop A.M., Cieply K.M., Munne S., Feingold E. (1999). Multicolour FISH analysis of the spermatozoa of a male heterozygous for a reciprocal translocation t(11;22)(q23;q11). *Hum Genet*, **104**, 412-7

Evsikov S. and Verlinsky Y. (1998). Mosaicism in the inner cell mass of human blastocysts. *Hum Reprod*, **13**, 3151-55

Evsikov S., Cieslak J., Verlinsky Y. (2000). Effect of chromosomal translocations on the development of preimplantation human embryos in vitro. *Fertil Steril*, **74**, 672-677

Fasouliotis S.J. and Schenker J.G. (1998). Preimplantation genetic diagnosis principles and ethics. *Hum Reprod*, **13**, 2238-2245.

Feldman B., Ebrahim S.A.D., Hazan S.L., Gyi K., Johnson M.P., Evans M.I. (2000). Routine prenatal diagnosis of aneuploidies by FISH studies in high-risk pregnancies. *Am J Med Gen*, **90**, 233-238

Fiddler M., Pergament D., Pergament E. (1999). Prenatal gene therapy: prospects and issues. *Prenat Diagn*, **15**, 1303-11.

Fiddler M, Pergament E. (1999). The role of the preimplantation geneticist in human cloning. *Prenat Diagn*, **19**, 1200-4

Fillipova G.N., Thienes C.P., Penn B.H. (2001). CTCF-binding sites flank CTG/CAG repeats and form a methylation –sensitiveinsulator at the DM1 locus. *Nat Genet*, 28, 335-341

Findlay I., Ray P., Quirke P. (1995a). Allelic drop-out and preferential amplification in single cell s and human blastomeres: implications for preimplantation genetic diagnosis. *Hum Reprod*, **10**, 1609-18 Findlay I., Urquhart A., Quirke P. et al (1995b). Simultaneous DNA fingerprinting, diagnosis of sex and single gene defect status from single cells. *Hun Reprod*, **10**, 1005-10

Findlay I., Mathews P. and Quirke P. (1999). Preimplantation genetic diagnosis using fluorescent polymerase chain reaction: results and future developments. *J Assist Reprod Genet*, **16**, 199-206

Findlay I. (2000). Preimplantation genetic Diagnosis. Br Med Bull, 56, 672-90

Fiorentino F., Magli M.C., Podini D., Ferraretti A.A., Vitale N., Gianaroli L. (2003). The minisequencing method: an alternative strategy for PGD of single gene disorders. *Mol Hum Reprod*, **9**, 399-410

Fisher J.M., Morton N.E., Harvey J.F, Jacobs P.A. (1991). Trisomy 18: studies of the parent and cell division of origin and the defect of aberrant recombination on nondisjunction. *Am J Hum Genet*, **556**, 669-675

Fisher J.M., Harvey J.F., Morton N.E., Jacobs P.A. (1995). Trisomy 18: studies of the parent and cell division of origin and the effect of aberrant recombination on nondisjunction. *Am J Hum Genet*, **56**, 669-75

FIVNAT - French In Vitro National (1995) Pregnancies and births resulting from in vitro fertilisation: French national registry, analysis of data 1986 to 1990. *Fertil Steril* 64, 746-756.

Flach G, Johnson MH, Braude PR, Taylor RAS, Bolton VS (1982) The transition from maternal to embryonic control in the 2-cell mouse embryo. *EMBO J*, **1**, 681-686 Fletcher J. (1978). Ethics and recombinant DNA research. *South Calif Law Rev.* **51**, 1131-40

Fleming T.P., Javed Q., Collins J., Hay M. (1993). Biogenesis of structural intercellular junctions during cleavage mouse embryo. *J Cell Sci Suppl*, **17**, 119-25

Fleming T.P., Ghassemifar M.R., Sheth B. (2000). Juctional complexes in the early mammalian embryo. *Semin Reprod Med*, **18**, 185-93

Focault F., Praz F., Jaulin C., Amor-Gueret M. (1996). Experimental limits of PCR analysis of (CA)n repeat alterations. *Trends Genet*, **12**, 450-452

Forrestier F., Hohlfeld P., Vial Y., Olin P., Andreux J.P., Tissot J.D. (1996). Blood smears and prenatal diagnosis. *Br J Haematol*, **95**, 278-80

Foster HA, Abeydeera LR, Griffin DK, Bridger JM. (2005). Non-random

chromosome positioning in mammalian sperm nuclei, with migration of the sex chromosomes during late spermatogenesis. *J Cell Sci*, **118**(Pt 9), 1811-20

Frayling (2002). Methods of molecular analysis: mutation detection in solid tumours. *Mol Pathol.*, **55**, 73-9.

Fritz B, Hallermann C, Olert J, Fuchs B, Bruns M, Aslan M, Schmidt S, Coerdt W, Muntefering H, Rehder H. (2001) Cytogenetic analyses of culture failures by comparative genomic hybridisation (CGH)-Re-evaluation of chromosome aberration rates in early spontaneous abortions. *Eur J Hum Genet*, **9**, 539-47.

Fu Y.H., Pizzuti A., Fenwick R.G. *et al* (1992). An unstable triplet repeat in a gene related to Myotonic muscular dystrophy. *Science*, **260**, 235-8

Fung J., Munne S., Duell T., Weier H.U.G. (1998). Rapid cloning of translocation breakpoints: from blood to YAC in 50 days. *J Biochem Mol Biol Biophys*, **1**, 181-92

Fuscoe JC, Collins CC, Pinkel D, Gray JW (1989) An efficient method for selecting unique clones from DNA libraries and its application to fluorescent staining of human chromosome 21 using in situ hybridisation. *Genomics*, **5**, 100-109

Gardner R.L. (1971). Manipulations on the blastocyst. Adv Biosc, 6, 279-96

Gardner R.J.M., Sutherland G.R. (1996). Chromosome abnormalities and genetic counselling. 2nd edn. Oxford University Press

Garner D.K., Lane M. (2005). Ex vivo embryo development and effects on gene expression and imprinting. *Reprod Fertil Dev*, **17**, 361-70

Gardner D.K. (1998). Development of serum-free media for the culture and transfer of human blastocysts. *Hum Reprod*, **13** (Suppl 4), 367-82

Garside W.T., Loer De Mola J.T., Bucci J.A. (1997). Sequential analysis of zona thickness during in vitro culture of human zygotes: correlation with embryo quality, age and implantation. *Mol Reprod Dev*, **47**, 99-104

Gaulden ME (1992) Maternal age effect: The enigma of Down syndrome and other trisomic conditions. *Mut Res*, **296**, 69-88.

Geifman-Holtzman O. and Fay K. (1998). Prenatal diagnosis of congenital myotonic dystrophy and counselling of the pregnant mother. *Am J Med Genet*, **78**, 250-253

Gianaroli L., Magli M.C., Munne S., Fiorentino A., Montanaro N., Ferraretti A.P. (1997a). Will preimplantation genetic diagnosis assist patients with a poor prognosis to achieve pregnancy? *Hum Reprod*, **12**, 1762-7.

Gianaroli L., Magli M.C., Ferraretti A.P., (1997b). Preimplantation genetic diagnosis increase the implantation rate in human in vitro fertilisation by avoiding the transfer of chromosomaly abnormal embryos. *Fertil Steril*, **69**, 1128-1131

Gianaroli L., Magli M.C., Ferraretti A.P., Munne S. (1999). Preimplantation diagnosis for aneuploidies in patients undergoing in vitro fertilization with a poor prognosis: identification of the categories for which it should be proposed. *Fertil Steril*, **72**, 837-44

Gianaroli L. (2000). Preimplantation Genetic Diagnosis: polar body and embryo biopsy. *Hum Reprod*, **15** (Suppl **4**), 69-75

Gianaroli L., Magli M.C., Ferraretti A.P. (2001a). The in vivo and in vitro efficiencies and efficacy of PGD for aneuploidy (2001). *Mol Cell Endocrinol*, **183**, S13-18

Gianaroli L., Magli M.C., Ferraretti A.P., Tabanelli C., Trombetta C., Boudjema E. (2001b). the role of PGD for aneuploidy. *Reprod Biomed Online*, **4**, 31-36

Gianaroli L, Magli MC, Ferraretti AP, Munné S, Balicchia B, Escudero T, Crippa A (2002). Possible interchromosomal effect in embryos generated by gametes from translocation carriers. *Hum. Reprod*, **17**, 3201-7.

Gibbons WE, Gitlin SA, Lanzendorf SE, Kaufmann RA, Slotnick RN, Hodgen GD (1995) Preimplantation genetic diagnosis for Tay-Sachs disease; Successful pregnancy after pre-embryo biopsy and gene amplification by polymerase chain reaction. *Fertil Steril*, **63**, 23-728.

Glassman A.B. (1997). Cytogenetics. An evolving role in the diagnosis and treatment of cancer. *Clin Lab Med.* **17**(1), 21-37

Gleincher N, Karande V. (2002). Gender selection for non-medical reasons. *Fertil. Sterility*, **78**, 460-462

Glenn CC, Driscoll DJ, Yang TP, Nicholls RD (1997). Genomic Imprinting: potential function and mechanisms revealed by the Prader-Willi and Angelman syndromes. *Mol Hum Reprod*, **3**, 321-32

Goel V., Glazier R., Summers A., Holzapfel S. (1998). Psychological aoutcomes foloowing maternal serum screening: a cohort study. *CMAJ*, **159**, 651-656

Goumy C., Bonnet-Dupeyron M.N., Cherasse Y., Laurichesse H., Jaffray J.Y., Lacroute G., Geneix A., Lemery D. and Vago P. (2004). Chorionic villus sampling (CVS) and fluorescence *in situ* hybridisation (FISH) for a rapid first-trimester prenatal diagnosis. *Prenat Diagn*, **24**, 249-256

Grati FR, Grimi B, Frascoli G, Di Meco AM, Liuti R, Milani S, Trotta A, Dulcetti F, Grosso E, Miozzo M, Maggi F, Simoni G. (2006). Confirmation of mosaicism and

uniparental disomy in amniocytes, after detection of mosaic chromosome abnormalities in chorionic villi. *Eur J Hum Genet*, **14**, 282-8.

Griffin DK, Handyside AH, Penketh RJ, Winston RM, Delhanty JD. (1991). Fluorescent in-situ hybridisation to interphase nuclei of human preimplantation embryos with X and Y chromosome specific probes. *Hum Reprod*, **6**, 101-5

Griffin D.K., Wilton L.J., Handyside A.H. (1992). Dual fluorescent in situ hybridisation for simultaneous detection of and Y chromosome-specific probes for the sexing of human preimplantation embryonic nuclei. *Hum Genet*, **89**, 18-22

Griffin D.K., Wilton L.J., Handyside A.H., Winston R., Delahnty J.D.A. (1993). Pregnancies following the diagnosis of sex in preimplantation embryos by FISH. *BMJ*, **306**, 1382-83

Griffin D.K., Handyside A.H., Harper J.C., Wilton L.J., Atkinson G., Soussis I., Wells D., Kontogianni E., Tarin J., Geber S. *et al* (1994). Clinical experience with PGD of sex by dual fluorescent hybridisation. *J Assist Reprod Genet*, **11**, 132-43

Griffin D.K., Abruzzo M.A., Millie E.A., Sheean L.A., Feingold E., Sherman S.L., Hassold T.J. (1995). Non-disjunction in human sperm: evidence for an effect of increasing paternal age. *Hum Mol Genet*, **4**, 2227-32

Griffin D.K. (1996). The incidence, origin and aetiology of human aneuploidy. *Intern Rev Cytol*, **167**, 263-296

Griffin D.K., Sanoudou D., Adamski E., McGiffert C., O'Brien P., Wienberg J., Ferguson-Smith M.A. (1998). Chromosome specifc CGH for determining the origin of intachromosomal duplications. *J Med Genet*, **35**, 37-41

Griffin D.K., Hyland P., Tempest H.G., Homa S.T. (2003). Should men undergoing ICSI be screened for chromosomal abnormalities in their sperm? *Hum Reprod*, 18, 229-35

Grifo JA, Boyle A, Fisher E, Lavy G, DeCherney AH, Ward DC, Sanyal MK (1990) Pre-embryo biopsy and analysis of blastomeres by in situ hybridisation. *Am J Obstet Gynecol*, **163**, 2013-2019.

Grifo JA, Tang YX, Cohen J, Gilbert F, Sanyal MK, Rosenwaks Z (1992) Pregnancy after embryo biopsy and co-amplification of DNA from X and Y chromosomes. *J Am Med Ass*, **268**, 727-729.

Grifo JA, Tang YX, Munne S, Alikani M, Cohen J, Rosenwaks Z (1994) Healthy deliveries from biopsied human embryos. *Hum Reprod*, **9**, 912-916

Grimshaw G.M., Szczepura A., Hulten M., McDonald F., Nevin N.C., Dhanjal S. (2003). Evaluation of moleculat tests for prenatal diagnosis of chromosome abnormalities. *Health Technology Assessment*, **7**, 1-56

Gutierez-Mateo C., Wells D., Benet J., Sanchez-Garcia J.F., Bermudez M.C., Belil I., Munne S., Navvaro J. (2004). Reliability of CGH to detect chromosome abnormalities in first polar bodies and MII oocytes. *Hum Reprod*, **19**, 2118-2125

Guttenbach M., Michemann H.W., Hiney B., Engel W., Smidt M. (1997). Segregation of sex chromosomes into sperm nuclei in a man with 47 XXY Klinefelter's karyotype: a FISH analysis. *Hum Genet*, **99**, 474-477

Handyside A.H., Pattinson J.K., Penketh R.J., Delhanty J.D.A. et al (1989). Biopsy of human preimplantation embryos and sexing by DNA amplification. *Lancet*, 1, 347-9 Handyside AH, Kontogianni EH, Hardy K, Winston RML (1990) Pregnancies from biopsied human preimplantation embryos sexed by Y-specific DNA amplification. *Nature*, **344**, 768-770.

Handyside AH (1991) Biopsy of human cleavage stage embryos and sexing by DNA amplification. In: Verlinsky Y, Strom C (eds) Proceedings of the first symposium on preimplantation genetics. Plenum, New York, pp75-83.

Handyside A.H. and Delahnty J.D.A. (1993). Cleavage stage biopsy of human embryos and diagnosis of X-linked recessive disease. In: Edwards R.G. (ed). *Preimplantation diagnosis of human genetic disease*, Cambridge University Press, 239-270

Handyside AH, Lesko JG, Tarin JJ, Winston RML, Hughes MR (1992) Birth of a normal girl after in vitro fertilisation and preimplantation diagnostic testing for cystic fibrosis. *N Engl J Med*, **327**, 905-909.

Handyside AH (1993) Diagnosis of inherited disease before implantation. *Reprod* Med Rev, 2, 51-61

Handyside A.H. (1996). Preimplantation genetic diagnosis today. *Hum Reprod*, 11 (Suppl 1), 139-55

Handyside A.H. and Delhanty J.D.A (1997). Preimplantation genetic diagnosis: strategies and surprises. *TIG*, **13**, 270-5

Handyside A.H. (1998). Clinical evaluation of preimplantation genetic diagnosis. *Prenat Diagn*, **18**, 1345-8

Handyside A.H., Robinson M.D., Simpson R.J., Omar M.B., Shaw M.A., Grudzinskas J.G., Rutherford A. (2004). Isothermal whole genome amplification from

single and small numbers of cells: a new era for preimplantation genetic diagnosis of inherited disease. *Mol Hum Reprod*, **10**, 767-772

Hardy K, Martin KL, Leese HJ, Winston RML, Handyside AH (1990) Human preimplantation development in vitro is not adversely affected by biopsy at the 8-cell stage. *Hum Reprod*, **5**, 708-714

Hardy K. and Handyside A.H. (1992). Biopsy of cleavage stage human embryos and diagnosis of single gene defects by DNA amplification. *Arch Pathol Lab Med*, **116**, 388-92

Hardy K., Winston R.M.L., Handyside A.H. (1993). Binucleate cells in human preimplantation embryos in vitro: failure of cytokinesis during cleavage. *J Reprod Fertil*, **98**, 549-558

Harley H.G., Rundle S.A., MacMillan J.C. *et al* (1993). Size of the unstable CTG repeat sequence in relationto the phenotype and parental transmission in mytonic dystrophy. *Am J Hum Genet*, **52**, 1164-74

Harnden DG. (1985) Inherited factors in leukaemia and lymphoma. Leuk Res. 9(6):705-7.

Harper P.S. (1975). Congenital myotonic dystrophy in Britai. II. Genetic Basis. Arch Dis Child, **50**, 514-21

Harper P.S. (1989). Myotonic Dystrophy, 2nd ed. (London: W.B. Saunders Co)

Harper JC, Coonen E, Raemakers FCS, Delhanty JDA, Handyside AH, Winston RML, Hopman AHN (1994a) Identification of the sex of human preimplantation embryos in two hours using an improved spreading method and fluorescent in-situ hybridisation (FISH) using directly labelled probes. *Hum Reprod*, **9**, 721-724.

Harper J and Handyside AH (1994b). The current status of preimplantation diagnosis. *Curr. Obstet. Gynecol*, **4**, 143-9.

Harper, J.C., Robinson, F., Duffy, S., Griffin, D.K., Handyside, A.H., Delhanty, J.D., Winston, R.M. (1994c). Detection of fertilization in embryos with accelerated cleavage by fluorescent in-situ hybridization (FISH). *Hum. Reprod*, **9**, 1733-7.

Harper J.C., Coonen E., Handyside A.H. Winston R.M.L., Delhanty J.D.A. (1995). Mosaicism of autosomes and sex chromosomes in morphologically normal, monospermic preimplantation human embryos. *Prenat Diagn*, **15**, 41-9

Harper J.C. and Delhanty J.D.A. (1996). Detection of chromosomal abnormalities in human preimplantation embryos using FISH. *J Assist Reprod Genet*, **13**(2), 137-9.

Harper J.C. and Wells D. (1999). Recent advances and future developments in PGD. *Prenat Diagn*, **19**, 1193-99

Harper J.C. and Delhanty J.D.A. (2000). Preimplantation genetic diagnosis. Curr Opin Obstet Gynaecol, 12, 67-72

Harper J.C., Delhanty J.D.A., Handyside A. (2000). Preimplantation genetic diagnosis. Wiley 2000

Harper J.C., Wells D., Piyamongkol W. Abou-Sleiman P., Apessos A., Ioulianos A., Davis M., Doshi A., Serhal P., Ranieri M., Rodeck C., Delhanty JDA. (2002). Preimplantation genetic diagnosis for single gene disorders: experience with five single gene disorders. *Prenat Diagn*, **22**, 525-33

Harper JC and Bui TH (2002). Pre-implantation genetic diagnosis. Best Pract. Res. Clin. Obstet. Gynaecol., 16, 659-70.

Harris S., Moncrieff C., Johnson K. (1996). Myotonic Dystrophy: will the real gene please step forward! *Hum Mol Genet*, **5**, 1417-23

Harrison R.H., Kuo H.C., Scriven P.N., Handyside A.H., Ogilvie C.M. (2000). Lack of cell cycle checkpoints in human cleavage stage embryos revealed by a clonal pattern of chromosomal mosaicism analysed by sequential multicolour FISH. *Zygote*, **8**, 217-224

Harton G.L., Tsipouras P., Sisson M.E., Starr K.M., Mahoney B.S, Fugger E.F., Schulman J.D., Kilpatrick M.W., Levinson G., Black S.H. (1996). Preimplantation genetic testing for Marfan syndrome. *Mol Hum Reprod*, **2**, 713-715

Hartwell L.H. and Weinert E.A. (1989). Checkpoints: controls that ensure the order of cell cycle events. *Science*, **246**, 629-34

Hassold T., Chen N., Funkhouser J. et al (1980). A cytogenetic study of 1000 spontaneous abortions. Ann Hum Genet, 44, 151-76

Hassold T. and Jacobs P.A. (1984). Trisomy in man. Annual Rev of Genet, 18, 69-97

Hassold T. and Chiu D. (1985). Maternal age specific rates of numerical chromosome abnormalities with special reference to trisomy. *Hum Genet*, **70**, 11-17

Hassold T.J. (1986). Chromosome abnormalities in human reproductive wastage. *Trends Genet*, **2**, 105-110.

Hassold T.J., Jacobs P.A., Leppert M., Sheldon M. (1987). Cytogenetic and molecular studies of trisomy 13. *J Med Genet*, **24**, 725-732.

Hassold T, Benham F, Leppert M (1988) Cytogenetic and molecular analysis of sexchromosome monosomy. *Am J Hum Genet*, **42**, 534-541

Hassold T.J., Warburton D., Klein J. and Stein Z. (1995). Recombination and maternal age-dependant nondisjunction: molecular study of trisomy 16. *Am J Hum Genet*, **57**, 867-874

Hassold T., Abruzzo M., Adkins K., Griffin D. et al (1996). Human aneuploidy: Incidence, Origin and Aetiology. *Environ Molecul Mutagen*, **28**, 167-75

Hassold T. (1998). Nondisjunction in human male. Curr Top Dev Biol, 37, 363-406

Hattori M, Yoshioka K, Sakaki Y. (1992) High-sensitive fluorescent DNA sequencing and its application for detection and mass-screening of point mutations. *Electrophoresis*. **13**, 560-5.

Hauge X.Y. and Litt M. (1993). A study of the origin of 'shadow bands' seen even when typing dinucleotide repeat polymorphisms by the PCR. *Hum Mol Genet*, **2**, 411-415

Hertig A.T., Rock J., Adams E.C. and Mulligan W.J. (1954). On the preimplantation stages of the human ovum: a description of four normal and four abnormal specimens ranging from the second to the fifth day of development. *Contrib Embryol*, **35**, 201-20 HFEA - Human Fertilisation and Embryology Authority, 12th Annual Report (2003) Paxton House, 30 Artillery St, London E1, UK.

Hilgenfeld E., Padilla H., Schrock E., Ried T. (1999). Analysis of B-cell neoplasias by SKY. *Curr Top Mocrobiol Immunol*, **95**, 20-32

Hixon M., Millie E., Judis L.A., Sherman S., Hassold T. (1998). FISH studies of the sperm of fathers of paternally derived cases of trisomy 21: no evidence of increase of aneuploidy. *Hum Genet*, **103**, 654-657

Hodge S. (1989). Waiting for amniocentesis. N Eng J Med, 320, 63-64

Holzgreve W., Tercanli S., Surbek D., Miny P. (1999). Invasive diagnostic methods. In *Fetal medicine – basic science and clinical practice* (Rodeck C.H. and Whittle M.J.). 417-434. Edinburgh: Churchill-Livingstone,1999.

Honda H., Miharu N., Ohasi Y. (1999). Analysis of segregation and an euploidy in two reciprocal translocation carriers, t(3;9)(q26.2;q32) and t(3;9)(p25;q32), by triple0colour fluorescence in situ hybridisation. *Hum Genet*, **105**, 428-36

Hook E.B., Hammeron J.L. (1977). The frequency of chromosome abnormalities detected in consecutive newborn studies – differences between studies – results by sex

and severity of phenotypic involvement. In: Hook E.B., Porter L.H. Population Cytogenetics, NY: Academic Press, 66-79

Hopman AH, Moesker O, Smeets AW, Pauwels RP, Vooijs GP, Ramaekers FC (1991). Numerical chromosome 1, 7, 9, and 11 aberrations in bladder cancer detected by in situ hybridisation. *Cancer Res*, **51**, 644-51

Houghton F.D. and Leese H.J. (2004). Metabolism and developmental competence of the preimplantation embryo. *Europ J Obstet Gynecol*, **115S**, S92-S96

Hu D.G., Webb G., Hussey N. (2004). Aneuploidy detection in single cells using DNA array-based CGH. *Mol Hum Reprod*, **10**, 283-9

Huang Q., Schantz S.P., Rao P.H., McCormick S.A., Chagnati R.S.K. (2000). Improving DOP-PCR-CGH for analysis of DNA copy numbers changes in tumors. *Genes Chrom Cancer*, **28**, 395-403

Hulten M.A., Dhanjal S. and Pertl B. (2003). Rapid and simple prenatal diagnosis of common chromosome disorders: advantages and disadvantages of the molecular methods FISH and QF-PCR. *Reproduction*, **126**, 279-297

International Human Genome Sequencing Consortium (2001). Initial sequencing and analysis of the human genome. *Nature*, **409**, 860-921

In't Veld P.A., Brandenburg H., Verhoeff A., Dhont M., Los F. (1995). Sex chromosomal abnormalities and intracytoplasmic sperm injection. *Lancet*, **346**, 773.

In't Veld P.A., Weber R.F., Los F.J., den Hollander N., Dhont M., Pieters M.H., van Hemel J.O. (1997). Two cases of Robertsonian translocations in oligozoospermic males and their consequences for pregnancies induced by intracytoplasmic sperm injection. *Hum Reprod*, **12**, 1642-1644

Inzunza J, Iwarsson E, Fridström M, Rosenlund B, Sjöblom P, Hillensjö T, Blennow E, Jones B, Nordenskjöld M, Ährlund-Richter L (1998) Application of single-needle blastomere biopsy in human preimplantation genetic diagnosis. *Prenat Diagn*, **18**, 1381-1388

Ioulianos A, Wells D, Harper JC, Delhanty JD. (2000). A successful strategy for preimplantation diagnosis of medium-chain acyl-CoA dehydrogenase (MCAD) deficiency. *Prenat Diagn*, **20**, 593-8

Iselius L., Lidsten J., Fraccaro et al (1983). The 11q22q translocation; A collaborative study of 20 new cases and analysis of 110 families. *Hum Genet*, **64**, 343-355

Isola JJ, Kallioniemi OP, Chu LW, Fuqua SA, Hilsenbeck SG, Osborne CK, Waldman FM. (1995) Genetic aberrations detected by comparative genomic

hybridization predict outcome in node-negative breast cancer. *Am J Pathol.* **147**, 905-11.

Iwarsson E., Ährlund-Richter L., Inzunza J., Fridström M., Rosenlund B., Hillensjö T., Sjöblom P., Nordenskjöld M., Blennow E. (1998a) Preimplantation genetic diagnosis of DiGeorge syndrome. *Mol Hum Reprod*, **4**, 871-875.

Iwarsson E., Ährlund-Richter L., Inzunza J., Rosenlund B., Fridström M., Hillensjö T., Sjöblom P., Nordenskjöld M., Blennow E., (1998b). Preimplantation genetic diagnosis of a large pericentric inversion of chromosome 5. *Mol Hum Reprod*, 4, 719-723.

Iwarsson E. Lundqvist M., Inzunza J., Sjöblom P., Nordenskjöld M., Blennow E, (1999). A high degree of aneuploidy in frozen-thawed human preimplantation embryos. *Hum Genet*, **104**, 376-82.

Iwarsson E., Malmgren H., Inzinza J., Rosenlud B., Sjoblom P. *et al* (2000). Highly abnormal cleavage divisions in preimplantation embryos from translocation carriers. *Prenat Diagn*, **20**, 1038-47

Jackson L. (2002). Cytogenetic and molecular cytogenetics. *Clin Obstetr Gynecol*, **45**, 622-639

Jacobs P.A., Angell R.R., Buchannan I.M. et al (1978). The origin of human triploids. Ann Hum Genet, **42**, 49

Jacobs PA (1981) Mutation rates of structural rearrangements in man. Am J Hum Genet 33, 44-54.

Jacobs PA, Szulman AE, Funkhouser J, Matsuura JS Wilson CC (1982) Human triploidy: relationship between parental origin of the additional haploid complement and development of partial hydatidiform mole. *Ann Hum Genet*, **46**, 223-231.

Jacobs PA, Hassold TJ, Whittington E, Butter G, Collyer S, Keston M, Lee M (1988) Klinefelter's syndrome: an analysis of the origin of the additional sex chromosome using molecular probes. *Ann Hum Genet*, **52**, 93-109.

Jacobs PA (1992) The chromosome complement of human gametes. *Oxf Rev Reprod Biol*, **14**, 47-72.

Jacobs PA, Broune C, Gregson N, Joyce C, White H (1992) Estimates of the frequency of chromosome anomalies detectable using moderate levels of banding. *J Med Genet*, **29**, 103-108.

James L and Varley J. (1996) Advances in cytogenetic analysis of solid tumours. *Chromosome Res.* **4**, 479-85. James R.M and West J.D. (1994). A chimaeric animal model for confined placental mosaicism. *Hum Genet*, **93**, 603-4

Jamieson M.E., Coutts J.R.T., Connor J.M. (1994). The chromosome constitution of human preimplantation embryos fertilised in vivo. *Hum Reprod*, **9**, 709-15

Jansen G., de Jong P.J., Amemiya C., Aslanidis C., Shaw D.J., Harley H.G., Brook J.D., Fenwick R. Tsilfidis C. (1992). Physical and genetic characterisation of the distal segment of the myotonic dystrophy area on 19q. *Genomics*, **13**, 509-17

JansenG., Willems P., Coerwinkel M., Nillesen W., Smeets H., Vits L., Brunner H., Wieringa B. (1994). Gonosomal mosaicism in myotonic dystrophy patients: involvement of mitotic events in (CTG)n repeat variation and selection against extreme expansion in sperm. *Am J Hum Genet*, **54**, 575-85

Jansen R. (2000). Germline passage of mitochondria: quantitative considerations and possible embryological sequelae. *Hum Reprod*, **15** (suppl 2), 112-128

Jeffreys A.J., Thein S.L., (1985). Hypervariable minisatellite regions in human DNA. *Nature*, **314**, 67-73

John H.A., Birnstiel M.L., Jones K.W. (1969). RNA-DNA hybrids at the cytological level. *Nature*, **223**, 582-87

Jones G.M., Trounson A.O., Lolatgis N, Wood C. (1998) Factors affecting the success of human blastocyst development and pregnancy following in vitro fertilization and embryo transfer. *Fertil Steril.*, **70**, 1022-1029

Joos S, Bergerheim US, Pan Y, Matsuyama H, Bentz M, du Manoir S, Lichter P. (1995) Mapping of chromosomal gains and losses in prostate cancer by comparative genomic hybridization. *Genes Chromosomes Cancer*, **14**, 267-76.

Joris H, De Vos A, Janssens R, Devroey P, Liebaers I, Van Steirteghem A. (2003).

Comparison of the results of human embryo biopsy and outcome of PGD after zona drilling using acid Tyrode medium or a laser. *Hum Reprod*, **18**, 1896-902.

Kallioniemi A., Kallioniemi O.P., Sudan D. (1992). Comparative genomic hybridisation for molecular cytogenetic analysis of solid tumours. *Science*, **258**, 818-21

Kallionemi OP, Kallionemi A, Piper J Isola J, Waldman FM, Gray JW, Pinkel D (1994) Optimizing comparative genomic hybridisation for analysis of DNA sequence copy number changes in solid tumours. *Genes Chrom Cancer*, **10**, 231-243.

Kalousek DK, Howard-Peebles PN, Olson SB, Barrett IJ, Dorfmann A, Black SH, Schulman JD, Wilson RD. (1991). Confirmation of CVS mosaicism in term placentae

and high frequency of intrauterine growth retardation association with confined placental mosaicism. *Prenat Diagn*, **11**, 743-50

Kalousek DK, Vekemans M. (1996). Confined placental mosaicism. *J Med Genet*, **33**, 529-33.

Kalousek D.K. (2000). Pathogenesis of chromosomal mosaicism and its effect on early human development. *Am J Med Genet*, **91**, 39-45

Kanavakis E. and Synodinos J.T. (2002). Preimplantation genetic diagnosis in clinical practice. *J Med Genet*, **39**, 6-11

Kaplan B., Wolf G., Kovalinskaya L. Verlinsky Y. (1995). Viability of embryos flowing second polar body removal in a mouse model. *J Assist Reprod Genet*, **12**, 742-749

Kappel B., Nielson J., Brogaard H., Mikkelsen M., Therkelssen A. (1987). Spontaneous abortion following mid-trimester amniocentesis. Clinical significance of placental perforation and blood-stained amniotic fluid. *Br Med J Obstet Gynaecol*, **94**, 50-54

Kent-First MG, Kol S, Muallem A, Blazer S, Itzkovitz-Eldor J (1996) Infertility in intracytoplasmic-sperm-injection derived sons. *Lancet*, **348**, 332.

Kiechle M, Jacobsen A, Schwarz-Boeger U, Hedderich J, Pfisterer J, Arnold N. (2001) Comparative genomic hybridization detects genetic imbalances in primary ovarian carcinomas as correlated with grade of differentiation. *Cancer*, **1**;**91**, 534-40.

Kilby M.D., Brackley K.J., Walters J.J., Morton J., Roberts E., Davison E.V. (2001). First-trimester prenatal diagnosis of a familiar sub-telomeric translocation. *Ultrasound Obstet Gynecol*, **17**, 531-533

Kimpton C.P., Gill P., WantonA., Urquhart A., Milican E.S., Adams M. (1993). Automated DNA profiling employing multiplex amplification of short tandem repeat loci. *PCR Methods Appl*, **3**, 13-22

King W., Proffitt J., Morrison L., Piper J., Lane D., Seelig S. (2000). The role of FISH technologies in molecular diagnostics and disease management. *Mol Diagn*, **5**, 309-319

Kirchhoff M., Gerders T., Rose H., Maahr J., Ottessen A.M., Lundsteen C (1998). Detection of chromosomal gains and losses in CGH analysis based on standard interval references. *Cytometry*, **31**, 162-173
Kirchhoff M., Gerders T., Maahr J., Rose H., Bentz M., Lundsteen C (1999). Deletions below 10Mb are detected in CGH by standard reference intervals. *Genes Chrom Cancer*, **25**, 410-3

Kirchhoff M., Rose H., Maahr J., Gerders T., Bugge M., Tommerup N., Tumer Z., Lespinasse J., Wirth J., Lundsteen C. (2000). High resolution comparative genomic hybridisation analysis reveals imbalances in ndyschromosomal patients with normal or apparently balanced conventional karyotypes. *Eur J Hum Genet*, **8**, 661-8

Kirchhoff M, Rose H, Lundsteen C (2001) High resolution comparative genomic hybridisation in clinical cytogenetics. *J Med Genet*, **38**, 740-4.

Kittler R., Stoneking M., Kayser M. (2002). A whole genome amplification method to generate long fragments fro mlow quantities of genomic DNA. *Anal Biochem*, **300**, 237-44

Klein C.A., Schmidt-Kittler O., Schardt J.A., Pantel K., Speicher M.R., Reithmuller G. (1999). Comparitive genomic hybridisation, loss of heterozygocity, and DNA sequence analysis of single cells. *Proc. Natl. Acad. Sci. USA*, **9**, 4494-4499.

Klesert TR, Cho DH, Clark JI, Maylie J, Adelman J, Snider L, Yuen EC, Soriano P, Tapscott SJ. (2000). Mice deficient in Six5 develop cataracts: implications for myotonic dystrophy. *Nat Genet*, **25**, 105-9.

Klingman I., Benadiva C., Alikani M., Munne S. (1996). The presence of multinucleated blastomeres in human embryos is correlated with chromosomal abnormalities. *Hum Reprod*, **11**, 1492-8

Knapp L.A. (2005). Denaturing gradient gel electrophoresis and its use in the detection of major histocompatibility complex polymorphism. *Tissue Antigens.*, **65**, 211-9.

Knight S.J.L., Regan R., Nicod A., Flint J. (1999). Subtle chromosomal rearrangements in children with unexplained mental retardation. *Lancet*, **354**, 1676-81 Knight S.J.L. and Flint J. (2000). Perfect endings: a review of subtelomeric probes and their use in clinical diagnosis. *J Med Genet*, **37**, 401-409

Koch M.C., Grimm T., Harley H.G., Harper P.S. (1991). Genetic Risks for children of women with myotonic dystrophy. *Am J Hum Genet*, **48**, 1084-91

Kokkola A, Monni O, Puolakkainen P, Larramendy ML, Victorzon M, Nordling S, Haapiainen R, Kivilaakso E, Knuutila S (1997) 17q12-21 amplicon, a novel recurrent genetic change in intestinal type of gastric carcinoma: a comparative genomic hybridization study. *Genes Chromosomes Cancer*, **20**, 38-43.

Kola I, Wilton L (1991) Preimplantation embryo biopsy: detection of trisomy in a single cell biopsied from a four-cell mouse embryo. *Mol Reprod Develop*, **29**, 16-21

Kontogianni E.H., Griffin D.K. and Handyside A.H. (1996). Identifying the sex of human preimplantation embryos in X-linked disease: amplification efficiency of Y-specific alphoid repeat from single with two lysis protocols. *J. Assist Reprod Genetic*, **13**, 125-132

Kontogianni EH, Hardy K, Handyside AH (1991) Co-amplification of X- and Yspecific sequences for sexing preimplantation human embryos. In: Verlinsky Y, Strom C (eds) *Preimplantation Genetics*. New York Plenum, pp139-145

Kontogianni EH, Griffin DK, Handyside AH (1996) Identifying the sex of human preimplantation embryos in X-linked disease: amplification efficiency of a Y-specific alphoid repeat from single blastomeres with two lysis protocols. *J Assist Reprod Genet*, **13**, 125-132.

Korn WM, Oide Weghuis DE, Suijkerbuijk RF, Schmidt U, Otto T, du Manoir S, Geurts van Kessel A, Harstrick A, Seeber S, Becher R. (1996) Detection of chromosomal DNA gains and losses in testicular germ cell tumors by comparative genomic hybridization. *Genes Chromosomes Cancer*, **17**, 78-87.

Kuliev A, Rechitsky S, Verlinksty O, Ivakhnenko V, Evsikov S, Wolf G, Angastiniotis M, Georghiou D, Kukharenko V, Strom C, Verlinsky Y (1998) Preimplantation genetic diagnosis of thalassemias. *J Assist Reprod Genet*, 15, 219-225.

Kuo W.L., Tenjin H., Segraves R., Pinkel D., Gray J. (1991). Detection of aneuploidy involving chromosomes 13, 18 or 21 by FISH to interphase and metaphase amniocytes. *Am J Obstet Gynecol*, **174**, 886-890

Kuo H.C., Ogilvie C.M., Handyside A.H. (1998). Chromosomal mosaicism in cleavage-stage human embryos and the accuracy of single-cell genetic analysis. *J Assist Reprod Genet*, **15**, 276-80.

Lahiri D.K. and Nurnberger J.I. (1991). A rapid non-enzymatic method for the preparation of HMW DNA from blood for RFLP studies. *Nucleic Acid Res*, **19**, 5444

Lapierre J-M., Cacheux V., Luton D., Collot N., Aurias A., Tachdjian G. (2000). Analysis of uncultured amniocytes by CGH: a prospective prenatal study. *Prenat Diagn*, **20**, 123-131

Lareu M.V., Barral S., Salas A., Carracedo A. (1998). Sequence variation of a variable short tandem repeat at the D18S535 locus. *Int J Leg Med*, **111**, 337-9

Larsen W. (1997).Human embryology. Second Edition. Churchill Livingstone Publ. pp17.

Larkin K, Fardaei M (2001). Myotonic dystrophy – a multigene disorder. *Brain Res Bull*, **56**, 389-95

Lasken R.S., Egholm M. (2003). Whole genome amplification: abundant supplies of DNA from precious samples or clinical specimens. *Trends Biotechnol*, **21**, 531-535

Laverge H., De Sutter P., Verschraegen-Spae M.R., De Paepe A., Dhont M. (1997). Triple colour fluorescent in-situ hybridization for chromosomes X, Y and 1 on spare human embryos. *Hum Reprod*, **12**, 809-14.

Laverge H, Van deer Elst J., De Sutter P., De Paepe A., Dhont M. (1998). Fluorescent in-situ hybridisation on human embryos showing cleavage arrest after freezing and thawing. *Hum Reprod*, **13**, 425-9

Laws-King A., Trounson A., Sathananthan A.H., Kola I. (1987). Fertilisation of human oocytes by micromanipulation of a single spermatozoa under the zona pellucida. *Fertil Steril*, **48**, 637-642.

Leese H.J. (1998). Human embryo culture: back to nature. *J Reprod Genet*, **15**, 446-8 Lejeune J, Gautier M., Turpin R. (1959). Étude des chromosomes somatiques de neuf enfants mongoliens. *Comp Rend Acad Sci*, **248**, 1721-1722.

LeMarie-Adkins R., Radke R., Hunt P.A. (1997). Lack of checkpoint control at the metaphase/anaphase transition: a mechanism of meotic non-disjunction in mammalian females. *J Cell Biol*, **139**, 1611-19

Lestou VS, Lomax BL, Barrett IJ, Kalousek DK (1999). Screening of human placentas for chromosomal mosaicism using comparative genomic hybridization. *Teratology*, **59**, 325-30

Lestou VS, Desilets V, Lomax BL, Barrett IJ, Wilson RD, Langlois S, Kalousek DK. (2000). Comparative genomic hybridization: a new approach to screening for intrauterine complete or mosaic aneuploidy. *Am J Med Genet*, **92**, 281-4.

Levin M., Mercola M. (1998). Gap junction are invlolved in the early generation of the left-right asymmetry. Dev Biol, 203, 90-105

Levran D., Dor J., Rudak E., Nebel L., Ben-Shlomo I., Ben-Rafael Z., Mashiach S. (1990). Pregnancy potential of human oocytes--the effect of cryopreservation. *N Engl J Med*, **323**, 1153-6. Levy B., Gerschin I.F., Desnick R.J., Babu A., Gelb B.D., Cotter P.D. (1997). Characterisation of a de novo chromosome rearrangement by CGH and FISH. *Cytogenet Cell Genet*, **76**, 68-71

Lewis CM, Pinel T, Whittaker JC, Handyside AH. (2001) Controlling misdiagnosis errors in preimplantation genetic diagnosis: a comprehensive model encompassing extrinsic and intrinsic sources of error. *Hum Reprod*, **16**, 43-50

Lichter P, Cremer T, Tang CJ, Watkins PC, Manuelidis L, Ward DC (1988) Rapid detection of human chromosome 21 aberrations by in situ hybridisation. *Proc Natl Acad Sci USA*, **85**, 9664-9668.

Lichter P, Cremer T (1992) Chromosome analysis by non-isotopic in situ hybridisation. In: Rooney DE, Czepulkowski BH (eds) Human Cytogenetics, A Practical Approach vol I Constitutional Analysis. Oxford Univ Press, UK, pp157-192. Liehr T, Brude E, Gillessen-Kaesbach G, Konig R, Mrasek K, von Eggeling F, Starke H. (2005). Prader-Willi syndrome with a karyotype 47,XY,+min(15)(pter->q11.1:) and maternal UPD 15--case report plus review of similar cases. *Eur J Med Genet*, **48**, 175-81

Ligon AH, Beaudet AL, Shaffer LG (1997) Simultaneous multilocus FISH analysis for detection of microdeletions in the diagnostic evaluation of developmental delay and mental retardation. *Am J Hum Genet*, 61, 51-59

Liquori CL, Ricker K, Moseley ML, Jacobsen JF, Kress W, Naylor SL, Day JW, Ranum LP. (2001). Myotonic dystrophy type 2 caused by a CCTG expansion in intron 1 of ZNF9. *Science*, 293, 864-7.

Lissens W and Sermon K (1997). Preimplantation genetic diagnosis: current status and new developments. *Hum. Reprod*, **12**, 1756-61.

Liu J., Lissens W., Devroey P., Van Steirteghem A., Liebaers I. (1994) Amplification of X- and Y chromosome specific regions from single human blastomeres by polymerase chain reaction for sexing of preimplantation embryos. *Hum Reprod*, **9**, 716-720

Liu J, Lissens W, Van Broeckhoven C, Lofgren A, Camus M, Liebaers I, Van Steirteghem A (1995) Normal pregnancy after preimplantation DNA diagnosis of a dytrophin gene deletion. *Prenat Diagn*, **15**, 351-358

Liu, J., Tsai, Y.L., Zheng, X.Z., Yazigi, R.A., Baramki, T.A., Compton, G., Katz, E. (1998a). Feasibility study of repeated fluorescent in-situ hybridization in the same

human blastomeres for preimplantation genetic diagnosis. *Mol. Hum. Reprod*, **4**, 972-7.

Liu, J., Tsai, Y.L., Zheng, X.Z., Baramki, T.A., Yazigi, R.A., Katz, E. (1998b). Potential use of repeated fluorescent in-situ hybridization in the same human blastomeres for preimplantation genetic diagnosis. *Fertil Steril*, **70**, 729-733.

Lizardi P.M., Huan X., Zhu Z., Bray-Ward P. Thomas D.C., Ward D.C. (1998). Mutation detection and single-molecule counting using isothermal rolling circle amplification. *Nature Genet*, **19**, 225-232

Lo Y.M.D., Corbetta N., Chamberlain P.F., Rai V., Sargent I.L., Redman C.W.G., Wainscoat J.S. (1997). Presence of fetal DNA in maternal plasma and serum. *Lancet*, **350**, 485-87

Lomax B., Tang S., Separovic E., Phillips D., Hillard E., Thomson T., Kalousek D.K. (2000). Comparative genomic hybridisation in combination with flow cytometry improves results of cytogenetic analysis of spontaneous abortions. *Am J Hum Genet*, **66**, 1516-1521.

Lomax BL, Kalousek DK, Kuchinka BD, Barrett IJ, Harrison KJ, Safavi H. (1994). The utilization of interphase cytogenetic analysis for the detection of mosaicism. *Hum Genet*, **93**, 243-7.

Long S.E., Willians C.V. (1982). A comparison of the chromosome complement of inner cell mass and trophoblast cells in day-10 pig embryos. *J Reprod Fertil*, **66**, 645-648

Longo FJ (1963) Sperm aster in rabbit zygotes: its structure and function. *J Cell Biol*, **69**, 539-547.

Lopata A., Kolhman D., Johnston I. (1983). The fine structure of normal and abnormal human embryos developed in culture. In Beier, H.M. and Lindner, H.R. (eds), *Fertilisation of the human egg in-vitro*. Springer-Verlag, Heidelberg, 189

Lynch P.M., Howard L.D., Pleasants L.M. (1994). Molecular genetics of familial colorectal cancer. *Cancer Bull*, **46**, 315-21

Ma S.T. (1995). Applications and limitations of polymerase chain reaction amplification. *Chest*, **108**, 1393-1404

Ma K, Inglis JD, Sharkey A, Bickmore WA, Hill RE, Prosser EJ, Speed RM, Thomson EJ, Jobling M, Taylor K (1993) A Y chromosome gene family with RNA-

binding protein homology: candidates for the azoospermia factor AZF controlling human spermatogenesis. *Cell* **75**, 1287-1295.

Madan K and Menko FH (1992) Interchromosomal insertions: a case report and a review. *Hum Genet*, **89**, 1-9.

Magge A.C., Hughes A.E. (1998). Segregation distortion in myotonic dystrophy. J Med Genet, 35, 1045-6

Magli M.C., Jones G.M., Gras L., Gianaroli L., Korman I., Trounson A.O. (2000). Chromosome mosaicism in day 3 aneuploid embryos that develop to morphologically normal blastocysts in vitro. *Hum Reprod*, **15**, 1781-6

Magli MC, Sandalinas M, Escudero T, Morrison L, Ferraretti AP, Gianaroli L, Munne S. (2001a). Double locus analysis of chromosome 21 for preimplantation genetic diagnosis of aneuploidy. *Prenat Diagn*, **21**, 1080-5

Magli M.C., Gianaroli L., Ferrareti A.P. (2001b). Chromosomal abnormalities in embryos. *Mol Cel Endocrinol*, **183**, S29-34

Mahadevaiah S.K., Odorisio T., Elliot D.J., Rattigan A., Laval S.H., McCarrey J.R., Cattanach B.M., Burgoyne P.S. (1998). Mouse homologues of the human AZF candidate gene RBM are expressed in spermatogonia and spermatids, and map to a Y chromosome deletion interval associated with a high incidence of sperm abnormalities. *Hum Mol Genet*, **7**, 715-27

Mahmood R., Brierly C., Faed M.J.W, Mills J.A. and Delhanty J.D.A. (2000). Mechanisms of maternal aneuploidy: FISH analysis of oocytes and polar bodies in patients undergoing assisted conception. *Hum Genet*, **106**, 620-6

Malmgren H., Sahlen S., Inzunza J., Aho M., Rosenlund B., Fridstrom M., Hovatta O., Ahrlund-Richter L., Nordenskjold M., Blennow E. (2002). Single cell CGH analysis reveals a high degree of mosaicism in human embryos from patients with balanced structural chromosome aberrations. *Mol Hum Reprod*, **8**, 502-10.

Malter H.E., Cohen J. (1989). Partial zona dissection of the human oocyte: a non-traumatic method using micromanipulation to assist zona pellucida penetration. *Fertil Steril*, **51**, 139-148.

Mankodi A., Logigian E., Callahan L. (2000). Myotonic dystrophy in transgenic mice expressing an expanded CUG repeat. *Science*, **289**, 1769-1773

Mankodi A. Thornton C.A. (2002). Myotonic Syndromes. Curr Op Neurol, 15, 545-552 Du Manoir S., Speicher M.R., Joos S., Schrock E., Popp S., Donher H., Kovacs G., Lichter P., Cremer T. (1993). Detection of complete and partial chromosome gains and losses by comparative genomic in situ hybridisation. *Hum Genet*, **90**, 590-610

Mansfield E.S. (1993). Diagnosis of Down Syndrome and other aneuploidies using quantitative polymerase chain reaction and small tandem repeat polymorphisms. *Hum Mol Genet*, **2**, 43-50

Márquez C., Cohen J, Munné S. (1998). Chromosome identification in human oocytes and polar bodies by spectral karyotyping. *Cytogenet Cell Genet*, **81**, 254-258.

Márquez C., Sandalinas M., Bahçe M., Al M., Munné S. (2000) Chromosome abnormalities in 1255 cleavage-stage human embryos. *Reprod BioMed Online* 1, 1-12.

Martin R.H., Mahadevan M.M., Taylor p.J., Longsimpson L., Fleetham J. (1986). Chromosomal analysis of unfertilised human oocytes. *J Reprod Fertil*, **54**, 423-26

Martin K.H., Templado C., Ko E., Rademaker A. (1990). Effect of culture conditions and media on the frequency of chromosomal abnormalities in human sperm chromosome complements. *Mol Reprod Dev*, **26**, 101-104

Martin R.H. (1996). The risk of chromosomal abnormalities following ICSI. Hum Reprod, 11, 924, 925

Martin L.S., Ben-Yoseph Y., Ebrahim S.A.D. (1996). Laboratory techniques for prenatal diagnosis. In Hillier S.G., Kitchener H., Neilson J.P. (eds) *Scientific Essentials in Reproductive Medicine*. Saunders, Philadelphia, 300-11

Martini E., Flaherty S.P., Swann N., Mathews C., Geraedts J.P.M. (2000). FISH analysis of six chromosomes in unfertilised human oocytes after polar body removal. *J Assist Reprod Genet*, **17**, 276-283

McKusick V.A. (1992). *Mendelian inheritance in man*. John Hopkins, University Press, Baltomore.

McGinniss MJ, Kazazian Jr HH, Stetton G, Petersen MB, Boman H, Engel E, Greenberg F, Hertz JM, Johnson A, Laca Z, Mikkelsen M, Patil SR, Schinzel AA, Tranebjaerg L, Antonarakis SE (1992) Mechanisms of ring chromosome formation in 11 cases of human ring chromosome 21. *Am J Hum Genet*, **50**, 15-28.

McNeil, N., Ried, T. Novel molecular cytogenetic techniques for identifying complex chromosomal rearrangements: technology and applications in molecular medicine.

Exp. Rev. Mol. Med., http://www-ermm.cbcu.cam.ac.uk, accessed on 14 September 2000.

Megerian G, Ludomirsky A. (1994). Role of cordocentesis in perinatal medicine. Curr Opin Obstet Gynecol., 6, 30-5.

Menezo Y., Veiga A., Benkhalifa M. (1998). Improved methods for blastocyst formation and culture. *Hum Reprod*, **13** (Suppl 4), 256-65

Menezzo Y.J., Chiuteau J., Torello J., Girard A., Veiga A. (1999). Birth weight and sex ratio after transfer at the blastocyst stage in humans. *Fertil Steril*, **72**, 221-4

Meriano J, Clark C, Cadesky K, Laskin CA. (2004). Binucleated and micronucleated blastomeres in embryos derived from human assisted reproduction cycles. *Reprod Biomed Online*, **9**, 511-20.

Meschede D, Louwen F, Eiben B, Horst J (1997) Intracytoplasmic sperm injection pregnancy with fetal trisomy 9p resulting from a balanced paternal translocation. *Hum Reprod*, **12**, 1913-1914

Metz C.B., Monroy A (1985). Biology of Fertilisation, vol 2, Biology of The Sperm. Academic Press Inc Ltd, London

Mikki Y., Swensen J., Shattuck-Eidens D. et al (1994). A strong candidate for the breast and ovarian cancer susceptibility gene BRAC1. *Science*, **266**, 66-71

Moch H, Presti JC Jr, Sauter G, Buchholz N, Jordan P, Mihatsch MJ, Waldman FM. (1996) Genetic aberrations detected by comparative genomic hybridization are associated with clinical outcome in renal cell carcinoma. *Cancer Res*, **1**, 27-30.

Monckton D.G., Wong L., Ashiwaza T., Caskey C.T. (1995). Somatic mosaicism, germiline expansions, germiline reversions and intergenerational reductions in myotonic dystrophy males: small pool of PCR analyses. *Hum Mol Genet*, **4**, 1-8

Monk M., Holding C. (1990). Amplification of a ß-haemoglobin sequence in individual human oocytes and polar bodies. *Lancet*, **335**, 985-988

Montag M, van der Ven K, Delacretaz G, Rink K, van der Ven H (1998) Laserassisted microdissection of the zona pellucida facilitates polar body biopsy. *Fertil Steril*, **69**, 539-542.

Monteil M., Rousseuax S., Chevret E., Cozzi J., Sele B. (1997). Increased aneuploidy frequency in spermatozoa from a Hodgkin's disease patient after chemotherapy and radiotherapy. *Cytogenet Cell Genet*, **76**, 134-8

Moore DH 2nd, Pallavicini M, Cher ML, Gray JW. (1997) A t-statistic for objective interpretation of comparative genomic hybridization (CGH) profiles. *Cytometry*, **1;28**, 183-90.

Muller U., Weber J.L., Berry P. (1993). Second polar body incorporated into a blastomere results in 46,XX/69,XXX mixoploidy. *J Med Genet*, **30**, 597-600

Mullis K.B. (1990). The unusual origin of the polymerase chain reaction. *Sci Am*, **264**, 56-65

Mullis K.B., Faloona F., Scharf S. (1986). Specifc enzymatic amplification of DNA in vitro: the polymerase chain reaction. *Cold Spring Harb Symp Quant Biol*, **51**, 263-73

Munne S. and Cohen J. (1993). Unsuitability of multinucleated human blastomeres for preimplantation genetic diagnosis. *Hum Reprod*, **8**, 1120-5

Munne S. and Estop A.M. (1993a). Chromosome analysis of human spermatozoa stored in vitro. *Hum Reprod*, **8**, 581-6.

Munne S., Lee A., Rosenwaks Z., Grifo J., Cohen J. (1993b). Diagnosis of major chromosomal aneuploidies in human preimplantation embryos. *Hum Reprod*, **8**, 2185-91

Munné S, Grifo J, Cohen J, Weier H-UG (1994a) Chromosome abnormalities in human arrested preimplantation embryos: A multiple-probe FISH study. *Am J Hum Genet*, **55**, 150-159.

Munne S., Weier H.U.G., Grifo J. (1994b). Chromosome mosaicism in human embryos. *Biol Reprod* **51**, 373-9

Munne S. Alikani M., Tomkin G. (1995a). Embryo morphology, developmental rates and maternal age are correlated with chromosomal abnormalities. *Fertil Steril*, **64**, 382-91

Munne S., Dailey T., Sultan K.M. (1995b). The use of first polar bodies for preimplantation diagnosis of aneuploidy. *Hum Reprod*, **10**, 1015-21

Munne S. and Weier U. (1996). Simultaneous enumeration of chromosomes 13, 18, 21, X and Y in interphase cells for preimplantation genetic diagnosis of aneuploidy. *Cytogenet Cell Genet*, **75**, 263-70

Munne S. Magli C., Adler A. (1997). Treatment-related chromosome abnormalities in human embryos. *Hum Reprod*, **12**, 780-4

Munne S. and Cohen J. (1998). Chromosome abnormalities in human embryos. *Hum Reprod Update*, **4**, 842-55.

Munne S., Scott R., Sable D. and Cohen J. (1998a). Pregnancies after pre-conception testing of Robertsonian translocations of maternal origin. *Fertil Steril*, **69**, 675-81

Munne S., Morrison L., Fung J. *et al* (1998b). Spontaneous abortions are reduced after pre-conception diagnosis of translocations. *J Assist Reprod Genet*, **15**, 290-6

Munné S, Magli C, Bahce M *et al.* (1998c). Preimplantation diagnosis of the aneuploidies most commonly found in spontaneous abortions and live births XY, 13, 14, 15, 16, 18, 21, 22. *Prenat Diagn*, **18**, 1459-1466.

Munné S., Márquez C., Reing A., Garrisi J. and Alikani M. (1998d). Chromosome Abnormalities in Embryos Obtained After Conventional In Vitro Fertilization and Intracytoplasmic Sperm Injection. *Fertil Steril*, **69**, 904-8

Munne S., Marquez C., Magli C. et al (1998e). Scoring criteria for preimplantation genetic diagnosis of numerical abnormalities for chromosomes X, Y, 13, 16, 18 and 21. *Mol Hum Reprod*, **4**, 863-70

Munné S, Bahce M, Schimmel T, Sadowy S, Cohen J (1998f) Case report, Chromatid exchange and predivision of chromatids as other sources of abnormal oocytes detected by preimplantation genetic diagnosis of translocations. *Prenat Diagn*, **18**, 1450-1458.

Munné S, Fung J, Cassel MJ, Márquez C, Weier HUG (1998g) Preimplantation genetic analysis of translocations: case-specific probes for interphase cell analysis. *Hum Genet*, **102**, 663-674

Munne S, Magli C., Cohen J. Morton P., Sadowy S., Gianaroli L., Tucker M., Marquez C., Sable D., Ferraretti A.P., Massey J.B., Scott R. (1999). Positive outcome after preimplantation diagnosis of aneuploidy in human embryos. *Hum Reprod*, 14, 2192-99

Munne S., Sandalinas M., Escudero T., Fung J., Gianaroli L., Cohen J. (2000). Outcome of preimplantation genetic diagnosis of translocations. *Fertil Steril*, **73**, 1209-18

Munné, S. (2001). Preimplantation genetic diagnosis of structural abnormalities. *Mol. Cell. Endocrinol*, **183**(supplement), S55-8.

Munne S. (2002). Preimplantation genetic diagnosis of numerical and structural chromosome abnormalities. *Rep BioMed Onl*, **4**, 183-96

Munne S. and Wells D. (2002). Preimplantation genetic diagnosis. *Curr Opin Obstet Gynecol*, **14**, 239-44

Munne S. (2003). Preimplantation genetic diagnosis and human implantation. A Review. *Placenta*, 24, S70-6

Munne S., Sandalinas M., Escudero T., Vellila E., Walmsley R., Sadowy S., Cohen J., Sable D. (2003a). Improved implantation after preimplantation genetic diagnosis of aneuploidy. *Reprod Biomed Online*, **4**, 223-232

Munne S., Bahce M., Sandalinas M., Escudero T., Marquez C., Vellila E., Colls P., Oter M., Alikani M., Cohen J. (2004). Differences in chromosome susceptibility to aneuploidy and survival to 1st trimester. *Reprod Biomed Online*, **8**, 81-90

Munne S, Chen S, Fischer J, Colls P, Zheng X, Stevens J, Escudero T, Oter M, Schoolcraft B, Simpson JL, Cohen J (2005). Preimplantation genetic diagnosis reduces pregnancy loss in women aged 35 years and older with a history of recurrent miscarriages. *Fertil Steril*, **84**, 331-5.

Murray J.D., Moran C., Boland M.P., Nancarrow C.D., Sutton R., Hoskinson R.M., Scaramuzzi R.J. (1986). Polyploid cells in blastocysts and early fetuses from Australian Merino sheep. *J Reprod Fertil*, **78**, 439-446

Nakagawa M., Yamada H., Higuchi I., Kaminishi Y., Mikki T., Johnson K., Osame M. (1994). A case of paternally inherited congenital myotonic dystrophy. *J Med Genet*, **31**, 397-400

Nakahori Y., Hamanao K., Iwaya M., Nakagome Y. (1991). Sex identification by polymerase chain reaction using X-Y homologous primer. *Am J Med Genet*, **39**, 472-473.

Nakaoka Y., Okamoto E., Miharu N., Ohama K. (1998). Chromosome analysis in human oocytes remaining unfertilised after *in vivo* insemination: effect of maternal age and fertilisation rate. *Hum Reprod*, **13**, 419-24

Navot D., Drews M.R., Bergh P.A. (1994). Age related decline in female fertility is not due to diminished capacity of the uterus to sustain embryo implantation. *Fertil Steril*, **61**, 97-101

Nederlof P.M., van der Flier S., Wiegant J., Raap A.K., Tanke H.J., Ploem J.S., van der Ploeg M. (1990). Multiple fluorescent in situ hybridisation. *Cytometry*, **11**, 126-131.

Nicolaidis P., Petersen M.B. (1998). Origin and mechanisms of non-disjunction in human autosomal trisomies. *Hum Reprod*, **13(2)**, 313-9

Nielsen J, Wohlert M (1990) Sex chromosome abnormalities found among 34,910 newborn children: results from a 13-year incidence study in Arhus, Denmark. *Birth Defects Orig Artic Ser*, **26**, 209-223

Nielsen J, Wohlert M (1991) Chromosome abnormalities found among 34910 newborn children: results from a 13-year incidence study in Arhus, *Denmark. Hum Genet*, **87**, 81-83.

Ogilvie M. and Scriven P.N. (2004). Preimplantation genetic diagnosis for reciprocal translocations. *Prenat Diagn*, **24**, 553-555

Orita M., Suzuki Y., Sekiya T., Hayashi K. (1989). Rapid and sensitive detection of point mutations and DNA polymorphisms using PCR. *Genomics*, **5**, 874-79

Otten AD, Tapscott SJ (1995). Triplet repeat expansion in myotonic dystrophy alters the adjacent chromatin structure. *Procl Natl Acad Sci USA*, **92**, 5465-9

Ou C-Y. Kwok S., Mitchell S.W. (1988). DNA amplification fro direct detection of HIV-1 in DNA of peripheral blood mononuclear cells. *Science*, **239**, 295-97

Palanker D., Ohad S., Lewis A. (1991). Technique for cellular microsurgery using 193-nm excimer laser. *Laser Surg Med*, **11**, 580-6

Palermo G, Joris H, Devroey P, Van Steirteghem AC (1992) Pregnancies after intracytoplasmic sperm injection of single spermatozoon onto an oocyte. *Lancet*, **340**, 17-18.

Palermo G., Colombero L.T., Schattman G.L., Davis O.K., Rosenwaks Z. (1996). Evolution of pregnancies and initial follow-up of newborns delivered after ICSI. JAMA, 276, 1893-7

Pellestor F., Sele B., Jalbert H. (1987). Chromosome analysis of spermatozoa from a male heterozygous for a 13;14 Robertsonian translocation. *Hum Genet*, **76**, 116-120

Pellestor F., Girardet A., Lefort G., Andreo B., Charlieu JP. (1995). The cytogenetic analysis of human zygotes and preimplantation embryos. *Hum Reprod Update*. 1, 581-5

Pellicer A., Rubio C., Vidal F., Minguez Y., Egozcue J., Simon C. (1999). *In vitro* fertilisation plus PGD in patients with recurrent miscarriage: an analysis of chromosome abnormalities in human preimplantation embryos. *Fert Steril*, **71**, 1033-39

Papadopoulos G. Templeton A.A., Fisk N., Randall J. (1989). The frequency of chromosomal anomalies in human preimplantation embryos after *in vitro* fertilisation. *Hum Reprod*, **4**, 91-8

333

Pardue M.L., Gall J.G. (1969). Molecular hybridisation of radioactive DNA to the DNA of cytological preparations. *Proc Natl Acad Sci USA*. **64**, 600-4

Penrose LS, Delhanty JDA (1961) Triploid cell cultures from a macerated foetus. Lancet, I, 1261-1262.

Pergament E. (2000). New molecular techniques for chromosome analysis. Baillieres Best Pract Res Clin Obstet Gynaecol, 4, 677-90

Pergament E., Chen P.X., Thangavelu M., Fiddler M. (2000). The clinical application of FISH in prenatal diagnosis. *Prenat Diagn*, **20**, 215-20

Pergament E (2000). New molecular techniques for chromosome analysis. Baillieres Best Pract Res Clin Obstet Gynaecol., 14, 677-90.

Pertl B., Weitgasser U., Kopp S., Kroisel P.M., Sherlock J., Adinolfi M. (1996). Rapid detection of trisomy 18 and 21 and sexing with QF multiplex PCR. *Hum Genet*, **98**, 55-59

Pertl B, Yau SC, Sherlock J, Davies AF, Mathew CG, Adinolfi M (1994) Rapid molecular method for prenatal detection of Down syndrome. *Lancet*, **343**, 1197-1198.

Pertl B., Kopp S., Kroisel P.M., Tului L., Adinolfi M. (1999). Rapid detection of chromosome aneuploidies by quantitative fluorescent PCR: first application on 247 chorionic villus sampling. *J Med Genet*, **36**, 300-303

Pickering S., Braude P., Johnson M.H., Cant A., Currie J. (1990). Transient cooling to room temperature can cause irreversible disruption of the meiotic spindle in the human oocyte. *Fertil Steril*, **54**, 102-108

Pickering S.J., Taylor A., Johnson M.H., Braude P.R. (1995). An analysis of multinucleated blastomere formation in human embryos. *Mol Hum Reprod*, 1, see *Hum Reprod*, 10, 1912-22

Pickering S., Polidoropoulos N., Caller J., Scriven P., Ogilvie C.M., Braude P., and the PGD study group. (2003). Strategies and outcomes of first 100 cycles of preimplantation genetic diagnosis at the Guy's and St. Thomas' Center. *Fertil. Steril.*, **79**, 81-90

Pinkel D., Straume T., Gray J.W. (1986). Cytogenetic analyis using quantitative, highsensitivity, fluorescence hybridization. *Proc Natl Acad Sci USA*, **83**, 2934-38

Piyamongkol W., Harper J.C., Sherlok J.K., Delhnty J.D.A. (2001a). A successful strategy for preimplantation genetic diagnosis of myotonic dystrophy using multiplex fluorescent PCR. *Prenat Diagn*, **21**, 223-232

Piyamongkol W, Harper JC, Delhanty JD, Wells D. (2001b). Preimplantation genetic diagnostic protocols for alpha- and beta-thalassaemias using multiplex fluorescent PCR. *Prenat Diagn*, **21**, 753-9

Piyamongkol W., Bermudez M., Harper J., Wells D. (2003). Detailed investigation of factors influencing efficiency and allele drop-out in single cell PCR: implications for PGD. *Mol Hum Reprod*, **9**, 411-420

Plachot M., Mandelbaum J., Junca A.M., Salat-Baroux J., Cohen J. (1985). Impairment of human embryo development after abnormal in vitro fertilization. *Ann* NY Acad Sci, **442**, 336-41.

Plachot M., Junca A.M., Cohen J. et al (1987). Chromosome investigation in early life II. Human preimplantation embryos. *Hum Reprod*, **2**, 29-35

Plachot M, Veiga A, Montaggut J, De Grouchy J, Calderon G, Lepretre S, Junca AM, Santalo J, Carles E, Mandelbaum J (1988) Are clinical and biological IVF parameters correlated with chromosomal disorders in early life: a multicentric study. *Hum Reprod*, **3**, 627-635.

Plachot M., Mandelbaum J., Junca A.M., de Grouchy J., Salat-Baroux J, Cohen J. (1989). Cytogenetic analysis and developmental capacity of normal and abnormal embryos after IVF. *Hum Reprod*, **4**(8 Suppl), 99-103.

Prescher G, Bornfeld N, Hirche H., Horsthemke B., Jockel KH, Becher R. (1996). Prognostic implications of monosomy 3 in uveal melanoma. *Lancet*, **4**;347, 1222-5.

Quilter C.R., Holman S., Theodorides D., Hastings R.J., Delhanty J.D.A. (2001). Aneuploidy screening in direct chorionic villus samples by FISH: the use of commercial probes in a clinical setting. *BJOG*, **108**, 215-218

Racowsky C. (2002). High rates of embryonic loss, yet high incidence of multiple births in human ART: is this paradoxical. *Therionology*, **57**, 87-96

Raeburn J.A. (1995). Preimplantation diagnosis raises a philosophical dilemma. *BMJ*, **311**, 540

Ranum LP, Rasmussen PF, Benzow KA, Koob MD, Day JW. (1998). Genetic mapping of a second myotonic dystrophy locus. *Nat Genet*, **19**, 196-8.

Ranum LP, Day JW (2004). Myotonic dystrophy: RNA pathogenesis comes into focus. Am J Hum Gen, 74, 793-804

Ravi V, Wong MK. (2006). Strategies and methodologies for identifying molecular targets in sarcomas and other tumors. *Curr Treat Options Oncol*, **6**,487-9

Ray P.F., Winston R.M.L., Handyside A.H. (1994). Single cell analysis for diagnosis of cystic fibrosis and Lesch-Nyhan syndrome in human embryos before implantation. *Miami Biotechnology European Symposium, Advances In Gene Technology: Molecular Biology and Human Genetics*, **5**, 46

Ray P.F. and Handyside A.H. (1996). Increasing the denaturation temperature during the first cycles of amplification reduces allele dropout from single cells for preimplantation genetic diagnosis. *Mol Hum Reprod*, **2**, 213-8

Ray P.F., Ao A., Taylor D.M., Harper J.C., Hnadyside A.H. (1998). Assessment of the reliability of single blastomere analysis for preimplantation diagnosis of the Δ F508 deletion causing cystic fibrosis in clinical practice. *Prenat Diagn*, **18**, 1402-12

Ray P.F, Harper J.C., Ao A., Taylor D.M., Winston R.M.L., Hughes M., Handyside A.H. (1999). Successful preimplantation genetic diagnosis for sex linked Lesch-Nyhan syndrome using specific diagnosis. *Prenat Diagn*, **19**, 1237-1241

Raynham H., Gibbons R., Flint J., Higgs D. (1996). The genetic basis of mental retardation. *Q J Med*, **89**, 169-75

Rechitsky S., Freidine M., Verlinsky Y., Strom C. (1996). Allele drop-out in sequential PCR and FISH analysis of single cells (cell recycling). *J Assist Reprod Genet*, **13**, 115-124

Rechitsky S., Strom C., Verlinsky O., Kuliev A., Verlinsky Y. (1998). Allele dropout in polar bodies and blastomeres. *J Assist Reprod Genet*, **15**, 253-7

Reijo R, Lee T, Salo P, Alagapaan R, Brown LG, Rosenberg M, S Rozen, Jaffe T, Straus D, Hovatta O, de la Chapelle A, Silber S, Page DC (1995) Diverse spermatogenic defects in humans caused by Y chromosome deletions encompassing a novel RNA-binding protein gene. *Nat Genet*, **10**, 383-393.

Reiss J., Cooper D.N. (1990). Application of the polymerase chain reaction to the diagnosis of human genetic disease. *Hum Genet*, **85**, 1-8

Reubinoff B.E., Sushan A. (1996). To biopsy or not to biopsy. *Hum Reprod*, 11, 2071-75

Renaud J.F., Desnuelle C., Hugues M., Serratrice G., Lazunski M. (1986). Expression of apamin receptor in muscles of patients with myotonic muscular dystrophy. *Nature*, **319**, 678-6880

Reubinoff BE, Abeliovich D, Werner M, Schenker JG, Safran A, Lewin A (1998) A birth in non-mosaic Klinefelter's syndrome after testicular fine needle aspiration,

intracytoplasmic sperm injection and preimplantation genetic diagnosis. *Hum Reprod*, **13**, 1887-1892

Richards R.I. (2001). Fragile and unstable chromosomes in cancer: causes and consequences. *Trends in Genet*, **17**, 339-345

Ricker K, Koch MC, Lehmann-Horn F, Pongratz D, Otto M, Heine R, Moxley RT 3rd. (1994). Proximal myotonic myopathy: a new dominant disorder with myotonia, muscle weakness, and cataracts. *Neurology*, **44**, 1448-52.

Ried T., Landes G., Dackwski W., Klinger K., Ward D.C. (1992). Multicolour FISH for the simultaneous detection of probe sets for chromosomes 13, 18, 21, X and Y in uncultured amniotic fluid cells. *Hum Mol Genet*, **1**, 307-13

Ried T, Petersen I, Holtgreve-Grez H, Speicher MR, Schrock E, du Manoir S, Cremer T. (1994) Mapping of multiple DNA gains and losses in primary small cell lung carcinomas by comparative genomic hybridization. *Cancer Res*, **1**;**54**, 1801-6.

Robertson JA. (2002). Sex selection for gender variety by preimplantation genetic diagnosis. *Fertil Steril*, **78**, 463

Robinson WP, Barrett IJ, Bernard L, Telenius A, Bernasconi F, Wilson RD, Best R, et al (1997). A meiotic origin of trisomy in confined placental mosaicism is correlated with presence of fetal uniparental disomy, high levels of trisomy in trophoblast and increased risk of fetal intrauterine growth restriction. *Am J Hum Genet*, **60**, 917–927

Rodeck C.H. and Whittle M.J. (1999). Invasive diagnostic methods. In Fetal Medicine. p417-434, Edinburgh: Churchill-Livingstone, 1999.

Rommens J, Kerem BS, Greer W, Chang P, Tsui LC, Ray P. (1990) Rapid nonradioactive detection of the major cystic fibrosis mutation. *Am J Hum Genet.* **46**, 395-6.

Ronne M., Kirpekar F., Poulsen B.S. (1990). R-banding and *in situ* hybridization localization of single copy genes on high resolution banded chromosomes. *Anticancer Res*, **10**, 375-7.

Rose J., Gerkens S., Lynch S. (1993). Genetic susceptibility in familial multiple sclerosis not linked to the myelin basic protein gene. *Lancet*, **341**, 1179-81

Roudebush W.E., Kim J.G., Minhas B.S., Dodson M.G. (1990). Survival and cell acquisitional rates after preimplantation embryo biopsy: use of two mechanical techniques and two mouse strains. *Am J Obstet and Gynaecol*, **162**, 1084-90

Ruangvutilert P., Delhanty J.D.A., Serhal P., Simopoulou M., Rodeck C.H., Harper J.C. (2000a). FISH analysis on day 5 post-insemination of human arrested and blastocyst stage embryos. *Prenat Diagn*, **20**, 552-60

Ruangvutilert P., Delhanty J.D., Rodeck C.H., Harper J.C. (2000b). Relative efficiency of FISH on metaphase and interphase nuclei from non-mosaic trisomic or triploid fibroblast cultures. *Prenat Diagn*, **20**, 159-62

Rubio C., Simon C., Vidal F., Rodrigo L., Pellicer A. (2003). Chromosomal abnormalities and embryo development in recurrent miscarriage couples. *Hum Reprod*, **18**, 182-88

Saiki RK, Scharf S, Faloona F, Mullis KB, Horn GT, Erlich HA, Arnheim N (1985) Emzymatic amplification of alpha-globin genomic sequences and restriction site analysis for diagnosis of sickle cell anaemia. *Science*, **230**, 1350-1354.

Salvado C.S., Trounson A.O., Cram D.S. (2004). Towards preimplantation genetic diagnosis of cystic fibrosis using microarrays. *Reprod Biomed Online*, **8**, 107-14

Sandalinas M., Sadowy S., Alikani M., Calderon G, Munne S. (2000). Some chromosomal abnormalities in early embryos preclude blastocyst formation in vitro. *Fertile Steril*, **74** (Abstract book), O-009

Sandalinas M., Sadowy S., Alikani M., Calderon G, Cohen J. and Munne S. (2001). Developmental ability of chromosomally abnormal human embryos to develop to the blastocyst stage. *Hum Reprod*, **16**, 1954-58

Sandalinas M, Marquez C, Munne S (2002). Spectral karyotyping of fresh, noninseminated oocytes. *Mol Hum Reprod.*, **8**, 580-5.

Santalo J, Badenas J., Munne S. *et al* (1992). The genetic risks of in vitro fertilization techniques: the use of an animal model. *J Assist Reprod Genet*, **9**, 462-74.

Santalo J., Veiga A., Calderon J. *et al* (1995). Evaluation of cytogenetic analysis for clinical preimplantation diagnosis. *Fertil Steril.* **64**, 44-50.

Santalo J., Grossman M., Egozcue J. (1996). Does Ca^{2+}/Mg^{2+} -free medium have an effect on the survival of the preimplantation mouse embryo after biopsy? *Hum Reprod Update*, **2**, 257-61

Sarkar G. and Sommer S.S. (1993). Removal of DNA contamination in PCR reagents by UV irradiation. *Methods Enzymol*, **218**, 381-88

SART - Society for Assisted Reproductive Technology and The American Fertility Society (1992) In vitro fertilisation-embryo transfer (IVF-ET) in the United States: 1990 results from the IVF-ET Registry. *Fertil Steril*, **57**, 15-24. Savulescu J, Dahl E. (2000). Sex selection and preimplantation diagnosis: a response to the Ethics Committee of the American Society of Reproductive Medicine. *Hum Reprod*, **15**, 1879-80

Scandura JM. (2005). Advances in the molecular genetics of acute leukemia. *Curr Oncol Rep*, **7**, 323-32

Schaffer A.J., Chung J., Heretis K., Wong A., Ledbetter D.H., Lese Martin C. (2004). Comparative genomic hybridisation-array analysis enhances the detection of aneuploidies and submicroscoic imbalances in spontaneous abortions. *Am J Hun Genet*, **74**, 1168-74

Schilling G, Dierlamm J, Hossfeld DK. (2005). Prognostic impact of cytogenetic aberrations in patients with multiple myeloma or monoclonal gammopathy of unknown significance. *Hematol Oncol.*, **23**, 102-7.

Schinzel A.A., Adelsberger P.A., Binkert F., Basaran S., Antonarakis S.E. (1992). No evidence for a paternal interchromosomal effect from analysis of the origin of nondisjunction in Down syndrome patients with concomitant familial chromosome rearrangements. *Am J Hum Genet*, **50**, 288-293.

Scholtes MCW, Behrend C, Dietzel-Dahmen J, van Hoogstraten DG, Marx K, Wohlers S, Verhoeven H, Zeilmaker GH (1998) Chromosomal aberrations in couples undergoing intracytoplasmic sperm injection, influence on implantation and ongoing pregnancy rates. *Fertil Steril*, **70**, 933-937.

Schoolcraft WB, Gardner DK. (2001). Blastocyst versus day 2 or 3 transfer. Semin Reprod Med, 19, 259-68

Schröck E., Manoir S. du, Veldman T., Schoell, Widberg J., Ferguson-Smith M.A., Nin Y., Ledbetter D., Bar-Am I., Soenksen D., Garini Y. and Ried T. (1996). Multicolour spectral karyotyping of human chromosomes. *Science*, **273**, 494-497.

Schröck E., Veldman T., Padill-Nash H. Nin Y. Ried T. (1997). Spectral karyotyping refines cytogenetic diagnostics of constitutional chromosomal abnormalities. *Hum Genet*, **101**, 255-62

Scriven P.N., Handyside A.H., Ogilvie C.M. (1998). Chromosome translocations: segregation modes and strategies for preimplantation genetic diagnosis. *Prenat Diagn*, **18**, 1437-49

Scriven P.N., Mahony F.O., Bickerstaff H., Braude P. *et al* (2000). Clinical pregnancy following blastomere biopsy and PGD for a reciprocal translocation carrier: analysis

of meiotic outcomes and embryo quality in two IVF cycles. *Prenat Diagn*, **20**, 587-592

Scriven, P.N., Flinter, F.A., Braude, P.R., Ogilvie, C.M. (2001). Robertsonian translocations-reproductive risks and indications for preimplantation genetic diagnosis. *Hum. Reprod*, **16**, 2267-73.

Seiler D., Kuhn E. (1970). Kalzium Transport der isolierten vesikel des sarkoplasmatischen retikulums von pateinten mit myotonica congenitaund myotonic dystrophica. *Schweiz Med Wochenschr*, **100**, 1374-76 (referenced from Brook *et al*, 1992)

Sermon K., Lissens W., Nagy Z.P., Joris H., Liebaers I. (1995). Simultaneous amplification of the two most common mutations of infantile Tay-Sachs disease in single blastomeres. *Hum Reprod*, **10**, 2214-17

Sermon K., Lissens W., Joris H., Seneca S., Desmyttere S., Devroey P., Liebaers I. (1997). Clinical application of preimplantation diagnosis for myotonic dystrophy. *Prenat Diagn*, **17**, 925-32

Sermon K, De Vos A, Van de Velde H, Seneca S, Lissens W, Joris H, Vandervorst M, Van Steirteghem A, Liebaers I (1998a) Fluorescent PCR and automated fragment analysis for the clinical application of preimplantation genetic diagnosis of myotonic dystrophy (Steinerts disease). *Mol Hum Reprod*, **4**, 791-796.

Sermon K, Goossens V, Seneca S, Lissens W, De Vos A, Vandervorst M, Van Steirteghem A, Liebaers I (1998b) Preimplantation diagnosis of Huntingtons disease (HD): clinical application and analysis of the HD expansion in affected embryos. *Prenat Diagn*, **18**, 1427-1436.

Sermon K, Seneca S, Vanderfaeillie A, Lissens W, Joris H, Vandervorst M, Van Steirteghem A, Liebaers I (1999) Preimplantation diagnosis for fragile X syndrome based on the detection of the non-expanded paternal and maternal CGG. *Prenat Diagn*, **19**, 1223-1230

Sermon K., Seneca S., De Rycke M. (2001). PGD in the lab for triplet repeat diseases – myotonic dystrophy, Huntington's disease and Fragile-X syndrome. *Mol Cell Endocrin*, **183** (Suppl 1), S77-85

Sermon K. (2002). Current concepts in preimplantation genetic diagnosis (PGD): a molecular biologist's view. *Hum Reprod Upd*, **8**, 11-20

Sermon K., Moutou C., Harper J., Geraedts J., Sriven P., Wilton L., Magli M.C., Michiels A., Viville S., Die D. (2004). ESHRE Preimplantation Genetic Diagnosis Consortium: data collection IV. *Hum Reprod*, **20**, 19-34

Shaffer L.G., Bejjani B.A. (2004). A cytogeneticists perspective on genomic microarrays. *Hum Reprod Update*, **10**, 221-226

Shapiro B.S., Harris D.C., Richter K.S. (2000). Predictive value of 72-hour blastomere cell number on blastocyst development and success of subsequent transfer based on the degree of blastocyst development. *Fertil Steril*, **73**, 582-586

Sharma V. and Litt M. (1992). Tetranucleotide repeat polymorphism at the D21S11 locus. *Hum Mol Genet*, **1**, 67

Shaw D.J. and Harper P.S. (1989). Myotonic Dystrophy: developments in molecular genetic. *Br Med Bull*, **45**, 745-759

Shaw D.J., McCurrach M., Rundle S.A., Harley H.G., Crow S.R., Sohn R., Thirion J.P., Harper P.S., Brook J.D. (1993). Genomic organisation and transcriptional units at the Myotonic dystrophy locus. *Genomics*, **18**, 673-9

Sherlock J., Halder A., Tutschek B., Delhanty J.D.A., Rodeck C., Adinolfi M. (1997). Prenatal detection of fetal aneuploidies using transcervical cell samples. *J Med Genet*, **34**, 302-305

Sherlock J., Cirigliano V., Petrou M., Tutschek B., Adinolfi M. (1998). Assessment of Quantitative Fluorescent Multiplex PCR performed on single cells. *Ann Hum Genet* **62**, 9-23.

Shin J.C., Ross H.L., Elias S., Simpson J.L., Bischoff F.Z. (1997). Detection of chromosomal aneuploidy in endometriosis by M-FISH. *Hum Genet*, **100**, 401-406

Simopoulou M., Harper J.C., Fragouli E., Mantzouratou A., Speyer B.E., Serhal P., Ranieri D.M., Doshi A., Henderson J., Rodeck C.H., Delhanty J.D. (2003). Preimplantation genetic diagnosis of chromosome abnormalities: implications from the outcome for couples with chromosomal rearrangements. *Prenat Diagn*, **23**, 652-62.

Simpson J.L., Elias S. (1995). Isolating fetal cells in the maternal circulation. *Hum* Reprod Update, 1, 409-18

Siracusa G, De Felici M, Salustri A (1985). The proliferative and meiotic history of mammalian female germ cells. In: Metz CB, Monroy A (eds) Biology of Fertilisation, vol 1, Model Systems and Oogenesis. Academic Press Inc Ltd, London, pp253-297.

Smidt-Jensen S. and Hahnemann N. (1984). Transabdominal fine needle biopsy from chorionic villi sampling in the first trimester. *Prenat Diagn*, **4**, 163-168

Snabes MC, Chong SS, Subramanian SB, Kristjansson K, DiSepio D, Hughes MR. (1994). Preimplantation single-cell analysis of multiple genetic loci by whole-genome amplification. *Proc Natl Acad Sci USA*, **91**, 6181-5

Speicher MR, du Manoir S, Schrock E, Holtgreve-Grez H, Schoell B, Lengauer C, Cremer T, Ried T. (1993) Molecular cytogenetic analysis of formalin-fixed, paraffinembedded solid tumors by comparative genomic hybridization after universal DNAamplification. *Hum Mol Genet*, **2**, 1907-14.

Speicher MR, Ballard SG, Ward DC (1996) Karyotyping human chromosomes by combinatorial multi-fluor FISH. *Nat Genet*, **12**, 368-375.

Spits C, Le Caignec C, De Rycke M, Van Haute L, Van Steirteghem A, Liebaers I, Sermon K.(2006). Optimization and evaluation of single-cell whole-genome multiple displacement amplification. *Hum Mutat.*, **27**, 496-503.

Staessen C. and Van Steirteghem (1998). The genetic constitution of multinuclear blastomeres and their derivative daughter blastomeres. *Hum Reprod*, **13**, 1625-31

Staessen C., Van Assche E., Joris H. *et al* (1999). Clinical experience of sex determination by fluorescent in situ hybridisation for preimplantation genetic diagnosis. *Mol Hum Reprod*, **5**, 382-9

Staessen C, Platteau P, Van Assche E, Michiels A, Tournaye H, Camus M, Devroey P, Liebaers I, Van Steirteghem A. (2004). Comparison of blastocyst transfer with or without preimplantation genetic diagnosis for aneuploidy screening in couples with advanced maternal age: a prospective randomized controlled trial. Hum Reprod, 19, 2849-58

Stephenson M.D. (1996). Frequency of factors associated with habitual abortion in 197 couples. *Fert Steril*, **66**, 24-9

Steptoe P.C. and Edwards R.G. (1978). Birth after the reimplantation of a human embryo. *Lancet*, 2(8085), 366

Stern C, Pertile M, Norris H, Hale L, Baker HW. (1999) Chromosome translocations in couples with in-vitro fertilization implantation failure. *Hum Reprod*, **14**, 2097-101 Stranc L.C., Evans J.A., Hamerton J.L. (1997). Chorionic villussampling ans amniocentesis for prenatal diagnosis. *Lancet*, **349**, 435-439

Strom C., Rechisky S., Cieslak J., et al (1997). Preimplantation genetic diagnosis of single gene disorders by two-step oocyte genetic analysis. *J Assist Reprod Genet*, **14**, 469-76

Strom CM, Ginsberg N, Rechitsky S, Cieslak J, Ivakhenko V, Wolf G, Lifchez A, Moise J, Valle J, Kaplan B, White M, Barton J, Kuliev A, Verlinsky Y (1998) Three births after preimplantation genetic diagnosis for cystic fibrosis with sequential first and second polar body analysis. *Am J Obstet Gynecol*, **178**,1298-1306.

Strom C.M., Levin R., Strom S., Masciangelo C., Kuliev A., Verlinsky Y. (2000a) Neonatal outcome of preimplantation genetic diagnosis by polar body removal: the first 109 infants. *Pediatrics*, **106**, 650-653.

Strom C.M., Strom S., Levine E., Ginsberg N., Barton J., Verlinsky Y. (2000b) Obstetric outcomes in 102 pregnancies after preimplantation genetic diagnosis. *Am J Obstet Gynecol*, **182**, 1629-1632.

Struan F.A.G., Steinlicht S., Nentwich U., Kern R., Barwinkel B., Tolle R. (2002). SNP genotyping on a genome wide amplified DOP-PCR template. *Nucleic Acid Res*, **30**, e125

Summers P.M., Campbell J.M., Miller M.W. (1988). Normal *in vitro* development of marmoset monkey embryos after trophectoderm biopsy. *Hum Reprod*, **3**, 389-90

Sutherland G.R. (1979). Heritable fragile sites on human chromosomes. I. Factors affecting expression in lymphocyte culture. *Am J Hum Genet*, **197**, 265-66

Sutherland G.R. (2003). Rare fragile sites. Cytogenet Genome Res, 100, 77-84

Sutherland G.R. and Baker E. (2000). The clinical significance of fragile sites on human chromosomes. *Clin Genet*, **58**, 157-161

Sutton H.E. (1980). An introduction to Human Genetics. Saunders College: Philadelphia

Tabet AC, Aboura A, Dauge MC, Audibert F, Coulomb A, Batallan A, Couturier-Turpin MH, Feldmann G, Tachdjian G. (2001) Cytogenetic analysis of trophoblasts by comparative genomic hybridization in embryo-fetal development anomalies. *Prenat Diagn*, **21**, 613-8.

Tabor A., Madsen M, O'Bell E.B. (1986). Randomised control trials of genetic amniocentesis in 4606 low risk women. *Lancet*, I, 1287-1292

Tapscott SJ (2000). Deconstructing myotonic dystrophy. Science, 289, 1701-2.

Tarin JJ, Conaghan J, Winston RML, Handyside AH (1992) Human embryo biopsy on the 2nd day after insemination for preimplantation diagnosis: removal of a quarter of embryo retards cleavage. *Fertil Steril*, **58**, 970-97

Telford NA, Watson AJ, Schultz GA (1990) Transition from maternal to embryonic control in early mammalian development: a comparison of several species. *Mol Reprod Dev*, **26**, 90-100.

Tempest H.G., Homa S.T., Dalakiouridou M., Christopikou D., Zhai X.P., Griffin D.K. (2004). The association between male infertility and sperm disomy: evidence for variation in disomy levels among individuals and a correlation between particular semen parameters and disomy specific chromosome pairs. *Reprod Biol Endocr*, 14, 82 Tesarik J, Kopecny V, Plachot M, Mandelbaum J (1986) Activation of nucleolar and extra-nucleolar RNA synthesis and changes in ribosomal content of human embryos developing in vitro. *J Reprod Fertil* 78, 463-470.

Tesarik J., Kopecny V., Plaschot M., Mandelbaum J. (1987). Ultrastructural and autoradiographic observations on multinucleated blastomeres of human cleaving embryos obtained by in-vitro fertilisation. *Hum Reprod*, **2**, 127-136

Testart J., Belaisch-Allart J., Forman R., Gazengel A., Strubb N., Hazout A., Frydman R. (1989). Influence of different stimulation treatments on oocyte characteristics and in-vitro fertilizing ability. *Hum Reprod*, **4**,192-7.

Testart J, Gautier E, Brami C, Rolet F, Sedbon E, Thebault (1996) Intracytoplasmic sperm injection in infertile patients with structural chromosome abnormalities. *Hum Reprod*, **11**, 2609-2612.

Tharapel A.T., Tharapel S.A. Bannerman R.M. (1989). Recurrent pregnancy losses and parental chromosome abnormalities: a review. *Br J Obstet Gynecol*, **92**, 899-914

The International Myotonic Dystrophy Consortium (IDMC) (2000). New nomenclature and DNA testing guidelines for Myotonic Dystrophy type I. *Neurology*, **54**, 1218-21

Therman E., Susman M.B. (1993). Reciprocal translocations. In: Human chromosome structure, behaviour and effects. 3rd edition, 273-87

Thornhill A.R., Holding C., Monk M. (1994). Cell recycling of a single cell to detect specific chromosomes and to investigate specific gene sequences. *Hum Reprod*, 9, 2150-2155.

Thornhill A.R. and Monk M. (1996). Cell recycling of a single human cell for preimplantation diagnosis of X-linked disease and dual sex determination. *Mol Hum Reprod*, **2**, 285-289.

Thornton C.A., Johnson K., Moxley R.T. (1994). DM patients have larger CTG expansions in skeletal muscle than in leukocytes. *Ann Neurol*, **35**, 104-107

Tiepolo L, Zuffardi O (1976) Localisation of factors controlling spermatogenesis in the non-fluorescent portion of the human Y chromosome long arm. *Hum Genet*, **34**, 119-124.

Tonnies H, Stumm M, Wegner RD, Chudoba I, Kalscheuer V, Neitzel H. (2001) Comparative genomic hybridization based strategy for the analysis of different chromosome imbalances detected in conventional cytogenetic diagnostics. *Cytogenet Cell Genet*, **93**, 188-94

Tonnies H. (2002). Modern molecular cytogenetic techniques in genetic diagnostics. *Trends Mol Med*, **8**, 246-50

Tournaye H, Devroey P, Liu J, Nagy Z, Lissens W, Van Steirteghem A (1994) Microsurgical epididymal sperm aspiration and intracytoplasmic sperm injection: a new effective approach to infertility as a result of congenital bilateral absence of the vas deferens. *Fertil Steril*, **61**, 1045-1051.

Tournaye H, Liu J, Nagy Z, Joris H, Wisanto A, Bonduelle M, Van der Elst J, Staessen C, Smitz J, Silber S (1995) Intracytoplasmic sperm injection (ICSI): the Brussels experience. *Reprod Fertil Dev*, 7, 269-279

Trounson AO, Mohr LR, Wood C, Leeton JF (1982) Effect of delayed insemination on in-vitro fertilization, culture and transfer of human embryos. *J Reprod Fertil*, **64**, 285-294.

Trussler J.L., Pickering S.J., Ogilvie C.M. (2004). Investigation of chromosomal imbalance in human embryos using CGH. *Reprod Biomed Online*,

Van Assche E., Straessen C., Vegetti W., Bonduelle M. (1999). Preimplantation genetic diagnosis and sperm analysis of fluorescence in situ hybridisation for the most common reciprocal translocation t(11;22). *Mol Hum Reprod*, **5**, 682-90

Van Blerkom J., Davis P.W., Lee J. (1995). ATP content of human oocytes and developmental potential and outcome after IVF and embryo transfer. *Hum Reprod*, 10, 415-424

Van Blerkom J (1998) Epigenetic influences on oocyte developmental competence: perifollicular vascularity and intrafollicular oxygen. *J Assist Reprod Genet*, **15**, 226-234.

Van Blerkom J. (2004). Mitochondria in human oogenesis and preimplantation embryogenesis: engines of metablosim, ionic regulation and developmental competence. *Reproduction*, **128**, 269-280

Van der Elst Y., Camus M., Van den Abbeel E., Maes R., Devroey P., Van Steirteghem A.C. (1995). Prospective randomized study on the cryopreservation of human embryos with dimethylsulfoxide or 1,2-propanediol protocols. *Fertil Steril*, **63**, 92-100

Vandervors M., Staessen C., Sermon K., De Vos A., Van de Helde H., Van Assche E., Bonduelle M., Vanderfaellie A., Lissens W., Tournaye H., Devroey P., Van Steirteghem A. and Liebaers I. (2000). The Brussels experience of more than five years of clinical preimplantation genetic diagnosis. *Hum Reprod Update*, **6**, 364-373

Van de Velde H, De Vos A, Sermon K, Staessen C, De Rycke M, Van Assche E, Lissens W, Vandervorst M, Van Ranst H, Liebaers I, Van Steirteghem A (2000) Embryo implantation after biopsy of one or two cells from cleavage-stage embryos with a view to preimplantation genetic diagnosis. *Prenat Diagn*, **20**, 1030-1037.

Van Hemel JO, Eussen HJ (2000) Interchromosomal insertions. Identification of five cases and a review. *Hum Genet*, **107**, 415-432.

Van Opstal D., Eussen H.J., Van Hemel J.O. (1993). Application of fluorescent *in situ* hybridisation for 'de novo' anomalies in prenatal diagnosis. *Prenat Diagn*, **13**, 825-832

Van Opstal D., Eussen H.J., Van Hemel J.O., van der Heide ., van der Berg C., In'T Veld P.A., Los F.J. (1995). A chromosome 21-specific cosmid cocktail for the detection of chromosome 21 aberrations in interphase nuclei. *Prenat Diagn*, **15**, 705-11

Van Roy N., Jauch A., Van Gele M., Laureys G., Versteeg R., De Paepe A., Cremer T., Speleman F. (1997). Comparative genomic hybridization analysis of human neuroblastomas: detection of distal 1p deletions and further molecular genetic characterization of neuroblastoma cell lines. Cancer Genet *Cytogenet*, **97**, 135-142.

Van der Elst J., Camus M., Van den Abbeel E. (1995). Prospective randomised study in the cryopreservation of human embryos with mdimethylsulfoxide or 1,2propanediol protocols. *Fertil Steril*, **63**, 92-100 Van Steirteghem A., Van den Abbeel E., Camus M. (1992). Cryopreservation of human embryos. *Bal Clin Obstet Gynecol*, **6**, 313-25

Veiga A., Santalo J., Vidal F., Calderon G., Gimenez C., Boada M., Egozcue J., Barrie P.N. (1994). Twin pregnancy after preimplantation diagnosis for sex selection: case report. *Hum Reprod*, **9**, 2156-2159

Veiga A., Sandalinas M., Benkhalifa M., Boada M., Carrera M., Santalo J., Barri P.N., Ménézo Y. (1997). Laser blastocyst biopsy for preimplantation genetic diagnosis in the human. *Zygote*, **5**, 351-354.

Veiga A., Gil Y., Boada M., Carrera M. (1999). Confirmation of diagnosis in preimplantation genetic diagnosis (PGD) through blastocyst culture: preliminary experience. *Prenat Diagn*, **19**, 1242-7.

Verlinsky Y., Ginsburg N., Lifchez J., Moise J., Strom C. (1990). Analysis of the first polar body: preconception genetic diagnosis. *Hum Reprod*, **5**, 826-29

Verlinsky Y., Rechitsky S., Evsikov S. Kuliev A. (1992). Preconception and preimplantation diagnosis for cystic fibrosis. *Prenat Diagn*, **12**, 102-110

Verlinsky Y. and Cieslak J. (1993). Embryological and technical aspects of preimplantation genetic diagnosis. In: Verlinsky Y, Kuliev AM (eds) *Preimplantation Diagnosis of Genetic Diseases: A New Technique in Assisted Reproduction*. Wiley-Liss, New York, pp49-67.

Verlinsky Y., Handyside A., Grifo J., Munne S. et al (1994). Preimplantation diagnosis of genetic and chromosomal diseases. *J Assist Reprod Genet*, **11**, 236-43 Verlinsky Y. and Kuliev A. (1996a). Preimplantation polar body diagnosis. *Biochem Mol Med*, **58**, 13-7

Verlinsky Y. and Kuliev A. (1996b). Preimplantation diagnosis of common aneuploidies in fertile couples of advanced maternal age. *Hum Reprod*, **11**, 2076-77

Verlinsky Y, Cieslak J, Freidine M, Ivakhnenko V, Wolf G, Kovalinskaya L, White M, Lifchez A, Kaplan B, Moise J, Valle J, Ginsberg N, Strom C, Kuliev A (1996) Polar body diagnosis of common aneuploidies by FISH. *J Assist Reprod Genet*, **13**, 157-162.

Verlinsky Y. (1999). Preimplantation diagnosis: an alternative to prenatal diagnosis of genetic and chromosomal disorders. Report of the Eighth Annual Meeting of the International Working Group on Preimplantation Genetics in association with the Ninth International Conference on Prenatal Diagnosis and Therapy, Los Angeles, June 7, 1998. *J Assist Reprod Genet*, **16**, 161-164.

Verlinsky Y., Evsikov S. (1999a). A simplified and efficient method for obtaining metaphase chromosomes from individual human blastomeres. *Fertil Steril*, **72**, 1127-1133.

Verlinsky Y., Evsikov S. (1999b). Karyotyping of human oocytes by chromosomal analysis of the second polar bodies. *Mol Hum Reprod*, **5**, 89-95.

Verlinsky Y, Cieslak J, Ivakhnenko V, Evsikov S, Wolf G, White M, Lifchez A, Kaplan B, Moise J, Valle J, Ginsberg N, Strom C, Kuliev A. (2001). Chromosomal abnormalities in the first and second polar body. *Mol Cell Endocrinol*, **22**, S47-9.

Verlinsky Y., Rechtsky S., verlinsky O., Sharapova T., Levy M., Kuliev A. (2002). Preimplantation diagnosis of neurofibromatosis. *Reprod Biomed Online*, **4**, 218-222

Visakorpi T, Kallioniemi AH, Syvanen AC, Hyytinen ER, Karhu R, Tammela T, Isola JJ, Kallioniemi OP. (1995) Genetic changes in primary and recurrent prostate cancer by comparative genomic hybridization. *Cancer Res.* **15**;55, 342-7.

Viville S. and Pergament D. (1998). Results of a survey of the legal status and attitudes towards preimplantation genetic diagnosis conducted in 13 different countries. *Prenat Diagn*, **18**, 1374-80

Vooijs M, Yu LC, Tkachuk D, Pinkel D, Gray JW (1993) Libraries for each human chromosome constructed from sorter-enriched chromosomes by using linker-adapter PCR. *Am J Hum Genet*, **52**, 586-597.

Voullaire L., Wilton L., Slater H., Williamson R. (1999). Detection of aneuploidy in single cells using CGH. *Prenat Diagn*, **19**, 846-51

Voullaire L., Slater H., Williamson R., Wilton L. (2000). Chromosome analysis of blastomeres from human embryos by using comparative genomic hybridisation. *Hum Genet*, **106**, 210-7

Voullaire L, Wilton L, McBain J, Callaghan T, Williamson R. (2002). Chromosome abnormalities identified by comparative genomic hybridization in embryos from women with repeated implantation failure. *Mol Hum Reprod*, **8**, 1035-41.

Walknowska J., Conte F.A., Grumbach M.M. (1969). Practical and theoretical implications of fetal/maternal lymphocyte transfer. *Lancet*, **1**, 1119-1122

Warburton D., Kline J., Stein Z. (1986). Cytogenetic abnormalities in spontaneous abortions of recognised conceptions. In Porter I.H., and Willey A., *Perinatal Genetics: Diagnosis and treatment*. Academic press, New York, 133-48

Warburton D. (1991). De novo balanced chromosome rearrangements and extra marker chromosomes identified at prenatal diagnosis: clinical significance and distribution of breakpoints. *Am J Hum Genet*, **49**, 995-1013

Ward R.H.T., Modell B., Petrou M. (1983). Method of sampling chorionic villi in 1st trimester of pregnancy under guidance of realtime ultrasound. *Br Med J*, **286**, 1542-49 Ward B.E., Gersen S.L., Carelli M.P. (1993). Rapid prenatal diagnosis of chromosomal aneuploidies by FISH: clinical experience with 4500 specimens. *Am J Hum Genet*, **52**, 854-65

Weber J.L, May P.E. (1989). Abundant class of human DNA polymorphisms which may be typed using the PCR. *Am J Hum Genet*, **44**, 388-96

Weber J, Looten R, Houdayer C, Stoppa-Lyonnet D, Viovy JL (2005). Improving sensitivity of electrophoretic heteroduplex analysis using nucleosides as additives: Application to the breast cancer predisposition gene BRCA2. *Electrophoresis.*, **27**, 1444-52.

Wells D. and Sherlock J. (1998). Strategies for preimplantation genetic diagnosis of single gene disorders by DNA amplification. *Prenat Diagn*, **18**, 1389-401

Wells D, Sherlock JK, Handyside AH, Delhanty JDA (1999) Detailed chromosomal and molecular genetic analysis of single cells by whole genome amplification and comparative genomic hybridisation (CGH). *Nucl Acids Res*, **27**, 1214-1218.

Wells D. and Delhanty JDA (2000). Comprehensive chromosomal analysis of human reproduction using whole genome amplification and single cell comparative genomic hybridisation. *Mol Hum Reprod*, **6**, 1055-62

Wells D. and Delhanty J.D.A (2001). Preimplantation genetic diagnosis: applications for molecular medicine. *Trends Mol Med*, **7**, 23-30

Wells D., Escudero T., Levy B., Hirschhorn K., Delhanty J.D., Munne S. (2002). First clinical application of comparative genomic hybridization and polar body testing for preimplantation genetic diagnosis of aneuploidy. *Fertil Steril*, **78**, 543-9.

Wells D. and Levy B. (2003). Cytogenetics in reproductive medicine: the contribution of comparative genomic hybridization (CGH). *BioEssays*, **25**, 289-300.

Wells D. (2004). Advances in preimplantation genetic diagnosis. Eur I Obstet Gynecol Reprod Biol, 115, S97-101

Wells D, Bermudez MG, Steuerwald N, Thornhill AR, Walker DL, Malter H, Delhanty JD, Cohen J. (2005). Expression of genes regulating chromosome segregation, the cell cycle and apoptosis during human preimplantation development. *Hum Reprod.*, **20**, 1339-48

Wennerholm U.B., Hamberger L., Nilsson L., Wennergren M., Wikland M., Bergh C. (1997). Obstetric and perinatal outcome of children conceived from cryopreserved embryos. *Hum Reprod*, **12**, 1819-1825.

Werlin L., Rodi I., DeCherney A., Marello E., Hill D., Munne S. (2003). PGD as both therapeutic and diagnostic tool in assisted reproduction technology. *Fert Steril*,

White M.B., Carvalho M., Derse D., O'Brien S.J., Dean M. (1992). Detecting single base substitutions as heteroduplex polymorphisms. *Genomics*, **12**, 301-306

Wiegant J., Kalle W., Mullenders L., Brookes S., Hoovers J.M.N., Dauwerse J.G., van Ommen G.J.B., Raap A.K. (1992). High resolution in situ hybridisation using DNA halo preparations. *Hum Mol Genet*, **1**, 587-591.

Wiegant J., Weismeijer C.C., Hoovers J.M.N., Schuuring E., d'Azzo A., Vrolijk J., Tanke H.J., Raap A.K. (1993). Multiple and sensitive fluorescence in situ hybridisation with rhodamin-, fluorescein- and coumarin-labelled DNAs. *Cytogenet Cell Genet* **63**, 73-76.

Wilcox A.J., Weinberg C.R., O'Connor J.F. (1988). Incidence of early loss pregnancy. *N Engl J Med*, **319**, 189-194

Willard HF (1985) Chromosome-specific organisation of human alpha satellite DNA. *Am J Hum Genet*, **37**, 524-532.

Willadsen S.M., Levron J., Munné S., Schimmel T., Márquez C., Scott R., Cohen J. (1999). Rapid visualisation of metaphase chromosomes in single human blastomeres after fusion with in-vitro matured bovine eggs. *Hum Reprod* **14**, 470-475

Wilson MG. (1985) Cytogenetics update for pediatricians. *Curr Probl Pediatr*. 15, 1-47.

Wilton L, Williamson R, McBain J, Edgar D, Voullaire L. (2001). Birth of a healthy infant after preimplantation confirmation of euploidy by comparative genomic hybridization. *N Engl J Med.*, **345**, 1537-41.

Wilton L. (2002) Preimplantation genetic diagnosis for an euploidy screening in early human embryos: a review. *Prenat Diagn*, **22**, 512-8.

Wilton L., Voullaire L., Sargeant P., Williamson R., McBain J. (2003). Preimplantation aneuploidy screening using comparative genomic hybridization or fluorescence in situ hybridization of embryos from patients with recurrent implantation failure. *Fertil Steril*, **80**, 860-8.

Winston N.J., Braude P.R., Pickering S.J., George M.A., Cant A., Currie J., Johnson M.H. (1991). The incidence of abnormal morphology and nucleocytoplasmic ratios in 2-, 3- and 5-day human pre-embryos. *Hum Reprod*, **6**, 17-24.

Wolstenholme J. (1996). Confined placental mosaicism for trisomies 2, 3, 7, 8, 9, 16 and 22: their incidence. Likely origins and mechanisms for cell lineage compartmentalisation. *Prenat Diagn*, **16**, 511-24

Wong I.H.N., Lo Y.M.D., Zhang J., Liew C.T., Wong N. (1999). Detection of aberrant p16 methylation in the plasma and serum of liver cancer patients. *Cancer Res*, **59**, 71-3

Worsham MJ, Miller DA, Devries JM, Mitchell AR, Babu VR, Surli V, Weiss L, Van Dyke DL (1989) A dicentric recombinant 9 derived from a paracentric inversion: phenotype, cytogenetics, and molecular analysis of centromeres. *Am J Hum Genet*, **44**, 115-123.

Xu K., Shi Z.M., Veek L.L., Hughes M.R., Rosenwaks Z. (1999). First unaffected pregnancy using preimplantation genetic diagnosis of sickle cell anaemia. *JAMA 281*, 1701-1706

Yurov Y.B., Soloviev I.V., Vorsanova S.G., Marcais B., Roizes G., Lewis R. (1996). High resolution multicolour fluorescence in situ hybridisation using cyanine and fluorescein dyes: rapid chromosome identification by directly fluorescently labelled alphoid DNA probes. *Hum Genet*, **97**, 390-398.

Zackai E.H., Emanuel B.S. (1980). Site-specific reciprocal translocation, t(11;22)(q23.3;11.2), in several unrelated families with 3:1 meiotic disjunction. Am J Med Genet, 7, 507-21

Zaragoza MV, Surti U, Redline RW, Millie E, Chakravarti A, Hassold TJ (2000) Parental origin and phenotype of triploidy in spontaneous abortions: predominance of diandry and association with the partial hydatidiform mole. *Am J Hum Genet*, 66, 1807-1820

Zeesman S, Carson N, Whelan DT (2002). Paternal transmission of the congenital form of myotonic dystrophy type 1: a new case and review of the literature. *Am J Med Genet*, 107, 222-6.

Zenzes MT, Casper RF. (1991) A morphologic study of unfertilized oocytes and abnormal embryos in human in vitro fertilization. *J In Vitro Fert Embryo Transf.*, **8**, 73-9.

Zhang, L., Cui, X., Schmitt, K., Hubert, R., Navidi, W., Arnheim, N. (1992). Whole genome amplification form a single cell-implications for genetic analysis. *Proc. Natl. Acad. Sci. U.S.A.*, **89**, 5847-51.

Zhang J, Meltzer P, Jenkins R, Guan XY, Trent J (1993) Application of chromosome microdissection probes for elucidation of bcr-abl fusion and variant Philadelphia chromosome translocations in chronic myelogenous leukemia. *Blood*, **81**, 3365-3371.

Ziegle JS, Su Y, Corcoran KP, Nie L, Mayrand PE, Hoff LB, McBride LJ, Kronick MN, Diehl SR. (1992) Application of automated DNA sizing technology for genotyping microsatellite loci. *Genomics*, **14**, 1026-31.

Zuhlke C., Atic J., Martorell L., Gembruch U., Kohl M., Gopel W., Schwiger E. (2000). Rapid detection of expansions by PCR and non-radioactive hybridisation: application from prenatal diagnosis of myotonic dystrophy. *Prenat Diagn*, **20**, 66-69 Zunz E, Abeliovich D, Halpern GJ, Magal N, Shohat M. (2004). Myotonic dystrophy-no evidence for preferential transmission of the mutated allele: a prenatal analysis. *Am J Med Genet A*, **127**, 50-3.

Chapter 8 Appendix

| Pt No. | EMB No | Cell | CGH Result | FISH Result (day 5) | Re-FISH Result (day 5) |
|-----------|-----------|------|---------------------|--|--|
| (age) | 110. | | (day 5) | X/Y/18 (cell no) | 9c/16c/22LSI (cell no) |
| | | a | No result | X, Y, 18,18 (1,5,6,7,8,9,10,11,13) X, Y, 18 (2,12) | 9, 9, 16, 16, 22, 22 (1,2,3,4,5,6,9,10,11, 12,13) |
| | 1.1 | b | rev ish XY | X, X, Y, Y, 18,18 (3,4) | 9, 9, 16, 16, 22, 22, 22 (7,8) |
| 1 (38) | | a | rev ish XY, dim(22) | X, 18 (1) X,X,X,X,X,X,X,18,18,18,18,18,18,18 (2) | 9, 9, 22 (1) 9,9,9,9,9, 16,16,16,16,16, 22,22,22,22 (2) |
| | 1.2 | b | No result | | |
| | 1.3 | a | rev ish XY | X, X, Y, Y, 18,18, 18, 18 (1) X, Y, 18,18 (2) | 9, 9, 16, 16, 22, 22 (1,2) |
| 2 | 2.1 | a | rev ish XY | X, Y, 18,18 (1,2,3,4) | 9, 9, 16, 16, 22, 22 (1,2,3,4) |
| (29) | | b | No result | | |

Table 8.1. Analytical CGH and FISH results from each cell for chapter 4

| Pt No. | EMB | Cell | CGH Result | FISH Result (day 5) | Re-FISH Result (day 5) |
|-----------|------|------|--|---|---|
| (age) | 190. | | (day 3) | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| 2 | | a | rev ish XY, dim(18) | Nullisomy for all chromosomes (1) X, Y, 18 (2) 18 (3) | 9, 22 (1,3) 9, 9, 16 (2 |
| (29) | 2.2 | b | No result | | |
| 3 | 3.1 | a | rev ish XY, enh(5pter, 9qter, 17), dim(4,19) | Y,18,18,18 (1,2) | 16, 22 (1) 9, 9, 16, 22 (2) |
| (35) | | b | No result | | |
| | 4.1 | a | rev ish XY | No cells to FISH | No cells to FISH |
| 4 (35) | | a | rev ish XX | X, X, 18, 18 (10,12,16,17,19,20,22, 26,32) X, 18, 18 (4,9,13,23,24,27,30,31) | 9, 9, 16, 16, 22, 22 (2,4,6, 12,14,16,21, 22,23,24,25,26,27, 29,30,31,32,33) |
| | 4.2 | b | rev ish XX, enh(1, 22), dim(16pter, 18) | X, X, 18 (2,6,8,14,28) X, 18 (11,15,18,29) 18, 18 (5,7,21) X, X, 18, 18, 18 (25,33) X, X, 18, 18, 18, 18 (1) X, X, X, 18, 18, 18, 18 (3) | 9, 9, 16, 16, 22 (7,8,9, 15) 9, 9, 16, 22 (10,11,13) 9, 9, 16, 22, 22 (17, 18, 19) 9, 9, 16, 22 (3) 9, 9, 16 (1,5,28) |

| Pt No. (age) | EMB No. | Cell | CGH Result (day 3) | FISH Result (day 5) | Re-FISH Result (day 5) |
|-----------------|------------|------|-----------------------|--------------------------|-------------------------------|
| (-8-) | | | (uuy 0) | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| 4 | 4.3 | a | rev ish XY | No cells to FISH | No cells to FISH |
| (35) | 4.4 | a | rev ish XY, enh(X) | No cells to FISH | No cells to FISH |
| | 6.1 | а | No result | No cells to FISH | No cells to FISH |
| 6 (30) | | a | rev ish XY | X, X, 18, 18 (1,2,3,4,5) | 9, 9, 16, 16, 22, 22 (1,3,4) |
| | 6.2 | b | rev ish XY | | 9, 9, 10, 10, 22 (2,3) |

| Pt No. | EMB | Cell | CGH Result | FISH Result (day 5) | Re-FISH Result (day 5) |
|-----------|-----|------|---|---|--|
| (age) | No. | | (day 3) | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| | | a | rev ish XY, enh(6p25-p21.1)* | X, X, Y, 18, 18 (1,2) X, 18, 18 (3,4,7,8) X, Y, 18, 18 (5,6) | 9, 9, 16, 16, 22, 22 (1,2,5,6,7) 9, 9, 16, 22, 22 (3) 9, 9, 16, 16, 22, 22, 22 (4) |
| 6 (30) | 6.3 | b | rev ish XY, enh(Y) | | 9, 9, 16, 16, 22 (8) |
| | 7.1 | a | rev ish XX | X, X, 18, 18 (3,6,7,8) X, X, 18 (1,2,4) X, X, 18, 18, 18 (5) | 9, 9, 16, 16, 22, 22 (1,2,3,5,7,8) 9, 9, 16, 22, 22 (4) 9, 9, 16, 16, 22 (6) |
| 7 | 7.2 | a | rev ish XYY, enh(1p36.2-36.1, 2q31-p25, 5, 7, 8, 18, 19, 20, 21 and Y), dim(1p31-q44, 2q32-q37, 3, 6, 11, 13, 14, 15) | X, X, X, 18, 18, 18 (1,3) X, X, X, X, X, Y, Y, 18,18, 18 (2) | 3c/11c/13LSI 3, 13, 13 (1) 3 (2) 3, 3, 3, 13 (3) |
| (39) | | b | No result | | |
| | 7.3 | a | rev ish XY, dim(22q11.1-q13) | X, X, 18, 18 (3,5,6,7,9,10,11,12, 13, 14,15,17,18,19, 20,21,22,23,24,25, 26,27,28,29,30) | 9, 9, 16, 16, 22, 22 (6,13,15,23,24) 9, 9, 16, 16,22,22,22 (7,14,18,19,26) 9, 9, 16, 22 (12,22) 9, 9 (17,20) |
| | | b | rev ish XY | X, 18, 18 (4,8) X, X, X, X, 18, 18 (1,2) X, X, X, X, 18, 18, 18, 18 (16) X, X, 18 (24) | 22 (3,8) 9, 9, 22 (4,11) 9, 16, 16, 22 (1) 9, 9, 9, 16, 22 (2) 9, 16, 16, 22, 22 (5) 9,9,9,9, 16,16,16,16, 22,22,22,22 (16) |

| Pt No. | EMB | Cell | CGH Result | FISH Result (day 5) | Re-FISH Result (day 5) |
|-----------|------|------|---|---|--|
| (age) | INO. | | (day 3) | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| | | 0 | rev ish XX | | <u>10c / 14q</u> |
| | | a | | X, X, 18, 18 (1,2,3) | 10, 14q, 14q (1,2) |
| | 8.1 | | ray ish XX | X, X, 18, 18, 18, 18 (4) | 10, 14q, 14q, 14q (3,4) |
| 8 (31) | | b | enh(4, 12p11.2-q24.3, 14q21-q32) dim(2q31-q37, 10) | | |
| | 8.2 | a | rev ish XX, enh(9 and 16) | X, X,18,18 (1,2,3,4,5,6,7,8,9,10) | 9, 9, 16, 16, 22, 22 (2,3,4,6,7,10) 22, 22 (1,8,9) 9, 22, 22 (5) |
| | 9.1 | a | rev ish XY, enh(Xp11.2-q22)* | X, Y, 18, 18 (1,2,3,4,5,6,7) | 9, 9, 16, 16, 22, 22 (1,2,3,4,5,6,7) |
| 9 (33) | 9.2 | a | rev ish XY, dim(Yq12)* | X, Y, 18, 18 (2,3,4,5,7,8,9,10,11, 12,1516,17,18,19,20 21,22,23,24,2526,27 28,29,30,31) X, Y, 18 (1,13,14) Y, 18, 18 (6) | 9, 9, 16, 16, 22, 22 (2,3,4,5,6,7,8,9, 10,11,12,14,15, 16,17,18,19,20, 21,22,23,24,25, 26,27,29,31) 9, 9, 16, 22, 22, 22 (1,30) 9, 9, 9, 16, 22, 22, 22 (13) |
| i | | | | | 9, 9, 16, 16, 22, 22, 22, 22 (28) |
| Pt No. | EMB | Cell | CGH Result | FISH Result (day 5) | Re-FISH Result (day 5) |
|-----------|---------------|------|---|---|--|
| (age) | 110. | | (day 5) | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| | 9.3 | a | rev ish XY, dim(15q15-q26, 16) | X, Y, 18, 18 (1,2,3,4) | 9, 9, 16, 16, 22, 22 (1,4) 9, 9, 16, 22, 22, 22 (2,3) |
| 9 (33) | 9.4 a rev ish | | rev ish XX, dim(11q23-q25)* | X, X, 18, 18 (1,3,5) X, X, X, X, X, 18, 18, 18, 18 (2,6) X, X (4) | 9, 9, 16, 16, 22, 22 (1,3,4,5) 9,9,9,9, 16,16,16,16, 22,22,22,22 (2,6) |
| | 9.5 | a | rev ish XY | X, Y, 18, 18 (1,3,4,5,7,8,9) Y, 18, 18 (2,6,10) 18 (11) | 9, 9, 16, 16, 22, 22 (1,2,3,4,5,6,7, 8,9,10) 9, 16, 22, 22 (11) |
| | 9.6 | a | rev ish XX, enh(2q21-q33, 3q11.1-q25), dim(1p36.1-p33, 16, 19 and 22) | X, X, 18, 18 (1,2,3,4,5,6,7,8,9,12,13 14,15,16,18,19,20,22) X, X, 18 (10) X, X, X, X, 18, 18, 18, 18 (11,17) | 9, 9, 16, 16, 22, 22 (2,3,4,5,7,8, 9,10,13,16, 17,19,20,21 22,23) |
| | | b | rev ish XX | X, X, X, 18, 18 (22) | 9, 9, 16, 22 (6,14,15) 9,9,9,9, 16,16,16,16, 22,22,22,22 (11,17) 9, 9, 16, 16, 22 (12,18) 9, 9, 16, 22, 22, 22, 22 (1) |

| Pt No. | EMB No | Cell | CGH Result | FISH Result (day 5) | Re-FISH Result (day 5) |
|------------|-----------|------|--|--|---|
| (450) | 110. | | (uay 5) | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| | 9.7 | a | rev ish XY, enh(1p36.3-q21), dim(1q31-q44) | X, Y, 18, 18 (3,5) X, X, Y, 18, 18 (1) | <u>1p / 1q / 16cep</u> 1p, 1q, 16, 16 (1) 1p, 1q, 1q, 1q, 16, 16 (2,5) |
| 9 | | b | rev ish XY, dim(1) | X, Y,Y, 18, 18 (2) Y, 18, 18 (4) | 1p, 1p, 1p, 16, 16 (3) 1p, 1p, 1q, 1q, 16 (4) |
| (00) | | | rev ish XY, | | <u>3cep/6cep/18cep</u> |
| | 9.8 | a | enh(17p13-q11, 18p11.3-q11.1), dim(3p26-p14) | X, Y, 18, 18 (1,2,3,4,5,6,7,8,9,10,12, 13,15,16,17) | 3, 3, 6, 6, 18, 18 (1,2,3,5,6,7,8,12, 13,16,17) |
| | | b | rev ish XY, enh(6) | X, Y,Y Y, 18, 18 (11) Y, Y, 18, 18 (14) | 3, 3, 6, 6, 6, 18, 18 (4,9,10) 3, 3, 6, 18, 18 (15) 3, 3, 6, 6, 18 (11) |
| 10 (32) | 10.1 | a | rev ish XX, enh(20) | X, X, 18, 18 (4,8,12,13,16,17,18) X, X, X, X, 18, 18 (1,2,3,5,6,11,14) X, X, X, 18, 18 (7,15,19) X, X, X, 18, 18, 18 (9) X, 18 (10) X, X, X, X, X, X, 18, 18, 18, 18 (17) | 9, 9, 16, 16, 22, 22 (1,2,3,4,5,6,7, 8,10,11,12,13, 15,16,17,18,19 9, 9, 16, 16, 16, 22, 22, 22 (9) 9, 9, 16, 16, 16, 22 (14) 9, 9, 9, 9, 16, 16, 16, 16, 22, 22, 22, 22 (17) |
| 11 (35) | 11.1 | a | rev ish XY, enh(Yp11.3-11.2)* | X, Y, 18, 18 (1,2,3,4,5,6,7,8,9,10) | 9, 9, 16, 16, 22, 22 (1,2,3,4,5,6,7, 8,9,10) |

| Pt No. | EMB | Cell | CGH Result | FISH Result (day 5) | Re-FISH Result (day 5) |
|------------|------|------|---|--|---|
| (age) | 140. | | (day 5) | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| 12 (30) | 12.1 | a | rev ish XX | X, X, 18, 18 (1,3) X, X, 18 (2) | 9, 9, 16, 16, 22, 22 (1) 9, 9, 16, 22, 22, 22, 22 (2) 9, 22, 22 (3) |
| | 12.2 | a | rev ish XY, enh(Y), dim(4pter) | X, Y, 18, 18 (1,2,3,4,5,11) X, Y, 18 (8,10) X, 18, 18 (6,7) 18 (9) Y, Y, 18, 18 (12) | 4cep/Ycep 4, 4, Y (1,2,3,4,5,8,10,11) 4, 4 (6,7) Nullisomy 4 and Y (9) 4, Y, Y (12) |
| | 12.3 | a | rev ish XY, enh(Yq11.1-q12), dim(16q21-q24) | X, Y, 18 (1) X, Y, 18, 18 (2) | <u>Xc/Yc/16q</u> X, Y, 16q (1,2) |
| 13 (36) | 13.1 | a | rev ish XY, enh(5) | No cells to FISH | No cells to FISH |

| Pt No. | EMB | Cell | CGH Result | FISH Result (day 5) | Re-FISH Result (day 5) |
|------------|------|------|---|---|---|
| (age) | N0. | | (day 3) | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| | | | rev ish XX, dim(19p13.3- | X, X, 18, 18 (1.2.3.4.7.9.12.14) | 9, 9, 16, 16, 22, 22 (1,2,3,7,9,12,14) |
| | | a | p13.2)* | 18, 18 (5) | 9, 16, 16, 22, 22, 22 (4) |
| | | | | X, X, X, X, 18, 18 (6) | 16, 22, 22, 22, 22 (5) |
| | 13.2 | | | X, X, X, 18, 18, 18 (8) | 22, 22 (6) |
| | | | rev ish XX | X, X, X, 18 (10) | 9, 9, 9, 16, 16, 16, 22, 22, 22 (8) |
| | | b | | X, 18, 18, 18 (11) | 9, 9, 16, 16, 22, 22, 22 (10) |
| | | | | X, X, 18 (13) | 9, 9, 16, 16, 22 (11) |
| 13 | | | | X, X, X, X, 18, 18, 18, 18 (15) | 9, 9, 9, 16, 16, 22, 22, 22 (13) |
| (36) | | | | | 9, 9, 9, 9, 16, 16, 16, 16, 22, 22, 22, 22 (15) |
| | | | rev ish XX | X, X, 18, 18 (1,2,3,4,5,6,8,9,10,12,15,16 | 9, 9, 16, 16, 22, 22 (1,2,3,4,5,6,8,9,10,11, |
| | | a | | 17,20,22,23,24,25,26,27, | 12,15,16,17,18,20,21, |
| | | | | 28,29,30,31,32,33,34,35) | 22,23,24,25,26,27,28, |
| l | 13.3 | | rev ish XX | X, X, X, 18, 18, 18 (13,14) | 29,30,31,32,33,34,35) |
| | | b | | X, X, X, 18 (11,21) | 9, 9, 9, 16, 16, 16, 22, 22, 22 (13, 14) |
| | | | | X, X, X, X, X, X, 18, 18, 18 18, 18 (7) | 9, 9, 9, 16, 16, 22, 22, 22, 22 (7) |
| | | | | X, X, 18, 18, 18, 18, 18 (18) | 9, 16, 22, 22 (19) |
| | | | | X, X, 18, 18, 18 (19) | |
| | | a | rev ish XX | No cells to FISH | No cells to FISH |
| 14 (28) | 14.1 | b | rev ish XX, enh(2,4, 9qter), dim(1, 16, 21) | | |

| Pt No. (age) | EMB No. | Cell | CGH Result (day 3) | FISH Result (day 5) | Re-FISH Result (day 5) |
|-----------------|------------|------|---|------------------------------------|--|
| | | | | X/Y/18 (cell no.) | 9c/16c/22LSI (cell no.) |
| | | a | rev ish XX, | X, X, 18, 18 (1,2,5,6) X, X (3) | 9, 9, 16, 16, 22, 22 (1) 9, 9, 16, 22, 22 (2) |
| | | | enh(1, 10, 16qter), | X, 18 (4) | 9, 9, 16, 22 (3,5) |
| 1 | 14.2 | | dim(8, 13, 21, 22) | _ | 9, 9, 16, 16, 16, 22, 22, 22 (4,6) |
| 14 (28) | | b | rev ish XX, enh(2,6), dim(9qter,13, 15, 16qter, 17) | | |
| | 14.3 | a | rev ish XX | No cells to FISH | No cells to FISH |